**Probability Surveys** Vol. 5 (2008) 169–345 ISSN: 1549-5787 DOI: 10.1214/08-PS137

### Stability of queueing networks<sup>\*</sup>

### Maury Bramson

University of Minnesota e-mail: bramson@math.umn.edu

Received June 2008.

# Contents

<ul> <li>1.1 The <i>M/M/</i>1 queue</li></ul>	172 173 180 183			
<ul> <li>1.2 Basic concepts of queueing networks</li></ul>	173 180 183			
<ul> <li>1.3 Queueing network equations and fluid models</li></ul>	180 183			
<ul> <li>1.4 Outline of lectures</li></ul>	183			
2 The classical networks         2.1 Main results         2.2 Stationarity and reversibility         2.3 Homogeneous nodes of Kelly type				
<ul> <li>2.1 Main results</li></ul>	186			
<ul> <li>2.2 Stationarity and reversibility</li> <li>2.3 Homogeneous nodes of Kelly type</li> </ul>	187			
2.3 Homogeneous nodes of Kelly type	191			
	195			
2.4 Symmetric nodes	200			
2.5 Quasi-reversibility	207			
Instability of subcritical queueing networks 220				
3.1 Basic examples of unstable networks	221			
3.2 Examples of unstable FIFO networks	227			
3.3 Other examples of unstable networks	237			
4 Stability of queueing networks	243			
4.1 Some Markov process background	246			
4.2 Results for bounded sets	257			
4.3 Fluid models and fluid limits				
4.4 Demonstration of stability	265			
4.5 Appendix	$265 \\ 280$			

<sup>\*</sup>This monograph is also being published as Springer Lecture Notes in Mathematics 1950, École d'Été de Probabilités de Saint-Flour XXXVI-2006. Copyright © 2008 Springer-Verlag Berlin Heidelberg.

		M. Bramson/Stability of queueing networks	170
5	Applications and some further theory		
	5.1	Single class networks	303
	5.2	FBFS and LBFS reentrant lines	307
	5.3	FIFO networks of Kelly type	310
	5.4	Global stability	318
	5.5	Relationship between QN and FM stability	325
Acknowledgments			336
Bibliography			
In		<b>342</b>	

# Chapter 1

# Introduction

Queueing networks constitute a large family of models in a variety of settings, involving "jobs" or "customers" that wait in queues until being served. Once its service is completed, a job moves to the next prescribed queue, where it remains until being served. This procedure continues until the job leaves the network; jobs also enter the network according to some assigned rule.

In these lectures, we will study the evolution of such networks and address the question: When is a network *stable*? That is, when is the underlying Markov process of the queueing network positive Harris recurrent? When the state space is countable and all states communicate, this is equivalent to the Markov process being positive recurrent. An important theme, in these lectures, is the application of fluid models, which may be thought of as being, in a general sense, dynamical systems that are associated with the networks.

The goal of this chapter is to provide a quick introduction to queueing networks. We will provide basic vocabulary and attempt to explain some of the concepts that will motivate later chapters. The chapter is organized as follows. In Section 1.1, we discuss the M/M/1 queue, which is the "simplest" queueing network. It consists of a single queue, where jobs enter according to a Poisson process and have exponentially distributed service times. The problem of stability is not difficult to resolve in this setting.

Using M/M/1 queues as motivation, we proceed to more general queueing networks in Section 1.2. We introduce many of the basic concepts of queueing networks, such as the discipline (or policy) of a network determining which jobs are served first, and the traffic intensity  $\rho$  of a network, which provides a natural condition for deciding its stability. In Section 1.3, we provide a preliminary description of fluid models, and how they can be applied to provide conditions for the stability of queueing networks.

In Section 1.4, we summarize the topics we will cover in the remaining chapters. These include the product representation of the stationary distributions of certain classical queueing networks in Chapter 2, and examples of unstable queueing networks in Chapter 3. Chapters 4 and 5 introduce fluid models and apply them to obtain criteria for the stability of queueing networks.

### 1.1. The M/M/1 queue

The M/M/1 queue, or simple queue, is the most basic example of a queueing network. It is familiar to most probabilists and is simple to analyze. We therefore begin with a summary of some of its basic properties to motivate more general networks.

The setup consists of a server at a workstation, and "jobs" (or "customers") who line up at the server until they are served, one by one. After service of a job is completed, it leaves the system. The jobs are assumed to arrive at the station according to a Poisson process with intensity 1; equivalently, the interarrival times of succeeding jobs are given by independent rate -1 exponentially distributed random variables. The service times of jobs are given by independent rate- $\mu$  exponentially distributed random variables, with  $\mu > 0$ ; the mean service time of jobs is therefore  $m = 1/\mu$ . We are interested here in the behavior of Z(t), the number of jobs in the queue at time t, including the job currently being served (see Figure 1.1).

The process  $Z(\cdot)$  can be interpreted in several ways. Because of the independent exponentially distributed interarrival and service times,  $Z(\cdot)$  defines a Markov process, with states  $0, 1, 2, \ldots, (M/M/1 \text{ stands for Markov input})$  and Markov output, with one server.) It is also a birth and death process on  $\{0, 1, 2, \ldots\}$ , with birth rate 1 and death rate  $\mu$ . Because of the latter interpretation, it is easy to compute the stationary (or invariant) probability measure  $\pi_m$  of  $Z(\cdot)$  when it exists, since the process will be reversible. Such a measure satisfies

$$\pi_m(n+1) = m\pi_m(n)$$
 for  $n = 0, 1, 2, \dots$ ,

since it is constant, over time, on the intervals [0, n] and  $[n + 1, \infty)$ . It follows that when m < 1,  $\pi_m$  is geometrically distributed, with

$$\pi_m(n) = (1-m)m^n, \quad n = 0, 1, 2, \dots$$
 (1.1)

All states clearly communicate with one another, and the process  $Z(\cdot)$  is positive recurrent. The mean of  $\pi_m$  is  $m(1-m)^{-1}$ , which blows up as  $m \uparrow 1$ . When  $m \ge 1$ , no stationary probability measure exists for  $Z(\cdot)$ . Using standard reasoning, one can show that  $Z(\cdot)$  is null recurrent when m = 1 and is transient when m > 1.

The behavior of  $Z(\cdot)$  that was observed in the last paragraph provides the basic motivation for these lectures, in the context of the more general queueing



FIG. 1.1. Jobs enter the system at rate 1 and depart at rate  $\mu$ . There are currently 2 jobs in the queue.

networks which will be introduced in the next section. We will investigate when the Markov process corresponding to a queueing network is stable, i.e., is positive Harris recurrent. As mentioned earlier, this is equivalent to positive recurrence when the state space is countable and all states communicate.

For M/M/1 queues, we explicitly constructed a stationary probability measure to demonstrate positive recurrence of the Markov process. Typically, however, such a measure will not be explicitly computable, since it will not be reversible. This, in particular, necessitates a new, more qualitative, approach for showing positive recurrence. We will present such an approach in Chapter 4.

### 1.2. Basic concepts of queueing networks

The M/M/1 queue admits natural generalizations in a number of directions. It is unnecessary to assume that the interarrival and service distributions are exponential. For general distributions, one employs the notation G/G/1; or M/G/1or G/M/1, if one of the distributions is exponential. (To emphasize the independence of the corresponding random variables, one often uses the notation GI instead of G.)

The single queue can be extended to a finite system of queues, or a *queueing network* (for short, *network*), where jobs, upon leaving a queue, line up at another queue, or *station*, or leave the system. The queueing network in Figure 1.2 is also a *reentrant line*, since all jobs follow a fixed route.

Depending on a job's previous history, one may wish to prescribe different service distributions at its current station or different routing to the next station. This is done by assigning one or more *classes*, or *buffers*, to each station. Except when stated otherwise, we label stations by  $j = 1, \ldots, J$  and classes by  $k = 1, \ldots, K$ ; we use C(j) to denote the set of classes belonging to station j and s(k) to denote the station to which class k belongs. In Figure 1.3, there are 3 stations and 5 classes. Classes are labelled here in the order they occur along the route, with  $C(1) = \{1, 5\}, C(2) = \{2, 4\}, \text{ and } C(3) = \{3\}.$ 

Other examples of queueing networks are given in Figures 1.4 and 1.5. Figure 1.4 depicts a network with 2 stations, each possessing 2 classes. The network is not a reentrant line but still exhibits *deterministic routing*, since each job entering the network at a given class follows a fixed route. When the individual



FIG. 1.2. A reentrant line with 3 stations.



FIG. 1.3. A reentrant line with 3 stations and 5 classes. The stations are labelled by j = 1, 2, 3; the classes are labelled by  $k = 1, \ldots 5$ , in the order they occur along the route.



FIG. 1.4. A queueing network having 3 routes, with 2 stations and 4 classes.

routes are longer, it is sometimes more convenient to replace the above labelling of classes by (i, k), where *i* gives the route that is followed and *k* the order of the class along the route.

Figure 1.5 depicts a network with 2 stations and 3 classes. The routing at class 2 is random, with each job served there having probability 0.4 of being sent to class 1 and probability 0.6 of being set to class 3. For queueing networks in general, we will assume that the interarrival times, service times, and routing of jobs at each class are given by sequences of i.i.d. random variables that are mutually independent (but whose distributions may depend on the class).

Queueing networks occur naturally in a wide variety of settings. "Jobs" can be interpreted as products of some sort of complex manufacturing process with multiple steps, as tasks that need to be performed by a computer or communication system, or as people moving about through a bureaucratic maze. Such networks can be quite complicated. A portion of the procedure in the manufacture of a semiconductor wafer is depicted by the reentrant line in Figure 1.6. Another simplified example, with classes emphasized, is given in Figure 1.7. Typically, such procedures can require hundreds of steps.



FIG. 1.5. A queueing network with 2 stations and 3 classes. The random routing at class 2 is labelled with the probability each route is taken.



FIG. 1.6. Part of the procedure in the manufacture of a semiconductor wafer. The procedure starts at the top and ends at the bottom. (Example courtesy of P. R. Kumar.)



FIG. 1.7. Part of the procedure in the manufacture of another semiconductor wafer. (Example courtesy of J. G. Dai.)

We will say that a queueing network is *multiclass* when at least one station has more than one class; otherwise, the network is *single class*. The term *Jackson network* is frequently used for a single class network with exponentially distributed interarrival and service times, and *generalized Jackson network* is used for a single class network with arbitrary distributions. (The networks in Figures 1.3–1.7 are all multiclass; the networks in Figures 1.1–1.2 are single class.) Unless otherwise specified, it will be assumed that there is a single server at each station *j*. This server will be assumed to be *non-idling* (or *work conserving*), that is, it remains busy as long as there are jobs present at any  $k \in C(j)$ . Stations are assumed to have infinite capacity, with jobs never being turned away. (That is, there is no limit to the allowed length of queues.)

As already indicated, the interarrival times, service times, and routing of jobs are given by sequences of independent random variables. Although their specific distributions will be relevant for some matters, their means will be of much greater importance. We therefore introduce here the following notation and terminology for the means; systematic notation for the random variables will be introduced in Chapter 4.

We denote by  $\alpha_k$  the rate at which jobs arrive at a class k from outside the network. When  $k \in \mathcal{A}$ , the subset of classes where external arrivals are allowed,  $\alpha_k$  is the reciprocal of the corresponding mean interarrival time; when  $k \notin \mathcal{A}$ , we set  $\alpha_k = 0$ . The vector  $\alpha = \{\alpha_k, k = 1, \ldots, K\}$  is referred to as the *external arrival rate*. (Throughout the lectures, we interpret vectors as column vectors unless stated otherwise.) We denote by  $m_k, m_k > 0$ , the *mean service time* at class k, and by M the diagonal matrix with  $m_k, k = 1, \ldots, K$ , as its entries; we set  $\mu_k = 1/m_k$ , which is the *service rate*. We also denote by  $P = \{P_{k,\ell}, k, \ell = 1, \ldots, K\}$  the *mean transition matrix* (or *mean routing matrix*), where  $P_{k,\ell}$  is the probability a job departing from class k goes to class  $\ell$ . (In Figure 1.5,  $P_{2,1} = 0.4$  and  $P_{2,3} = 0.6$ .) In these lectures, we will be interested in *open queueing networks*, that is, those networks for which the matrix

$$Q \stackrel{\text{def}}{=} (I - P^T)^{-1} = I + P^T + (P^T)^2 + \cdots$$
(1.2)

is finite. (" $^{T}$ " denotes the transpose.) This means that jobs at any class are capable of ultimately leaving the network. *Closed queueing networks*, where neither arrivals to, nor departures from, the network are permitted, are not discussed here, although there are certain similarities between the two cases.

#### Disciplines for multiclass queueing networks

We have so far neglected an important aspect of multiclass queueing networks. Namely, when more than one job is present at a station, what determines the order in which jobs are served? There are numerous possibilities. As we will see in these lectures, the choice of the service rule, or *discipline* (also known as *policy*), can have a major impact on the evolution of the queueing network.

Perhaps the most natural discipline is *first-in*, *first-out* (FIFO), where the first job to arrive at a station (or "oldest" job) receives all of the service, irrespective of its class. (If the job later returns to the station, it starts over again as the "youngest" job. The jobs originally at a class are assigned some arbitrary order.) Another widely used discipline is *processor sharing* (PS), where the service at a station is simultaneously equally divided between all jobs presently there. PS can be thought of as a limiting *round robin* discipline, where a server alternates among the jobs currently present at the station, dispensing a fixed small amount of service until it moves on to the next job.

The FIFO and PS disciplines are egalitarian in the sense that no job, because of its class, is favored over another. The opposite is the case for *static buffer priority* (SBP) disciplines, where classes are assigned a strict ranking, and jobs of higher ranked (or priority) classes are always served before jobs of lower ranked classes, irrespective of when they arrived at the station. Within a class, the jobs are served in the order of their arrival there. The disciplines are called *preemptive resume*, or more simply, *preemptive*, if arriving higher ranked jobs interrupt lower ranked jobs currently in service; when the service of such higher ranked jobs has been completed, service of lower ranked jobs continues where it left off. If the service of lower ranked jobs is not interrupted, then the discipline is *nonpreemptive*.

For reentrant lines, two natural SBP disciplines are *first-buffer-first-served* (FBFS) and *last-buffer-first-served* (LBFS). For FBFS, jobs at earlier classes along the route have priority over later classes. For example, jobs in class 1, in the network given in Figure 1.3, have priority over those in class 5, and jobs in class 2 have priority over those in class 4. For LBFS, the priority is reversed, with jobs at later classes along the route having priority over earlier classes.

In these lectures, we will concentrate on *head-of-the-line* (HL) disciplines, where only the first job in each class of a station may receive service. (This property is frequently referred to as FIFO within a class.) FIFO and SBP disciplines are HL, but PS is not. The single class networks we consider will always

be assumed to be HL unless explicitly stated otherwise. A major simplifying feature of HL disciplines is that the total amount of "work" (or "effort") already devoted to partially served jobs does not build up. Since large numbers of jobs will therefore tend not to complete service at a station at close times, this avoids "bursts" of jobs from being suddenly routed elsewhere in the network.

Another non-HL discipline is *last-in*, *first-out* (LIFO), where the last job to arrive at a station is served first. Other plausible non-HL disciplines from a somewhat different setting are *first-in-system*, *first-out* (FISFO) and *last-in-system*, *first-out* (LISFO). FISFO is the same as FIFO, except that the first job to enter the queueing network (rather than the station) has priority over other jobs; the discipline may be preemptive or nonpreemptive. LISFO gives priority to the last job to enter the network.

Throughout these lectures, the term queueing network will indicate that a discipline, such as FIFO or FBFS, has already been assigned to the system. In much of the literature, the discipline is specified afterwards. This linguistic difference does not affect the theory, of course.

For the M/M/1 queue in the previous section, the interarrival and service times were assumed to be exponentially distributed. As a consequence, the process  $Z(\cdot)$  counting the number of jobs is Markov. For queueing networks with exponentially distributed interarrival and service times, the situation is often analogous. For instance, for preemptive SBP disciplines, the vector valued process  $Z(t) = \{Z_k(t), k = 1, \ldots, K\}$  counting the number of jobs at each class is Markov; the same is true for PS. For the FIFO and LIFO disciplines, the process will be Markov if one appends additional information giving the order in which each job entered the class. In all of these cases, the state space is countable, so one can apply standard Markov chain theory. We will denote by  $X(\cdot)$  the corresponding Markov processes.

The situation becomes more complicated when the interarrival and service times are not exponentially distributed, since the residual interarrival and service times need to be appended to the state in order for the process to be Markov. The resulting state space is uncountable, and so a more general theory, involving Harris recurrence, is required for the corresponding Markov process. We will give a careful construction of such processes and will summarize the needed results on positive Harris recurrence at the beginning of Chapter 4. We avoid this issue until then, since the material in Chapters 2 and 3 typically does not involve Markov processes on an uncountable state space. (The sole exception is the uncountable state space extension for symmetric processes at the end of Section 2.4.) In a first reading of the lectures, not too much will be lost by assuming the interarrival and service times are exponentially distributed.

### Traffic intensity, criticality, and stability

The mean quantities  $\alpha$ , M, and P introduced earlier perform an important role in determining the long-time behavior of a queueing network. To provide motivation, we first consider reentrant lines.

The (long term) rate at which jobs enter a reentrant line is  $\alpha_1$ , and the mean time required to serve a job over all of its visits to a station j is  $\sum_{k \in \mathcal{C}(j)} m_k$ . So, the rate at which future "work" for the station j enters the network is

$$\rho_j = \alpha_1 \sum_{k \in \mathcal{C}(j)} m_k. \tag{1.3}$$

We recall that a queueing network is defined to be stable when its underlying Markov process is positive Harris recurrent. When the state space is countable and all states communicate, this is equivalent to the Markov process being positive recurrent. It is intuitively clear that, in order for a reentrant line to be stable,  $\rho_j \leq 1$ , for all j, is necessary. Otherwise, by the law of large numbers, the work in the system, corresponding to j, will increase linearly to infinity as  $t \to \infty$ , and so the same should be true for the total number of jobs in the system. (When the state space is countable, it follows from standard Markov chain theory that a stable network must be empty a fixed fraction of the time, as  $t \to \infty$ , from which it follows that, in fact,  $\rho_j < 1$  must hold for all j.)

A natural condition for stability (assuming all states communicate), is that  $\rho_j < 1$  for all j. This turns out, in fact, not to be sufficient for multiclass queueing networks, as we will see in Chapter 3. Not surprisingly, the much stronger condition

$$\sum_{j} \rho_j < 1 \tag{1.4}$$

suffices for stability, since whenever the network is not empty, the total work in the network tends to decrease faster than it increases. (This will follow from Proposition 4.5.)

The situation is analogous for queueing networks with general routing, after the correct quantities have been introduced. We set

$$\lambda = Q\alpha, \tag{1.5}$$

and refer to the vector  $\lambda$  as the *total arrival rate*. (Recall that vectors are to be interpreted as column vectors.) Its components  $\lambda_k$ ,  $k = 1, \ldots, K$ , are the rates at which jobs enter the K classes; they each equal  $\alpha_1$  for reentrant lines. The *traffic equations* 

$$\lambda_{\ell} = \alpha_{\ell} + \sum_{k} \lambda_{k} P_{k,\ell}, \qquad (1.6)$$

or, in vector form,  $\lambda = \alpha + P^T \lambda$ , are equivalent to (1.5), and are useful in certain situations. Employing *m* and  $\lambda$ , we define the *traffic intensity*  $\rho_j$  at station *j* to be

$$\rho_j = \sum_{k \in \mathcal{C}(j)} m_k \lambda_k. \tag{1.7}$$

 $(\rho_j \text{ is also known as the nominal load.})$  The traffic intensity is the rate at which work for the station j enters the network, and reduces to (1.3) for reentrant lines. We write  $\rho$  for the corresponding traffic intensity vector.

We continue the analogy with reentrant lines, and say that a station j is subcritical if  $\rho_j < 1$ , critical if  $\rho_j = 1$ , and supercritical if  $\rho_j > 1$ . When all stations of a network are either subcritical or are critical, we refer to the network as being subcritical or critical. We will sometimes abbreviate these conditions by writing  $\rho < e$  and  $\rho = e$ , where  $e = (1, \ldots, 1)^T$ ; we similarly write  $\rho \leq e$ , when  $\rho_j \leq 1$  for each j. When at least one station is supercritical, we refer to the network as being supercritical. As in the reentrant line setting, a supercritical queueing network will not be stable, and the number of jobs in the network will increase linearly as  $t \to \infty$ . This is a bit tedious to show directly; it will follow quickly using fluid limits in Proposition 5.21. Of course, since a reentrant line with  $\rho < e$  need not be stable, the same is the case for queueing networks with general routing. We will show in Chapters 4 and 5, though, that the condition  $\rho < e$  is sufficient for the stability of queueing networks under various disciplines. Also, (1.4) suffices for stability irrespective of the discipline; a variant of this is shown in Example 1 at the end of Section 4.4.

In Chapters 4 and 5, we will also establish criteria for when a queueing network is *e-stable*. By this, we will mean that the underlying Markov process is ergodic, i.e., it possesses a stationary distribution  $\pi$  to which the distribution at time *t* converges in total variation norm as  $t \to \infty$ , irrespective of the initial state. When the state space is countable, this is equivalent to the probabilities converging to  $\pi(x)$  at each point *x*. As we will see, results on stability can be modified to results on *e*-stability with little additional work.

We have so far not used the term "unstable". Somewhat different definitions exist in the literature; in each case, they mean more than just "not stable". For us, a queueing network will be *unstable* if, for some initial state, the number of jobs in the network will, with positive probability, go to infinity as  $t \to \infty$ . (When the network has only a finite number of states with fewer than a given number of jobs, and all states communicate with one another, this is equivalent to saying that for each initial state, the number of jobs in the network goes to infinity almost surely as  $t \to \infty$ .) According to the paragraph before the last, a supercritical queueing network will be unstable; in fact, the number of jobs in the network grows linearly in time. This will typically be the case for unstable networks, such as those given in Chapter 3.

#### 1.3. Queueing network equations and fluid models

One of the main themes in these lectures will be the application of fluid models to study the stability of queueing networks. Fluid models will be introduced and studied in detail in Chapter 4; they will then by applied to specific disciplines in Chapter 5. We provide here some of the basic motivation for fluid models.

In the last section, we gave various examples of queueing networks. Such systems are frequently complicated. They can be interpreted as Markov processes, but to derive specific results, one needs a means of expressing the properties of the specific queueing network. *Queueing network equations* provide an analytic formulation for this. After deriving these equations and taking appropriate limits, one obtains the corresponding fluid models. Queueing network equations tie together random vectors, such as A(t), D(t), T(t), and Z(t), that describe the evolution of a queueing network. The individual coordinates of these K dimensional vectors correspond, respectively, to the cumulative number of arrivals  $A_k(t)$  and departures  $D_k(t)$  at a class k up to time t, the cumulative time  $T_k(t)$  spent serving this class up to time t, and the number of jobs  $Z_k(t)$  at this class at time t; we introduced Z(t) in the last section. Examples of queueing network equations are

$$A(t) = E(t) + \sum_{k} \Phi^{k}(D_{k}(t)), \qquad (1.8)$$

$$Z(t) = Z(0) + A(t) - D(t),$$
(1.9)

$$D_k(t) = S_k(T_k(t)), \quad k = 1, \dots, K.$$
 (1.10)

We are employing here the following terminology. The vector E(t) is the cumulative number of jobs arriving by time t at each class from outside the network (i.e., *external arrivals*). When the interarrival times are exponentially distributed,  $E(\cdot)$  will be a Poisson process. The vector  $\Phi^k(d_k)$  is the number of the first  $d_k$  departures from class k that are routed to each of the K classes;  $S_k(t_k)$  is the cumulative number of departures from class k after  $t_k$  units of service there. We will denote by  $\Phi$  and S the matrix and vector corresponding to these quantities.

The quantities  $E(\cdot)$ ,  $S(\cdot)$ , and  $\Phi(\cdot)$  should be thought of as random, but known input into the system, from which the evolution of  $A(\cdot)$ ,  $D(\cdot)$ ,  $T(\cdot)$ , and  $Z(\cdot)$  will be determined via (1.8)–(1.10) and other equations. The middle equation is easiest to read, and just says that the number of jobs at time t is equal to the original number, plus arrivals, and minus departures. The first equation says the total number of arrivals is equal to the number of external arrivals plus the number of arrivals from other classes; the last equation gives the number of departures at a class as a function of the time spent serving jobs there. We note that the first two equations hold irrespective of the discipline, whereas the last equation requires the discipline to be HL.

Other choices of variables, in addition to  $A(\cdot), D(\cdot), T(\cdot)$  and  $Z(\cdot)$ , are frequently made. We will include other variables, such as the immediate workload  $W(\cdot)$ , in our detailed treatment in Section 4.3. Often, different formulations of the queueing network equations are equivalent, with the exact format being chosen for convenience. One can, for example, eliminate A(t) and D(t) in (1.8)–(1.10), and instead employ the single equation

$$Z(t) = Z(0) + E(t) + \sum_{k} \Phi^{k}(S_{k}(T_{k}(t))) - S(T(t)), \qquad (1.11)$$

if one is just interested in the evolution of Z(t) (which is most often the case). Note that, for multiclass networks, neither (1.8)-(1.10) nor (1.11) supplies enough information to solve for the unknown variables, since the discipline has not been specified, and so  $T(\cdot)$  has not been uniquely determined. To specify the discipline, an additional equation (or equations) is required. For single class networks, this complication is not present. As an elementary example, note that for the M/M/1 queue, (1.11) reduces to the simple scalar equation

$$Z(t) = Z(0) + E(t) - S\left(\int_0^t \mathbf{1}\left\{Z(s) > 0\right\} ds\right),$$
(1.12)

since departing jobs are not rerouted into the queue and the network is nonidling.

Fluid model equations are the deterministic analog of queueing network equations, with one replacing the random quantities  $E(\cdot), S(\cdot)$ , and  $\Phi(\cdot)$  by their corresponding means. The fluid model equations corresponding to (1.8)-(1.10)are then

$$A(t) = \alpha t + P^T D(t), \qquad (1.13)$$

$$Z(t) = Z(0) + A(t) - D(t), (1.14)$$

$$D_k(t) = \mu_k T_k(t), \quad k = 1, \dots, K,$$
 (1.15)

where  $\alpha, P$ , and  $\mu_k$  were defined in the previous section. By employing the matrix M, one can also write (1.15) as

$$D(t) = M^{-1}T(t). (1.15')$$

Similarly, the fluid model equation corresponding to (1.11) is

$$Z(t) = Z(0) + \alpha t + (P^T - I)M^{-1}T(t).$$
(1.16)

For the M/M/1 queue, this reduces to the scalar equation

$$Z(t) = Z(0) + \alpha t - \mu \int_0^t \mathbf{1} \{Z(s) > 0\} \, ds.$$
(1.17)

Fluid model equations can be thought of as belonging to a *fluid network* which is the deterministic equivalent of the given queueing network. Jobs are replaced by continuous fluid mass (or "job mass"), which follows the same routing as before. The constant rate at which such mass enters the network is given by the vector  $\alpha$ . The rate at which mass is served for a class is  $\mu_k$  and the service time per unit mass is  $m_k = 1/\mu_k$ . A set of fluid model equations, as in (1.13)–(1.15), is referred to collectively as a *fluid model*.

The solutions of fluid models are frequently much easier to analyze than are the corresponding queueing network equations. As we will see in Chapter 4, fluid model equations can be derived from the corresponding queueing network equations by taking limits of  $Z(\cdot)$  and the other quantities after hydrodynamic scaling. (That is, "law of large numbers" scaling of the form Z(st)/s as  $s \to \infty$ .) Fluid models will provide a valuable tool for demonstrating the stability of queueing networks, as we will see in Chapters 4 and 5. Fluid models are also an important tool for studying *heavy traffic limits*, which lie outside the scope of these lectures. To analyze the stability of a queueing network, one introduces the following notion of stability for a fluid model. A fluid model is said to be *stable* if there exists an N > 0, so that for any solution of the fluid model equations, its  $Z(\cdot)$  component satisfies

$$Z(t) = 0 \quad \text{for } t \ge N |Z(0)|.$$
 (1.18)

(Here,  $|\cdot|$  denotes the  $\ell^1$  norm.) We will show in Chapter 4 that, under certain conditions on the interarrival and service times, a queueing network will be stable if its fluid model is stable.

In the special case of the fluid model equation (1.17) corresponding to the M/M/1 queue, it is easy to explicitly solve for  $Z(\cdot)$ . For  $\mu > \alpha$ , the solution drifts linearly to 0 at rate  $\mu - \alpha$ , after which it remains there, and so

$$Z(t) = 0 \quad \text{for } t \ge Z(0)/(\mu - \alpha),$$
 (1.19)

which is a special case of (1.18). This behavior of the solution of equation (1.17) corresponding to a subcritical M/M/1 queue is not surprising, since the solution  $Z(\cdot)$  of (1.12) possesses the same negative drift  $\alpha - \mu$  when Z(t) > 0. For more general networks, one typically constructs a Lyapunov function with respect to which the fluid model solution  $Z(\cdot)$  of (1.16) exhibits a uniformly negative drift until hitting 0.

Despite the utility of fluid models, one needs to exercise some caution in their application. In particular, a fluid model need not have a unique solution for a given initial condition, since solutions might bifurcate. As we will see in Section 4.3, this can be the case even for certain standard disciplines. (Such behavior might occur at times when there are two or more empty multiclass stations.) On account of this, thinking of fluid model equations as belonging to a fluid network loses some of its appeal. In practice, it is often better to think directly in terms of the fluid model which is defined by the appropriate system of equations.

### 1.4. Outline of lectures

We conclude the introduction with an outline of the subject matter we will be covering. The material can be broken into three parts, Chapter 2, Chapter 3, and Chapters 4 and 5, each with its own distinct character.

Chapter 2 discusses the "classical" queueing networks introduced in [BaCMP75] and the accompanying theory of quasi-reversible queueing networks in [Ke79]. The examples considered in [BaCMP75] include networks with the FIFO, PS, and LIFO disciplines that were introduced in Section 1.2, and an infinite server network, which we will introduce in Chapter 2. Exponentially distributed interarrival times, and in some cases, exponentially distributed service times, are assumed. For  $\rho < e$ , these networks are stable and explicit product-like formulas are given for their stationary distributions. This special structure is a generalization of that for the M/M/1 queue. Independent work of F. P. Kelly, leading to the book [Ke79], employed quasi-reversibility to show

that these explicit formulas hold in a more general setting. These results have strongly influenced the development of queueing theory over the past several decades.

In the previous sections, we mentioned that even when  $\rho < e$ , a queueing network may be unstable. This behavior came as a surprise in the early 1990's. Various examples of instability have since been given in different settings, primarily in the context of SBP and FIFO disciplines. At this point, there is no comprehensive theory, and in fact, not much is known, in general, about how the Markov processes for such networks go to infinity as  $t \to \infty$ . Chapter 3 presents the best known examples of unstable subcritical queueing networks in more-orless chronological order, with an attempt being made to provide some feeling for the development of the subject. Examples include those from [LuK91], [RyS92], [Br94a], [Se94], and [Du97].

In Chapter 4, we give general sufficient conditions for the stability of a queueing network. The main condition is that the fluid model of the queueing network be stable; general conditions on the interarrival and service times of the queueing network are also needed. In the previous section, we gave a brief discussion of how the fluid model equations are obtained from the queueing network equations that describe the evolution of the queueing network. The material in Chapter 4 is largely based on [Da95] and the sources employed there. We go into considerable detail on the arguments leading to the main result, Theorem 4.16, because we feel that it is important to have this material accessible in one place.

The first part of Chapter 5 consists of applications of Theorem 4.16, where stability is demonstrated, under  $\rho < e$ , for a number of disciplines. In Sections 5.1, 5.2, and 5.3, the stability of single class networks, SBP reentrant lines with FBFS and LBFS priority schemes, and FIFO networks, with constant mean service times at a station, are demonstrated. In each case, the procedure is to demonstrate the stability of a fluid model; the stability of the queueing network then follows by applying the above theorem.

Sections 5.4 and 5.5 are different in nature from the previous three sections. Section 5.4 is concerned with the question of global stability. That is, when is a queueing network stable, irrespective of the particular discipline that is applied? Again applying Theorem 4.16, a queueing network will be globally stable if its fluid model is. For two-station fluid models with deterministic routing, a complete theory is given. Section 5.5 investigates the converse of Theorem 4.16, namely the necessity of fluid model stability for the stability of a given queueing network. It turns out that there is not an exact correspondence between the two types of stability, as examples show. Robust conditions for the necessity of fluid model stability are presently lacking. The material in Chapter 5 is taken from a member of sources, including [Br96a], [DaWe96], [Br99], [DaV00], and [DaHV04].

We conclude this section with some basic notation and conventions. We let  $\mathbf{Z}_+$  and  $\mathbf{R}_+$  denote the positive integers and real numbers;  $\mathbf{Z}_{+,0}$  and  $\mathbf{R}_{+,0}$  will denote the sets appended with  $\{0\}$ ; and  $\mathbf{Z}_+^d$ ,  $\mathbf{R}_+^d$ ,  $\mathbf{Z}_{+,0}^d$ , and  $\mathbf{R}_{+,0}^d$  will denote their d dimensional equivalents. For  $x, y \in \mathbf{R}^d$ ,  $x \leq y$  means  $x_k \leq y_k$  for

each coordinate k; x < y means  $x_k < y_k$  for each coordinate. Unless indicated otherwise,  $|\cdot|$  denotes the  $\ell^1$  (or sum) norm, e.g.,  $|x| = \sum_{i=1}^d |x_i|$  for  $x \in \mathbf{R}^d$ . By  $a \lor b$  and  $a \land b$ , we mean max $\{a, b\}$  and min $\{a, b\}$ . For  $x \in \mathbf{R}$ , by  $\lfloor x \rfloor$  and  $\lceil x \rceil$ , we mean the integer part of x and the smallest integer that is at least x. By  $a(t) \sim b(t)$ , we mean  $a(t)/b(t) \to 1$  as  $t \to \infty$ , and by  $a \approx b$ , that a and b are approximately the same in some weaker sense.

When convenient, we will refer to a countable state Markov process as a Markov chain. (In the literature, Markov chain often instead refers to a discrete time Markov process.) We say that a Markov chain is positive recurrent if each state is positive recurrent and all states communicate. (The second condition is sometimes not assumed in the literature.) As already mentioned, instead of the term "queueing network", we will often employ the shorter "network", when the context is clear. Throughout these lectures, continuous time stochastic processes will be assumed to be right continuous unless indicated otherwise.

### Chapter 2

# The classical networks

In this chapter, we discuss two families of queueing networks whose Markov processes are positive recurrent when  $\rho < e$ , and whose stationary distributions have explicit product-like formulas. The first family includes networks with the FIFO discipline, and the second family includes networks with the PS and LIFO disciplines, as well as infinite server (IS) networks. We introduced the first three networks in Section 1.2; we will define infinite server networks shortly. These four networks are sometimes known as the "classical networks". Together with their generalizations, they have had a major influence on the development of queueing theory because of the explicit nature of their stationary distributions. For this reason, we present the basic results for the accompanying theory here, although only the FIFO discipline is HL. These results are primarily from [BaCMP75], and papers by F. P. Kelly (such as [Ke75] and [Ke76]) that led to the book [Ke79].

In Section 2.1, we state the main results in the context of the above four networks. We first characterize the stationary distributions for networks consisting of a single station, whose jobs exit from the network when service is completed, without being routed to another class. We will refer to such a station as a *node*. We then characterize the stationary distribution for networks with multiple stations and general routing. Since all states will communicate, the Markov processes for the networks will be positive recurrent, and hence the networks will be stable.

In the remainder of the chapter, we present the background for these results and the accompanying theory. In Section 2.2, we give certain basic properties of stationary and reversible Markov processes on countable state spaces that we will use later. Sections 2.3 and 2.4 apply this material to obtain generalizations of the node-level results in Section 2.1 to the two families of interest to us. The first family, homogeneous nodes, includes FIFO nodes under certain restrictions, and the second family, symmetric nodes, includes PS, LIFO, and IS nodes.

The concept of quasi-reversibility is introduced in Section 2.5. Using quasireversibility, the stationary distributions of certain queueing networks can be written as the product of the stationary distributions of nodes that correspond to the stations "in isolation". These queueing networks include the FIFO, PS, LIFO, and IS networks, and so generalize the network-level results in Section 2.1. Except for Theorem 2.9, all of the material in this chapter is for queueing networks with a countable state space.

The main source of the material in this chapter is [Ke79]. Section 2.2 is essentially an abridged version of the material in Chapter 1 of [Ke79]. Most of the material in Sections 2.3–2.5 is from Sections 3.1–3.3 of [Ke79], with [Wa88], [ChY01], [As03], and lecture notes by J.M. Harrison having also been consulted. The order of presentation here, starting with nodes in Sections 2.3 and 2.4 and ending with quasi-stationarity in Section 2.5, is different.

### 2.1. Main results

In this section, we will give explicit formulas for the stationary distributions of FIFO, PS, LIFO, and infinite server networks. Theorems 2.1 and 2.2 state these results for individual nodes, and Theorem 2.3 does so for networks. In Sections 2.3–2.5, we will prove generalizations of these results.

The range of disciplines that we consider here is of limited scope. On the other hand, the routing that is allowed for the network-level results will be completely general. As in Chapter 1, routing will be given by a mean transition matrix  $P = \{P_{k,\ell}, k, \ell = 1, \ldots, K\}$  for which the network is open.<sup>1</sup> For all queueing networks considered in this section, the interarrival times are assumed to be exponentially distributed. When the service times are also exponentially distributed, the evolution of these queueing networks can be expressed in terms of a countable state Markov process. By enriching the state space, more general service times can also be considered in the countable state space setting. This will be useful for the PS, LIFO, and IS networks.

In Section 1.1, we introduced the M/M/1 queue with external arrival rate  $\alpha = 1$ . By employing the reversibility of its Markov process when m < 1, we saw that its stationary distribution is given by the geometric distribution in (1.1). Allowing  $\alpha$  to be arbitrary with  $\alpha m < 1$ , this generalizes to

$$\pi(n) = (1 - \alpha m)(\alpha m)^n \quad \text{for } n = 0, 1, 2, \dots$$
(2.1)

For a surprisingly large group of queueing networks, generalizations of (2.1) hold, with the stationary distribution being given by products of terms similar to those on the right side of (2.1).

We first consider queueing networks consisting of just a single node. That is, jobs at the unique station leave the network immediately upon completion of service, without being routed to other classes. Classes are labelled  $k = 1, \ldots, K$ .

 $<sup>^{1}</sup>$ In the literature (such as in [Ke79]), deterministic routing is frequently employed. For these lectures, we prefer to use random routing, which was used in [BaCMP75] and has been promulgated in its current form by J. M. Harrison. By employing sufficiently many routes, one can show the two approaches are equivalent. We find the approach with random routing to be notationally more flexible. The formulation is also more amenable to problems involving dynamic scheduling, which we do not cover here.

The nodes of interest to us in this chapter fall into two basic families, depending on the discipline.

The first family, homogeneous nodes, will be defined in Section 2.3. FIFO nodes, which are the canonical example for this family, will be considered here. For homogeneous nodes, including FIFO nodes, we need to assume that the mean service times  $m_k$  at all classes k are equal. In order to avoid confusion with the vector m, we label such a service time by  $m^s$  (with "s" standing for "station"). We will refer to such a node as a *FIFO node of Kelly type*. In addition to assuming the interarrival times are exponentially distributed, we assume the same is true for the service times.

The state x of the node at any time will be specified by an n-tuple of the form

$$(x(1), \dots, x(n)),$$
 (2.2a)

where n is the number of jobs in the node and

$$x(i) \in \{1, \dots, K\}$$
 for  $i = 1, \dots, n$  (2.2b)

gives the class of the job in the  $i^{\text{th}}$  position in the node. We interpret  $i = 1, \ldots, n$ as giving the order of arrival of the jobs currently in the node; because of the FIFO discipline, all service is directed to the job at i = 1. The state space  $S_0$  will be the union of these states. The stochastic process  $X(t), t \ge 0$ , thus defined will be Markov with a countable state space. For consistency with other chapters, we interpret vectors as column vectors, although this is not needed in the present chapter (since matrix multiplication is not employed).

All states communicate with the empty state. So, if  $X(\cdot)$  has a stationary distribution, it will be unique. Since there is only a single station, a stationary distribution will exist when the node is subcritical. Theorem 2.1 below gives an explicit formula for the distribution. As in Chapter 1,  $\alpha_k$  denotes the external arrival rates at the different classes k;  $\rho$  denotes the traffic intensity and is in the present setting given by the scalar

$$\rho = m^s \sum_k \alpha_k. \tag{2.3}$$

As elsewhere in this chapter, when we say that a node (or queueing network) has a stationary distribution, we mean that its Markov process, on the chosen state space, has this distribution. (For us, "distribution" is synonymous with the somewhat longer "probability measure".)

**Theorem 2.1.** Each subcritical FIFO node of Kelly type has a stationary distribution  $\pi$ , which is given by

$$\pi(x) = (1 - \rho) \prod_{i=1}^{n} m^{s} \alpha_{x(i)}, \qquad (2.4)$$

for  $x = (x(1), \dots, x(n)) \in S_0$ .

The stationary distribution  $\pi$  in Theorem 2.1 can be described as follows. The probability of there being a total of n jobs in the node is  $(1 - \rho)\rho^n$ . Given a total of n jobs, the probability of there being  $n_1, \ldots, n_K$  jobs at the classes  $1, \ldots, K$ , with  $n = n_1 + \cdots + n_K$  and no attention being paid to their order, is

$$\rho^{-n} \begin{pmatrix} n \\ n_1, \dots, n_K \end{pmatrix} \prod_{k=1}^K (m^s \alpha_k)^{n_k}.$$
 (2.5)

Moreover, given that there are  $n_1, \ldots, n_K$  jobs at the classes  $1, \ldots, K$ , any ordering of the different classes of jobs is equally likely. Note that since all states have positive probability of occurring, the process  $X(\cdot)$  is positive recurrent. Consequently, the node is stable.

The other family of nodes that will be discussed in this chapter, symmetric nodes, will be defined in Section 2.4. Standard members of this family are PS, LIFO, and IS nodes. The PS and LIFO disciplines were specified in Chapter 1. In an *infinite server* (IS) node, each job is assumed to start receiving service as soon as it enters the node, which it receives at rate 1. One can therefore think of there being an infinite number of unoccupied servers available to provide service, one of which is selected whenever a job enters the node. All other disciplines studied in these lectures will have only a single server at a given station. We note that although the PS discipline is not HL, it is related to the HLPPS discipline given at the end of Section 5.3.

We consider the stationary distributions of PS, LIFO and IS nodes. As with FIFO nodes, we need to assume that the interarrival times of jobs are exponentially distributed. If we assume that the service times are also exponentially distributed, then the process  $X(\cdot)$  defined on  $S_0$  will be Markov. As before, we interpret the coordinates  $i = 1, \ldots, n$  in (2.2) as giving the order of jobs currently in the node. For LIFO nodes, this is also the order of arrival of jobs there. For reasons relating to the definition of symmetric nodes in Section 2.4, we will instead assume, for PS and IS nodes, that arriving jobs are, with equal probability 1/n, placed at one of the n positions of the node, where n is the number of jobs present after the arrival of the job. Since in both cases, jobs are served at the same rate irrespective of their position in the node, the processes  $X(\cdot)$  defined in this manner are equivalent to the processes defined by jobs always arriving at the rear of the node.

The analog of Theorem 2.1 holds for PS, LIFO, and IS nodes when the service times are exponentially distributed. We no longer need to assume that the service times have the same means, so in the present setting, the traffic intensity  $\rho$  is given by

$$\rho = \sum_{k} m_k \alpha_k. \tag{2.6}$$

For subcritical PS and LIFO nodes, the stationary distribution  $\pi$  is given by

$$\pi(x) = (1-\rho) \prod_{i=1}^{n} m_{x(i)} \alpha_{x(i)}, \qquad (2.7)$$

for  $x = (x(1), \ldots, x(n)) \in S_0$ . For any IS node, the stationary distribution  $\pi$  is given by

$$\pi(x) = \frac{e^{-\rho}}{n!} \prod_{i=1}^{n} m_{x(i)} \alpha_{x(i)}.$$
(2.8)

The stability of the infinite server node for all values of  $\rho$  is not surprising, since the total rate of service at the node is proportional to the number of jobs presently there.

For PS, LIFO, and IS nodes, an analogous result still holds when exponential service times are replaced by service times with more general distributions. One employs the "method of stages", which is defined in Section 2.4. One enriches the state space  $S_0$  to allow for different stages of service for each job, with a job advancing to its next stage of service after service at the previous stage has been completed. After service at the last stage has been completed, the job leaves the node. Since the service times at each stage are assumed to be exponentially distributed, the corresponding process  $X(\cdot)$  for the node, on this enriched state space  $S_e$ , will still be Markov. On the other hand, the total service time required by a given job will be the sum of the exponential service times at its different stages, which we take to be i.i.d. Such service times are said to have *Erlang distributions*.

Using the method of stages, one can extend the formulas (2.7) and (2.8), for the stationary distributions of PS, LIFO, and IS nodes, to nodes that have service distributions which are countable mixtures of Erlang distributions. This result is stated in Theorem 2.2. The state space here for the Markov process  $X(\cdot)$ of the node is  $S_e$ , which is defined in Section 2.4. The analog of this extension for FIFO nodes is not valid.

**Theorem 2.2.** Each subcritical PS node and LIFO node, whose service time distributions are mixtures of Erlang distributions, has a stationary distribution  $\pi$ . The probability of there being n jobs in the node with classes  $x(1), \ldots, x(n)$  is given by (2.7). The same is true for any IS node with these service time distributions, but with (2.8) replacing (2.7).

Mixtures of Erlang distributions are dense in the set of distribution functions, so it is suggestive that a result analogous to Theorem 2.2 should hold for service times with arbitrary distributions. This is in fact the case, although one needs to be more careful here, since one needs to replace  $S_e$  with an uncountable state space which specifies the residual service times of jobs at the node. More detail on this setting is given at the end of Section 2.4. Because of the technical difficulties for uncountable state spaces, (2.7) and (2.8) are typically stated for mixtures of Erlang distributions or other related distributions on countable state spaces. Moreover, quasi-reversibility, which we discuss shortly, employs a countable state space setting.

So far in this section, we have restricted our attention to nodes. As mentioned at the beginning of the section, the results in Theorems 2.1 and 2.2 extend to analogous results for queueing networks, which are given in Theorem 2.3, below. FIFO, PS, LIFO, and IS queueing networks are the analogs of the respective

nodes, with jobs at individual stations being subjected to the same service rules as before, and, upon completion of service at a class k, a job returning to class  $\ell$  with probability  $P_{k,\ell}$ , which is given by the mean transition matrix P. In addition to applying to these queueing networks, Theorem 2.3 also applies to networks that are mixtures of such stations, with one of the above four rules holding for any particular station. We also note that the formula in Theorem 2.3 holds for Jackson networks, as a special case of FIFO networks. The product formula for Jackson networks in [Ja63] predates those for the other networks.

In Theorem 2.3, we assume that the service times are exponentially distributed when the station is FIFO, and are mixtures of Erlang distributions in the other three cases. The distribution function  $\pi$  is defined on the state space

$$S = S^1 \times \dots \times S^J,$$

where  $S^j = S_0$  if j is FIFO and  $S^j = S_e$  for the other cases, and  $\pi^j$  is defined on  $S^j$ . (The choice of K in each factor depends on  $S^j$ .)

**Theorem 2.3.** Suppose that each station j of a queueing network is either FIFO of Kelly type, PS, LIFO, or IS. Suppose that in the first three cases, the station is subcritical. Then, the queueing network has a stationary distribution  $\pi$  that is given by

$$\pi(x) = \prod_{j=1}^{J} \pi^{j}(x^{j}), \qquad (2.9)$$

for  $x = (x^1, \ldots, x^J)$ . Here, each  $\pi^j$  is either of the form (2.4), (2.7), or (2.8), depending on whether the station j is FIFO of Kelly type, PS or LIFO, or IS, and  $\alpha_k$  in the formulas is replaced by  $\lambda_k$ .

Theorem 2.3 will be a consequence of Theorems 2.1 and 2.2, and of the quasi-reversibility of the nodes there. Quasi-reversibility will be introduced in Section 2.5. Using quasi-reversibility, it will be shown, in Theorem 2.11, that the stationary distributions of certain queueing networks can be written as the product of the stationary distributions of nodes that correspond to the individual stations "in isolation". This will mean that when service of a job at a class is completed, the job will leave the network rather than returning to another class (either at the same or a different station). The external arrival rates  $\alpha_k$  at classes are replaced by the total arrival rates  $\lambda_k$  of the network in order to compensate for the loss of jobs that would return to them. Quasi-reversibility can be applied to queueing networks whose stations are FIFO, PS, LIFO, or IS. By employing Theorem 2.11, one obtains Theorem 2.3 as a special case of Theorem 2.12.

### 2.2. Stationarity and reversibility

In this section, we will summarize certain basic results for countable state, continuous time Markov processes. We define stationarity and reversibility, and

provide alternative characterizations. Proposition 2.6, in particular, will be used in the remainder of the chapter.

The Markov processes X(t),  $t \ge 0$ , we consider here will be assumed to be defined on a countable state space S. The space S will be assumed to be irreducible, that is, all states communicate. None of the states will be instantaneous; we will assume there are only a finite number of transitions after a finite time, and hence no explosions. Sample paths will therefore be right continuous with left limits. The transition rate between states x and y will be denoted by q(x, y); the rate at which a transition occurs at x is therefore  $q(x) \stackrel{\text{def}}{=} \sum_{y \in S} q(x, y)$ . The *embedded jump chain* has mean transition matrix  $\{p(x,y), x, y \in S\}$ , where  $p(x, y) \stackrel{\text{def}}{=} q(x, y)/q(x)$  is the probability  $X(\cdot)$  next visits y from the state x. The time  $X(\cdot)$  remains at a state x before a transition occurs is exponentially distributed with mean 1/q(x).

A stochastic process X(t),  $t \ge 0$ , is said to be *stationary* if  $(X(t_1), \ldots, X(t_n))$ has the same distribution as  $(X(t_1 + u), \ldots, X(t_n + u))$ , for each nonnegative  $t_1, \ldots, t_n$  and u. Such a process can always be extended to  $-\infty < t < \infty$  so that it is stationary as before, but with  $t_1, \ldots, t_n$  and u now being allowed to assume any real values. When  $X(\cdot)$  is a Markov process, it suffices to consider just n = 1 in order to verify stationarity.

A stationary distribution  $\pi = \{\pi(x), x \in S\}$  for a Markov process  $X(\cdot)$  satisfies the balance equations

$$\pi(x) \sum_{y \in S} q(x, y) = \sum_{y \in S} \pi(y) q(y, x) \text{ for } x \in S,$$
(2.10)

which say that the rates at which mass leaves and enters a state x are the same. We are assuming here that  $\sum_{x \in S} \pi(x) = 1$ . Since all states are assumed to communicate,  $\pi$  will be unique. If  $\pi$  exists, then it is the limit of the distributions of the Markov process starting from any initial state. If a measure  $\pi$  satisfying (2.10) with  $\sum_{x \in S} \pi(x) < \infty$  exists, then it can be normalized so that  $\sum_{x \in S} \pi(x) = 1$ . If  $\sum_{x \in S} \pi(x) = \infty$ , then there is no stationary distribution, and for all  $x, y \in S$ ,

$$P(X(t) = \ell \mid X(0) = x) \to 0 \quad \text{as } t \to \infty.$$

(See, e.g., [Re92] for such basic theory.)

A stochastic process  $X(t), -\infty < t < \infty$ , is said to be *reversible* if  $(X(t_1), \ldots, X(t_n))$  has the same distribution as  $(X(u-t_1), \ldots, X(u-t_n))$ , for each  $t_1, \ldots, t_n$  and u. This condition says that the process is stochastically indistinguishable, whether it is run forward or backwards in time. It is easy to see that if  $X(\cdot)$  is reversible, then it must be stationary. When  $X(\cdot)$  is Markov, it suffices to consider n = 2 in order to verify reversibility. The Markov property can be formulated as saying that the past and future states of the process  $X(\cdot)$  are independent given the present. It follows that the *reversed process*  $\hat{X}(t) \stackrel{\text{def}}{=} X(-t)$  is Markov exactly when  $X(\cdot)$  is. If  $X(\cdot)$  has stationary measure  $\pi$ , then  $\pi$  is also stationary

for  $\hat{X}(\cdot)$  and the transition rates of  $\hat{X}(\cdot)$  are given by

$$\hat{q}(x,y) = \frac{\pi(y)}{\pi(x)}q(y,x) \text{ for } x, y \in S.$$
 (2.11)

A stationary Markov process  $X(\cdot)$  with distribution  $\pi$  is reversible exactly when it satisfies the *detailed balance equations* 

$$\pi(x)q(x,y) = \pi(y)q(y,x) \quad \text{for } x, y \in S.$$

$$(2.12)$$

This condition says that the rate at which mass moves from x to y is the same rate at which it moves in the reverse direction. This condition need not, of course, be satisfied for arbitrary stationary distributions. When (2.12) holds, it often enables one to express the stationary distribution in closed form, as, for example, for the M/M/1 queue in Section 1.1. It will always hold for the stationary distribution of any birth and death process, and, more generally, for the stationary distribution of any Markov process on a tree. By summing over  $\ell$ , one obtains the balance equations in (2.10) from (2.12).

An alternative characterization of reversibility is given by Proposition 2.4. We will not employ the proposition elsewhere, but state it because it provides useful intuition for the concept.

**Proposition 2.4.** A stationary Markov process is reversible if and only if its transition rates satisfy

$$q(x_1, x_2)q(x_2, x_3) \cdots q(x_{n-1}, x_n)q(x_n, x_1) = q(x_1, x_n)q(x_n, x_{n-1}) \cdots q(x_3, x_2)q(x_2, x_1),$$
(2.13)

for any  $x_1, x_2, ..., x_n$ .

The equality (2.13) says that the joint transition rates of the Markov process are the same along a path if it starts and ends at the same point, irrespective of its direction along the path.

*Proof of Proposition* 2.4. The "only if" direction follows immediately by plugging (2.12) into (2.13).

For the "if" direction, fix  $x_0$ , and define

$$\pi(x) = \prod_{i=1}^{n} [q(x_{i-1}, x_i)/q(x_i, x_{i-1})], \qquad (2.14)$$

where  $x_n = x$  and  $x_0, x_1, \ldots, x_n$  is any path from  $x_0$  to x, with  $q(x_i, x_{i-1}) > 0$ . One can check using (2.13) that the right side of (2.14) does not depend on the particular path that is chosen, and so  $\pi(x)$  is well defined. To see this, let  $\mathscr{P}_1$ and  $\mathscr{P}_2$  be any two paths from  $x_0$  to x, and  $\hat{\mathscr{P}}_1$  and  $\hat{\mathscr{P}}_2$  be the corresponding paths in the reverse directions. Then, the two paths from  $x_0$  to itself formed by linking  $\mathscr{P}_1$  to  $\hat{\mathscr{P}}_2$ , respectively  $\mathscr{P}_2$  to  $\hat{\mathscr{P}}_1$ , satisfy (2.13), from whence the uniqueness in (2.14) will follow. Assume that for given y, q(y, x) > 0. Multiplication of both sides of (2.14) by q(x, y) implies that

$$\pi(x)q(x,y) = q(y,x) \left(\prod_{i=1}^{n} q(x_{i-1},x_i)/q(x_i,x_{i-1})\right) (q(x,y)/q(y,x))$$
  
=  $q(y,x)\pi(y)$ .

This gives (2.12), since the case q(x, y) = q(y, x) = 0 is trivial. Since the process is assumed to be stationary,  $\sum_x \pi(x) < \infty$ , and so  $\pi$  can be scaled so that  $\sum_x \pi(x) = 1$ .

The following result says that under certain modifications of the transition rates q(x, y), a reversible Markov process will still be reversible. It will not be needed for later work, but has interesting applications.

**Proposition 2.5.** Suppose that the transition rates of a reversible Markov process  $X(\cdot)$ , with state space S and stationary distribution  $\pi$ , are altered by changing q(x, y) to q'(x, y) = bq(x, y) when  $x \in A$  and  $y \notin A$ , for some  $A \subseteq S$ . Then, the resulting Markov process  $X'(\cdot)$  is reversible and has stationary distribution

$$\pi'(x) = \begin{cases} c\pi(x) & \text{for } x \in A, \\ cb\pi(x) & \text{for } x \notin A, \end{cases}$$
(2.15)

where c is chosen so that  $\sum_{x} \pi'(x) = 1$ . In particular, when the state space is restricted to A by setting b = 0, then the stationary distribution of  $X'(\cdot)$  is given by

$$\pi'(x) = \pi(x) \Big/ \sum_{y \in A} \pi(y) \quad \text{for } x \in A.$$
(2.16)

*Proof.* It is easy to check that q' and  $\pi'$  satisfy the detailed balance equations in (2.12).

The following illustration of Proposition 2.5 is given in [Ke79].

**Example 1.** Two queues with a joint waiting room. Suppose that two independent M/M/1 queues are given, with external arrival rates  $\alpha_i$  and mean service times  $m_i$ , and  $\alpha_i m_i < 1$ . Let  $X_i(t)$  be the number of customers (or jobs) in each queue at time t. The Markov processes are each reversible with stationary distributions as in (2.1). It is easy to check that the joint Markov process  $X(t) = (X_1(t), X_2(t)), -\infty < t < \infty$ , is reversible, with stationary distribution

$$\pi(n_1, n_2) = (1 - \alpha_1 m_2)(1 - \alpha_2 m_2)(\alpha_1 m_1)^{n_1} (\alpha_2 m_2)^{n_2} \quad \text{for } n_i \in \mathbf{Z}_{+,0}.$$

Suppose now that the queues are required to share a common waiting room of size N, so that a customer who arrives to find N customers already there leaves without being served. This corresponds to restricting  $X(\cdot)$  to the set A of states with  $n_1 + n_2 \leq N$ . By Proposition 2.5, the corresponding process  $X'(\cdot)$ is reversible, and has stationary measure

$$\pi'(n_1, n_2) = \pi(0, 0)(\alpha_1 m_1)^{n_1} (\alpha_2 m_2)^{n_2}$$
 for  $(n_1, n_2) \in A$ .

It is often tedious to check the balance equations (2.10) in order to determine that a Markov process  $X(\cdot)$  is stationary. Proposition 2.6 gives the following alternative formulation. We abbreviate by setting

$$q(x) = \sum_{y \in S} q(x, y), \quad \hat{q}(x) = \sum_{y \in S} \hat{q}(x, y), \quad (2.17)$$

where q(x, y) are the transition rates for  $X(\cdot)$  and  $\hat{q}(x, y) \geq 0$  are for the moment arbitrary. When  $X(\cdot)$  has stationary distribution  $\pi$  and  $\hat{q}(x, y)$  is given by (2.11), it is easy to check that

$$q(x) = \hat{q}(x) \quad \text{for all } x. \tag{2.18}$$

The proposition gives a converse to this. As elsewhere in this section, we are assuming that S is irreducible.

**Proposition 2.6.** Let X(t),  $-\infty < t < \infty$ , be a Markov process with transition rates  $\{q(x, y), x, y \in S\}$ . Suppose that for given quantities  $\{\hat{q}(x, y), x, y \in S\}$  and  $\{\pi(x), x \in S\}$ , with  $\hat{q}(x, y) \ge 0$ ,  $\pi(x) > 0$ , and  $\sum_x \pi(x) = 1$ , that q,  $\hat{q}$ , and  $\pi$  satisfy (2.11) and (2.18). Then,  $\pi$  is the stationary distribution of  $X(\cdot)$  and  $\hat{q}$  gives the transition rates of the reversed process.

*Proof.* It follows, by applying (2.11) and then (2.18), that

$$\sum_{x \in S} \pi(x)q(x,y) = \pi(y) \sum_{x \in S} \hat{q}(y,x) = \pi(y)\hat{q}(y) = \pi(y)q(y).$$

So,  $\pi$  is stationary for  $X(\cdot)$ . The transition rates of the reversed process are therefore given by (2.11).

Proposition 2.6 simplifies the computations needed for the demonstration of stationarity by replacing the balance equations, that involve a large sum and the stationary distribution  $\pi$ , by two simpler equations, (2.18), which involves just a large sum, and (2.11), which involves just  $\pi$ . On the other hand, the application of Proposition 2.6 typically involves guessing  $\hat{q}$  and  $\pi$ . In situations where certain choices suggest themselves, the proposition can be quite useful. It will be used repeatedly in the remainder of the chapter.

### 2.3. Homogeneous nodes of Kelly type

FIFO nodes of Kelly type belong to a larger family of nodes whose stationary distributions have similar properties. We will refer to such a node as a *homogeneous node of Kelly type*. Such nodes are defined as follows.

Consider a node with K classes. The state  $x \in S_0$  of the node at any time is specified by an n-tuple as in (2.2), when there are n jobs present at the node. The ordering of the jobs is assumed to remain fixed between arrivals and service completions of jobs. When the job in position *i* completes its service, the position of the job is filled with the jobs in positions  $i + 1, \ldots, n$  moving up to positions  $i, \ldots, n-1$ , while retaining their previous order. Similarly, when a job arrives at the node, it is assigned some position i, with jobs previously at positions  $i, \ldots, n$  being moved back to positions  $i+1, \ldots, n+1$ . Each job requires a given random amount of service; when this is attained, the job leaves the node. As throughout this chapter, interarrival times are required to be exponentially distributed. As elsewhere in these lectures, all interarrival and service times are assumed to be independent.

We will say that such a node is a *homogeneous node* if it also satisfies the following properties:

- (a) The amount of service required by each job is exponentially distributed with mean  $m_k$ , where k is the class of the job.
- (b) The total rate of service supplied at the node is  $\phi(n)$ , where n is the number of jobs currently there.
- (c) The proportion of service that is directed at the job in position i is  $\delta(i, n)$ . Note that this proportion does not depend on the class of the job.
- (d) When a job arrives at the node, it moves into position i, i = 1, ..., n, with probability  $\beta(i, n)$ , where n is the number of jobs in the node including this job. Note that this probability does not depend on the class of the job.

When the mean  $m_k$  does not depend on the class k, we will say that such a node is a *homogeneous node of Kelly type*. We will analyze these nodes in this section. We use  $m^s$  when the mean service times of a node are constant, as we did in Section 2.1.

The rate at which service is directed to a job in position i is  $\delta(i, n)\phi(n)$ . So, the rate at which service at the job is completed is  $\delta(i, n)\phi(n)/m^s$ . We will assume that  $\phi(n) > 0$ , except when n = 0. The rate at which a job arrives at a class k and position i from outside the node is  $\alpha_k\beta(i, n)$ . We use here the mneumonics  $\beta$  and  $\delta$  to suggest births and deaths at a node. Of course,  $\sum_{i=1}^{n} \beta(i, n) = \sum_{i=1}^{n} \delta(i, n) = 1$ .

We have emphasized in the above definition that the external arrival rates and service rates  $\beta$  and  $\delta$  do not depend on the class of the job. This is crucial for Theorem 2.7, the main result in this section. This restriction will also be needed in Section 2.4 for symmetric nodes, as will be our assumption that the interarrival times are exponentially distributed. On the other hand, the assumptions that the service times be exponential and that their means  $m_k$  be constant, which are needed in this section, are not needed for symmetric nodes. We note that by scaling time by  $1/m^s$ , one can set  $m^s = 1$ , although we prefer the more general setup for comparison with symmetric nodes and for application in Section 2.5.

In Section 2.5, we will be interested in homogeneous queueing networks of Kelly type. *Homogeneous queueing networks* are defined analogously to homogeneous nodes. Jobs enter the network independently at the different stations according to exponentially distributed random variables, and are assigned positions at these stations as in (d). Jobs at different stations are served independent.

dently, as in (a)–(c), with departing jobs from class k being routed to class  $\ell$ with probability  $P_{k,\ell}$  and leaving the network with probability  $1 - \Sigma_{\ell} P_{k,\ell}$ . Jobs arriving at a class  $\ell$  from within the network are assigned positions as in (d), according to the same rule as was applied for external arrivals. The external arrival rates and the quantities in (a)–(d) are allowed to depend on the station. When the mean service times  $m_k$  are assumed to depend only on the station j = s(k), we may write  $m_j^s$ ; we refer to such networks as homogeneous queueing networks of Kelly type.

The most important examples of homogeneous nodes are FIFO nodes. Here, one sets  $\phi(n) \equiv 1$ ,

$$\beta(i,n) = \begin{cases} 1 & \text{for } i = n, \\ 0 & \text{otherwise,} \end{cases}$$

and

$$\delta(i,n) = \begin{cases} 1 & \text{for } i = 1, \\ 0 & \text{otherwise,} \end{cases}$$

for  $n \in \mathbb{Z}_+$ . That is, arriving jobs are always placed at the end of the queue and only the job at the front of the queue is served. Another example is given in [Ke79], where arriving jobs are again placed at the end of the queue, but where L servers are available to serve the first L jobs, for given L. In this setting,  $\phi(n) = L \wedge n, \beta$  is defined as above, and

$$\delta(i,n) = \begin{cases} 1/n & \text{for } i \le n \le L, \\ 1/L & \text{for } i \le L < n, \\ 0 & \text{for } i > L. \end{cases}$$

The main result in this section is Theorem 2.7, which is a generalization of Theorem 2.1. Since the total service rate  $\phi$  that is provided at the node can vary, the condition

$$B \stackrel{\text{def}}{=} \sum_{n=0}^{\infty} \left( \rho^n \middle/ \prod_{i=1}^n \phi(i) \right) < \infty$$
(2.19)

replaces the assumption in Theorem 2.1 that the node is subcritical. Here,  $\rho = m^s \sum_{k=1}^{K} \alpha_k$  is the traffic intensity.

**Theorem 2.7.** Suppose that a homogeneous node of Kelly type satisfies  $B < \infty$  in (2.19). Then, it has a stationary distribution  $\pi$  that is given by

$$\pi(x) = B^{-1} \prod_{i=1}^{n} (m^{s} \alpha_{x(i)} / \phi(i)), \qquad (2.20)$$

for  $x = (x(1), \dots, x(n)) \in S_0$ .

As was the case in Theorem 2.1, the structure of the stationary distribution  $\pi$  in Theorem 2.7 exhibits independence at multiple levels. The probability of there being a total of n jobs at the node is  $\rho^n / B \prod_{i=1}^n \phi(i)$ . Given a total of n jobs

at the node, the probability of there being  $n_1, \ldots, n_k$  jobs of classes  $1, \ldots, K$ , respectively, is given by the multinomial distribution in (2.5). Moreover, the ordering of the different classes of jobs is equally likely. As was the case in Theorem 2.1, all states communicate with the empty state, and the Markov process  $X(\cdot)$  for the node is positive recurrent.

### Demonstration of Theorem 2.7

~

The proof of Theorem 2.7 that we will give is based on Proposition 2.6. In order to employ the proposition, we need to choose quantities  $\hat{q}$  and  $\pi$  so that (2.11) and (2.18) are satisfied for them and the transition rates q of the Markov process  $X(\cdot)$  for the node. It will then follow from Proposition 2.6 that  $\pi$  is the stationary distribution for the node and  $\hat{q}$  gives the transition rates for the reversed process  $\hat{X}(\cdot)$ . A similar argument will be used again for symmetric nodes in Section 2.4 and for networks consisting of quasi-reversible nodes in Section 2.5. We will summarize a more probabilistic argument for Theorem 2.7 at the end of the section.

In order to demonstrate Theorem 2.7, we write q and our choices for  $\hat{q}$  and  $\pi$  explicitly in terms of  $\alpha$ ,  $\beta$ ,  $\delta$ , and  $\phi$ . To be able to reuse this argument in Section 2.4 for symmetric nodes, we write  $m_k$  for the mean service times, which in the present case reduces to the constant  $m^s$ .

The nonzero transition rates q(x, y) take on two forms, depending on whether the state is obtained from x by the arrival or exit of a job. In the former case, we write  $y = a_{k,i}(x)$  if a class k job arrives at position i; in the latter case, it follows that  $x = a_{k,i}(y)$ , if i is the position of the exiting class k job. One then has

$$q(x,y) = \begin{cases} \alpha_k \beta(i, n_y) & \text{for } y = a_{k,i}(x), \\ m_k^{-1} \delta(i, n_x) \phi(n_x) & \text{for } x = a_{k,i}(y), \end{cases}$$
(2.21)

where  $n_x$  and  $n_y$  are the number of jobs at the node for states x and y.

Finding the transition rates  $\hat{q}(x, y)$  of the reversed process involves some guessing, motivated by our idea of what  $\hat{X}(\cdot)$  should look like. It is reasonable to guess that  $\hat{X}(\cdot)$  is also the Markov process for a homogeneous node. The external arrival rates  $\alpha_k$  and  $\hat{\alpha}_k$  will then be the same for both processes, since arrivals for  $\hat{X}(\cdot)$  correspond to exits for  $X(\cdot)$ , and under the stationary distribution  $\pi$ , the two rates must be the same. The mean service times  $m_k$  and  $\hat{m}_k$  will also be the same. It is reasonable to guess that  $\hat{\phi}(n) = \phi(n)$ ; this would be the case if  $X(\cdot)$  were reversible, as it is for the M/M/1 queue. Running  $X(\cdot)$  backwards in time mentally, it is also tempting to set

$$\beta(i,n) = \delta(i,n), \ \delta(i,n) = \beta(i,n) \text{ for } n \in \mathbf{Z}_+, \ i \le n.$$

For instance, if the original node is FIFO, then jobs arrive at i = n and exit at i = 1; if the node is run backwards in time, jobs arrive at i = 1 and exit at i = n. Substitution of these choices for  $\alpha$ ,  $\beta$ ,  $\delta$ ,  $\phi$ , and m in (2.21) yields

$$\hat{q}(x,y) = \begin{cases} \alpha_k \delta(i, n_y) & \text{for } y = a_{k,i}(x), \\ m_k^{-1} \beta(i, n_x) \phi(n_x) & \text{for } x = a_{k,i}(y). \end{cases}$$
(2.22)

We still need to choose our candidate for the stationary distribution  $\pi$  of both  $X(\cdot)$  and  $\hat{X}(\cdot)$ . The equality (2.11), that is needed for Proposition 2.6, is equivalent to

$$\pi(x)q(x,y) = \pi(y)\hat{q}(y,x)$$
(2.23)

holding whenever  $y = a_{k,i}(x)$  or  $x = a_{k,i}(y)$ . When  $y = a_{k,i}(x)$  and q(x, y) > 0, substitution of (2.21) and (2.22) into (2.23) implies that

$$\pi(y)/\pi(x) = m_k \alpha_k / \phi(n_y). \tag{2.24}$$

The case  $x = a_{k,i}(y)$  yields the same equality, but with the roles of x and y reversed. Reasoning backwards, it is not difficult to see that (2.23) also follows from (2.24).

Set  $x = (x(1), \ldots, x(n))$ , for  $n \in \mathbb{Z}_{+,0}$ . One can repeatedly apply (2.24) by removing jobs from x one at a time, starting from the last, until the empty state is reached. We therefore choose  $\pi$  so that

$$\pi(x) = B^{-1} \prod_{i=1}^{n} (m_{x(i)} \alpha_{x(i)} / \phi(i)), \qquad (2.25)$$

where the normalizing constant  $B = 1/\pi(\emptyset)$ . Under (2.25), (2.24) must hold. We have therefore verified (2.11) for this choice of  $\pi$ . In particular, this holds for  $m_k \equiv m^s$ , as in Theorem 2.7.

In order to employ Proposition 2.6, we also need to verify (2.18). One can check that

$$\sum_{y} q(x,y) = \sum_{k} \alpha_{k} + \phi(n_{x}) \sum_{i} m_{x(i)}^{-1} \delta(i,n_{x}).$$
(2.26)

The first sum on the right side of (2.26) follows by summing the top line of (2.21) over all *i* and *k*. The relationship  $x = a_{k,i}(y)$  implies that x(i) = k, and so the last sum in (2.26) follows by summing the last line of (2.21) over all *i*. Using the same reasoning, one obtains the formula

$$\sum_{y} \hat{q}(x,y) = \sum_{k} \alpha_{k} + \phi(n_{x}) \sum_{i} m_{x(i)}^{-1} \beta(i,n_{x})$$
(2.27)

from (2.22). As before, the first sum on the right side is obtained from arriving jobs and the last sum is obtained from exiting jobs.

We see from (2.26) and (2.27) that a sufficient condition for (2.18) is that

$$\sum_{i} m_{x(i)}^{-1} \delta(i, n_x) = \sum_{i} m_{x(i)}^{-1} \beta(i, n_x).$$
(2.28)

In Theorem 2.7,  $m_k \equiv m^s$ , which factors outside of the sum on both sides of (2.28). Since the resulting sums both equal 1, (2.28), and hence (2.18), holds in this setting. Note that this is the only point in the argument at which we need  $m_k$  to be constant.

We have shown that both (2.11) and (2.18) are satisfied for q and q' given by (2.21) and (2.22), and  $\pi$  given by (2.25), under the assumptions in Theorem

2.7. It therefore follows from Proposition 2.6 that  $\pi$  is the stationary distribution for the Markov process with transition rates q. This implies Theorem 2.7.

### Some observations

One can generalize the above proof of Theorem 2.7 so that it applies to homogeneous queueing networks of Kelly type, rather than to just homogeneous nodes of Kelly type as in the theorem. Then, the stationary distribution  $\pi$  can be written as the product of stationary distributions  $\pi^j$  of nodes corresponding to the individual stations, when they operate "in isolation". This is done in Section 3.1 of [Ke79]. We prefer to postpone the treatment of homogeneous networks until Section 2.5, where they are considered within the context of quasi-reversibility.

One can give a more probabilistic proof of Theorem 2.7 that is based on the following intuitive argument. The rates  $\beta(i, n)$ ,  $\delta(i, n)$  and  $\phi(n)$  governing the arrival and service rates of jobs, as well as the mean service time  $m^s$ , do not distinguish between classes of jobs. Jobs are therefore served as they would be for an M/M/1 queue modified to have the total rate of service  $\phi(n)$ , when there are n jobs, and having the arrival rate  $\sum_k \alpha_k$ . By randomly choosing the class of each job, with probability  $\alpha_k / \sum_k \alpha_k$  for each k, at either time 0 or at some later time t, the distributions at time t of the two resulting processes will be the same. If the stationary distribution for the modified M/M/1 queue is chosen as its initial distribution, the resulting distribution  $\pi'$  for the K classes will therefore also be stationary.

One can show, by using reversibility, that the probability of there being n jobs for the stationary distribution of the modified M/M/1 queue is  $\rho^n/B \prod_{i=1}^n \phi(i)$ , where B is as in (2.19). Because of the random way in which the classes of jobs are chosen above for  $\pi'$ , the remaining properties in the alternative characterization of  $\pi$  in (2.20), that are given after the statement of Theorem 2.7, also hold. Therefore,  $\pi' = \pi$ , as desired.

We also note the following consequence of the proof of Theorem 2.7, that is a special case of phenomena that will be discussed in Section 2.5. (It also follows from the alternative argument that was sketched above.) The transition rates  $\hat{q}$  of the reversed Markov process  $\hat{X}(\cdot)$  in (2.22) are the rates for a homogeneous node of Kelly type. For this reversed node, arrivals are therefore given by K independent Poisson processes for the different classes, that are independent of the initial state. These arrivals correspond to exiting jobs for the original homogeneous node. It follows that the K different exit processes for the classes are also independent Poisson processes that are independent of any future state of the node.

### 2.4. Symmetric nodes

PS, LIFO, and IS nodes all belong to the family of symmetric nodes. They are defined similarly to the homogeneous nodes of Kelly type in the previous section, with a few major differences. The basic framework is the same, with state space  $S_0$  given by (2.2) and existing jobs being reordered as before upon the arrival and departure of jobs at the node. Moreover, the interarrival times are exponentially distributed.

In order for such a node to be a *symmetric node*, we require that it also satisfy the following properties:

- (a) The amount of service required by each job is exponentially distributed with mean  $m_k$ . We will soon allow more general distributions, but this will require us to extend the state space.
- (b) The total rate of service supplied at the node is  $\phi(n)$ , where n is the number of jobs currently there.
- (c) The proportion of service that is directed at the job in position i is  $\beta(i, n)$ . Note that the proportion does not depend on the class of the job.
- (d) When a job arrives at the node, it moves into position i, i = 1, ..., n, with probability  $\beta(i, n)$ , where n is the number of jobs in the node including this job. This function is the same as that given in (c).

In Section 2.5, we will also be interested in symmetric queueing networks. These networks are defined analogously, with properties (a)–(d) being assumed to hold at each station, and departing jobs from a class k being routed to a class  $\ell$  with probability  $P_{k,\ell}$ . More detail is given in Section 2.3 for homogeneous networks, where the procedure is the same.

The properties (a)–(d) given here are more restrictive than the properties (a)–(d) in the previous section in that we now assume, in the notation of Section 2.3, that  $\delta = \beta$ , in parts (c) and (d). These properties are more general in that the service time means  $m_k$  need no longer be equal at different classes. After comparing the stationary distributions of these nodes with those of Section 2.3, we proceed to generalize the exponential distributions of the service times in (a) in two steps, first to mixtures of Erlang distributions, and then to arbitrary distributions.

The PS, LIFO, and IS nodes are standard examples of symmetric nodes. For the PS discipline, one sets

$$\beta(i,n) = 1/n \text{ for } i \leq n,$$

for  $n \in \mathbf{Z}_+$ , and for LIFO, one sets

$$\beta(i,n) = \begin{cases} 1 & \text{for } i = n, \\ 0 & \text{otherwise.} \end{cases}$$

In both cases,  $\phi(n) \equiv 1$ . For IS nodes,  $\phi(n) = n$  and

$$\beta(i, n) = 1/n \text{ for } i \leq n.$$

The analog of Theorem 2.7 holds for symmetric nodes when  $B < \infty$ , for B in (2.19), with the formula for the stationary distribution  $\pi$ ,

$$\pi(x) = B^{-1} \prod_{i=1}^{n} (m_{x(i)} \alpha_{x(i)} / \phi(i)), \qquad (2.29)$$

for  $x = (x(1), \ldots, x(n))$ , replacing (2.20). One can check that the same argument as before is valid. One applies Proposition 2.6, for which one needs to verify (2.11) and (2.18). As before, the formulas for q and  $\hat{q}$  are given by (2.21) and (2.22), but with  $\delta = \beta$ . The formula for  $\pi$  is given by (2.25).

The argument we gave for (2.11) involved no restrictions on  $m_k$ , and therefore holds in the present context as well. The argument we gave for (2.18) consisted of equating (2.26) and (2.27), and therefore verifying (2.28). Previously, (2.28) held since  $m_k^{-1}$  was constant, and so could be factored out of both sums. It is now satisfied since  $\delta = \beta$ , and so the summands are identical. One can therefore apply Proposition 2.6, from which the analog of Theorem 2.7 for symmetric nodes follows, with (2.29) replacing (2.20).

### The method of stages

As mentioned earlier, the assumption that the service times be exponentially distributed is not necessary for symmetric nodes. By applying the *method of stages*, one can generalize the service time distributions to mixtures of Erlang distributions. (These are gamma distributions that are convolutions of identically distributed exponential distributions.) We will show that the analog of the formula (2.29) for the stationary distribution continues to hold in this more general setting.

In order for the stochastic process corresponding to the node to remain Markov for more general service times, we need to enrich the state space  $S_0$ . For this, we replace each coordinate x(i) in (2.2) by the triple  $\mathbf{x}(i) = (x(i), s(i), v(i))$ , where

$$x(i) \in \{1, \dots, K\}, \quad s(i) \in \mathbf{Z}_+, \quad v(i) \in \{1, \dots, s(i)\}.$$
 (2.30)

Such a triple gives the *refined class* of a job, with x(i) denoting its class as before. The third coordinate v(i) gives the current *stage* of a job, with s(i) denoting the total number of stages the job visits before leaving the node. The state space  $S_e$  will consist of such *n*-tuples  $x = (\mathbf{x}(1), \ldots, \mathbf{x}(n)), n \in \mathbf{Z}_{+,0}$ , under a mild restriction to ensure all states are accessible.

The basic dynamics of the node are the same as before, which satisfies the properties for symmetric nodes given at the beginning of the section, including properties (b)–(d); we are generalizing here the assumption in (a). Instead of entering the node at a class k with rate  $\alpha_k$ , jobs enter at a refined class (k, s, s) with rate  $\alpha_k p_k(s)$ , where  $\sum_s p_k(s) = 1$ . Once at a refined class (k, s, v), such a job moves to (k, s, v - 1) after completing its service requirement, which is exponentially distributed with mean  $m_k(s)$ . After a job completes its service at the stage v = 1, it leaves the node. The current stage v of a job can therefore be thought of as the residual number of stages remaining before the job leaves the node. We note that the proportion of service that is directed at a job in position i is  $\beta(i, n)$ , which does not depend on its class or refined class.

The distribution of the service time that is required for the job between entering and leaving class k is a mixture of Erlang distributions, and has mean

$$m_k \stackrel{\text{def}}{=} \sum_s sp_k(s)m_k(s). \tag{2.31}$$

The state space  $S_e$  mentioned earlier is defined to consist of *n*-tuples whose components (k, s, v) satisfy  $p_k(s) > 0$ , in order to exclude inaccessible states. Under this restriction, all states will communicate. The state space is of course countable. When  $p_k(1) = 1$  at all k, the service times are all exponentially distributed and the model reduces to the one considered at the beginning of the section. As with homogeneous and symmetric nodes with exponentially distributed service times, the networks corresponding to symmetric nodes with stages can be defined in the natural way.

We wish to show that the nodes just defined have stationary distributions  $\pi$  that generalize (2.29). This result is stated in Theorem 2.8.

**Theorem 2.8.** Suppose that the service times of a symmetric node are mixtures of Erlang distributions, and that the node satisfies  $B < \infty$  in (2.19). Then, the node has a stationary distribution  $\pi$  that is given by

$$\pi(x) = B^{-1} \prod_{i=1}^{n} (p_{x(i)}(s(i))m_{x(i)}(s(i))\alpha_{x(i)}/\phi(i)), \qquad (2.32)$$

where  $x = (\mathbf{x}(1), \dots, \mathbf{x}(n)) \in S_e$  and  $\mathbf{x}(i) = (x(i), s(i), v(i))$ , for  $i = 1, \dots, n$ .

As was the case in Theorem 2.7 and in (2.29), the structure of the stationary distribution  $\pi$  is Theorem 2.8 exhibits independence at multiple levels. The probability of there being a total of n jobs at the node is  $\rho^n / B \prod_{i=1}^n \phi(i)$ . Given a total of n jobs at the node, the probability of there being  $n_1, \ldots, n_k$  jobs at the classes  $1, \ldots, K$  is given by the multinomial

$$\rho^{-n} \begin{pmatrix} n \\ n_1, \dots, n_K \end{pmatrix} \prod_{k=1}^K (m_k \alpha_k)^{n_k}.$$
 (2.33)

The ordering of these classes is equally likely. Note that none of these quantities depends on the particular service time distributions, except for the means  $m_k$ .

The stationary distribution also has the following refined structure. Given the class of the job at each position i, the probability of the job at a given position, whose class is k, having refined class (k, x, v) is

$$p_k(s)m_k(s)/m_k,$$

and these events are independent at different i. Summing over all stages strictly greater than v and over all s, while keeping everything else fixed, this implies that the conditional probability of the job at position i being in a strictly earlier stage than v is

$$m_k^{-1} \sum_s (s-v) p_k(s) m_k(s).$$
 (2.34)

Note that (2.34) depends on the actual service time distributions, and not just on their means.

### Demonstration of Theorem 2.8

In order to demonstrate Theorem 2.8, we employ Proposition 2.6. To do so, we need to verify (2.11) and (2.18) for the transition rates q of the Markov process on  $S_e$  corresponding to the node, with an appropriate choice of the quantities  $\hat{q}$  and  $\pi$ .

In order to specify q and  $\hat{q}$ , we modify the function  $a_{k,i}(\cdot)$  we used for exponential service times on the state space  $S_0$ . Here,  $a_{k,s,i}(x)$  will denote the state y obtained from state x by the arrival of a job at position i, with refined class (k, s, s). Since jobs exit from the node at refined classes of the form (k, s, 1) (rather than at (k, s, s)), we need additional notation. With an eye on defining  $\hat{q}$ , we denote by  $\hat{a}_{k,s,i}(x)$  the state y that is obtained from x by inserting a job with refined class (k, s, 1) at i; the positions of jobs already at the node are shifted in the usual way. We also denote by  $\mathfrak{s}_i(x)$  the state y obtained from a state x satisfying  $2 \leq v(i) \leq s(i)$ , when the stage at i advances to v(i) - 1. ( $\mathfrak{s}_i(x)$  is not defined for other x.)

Using this notation, one can check that q is given by

$$q(x,y) = \begin{cases} \alpha_k p_k(s)\beta(i,n_y) & \text{for } y = a_{k,s,i}(x), \\ (m_k(s))^{-1}\beta(i,n_x)\phi(n_x) & \text{for } x = \hat{a}_{k,s,i}(y), \\ (m_k(s))^{-1}\beta(i,n_x)\phi(n_x) & \text{for } y = \mathfrak{s}_i(x), \end{cases}$$
(2.35)

with q(x, y) = 0 otherwise. Employing the same motivation as in Section 2.3, we choose  $\hat{q}$  so that

$$\hat{q}(x,y) = \begin{cases} \alpha_k p_k(s)\beta(i,n_y) & \text{for } y = \hat{a}_{k,s,i}(x), \\ (m_k(s))^{-1}\beta(i,n_x)\phi(n_x) & \text{for } x = a_{k,s,i}(y), \\ (m_k(s))^{-1}\beta(i,n_y)\phi(n_y) & \text{for } x = \mathfrak{s}_i(y), \end{cases}$$
(2.36)

with  $\hat{q}(x, y) = 0$  otherwise. The transition function  $\hat{q}$  is the same as q, except that jobs arrive at the stage v = 1, exit at v = s(i), with changes in stage occurring from v - 1 to v, for  $2 \le v \le s(i)$ . We choose  $\pi$  as in (2.32).

The assumptions for Proposition 2.6 can be verified as they were in Section 2.3 for the space  $S_0$ , with only a small change in argument. The argument for (2.11) is the same when either  $y = a_{k,s,i}(x)$  or  $x = \hat{a}_{k,s,i}(y)$ . For  $y = a_{k,s,i}(x)$ , the equality (2.24) is replaced by its analog

$$\pi(y)/\pi(x) = m_k(s)p_k(s)\alpha_k/\phi(n_y).$$

When  $y \in \mathfrak{s}_i(x)$ , one has

$$\pi(y)/\pi(x) = q(x,y)/\hat{q}(y,x) = 1, \qquad (2.37)$$

in which case (2.11) is obvious. So, (2.11) holds in all cases. (Note that for the homogeneous nodes in Section 2.3 with distinct  $\beta$  and  $\delta$ , the analog of (2.37) does not hold, and so the method of stages employed here for generalizing the exponential distributions will not work.)
The formula (2.18) holds for the same reasons as before, except that one now has

$$\sum_{y} q(x,y) = \sum_{y} \hat{q}(x,y) = \sum_{k} \alpha_{k} + \phi(n_{x}) \sum_{i} (m_{x(i)}(s(i)))^{-1} \beta(i,n_{x}),$$

with  $m_{x(i)}(s(i))$  replacing  $m_{x(i)}$ . So, the assumptions for Proposition 2.6 hold. Application of the proposition therefore implies Theorem 2.8.

One can generalize the above argument so that it applies to symmetric queueing networks. Then, the stationary distribution  $\pi$  can be written as the product of stationary distributions  $\pi^{j}$  of nodes corresponding to the individual stations. As in the previous section, we choose to postpone the treatment of symmetric networks until Section 2.5, where they are considered within the context of quasi-reversibility.

#### Extensions to general distributions

We have employed the method of stages to generalize the formula (2.29), for the stationary distribution of symmetric nodes with exponentially distributed service times, to the formula (2.32), which holds for service times that are mixtures of Erlang distributions. The method of stages can also be employed to construct service times with other distributions. This approach is employed, for example, in Section 3.6 of [Wa88] and in Section 3.4 of [As03], where the more general *phase-type distributions* are constructed. [As03] also gives further background on the problem.

Let  $\mathcal{H} = \bigcup_{N=1}^{\infty} \mathcal{H}_N$ , where  $\mathcal{H}_N$  denotes the family of mixtures of Erlang distributions, but with the restriction that  $m_k(s) = 1/N$  for all k and s. It is not difficult to show that  $\mathcal{H}$  is dense in the set of distribution functions, with respect to the weak topology; this result is given in Exercise 3.3.3 in [Ke79]. The basic idea is that the sum of Ns i.i.d. copies of an exponential distribution, with mean 1/N, has mean s and variance s/N, and so, for large N, is concentrated around s. For large enough N, one can therefore approximate a given service time distribution function  $F_k$  as closely as desired, by setting

$$p_k^N(s) = F_k^N(s/N) - F_k^N((s-1)/N), \quad \text{for } s \in \mathbf{Z}_+,$$
(2.38)

equal to the probability that a job chooses a refined class with s stages, when it enters class k. Here,  $F_k^N$  is the distribution function satisfying  $F_k^N(s') = F_k(s')$ , for  $Ns' \in \mathbb{Z}_+$ , and which is constant off this lattice. For the same reason, the phase-type distributions that were mentioned in the previous paragraph are also dense.

Since the family  $\mathcal{H}$  of mixtures of Erlang distributions is dense, it is tempting to infer that a stationary distribution will always exist for a symmetric node with any choice of service time distributions  $F_k$  satisfying (2.19), with  $m_k$  replacing  $m^s$  in the definition of  $\rho$ , and that this distribution has the same product structure as given below the statement of Theorem 2.8. Such a result holds, although the state space needs to be extended so that the last component v of the refined class (k, s, v) of a job can now take on any value in (0, s]; this corresponds to the residual service time of that job. The resulting state space  $S_{\infty}$  for the Markov process is uncountable; this causes technical problems which we discuss at the end of the section. Since the space is uncountable, it is most natural to formulate the result in a manner similar to that given below Theorem 2.8.

**Theorem 2.9.** Suppose that a symmetric node satisfies  $B < \infty$  in (2.19). Then, it has a stationary distribution  $\pi$  on  $S_{\infty}$  with

$$\pi(x(i) = k(i), \ i = 1, \dots, n) = B^{-1} \prod_{i=1}^{n} (m_{k(i)} \alpha_{k(i)} / \phi(i)).$$
(2.39)

Conditioned on any such set, the residual service times of the different jobs are independent, with the probability that a job of a class k has residual service time at most r being

$$F_k^*(r) = \frac{1}{m_k} \int_0^r (1 - F_k(s)) ds.$$
(2.40)

One can motivate (2.39) by applying Theorem 2.8 to a sequence of nodes indexed by N, with  $p_k^N$  for each class k being given by (2.38). Since  $F_k^N \Rightarrow F_k$ and  $m_k^N \to m_k$  as  $N \to \infty$ , one should expect (2.39) to follow from (2.32). In order to motivate (2.40), one can reason as follows. Applying Theorem 2.8, one can check that, under the stationary distribution  $\pi^N$  and conditioned on the job at a given position being k, the probability that the stage there is strictly greater than v is

$$\sum_{s>v} (s-v) p_k^N(s) \bigg/ \sum_s s p_k^N(s).$$

One can also check that  $\sum_{s>v}(s-v)p_k^N(s) = \sum_{s>v}\bar{F}_k^N(s/N)$ , where  $\bar{F}_k^N(s) = 1 - F_k^N(s)$ . So, the above quantity equals

$$\frac{1}{N} \sum_{s>v} \bar{F}_k^N(s/N) \bigg/ \frac{1}{N} \sum_s \bar{F}_k^N(s/N) \bigg|$$

By setting v = Nr,  $r \ge 0$ , and applying the Monotone Convergence Theorem to the numerator and denominator separately, one obtains the limit

$$\frac{1}{m_k} \int_r^\infty \bar{F}_k(s) ds. \tag{2.41}$$

On the other hand, the same reasoning as above (2.38) implies that, for large N, the stage v scaled by N typically approximates the residual service time. So, (2.41) will also give the limiting distribution of the residual service times as  $N \to \infty$ . Taking the complementary event, one obtains (2.40) from (2.41).

The same reasoning as above implies (2.40) is also the probability that the amount of service that has been received by a job is at least r. We point out that  $F_k^*$ , as in (2.40), is the distribution of the residual time for the stationary distribution of a renewal process, with lifetime distribution  $F_k$ .

The above reasoning, although suggestive, is not rigorous. In particular, implicit in the explanations for both (2.39) and (2.40) is the assumption that the stationary distribution  $\pi$  and the residual service time distributions  $F_1^*, \ldots, F_K^*$ are continuous in  $F_1, \ldots, F_K$ . A rigorous justification for Theorem 2.9 is given in [Ba76]. There, the Markov processes  $X^N(\cdot)$  corresponding to the above sequences of nodes are constructed on a common uncountable state space S, where the residual times of the jobs are included in the state. The Markov process that corresponds to the node in Theorem 2.9 is expressed as a weak limit of the processes  $X^N(\cdot)$ ; this provides a rigorous justification for the convergence of  $\pi^N$ and  $F_1^N, \ldots, F_K^N$  that is needed for the theorem. [Ba76] in fact demonstrates the analog of Theorem 2.9 in the more general context of symmetric queueing networks.

The above uncountable state space setting requires a more abstract framework than one typically wishes for a basic theory of symmetric networks. The countable state space setting is typically employed in the context of either mixtures of Erlang distributions, the more general phase-type distributions, or some other dense family of distributions. (See, for example, Section 3.4 of [As03], for more detail.) Quasi-reversibility, which we discuss in the next section, also employs a countable state space setting.

#### 2.5. Quasi-reversibility

In Sections 2.3 and 2.4, we showed that the stationary distributions of homogeneous nodes of Kelly type and symmetric nodes are of product form. Employing quasi-reversibility, it will follow that the stationary distributions of the corresponding queueing networks are also of product form, with the states at the individual stations being independent and the distributions there being given by Theorems 2.7 and 2.8.

Quasi-reversibility has two important consequences. When a queueing network can be decomposed in terms of nodes that are quasi-reversible, the stationary distribution of the network can be written as the product of the stationary distributions of these individual nodes. It will also follow from the "duality" present in quasi-reversibility that the exit processes of such networks are independent Poisson processes, a property that is inherited from the processes of external arrivals of the network. Quasi-reversibility does not depend on the routing in a network, but holds only under certain disciplines, like those mentioned in the first paragraph.

In this section, in order to avoid confusion, we will say that a departing job from a class that leaves the network *exits* from the network (as opposed to being routed to another class). For nodes, such as in the two previous sections, departures and exits are equivalent.

Before introducing quasi-reversibility, we first motivate the basic ideas with a finite sequence of M/M/1 queues that are placed in tandem:

$$\rightarrow 1 \rightarrow 2 \rightarrow \cdots \rightarrow k \rightarrow \cdots \rightarrow K \rightarrow \quad . \tag{2.42}$$

Jobs are assumed to enter the one-class station, with j = k = 1, according to a rate- $\alpha$  Poisson process, and are served in the order of their arrival there. Upon leaving station 1, jobs enter station 2 and are served there, and so on, until leaving the network after having been served at station K. All jobs are assumed to have exponentially distributed service times which are independent, with means  $m_k$  so that  $\alpha m_k < 1$ .

An M/M/1 queue with external arrival rate  $\alpha$  and mean service time m has stationary distribution given by (2.1), if  $m\alpha < 1$ . Under this distribution, the corresponding Markov process X(t),  $-\infty < t < \infty$ , is reversible, and so is stochastically equivalent to its reversed process  $\hat{X}(t) = X(-t)$ . In particular, the stationary process  $X_1(\cdot)$  for the number of jobs at station 1 is stochastically equivalent to its reversed process  $\hat{X}_1(\cdot)$  in terms of the arrival and departure of jobs, with the former corresponding to an increase and the latter a decrease of  $\hat{X}_1(\cdot)$ , jobs arrive according to a rate- $\alpha$  Poisson process. But, each arrival of a job for  $\hat{X}_1(\cdot)$ , at time t, corresponds to the departure of a job for  $X_1(\cdot)$ , at time -t. It follows that the departure process of jobs for  $X_1(\cdot)$  is a Poisson rate- $\alpha$  process. Moreover, departures preceding any given time  $t_1$  are independent of  $X(t_1)$ . What we are observing here, is that the specific nature of the Poisson *input* into station 1 results in an *output* of the same form.

Let  $X_2(\cdot)$  denote the process for the number of jobs at station 2, and assume that  $X_2(t_0)$  has the stationary distribution (2.1), with  $m = m_2$ , for a given  $t_0$ and is independent of  $X_1(t)$  for  $t \ge t_0$ . The arrival process of  $X_2(\cdot)$  is also the departure process of  $X_1(\cdot)$ , which is a rate- $\alpha$  Poisson process, by the previous paragraph. It follows that  $X_2(\cdot)$  is also the stationary process of an M/M/1queue. Its arrivals, up to a given time  $t_1$ , with  $t_1 \ge t_0$ , are independent of  $X_1(t_1)$ by the previous paragraph. Consequently,  $X_1(t_1)$  and  $X_2(t_1)$  are independent, each with the distribution in (2.1), with  $m_1$  and  $m_2$  replacing the mean m. Since  $t_1$  was arbitrary, the joint process  $(X_1(\cdot), X_2(\cdot))$  is Markov and stationary, for all  $t \ge t_0$ . Since  $t_0$  was arbitrary,  $(X_1(\cdot), X_2(\cdot))$  in fact defines a stationary Markov process over all t.

Continuing in this manner, one obtains a stationary Markov process  $X(t) = (X_1(t), \ldots, X_K(t)), -\infty < t < \infty$ , whose joint distribution at any time is a product of distributions of the form in (2.1). One therefore obtains the following result.

**Theorem 2.10.** Assume that the interarrival time and the service times of the sequence of stations depicted in (2.42) are exponentially distributed, with  $\alpha m_k < 1$  for each  $k = 1, \ldots, K$ . Then, the network has a stationary distribution  $\pi$ , which is given by

$$\pi(n_1, \dots, n_K) = \prod_{k=1}^K (1 - \alpha m_k) (\alpha m_k)^{n_k}, \qquad (2.43)$$

for  $n_k \in \mathbf{Z}_{+,0}$ .

We note that although the components of the stationary distribution given by (2.43) are independent, this is not at all the case for the components  $X_k(\cdot)$  of the corresponding stationary Markov process  $X(\cdot)$ . In particular, a departure at station k coincides with an arrival at station k + 1.

Results leading up to Theorem 2.10 and the above proof are given in [Ja54], [Bu56], and [Re57]. More detail on the background of the problem is given on page 212 of [Ke79].

The same formula as in (2.43) holds when the routing in (2.42) is replaced by general routing, if the total arrival rates  $\lambda_k$  are substituted for  $\alpha$ . More precisely, suppose that a subcritical Jackson network (i.e., a single class network with exponentially distributed interarrival and service times) has external arrival rates  $\alpha = \{\alpha_k, k = 1, ..., K\}$  and mean routing matrix  $P = \{P_{k,\ell}, k, \ell = 1, ..., K\}$ . Then, it has the stationary distribution  $\pi$ , with

$$\pi(n_1, \dots, n_K) = \prod_{k=1}^K (1 - \lambda_k m_k) (\lambda_k m_k)^{n_k}, \qquad (2.44)$$

for  $n_k \in \mathbb{Z}_{+,0}$ . This result is no longer as easy to see as is (2.43); it was shown in the important work [Ja63]. The result will follow as a special case of Theorems 2.3 and 2.12.

#### Basics of quasi-reversibility

The "input equals output" behavior of the network in (2.42) was central to our ability to write the stationary distribution of the network as a product of the stationary distributions at its individual stations. *Quasi-reversibility* generalizes this concept, and leads to similar results for more general families of networks. Quasi-reversibility was first identified in [Mu72] and has been extensively employed in work by F.P. Kelly. The property can be defined in different equivalent ways; we use the following analytic formulation.

We consider a node for which arrivals at its classes  $k = 1, \ldots, K$  are given by independent Poisson processes, with intensities  $\alpha = \{\alpha_k, k = 1, \ldots, K\}$ , that do not depend on the state of the node at earlier times, and for which the exits occur only one at a time and do not coincide with an arrival. The evolution of the node is assumed to be given by a Markov process  $X(\cdot)$  with stationary distribution  $\pi$  defined on a countable state space S. Assume that all states communicate. Also, let q denote the transition function of  $X(\cdot)$  and  $\hat{q}$  the transition function of the reversed process  $\hat{X}(\cdot)$  satisfying (2.11).

Under the above assumptions, any change in the state of the node due to a transition from x to y must be due to an increase by 1 in the number of jobs at some class k, for which we write  $y \in A_k(x)$ ; a decrease by 1 at some k, for which we write  $y \in E_k(x)$ ; or a transition that involves neither an increase nor a decrease, for which we write  $y \in I(x)$ , and which we refer to as an *internal transition*. Note that  $y \in I(x)$  and  $x \in I(y)$  are equivalent. An example of an internal transition is the "advance in stage"  $y = \mathfrak{s}_i(x)$  in Section 2.4, although in the current setting far more general changes of state are allowed, including the simultaneous swapping of positions by many jobs. General changes of state are also allowed with the arrival or exit of a job.

We will say the node is *quasi-reversible* if for each class k and state x,

$$\sum_{y \in A_k(x)} \hat{q}(x, y) = \beta_k \tag{2.45}$$

for some  $\beta_k \geq 0$ . The equality (2.45) says that the rate of arrivals at each class k for the reversed process  $\hat{X}(\cdot)$  does not depend on the state x. It is equivalent to the apparently stronger

$$\sum_{y \in A_k(x)} \hat{q}(x, y) = \sum_{y \in A_k(x)} q(x, y) = \alpha_k$$
(2.46)

for each k and x, which states that  $\beta_k = \alpha_k$ . (Note that the last equality follows automatically from the definition of  $\alpha_k$ .)

y

To see (2.46), we note that by (2.45), the arrival times of  $\hat{X}(\cdot)$  form independent rate- $\beta_k$  Poisson processes at the K classes. The same reasoning that was applied to the sequence of M/M/1 queues in (2.42) implies that the exit times of  $X(\cdot)$  also form independent rate- $\beta_k$  Poisson processes. By assumption, only one exit occurs at each such time. Moreover, under the stationary distribution  $\pi$ , the rates at which jobs enter and leave a class are the same. Since the former is  $\alpha_k$ , this implies  $\beta_k = \alpha_k$ , as needed for (2.46).

In the preceding argument, we have shown that the exit processes for  $X(\cdot)$  form independent rate- $\alpha_k$  Poisson processes. Comparison with  $\hat{X}(\cdot)$  also shows that exits for  $X(\cdot)$  preceding any given time  $t_1$  are independent of  $X(t_1)$ . These are important properties of quasi-reversible nodes. We have already employed them in the proof of Theorem 2.10.

The term quasi-reversible can also be applied to a queueing network rather than just to a node, with equation (2.45) again being employed as the defining property. (One should interpret  $A_k(x)$  in terms of external arrivals at k.) In this setting, the stronger (2.46) need not hold, since departures from a class, that are not exits, may occur because of a job moving to another class within the network, and the reasoning in the paragraph below (2.46) is not valid. Nevertheless, external arrivals for the reversed process  $\hat{X}(\cdot)$  correspond to jobs exiting the network for  $X(\cdot)$ . The same reasoning that was employed for quasi-reversible nodes therefore implies that the exiting processes at the classes k are independent rate- $\beta_k$  Poisson processes.

Although we will not use this here, we also note that the *partial balance* equations

$$\pi(x) \sum_{y \in A_k(x)} q(x, y) = \sum_{y \in A_k(x)} \pi(y) q(y, x), \qquad (2.47)$$

for each k and x, are equivalent to (2.46), and hence to the quasi-reversibility of a node. This follows immediately from the definition of  $\hat{q}$  in (2.11) and the assumption  $\pi(x) \neq 0$  for all x. These equations are weaker than the detailed balance equations, which correspond to reversibility, but include information not in the balance equations. The partial balance equations are often used as an alternative to quasi-reversibility.

#### Construction of networks from quasi-reversible nodes and applications

The main result on quasi-reversibility is Theorem 2.11, which states that when a queueing network satisfies certain conditions involving quasi-reversible nodes, its stationary distribution can be written as the product of the stationary distributions of these nodes. These nodes typically correspond to the stations of the network in a natural way. Such a queueing network is itself quasireversible. Examples of these queueing networks are the sequence of M/M/1queues in (2.42), Jackson networks, the homogeneous networks of Kelly type that were defined in Section 2.3, and the symmetric networks that were defined in Section 2.4.

We will consider queueing networks in the following framework. The network will consist of J stations and K classes on a countable state space S of the form

$$S = S^1 \times \dots \times S^J,$$

where  $S^j$  is the state space corresponding to the  $j^{\text{th}}$  station. We will typically write  $x = (x^1, \ldots, x^J)$  for  $x \in S$ , where  $x^j \in S^j$ . For concreteness, we will assume that for each j,  $S^j$  is one of the two spaces  $S_0$  and  $S_e$  that were employed in the last two sections, although the theory holds more generally. As usual, the queueing network is assumed to have transition matrix  $P = \{P_{k,\ell}, k, \ell = 1, \ldots, K\}$  and external arrival rates  $\alpha = \{\alpha_k, k = 1, \ldots, K\}$ . Recall that  $\lambda = Q\alpha$ denotes the total arrival rate, and satisfies the traffic equations given in (1.6).

We will employ notation similar to what was used earlier in the section, with  $A_k(x), E_k(x), I_j(x)$ , and  $R_{k,\ell}(x)$  denoting the states y obtained from x by the different types of transitions. As before,  $A_k(x), E_k(x)$ , and  $I_j(x)$  will denote the states obtained by an arrival into the network at k, an exit from the network at k, and an internal state change at j. For  $y \in A_k(x)$  or  $y \in E_k(x)$ , we will require that  $y^j = x^j$  for  $j \neq s(k)$ , and that the number of jobs at k increase or decrease by 1, and elsewhere remain the same. For  $y \in I_j(x)$ , we will require that  $y^j = x^j$  for  $j \neq s(k)$ , and that the number of jobs at each class remain the same. We let  $R_{k,\ell}(x)$  denote the set of y obtained from x by a job returning to class  $\ell$  after being served at class k. We require that  $y^j = x^j$  for  $j \neq s(k)$  and that the number of jobs at k increase by 1, and elsewhere remain the same. Such as  $y^j = x^j$  for  $j \neq s(k)$  and that the number of jobs at  $y^j = x^j$  for  $j \neq s(k)$  and the equivalent that  $y^j = x^j$  for  $j \neq s(k)$  and the equivalent that  $y^j = x^j$  for  $j \neq s(k)$  and the equivalent that  $y^j = x^j$  for  $j \neq s(k)$  and  $j \neq s(\ell)$ , and that the number of jobs at k decrease by 1, at  $\ell$  increase by 1, and elsewhere remain the same.

The queueing networks we will consider will be assumed to satisfy properties (2.48)-(2.52), which are given in terms of prechosen quasi-reversible nodes. Before listing these properties, we provide some motivation, recalling the sequence of 1-class stations given in (2.42), with the stationary distribution in (2.43). When a given station j, with j = k, is viewed "in isolation", it evolves as an M/M/1 queue with mean service time  $m_k$  and external arrival rate  $\alpha$ , and has as its stationary distribution the stationary distribution of the corresponding M/M/1 queue. The stationary distribution of the sequence of stations is given by the product of the stationary distributions of the individual queues. Because of the specific structure of the network, Poisson arrivals into a given station result in Poisson departures, which then serve as Poisson arrivals for the next station. This property allowed us to view the stations "in isolation". We will show that queueing networks satisfying properties (2.48)-(2.52) will have stationary distributions that are the product of the stationary distributions of the quasi-reversible nodes given there. Each such node can be interpreted as the corresponding station evolving "in isolation". Here, "in isolation" will also mean that routing between classes at the same station is not permitted. The network in the previous paragraph, consisting of a sequence of 1-class stations, will be a special case of this more general setup.

Because of the more abstract setting now being considered, we will not explicitly follow the evolution of individual jobs; instead, we will think of the quasi-reversible nodes as "black boxes", which have a given output for a given input, with a corresponding stationary distribution. Rather than employ the Poisson-in, Poisson-out property directly, we will use the definition of quasireversibility in (2.45). The external arrival rates  $\alpha_k^j$  for the classes at a given node j will be given by the total arrival rate  $\lambda_k$  for the corresponding class in the network. This will be consistent with jobs always leaving the node after being served, without being routed to another class.

The nodes we employ are assumed to have state spaces  $S^j$ ,  $j = 1, \ldots, J$ , which are the components of the state space S for the network. Therefore, for  $x = (x^1, \ldots, x^J) \in S$  with  $x^j \in S^j$ ,  $j = 1, \ldots, J$ , one can also interpret  $x^j$  as the state of the corresponding node. Jobs at a given node will have classes  $k \in \mathcal{C}(j)$ , which are in one-to-one correspondence with the classes of the correspondingly labelled station in the network. For  $x^j \in S^j$  and  $k \in \mathcal{C}(j)$ , we employ notation introduced earlier in the section for quasi-reversible nodes, with  $A_k^j(x^j)$ ,  $E_k^j(x^j)$ , and  $I^j(x^j)$  denoting those states  $y^j$  obtained from  $x^j$  by an arrival or exit at class k, or by an internal state change. We let  $q^j$ ,  $j = 1, \ldots, J$ , denote the transition rates for the Markov processes  $X^j(\cdot)$  of the nodes. The nodes are assumed to be quasi-reversible, with (2.45) being satisfied by  $\hat{q}^j$ , the transition rates of the reversed processes  $\hat{X}^j(\cdot)$ . As mentioned earlier, the external arrival rates of the nodes are given by  $\alpha_k^j = \lambda_k$ . As earlier in the section, we will assume that all states of a given node communicate.

We will assume that the transition rates q(x, y) for the queueing network can be written in terms of the rates  $q^j(x^j, y^j)$  for the nodes as follows. For  $y \in A_k(x)$ , we assume that

$$q(x,y) = (\alpha_k/\lambda_k)q^j(x^j, y^j).$$
(2.48)

Setting  $p_k^j(x^j, y^j) = q^j(x^j, y^j)/\lambda_k$ , this can be written as

$$q(x,y) = \alpha_k p_k^j(x^j, y^j), \qquad (2.48')$$

where  $\sum_{y^j \in A_k^j(x^j)} p_k^j(x^j, y^j) = 1$  holds. (Here and later on, when j and k appear together, we implicitly assume that  $k \in \mathcal{C}(j)$ .) For  $y \in E_k(x)$ , we assume that

$$q(x,y) = q^{j}(x^{j}, y^{j})P_{k,0}, \qquad (2.49)$$

where  $P_{k,0} \stackrel{\text{def}}{=} 1 - \sum_{\ell} P_{k,\ell}$ . For  $y \in R_{k,\ell}(x)$ , we require the existence of an "intermediate" state z between x and y, with  $z \in E_k(x)$  and  $y \in A_\ell(z)$ , and such that

$$q(x,y) = q^{j}(x^{j}, z^{j})q^{h}(z^{h}, y^{h})(P_{k,\ell}/\lambda_{\ell}), \qquad (2.50)$$

where  $h = s(\ell)$ . One can also write this as

$$q(x,y) = q^{j}(x^{j}, z^{j}) P_{k,\ell} p_{\ell}^{h}(z^{h}, y^{h}).$$
(2.50')

For  $y \in I_j(x)$ , we assume that

$$q(x,y) = q^{j}(x^{j}, y^{j}), \qquad (2.51)$$

and finally, on the complement of the above sets, we assume that

$$q(x,y) = 0. (2.52)$$

The equations (2.48)-(2.52) have the following interpretation in terms of the transition rates of the queueing network. The J different stations operate independently of one another, except for the movement of jobs between them. So, the transition rates in (2.48'), (2.49), and (2.51) depend on  $x^j$  and  $y^j$  instead of on the entire states x and y. In (2.50'), after a class k job is served, it moves to class  $\ell$  with probability  $P_{k,\ell}$ , with the probability of the new state y depending on just  $z^h$  and  $y^h$ . When  $h \neq j$ , z is automatically given by  $z^j = y^j$ ,  $z^h = x^h$ , and  $z^{j'} = x^{j'} = y^{j'}$  for other values j'. The transition rates  $q^j$  in each display are those of node j, which does not permit returns. This node can be thought of as the one obtained from the corresponding station by replacing transitions to and from each class k by external arrivals and exits at the same rates.

We now employ the above terminology to state Theorem 2.11. When its hypotheses are satisfied, the theorem enables us to write the stationary distribution of a queueing network as the product of the stationary distributions of the corresponding nodes.

**Theorem 2.11.** Suppose that the transition rates q(x, y) of a queueing network satisfy (2.48)–(2.52), where the nodes with the transition rates  $q^j(x^j, y^j)$  are quasi-reversible with stationary distributions  $\pi^j$ . Then, the queueing network has stationary distribution  $\pi$  given by

$$\pi(x) = \prod_{j=1}^{J} \pi^{j}(x^{j}), \qquad (2.53)$$

where  $x = (x^1, \ldots, x^J)$ . Moreover, the queueing network is itself quasi-reversible.

The proof of Theorem 2.11 will be given in the next subsection. We first note the following consequences of Theorem 2.11 and quasi-reversibility.

As an elementary illustration of Theorem 2.11, we return to the "sequence of M/M/1 queues" in (2.42). Equations (2.48)–(2.52) all hold in this setting, if  $q^j$  are the transition rates for the M/M/1 queues with  $\alpha^j = \alpha$  and  $m^j = m_j$ . All of these equations are easy to see and are nondegenerate in only a few cases. In (2.48),  $q(x, y) \neq 0$  only for k = j = 1 and, in (2.49),  $q(x, y) \neq 0$  only for k = K = J. In (2.50'), with  $1 \leq k < K$ , one has  $P_{k,k+1} = p_{k+1}^{k+1}(z^{k+1}, y^{k+1}) = 1$  if z is chosen by removing a class k job from x; (2.51) is vacuous in this setting. Moreover, since the M/M/1 queues are reversible, they are quasi-reversible.

These queues are assumed to be subcritical and have stationary distributions given by (2.1). Theorem 2.10 therefore follows as a special case of Theorem 2.11.

Equations (2.48)–(2.52) also hold for the more general Jackson networks (which are, in turn, special cases of FIFO networks of Kelly type). Again,  $q^j$  are the transition rates for M/M/1 queues, this time with  $\alpha^j = \lambda_j$  and  $m^j = m_j$ . The equations are similar to those for the previous example, except that the mean transition matrix P is general, and so (2.48)–(2.50') may be nonzero for arbitrary k. The formula for the stationary distribution in (2.44) is consequently an easy application of Theorem 2.11.

We now generalize Theorem 2.3 of Section 2.1. For homogeneous networks of Kelly type and symmetric networks, equations (2.48)–(2.52) all hold if the corresponding nodes are chosen in the natural way. Namely, each such node, for  $j = 1, \ldots, J$ , is obtained from the corresponding station by replacing transitions involving routing from one class to another by exits from the network, and by increasing the rate of external arrivals at each class k from  $\alpha_k$  to  $\lambda_k$  to compensate for this. Then, (2.48) and (2.49) are immediate. In (2.50'), the state z is chosen by removing the served job at k from x. The equality (2.50') then follows since a transition from x to y, with  $y \in R_{k,\ell}(x)$ , consists of a service completion at k, followed by the routing of the corresponding job to class l of a station h, with the job then being assigned a position i according to the rule  $p_{\ell}^{h}$ . When  $S^{j} = S_{0}$ , the transition q(x, y) in (2.51) does not occur; when, for a symmetric node,  $S^{j} = S_{e}$ , the transition corresponds to the advance of a stage. In either case, (2.51) is clear. Moreover, on account of (2.22) and (2.36) in Sections 2.3 and 2.4,

$$\sum_{y^j \in A_k^j(x^j)} \hat{q}^j(x^j, y^j) = \alpha_k \quad \text{for all } x^j \in S^j,$$
(2.54)

and so each such node is quasi-reversible. Note that this characterization continues to hold for networks that are of mixed type, with some stations being homogeneous of Kelly type and others being symmetric.

Recall that in Theorems 2.7 and 2.8, we saw that such nodes themselves have stationary distributions that are of product form, as given in (2.20) and (2.32). Theorem 2.7 was stated, for homogeneous nodes, in the context of service times that are exponentially distributed, and Theorem 2.8 was stated, for symmetric nodes, in the context of mixtures of Erlang distributions. Combining these results with Theorem 2.11, we therefore obtain Theorem 2.12. As in Theorems 2.7 and 2.8, when the node is homogeneous,  $x^j \in S_0$  is assumed, whereas when the node is symmetric,  $x^j \in S_e$ . In either case, we employ the condition

$$B_j \stackrel{\text{def}}{=} \sum_{n=0}^{\infty} \left( \rho_j^n \middle/ \prod_{i=1}^n \phi(i) \right) < \infty, \tag{2.55}$$

with  $\rho_j = \sum_{k \in \mathcal{C}(j)} m_k \lambda_k$ .

**Theorem 2.12.** Suppose that each station j of a queueing network is either homogeneous of Kelly type or is symmetric, and satisfies (2.55). Then, the queueing network has a stationary distribution  $\pi$  that is given by

$$\pi(x) = \prod_{j=1}^{J} \pi^{j}(x^{j}), \qquad (2.56)$$

where each  $\pi^{j}$  is either of the form (2.20) or (2.32), depending on whether the station j is homogeneous of Kelly type or is symmetric, and  $\alpha_{k}$  in these formulas is replaced by  $\lambda_{k}$ .

Theorem 2.11 shows that the stationary distribution of a queueing network that is composed of stations corresponding to quasi-reversible nodes has the product structure given in (2.53). Nevertheless, the processes that are associated with the stations are not independent. One can see this easily for the example at the beginning of the section consisting of a sequence of M/M/1 queues: a departure from one queue coincides with an arrival to the next. Similarly, the combined arrival processes at different classes (i.e., arrivals from other classes as well as external arrivals) are typically not independent, nor are the departure processes.

These processes are not to be confused with the processes of external arrivals or the processes of jobs exiting from the network, which are in either case independent. We note, though, that under the stationary distribution for a queueing network composed of quasi-reversible nodes, the conditional distribution at a station found by an arriving job is the same as the stationary distribution there. This is clearly the case for external arrivals, but is also true for arrivals in general. This is shown on page 70 of [Ke79] by considering the reversed Markov process for the network.

#### Demonstration of Theorem 2.11

In order to demonstrate Theorem 2.11, we will employ Proposition 2.6. We therefore need a candidate  $\hat{q}$  for the transition rates of the reversed process  $\hat{X}(\cdot)$  for the network under its stationary distribution. Letting  $\hat{q}^j$  denote the reversed transition rates corresponding to  $q^j$ , we define  $\hat{q}$  using the following analogs of (2.48)–(2.52). We set

$$\hat{q}(x,y) = (\hat{\alpha}_k/\lambda_k)\hat{q}^j(x^j,y^j) = P_{k,0}\hat{q}^j(x^j,y^j) \text{ for } y \in A_k(x),$$
 (2.57)

$$\hat{q}(x,y) = \hat{q}^{j}(x^{j},y^{j})\hat{P}_{k,0} = \hat{q}^{j}(x^{j},y^{j})(\alpha_{k}/\lambda_{k}) \quad \text{for } y \in E_{k}(x).$$
(2.58)

For  $y \in R_{k,\ell}(x)$ , we set

$$\hat{q}(x,y) = \hat{q}^{j}(x^{j},z^{j})\hat{P}_{k,\ell}\frac{\hat{q}^{h}(z^{h},y^{h})}{\sum_{w^{h}\in A_{\ell}^{h}(z^{h})}\hat{q}^{h}(z^{h},w^{h})} = \hat{q}^{j}(x^{j},z^{j})\hat{q}^{h}(z^{h},y^{h})(P_{\ell,k}/\lambda_{k}).$$
(2.59)

We also set

$$\hat{q}(x,y) = \hat{q}^{j}(x^{j},y^{j}) \text{ for } x \in I_{j}(y),$$
(2.60)

$$\hat{q}(x,y) = 0$$
 for other values of y. (2.61)

Here, we are setting

$$\hat{\alpha}_k = \lambda_k P_{k,0}, \quad \hat{P}_{k,\ell} = \lambda_\ell P_{\ell,k} / \lambda_k, \quad \hat{P}_{k,0} = \alpha_k / \lambda_k.$$
(2.62)

The term  $\lambda_k P_{k,0}$  is the rate at which jobs exit from the original network at class k, and so should be the rate they enter the reversed network at k. The second equality is obtained by reversing the direction of the mean transition matrix P; the third equality is obtained by setting  $\hat{P}_{k,0} = 1 - \sum_{\ell} \hat{P}_{k,\ell}$ , and applying the previous equality together with (1.6). We have implicitly set  $\hat{\lambda}_k = \lambda_k$  in (2.57). The second equality in (2.59) needs to be justified; it follows from (2.62) together with

$$\sum_{w^{h} \in A^{h}_{\ell}(z^{h})} \hat{q}^{h}(z^{h}, w^{h}) = \lambda_{\ell}.$$
(2.63)

Since each node is assumed to be quasi-reversible, (2.63) follows from (2.46) and  $\alpha_{\ell}^{h} = \lambda_{\ell}$ . In the proof of Theorem 2.11, we will also employ

$$\sum_{k} \hat{\alpha}_{k} = \sum_{k} \alpha_{k}, \qquad (2.64)$$

which follows from the definition of  $\hat{\alpha}_k$  and (1.6).

*Proof of Theorem 2.11.* The quasi-reversibility of the queueing network follows immediately from the first equality in (2.57) and from (2.63), since

$$\sum_{\substack{\in A_k(x)}} \hat{q}(x,y) = (\hat{\alpha}_k/\lambda_k) \sum_{y^j \in A_k^j(x^j)} \hat{q}^j(x^j, y^j) = (\hat{\alpha}_k/\lambda_k)\lambda_k = \hat{\alpha}_k,$$

which does not depend on x.

y

The remainder of the proof is devoted to showing that the distribution  $\pi$  in (2.53) is stationary. We wish to show that  $\pi$  satisfies

$$\pi(x)q(x,y) = \pi(y)\hat{q}(y,x) \quad \text{for all } x, y \in S$$
(2.65)

and

$$q(x) = \hat{q}(x) \quad \text{for all } x \in S, \tag{2.66}$$

where  $\hat{q}$  is defined in (2.57)–(2.61). These are restatements of (2.11) and (2.18), and together with Proposition 2.6 imply that  $\pi$  is stationary. Since all states are assumed to communicate, this is the unique such distribution.

Demonstration of (2.65). In order to verify (2.65), one needs to check the different cases given by the formulas for q in (2.48)–(2.52). Each is straightforward, with the most involved case being  $y \in R_{k,\ell}$ . To check (2.65) for  $y \in R_{k,\ell}(x)$ , note that by (2.50) and the second part of (2.59), (2.65) reduces to

$$\pi^{j}(x^{j})\pi^{h}(x^{h})q^{j}(x^{j},z^{j})q^{h}(z^{h},y^{h}) = \pi^{j}(y^{j})\pi^{h}(y^{h})\hat{q}^{j}(y^{j},z^{j})\hat{q}^{h}(z^{h},x^{h}) \quad (2.67)$$

216

after cancelling the common terms  $\pi^{j'}(x^{j'})$ , with  $j' \neq j$  and  $j' \neq h$ , and  $P_{k,\ell}/\lambda_{\ell}$ . This equality follows immediately from the definition of  $\hat{q}^j$  and  $\hat{q}^h$  in (2.11).

For the cases where  $y \in A_k(x)$  and  $y \in E_k(x)$ , (2.65) reduces to analogs of (2.67), which are somewhat simpler since only one node rather than two is involved. The case  $y \in I_j(x)$  follows from the definition of  $\hat{q}^j$ . For pairs x and y not covered in the preceding four cases,  $q(x, y) = \hat{q}(y, x) = 0$  by (2.52) and (2.61). So, (2.65) holds in this last case as well.

Demonstration of (2.66). This part requires more work. We will show that

$$q(x) = \sum_{k} \alpha_k - \sum_{k} \lambda_k + \sum_{j} q^j(x^j)$$
(2.68)

and

$$\hat{q}(x) = \sum_{k} \hat{\alpha}_k - \sum_{k} \lambda_k + \sum_{j} \hat{q}^j(x^j).$$
(2.69)

The first sums in the two equalities are equal by (2.64), and the last sums are equal since (2.18) holds for each node. So, (2.68) and (2.69) together imply (2.66).

We first show (2.68). We rewrite q(x) as

$$q(x) = \left(\sum_{k} \sum_{y \in A_k(x)} + \sum_{k} \sum_{y \in E_k(x)} + \sum_{k,\ell} \sum_{y \in R_{k,\ell}(x)} + \sum_{j} \sum_{y \in I_j(x)} \right) q(x,y), \quad (2.70)$$

and analyze the different parts. By (2.48'), the first double sum on the right equals

$$\sum_{k} \alpha_k \sum_{y^j \in A^j_k(x^j)} p^j_k(x^j, y^j) = \sum_k \alpha_k.$$
(2.71)

By (2.49), the second double sum equals

$$\sum_{k} \sum_{y^{j} \in E_{k}^{j}(x^{j})} q(x^{j}, y^{j}) P_{k,0}.$$
(2.72)

By (2.50'), the third double sum equals

$$\sum_{k} \sum_{z^{j} \in E_{k}^{j}(x^{j})} q^{j}(x^{j}, z^{j}) \sum_{\ell} P_{k,\ell} \sum_{y^{h} \in A_{\ell}^{h}(z^{h})} p_{\ell}^{h}(z^{h}, y^{h})$$

$$= \sum_{k} \sum_{z^{j} \in E_{k}^{j}(x^{j})} q^{j}(x^{j}, z^{j}) \sum_{\ell} P_{k,\ell}.$$
(2.73)

By (2.51), the last double sum equals

$$\sum_{j} \sum_{y^{j} \in I^{j}(x^{j})} q^{j}(x^{j}, y^{j}).$$
(2.74)

Summation of (2.72)–(2.74) gives

$$\left(\sum_{k}\sum_{y^{j}\in E_{k}^{j}(x^{j})}+\sum_{j}\sum_{y^{j}\in I^{j}(x^{j})}\right)q^{j}(x^{j},y^{j}).$$
(2.75)

Also, note that

$$\sum_{k} \sum_{y^j \in A_k^j(x^j)} q^j(x^j, y^j) = \sum_{k} \alpha_k^j = \sum_{k} \lambda_k.$$
(2.76)

The sum of (2.75) and the left side of (2.76) is just  $\sum_j q^j(x^j)$ . On the other hand, the right side of (2.70) is equal to the sum of the left side of (2.71) and (2.75). So, q(x) is equal to the sum of  $\sum_k \alpha_k$  and (2.75), whereas  $\sum_j q^j(x^j)$  is equal to the sum of  $\sum_k \lambda_k$  and (2.75). Solving for this last term implies (2.68).

The argument for (2.69) is similar, and we employ the analog of the decomposition in (2.70), but for  $\hat{q}(x)$  instead of q(x). By the first equality in (2.57) and (2.63),

$$\sum_{k} \sum_{y \in A_{k}(x)} \hat{q}(x,y) = \sum_{k} (\hat{\alpha}_{k}/\lambda_{k}) \sum_{y^{j} \in A_{k}^{j}(x^{j})} \hat{q}^{j}(x^{j},y^{j}) = \sum_{k} \hat{\alpha}_{k}.$$
 (2.77)

Also, using the first equalities in (2.58) and (2.59), and (2.60), the same reasoning as that leading to (2.75) implies that the sum of the terms corresponding to the last three double sums in (2.70) is

$$\left(\sum_{k}\sum_{y^{j}\in E_{k}^{j}(x^{j})}+\sum_{j}\sum_{y^{j}\in I^{j}(x^{j})}\right)\hat{q}^{j}(x^{j},y^{j}).$$
(2.78)

On the other hand, it follows from (2.63) that

$$\sum_{k} \sum_{y^{j} \in A_{k}(x^{j})} \hat{q}^{j}(x^{j}, y^{j}) = \sum_{k} \lambda_{k}.$$
(2.79)

The sum of (2.78) and the left side of (2.79) is just  $\sum_{j} \hat{q}^{j}(x^{j})$ . Employing (2.77), (2.78), and (2.79) as we did (2.71), (2.75), and (2.76), the same reasoning as before implies (2.69).

#### Another proof for Theorem 2.11

Another proof for Theorem 2.11, that is more probabilistic, is given in [Wa82] and [Wa83] (see also [Wa88]). The basic idea of the proof is to modify the queueing network by imposing an  $\epsilon$  delay, with  $\epsilon > 0$ , on all routing between classes. The corresponding stochastic process will be easier to analyze. It will not be Markov, but will have a distribution that is of product form and is invariant over time, and is the same for all values of  $\epsilon$ . The limiting process as  $\epsilon \downarrow 0$  will be the Markov process for the original queueing network, and its stationary distribution will be this distribution.

We now sketch the argument. Consider the J quasi-reversible nodes that are associated with the queueing network as in (2.48)–(2.52), but which have external arrival rates  $\alpha_k$  instead of  $\lambda_k$ . We form a new network from these nodes by assuming that when jobs leave a node j from class k, they are routed back to class  $\ell$  of node h with probability  $P_{k,\ell}$ , but with a fixed deterministic delay  $\epsilon > 0$ . During this delay, such jobs are assumed to not affect the transitions within the nodes, which now play the role of individual stations within the network.

One can construct the corresponding stochastic process inductively over time intervals of length  $\epsilon$ , starting with  $[0, \epsilon]$ . One argues by first assuming that (a) the initial states at the J stations are independent of one another and are given by the stationary distributions of the isolated nodes with external arrival rates  $\lambda_{\ell}$  and (b) over the time interval  $(0, \epsilon]$ , the jobs returning to classes  $\ell$  constitute independent Poisson processes having rates  $\lambda_k P_{k,\ell}$ , which are independent of the initial states in (a). Jobs from outside the system arrive at class  $\ell$  at rate  $\alpha_{\ell}$ , and so by (b) and the traffic equations (1.6), the combined arrivals at  $\ell$  from these two sources of jobs are Poisson processes with rates  $\lambda_{\ell}$  and are independent of one another. Because of the  $\epsilon$  delay for returning jobs, jobs departing from nodes over  $(0, \epsilon]$  will not return over this period, and so do not affect arrivals at  $\ell$ .

On account of these arrival processes, the processes at the stations will be stationary over  $(0, \epsilon]$  and independent of one another. Since the corresponding nodes are quasi-reversible, jobs depart from the classes k according to independent rate- $\lambda_k$  Poisson processes over this period, which are independent of the states of the stations at time  $\epsilon$ . Because of the deterministic  $\epsilon$  delay required for returns, these jobs return to the classes  $\ell$  as Poisson processes with rates  $\lambda_k P_{k,\ell}$ , over the period  $(\epsilon, 2\epsilon]$ . Consequently, the analogs of conditions (a) and (b) hold over the time interval  $[\epsilon, 2\epsilon]$ .

Iteration over the time intervals  $(\epsilon, 2\epsilon]$ ,  $(2\epsilon, 3\epsilon]$ , ... produces a stochastic process on  $[0, \infty)$  whose states at different stations at any fixed time are independent of one another, and whose distributions are the same as the stationary distributions of the corresponding isolated nodes. This process can also be extended to all times  $t \in (-\infty, \infty)$ .

Letting  $\epsilon \downarrow 0$ , the sequence of these processes will converge to the Markov process corresponding to the original queueing network. Since each of these processes has the same joint distribution at any given time, this distribution will be stationary for the limiting Markov process. Since this distribution has the desired product form, this reasoning implies (2.53) of Theorem 2.11. By considering the exit processes of the sequence of processes, one can also show that the original queueing network is quasi-reversible.

### Chapter 3

# Instability of subcritical queueing networks

Until the early 1990's, the understanding of multiclass queueing networks was sketchy. In particular, relatively little thought had been given to the stability of such networks. The network, of course, needs to be subcritical. On the other hand, the "classical networks" considered in Chapter 2 are all stable when they are subcritical. Does this behavior hold in general, assuming all states of the corresponding Markov process communicate? Since one typically cannot explicitly compute the stationary distribution of a network, the direct approach in Chapter 2 needs to be replaced.

If stability holds universally for subcritical networks (or, in a broad enough setting), one should expect a reasonably simple proof of this; the argument would presumably be elementary because of its robustness. If, however, such a result holds in some settings but not in others, a general theory (if one exists) might be complicated. We now know through various examples that stability does not always hold. Whether a general theory is possible is still an open question. In this chapter, we present a number of examples exhibiting different situations in which the number of jobs in the network goes to infinity as  $t \to \infty$ . Chapter 4 will be devoted to positive results on the stability of subcritical networks.

This chapter is broken into three sections, according to the order of appearance and content of the examples. In Section 3.1, the first basic examples of unstable subcritical queueing networks are given. These consist of examples in [LuK91] and [RyS92] for static priority disciplines and [KuS90] for a clearing policy. In Section 3.2, examples of unstable subcritical FIFO networks are given, which consist of examples in [Br94a; Br94a] and [Se94]. Section 3.3 discusses examples for other disciplines that illustrate features of interest. They consist of an unstable network of Kelly type from [DaWe96], and examples from [Du97] and [BaB99], where the region of stability of certain networks is examined in greater detail. Somewhat different definitions of "unstable" exist in the literature. For us, a queueing network will be unstable if, for some initial state, the number of jobs in the network will, with positive probability, go to infinity as  $t \to \infty$ . This was mentioned in Section 1.2. When the network has only a finite number of states possessing fewer than a given number of jobs, and all states communicate with one another, this is equivalent to saying that, for each initial state, the number of jobs in the network goes to infinity almost surely as  $t \to \infty$ . Both conditions will be satisfied for the examples in this chapter with Poisson arrivals and exponential service times. Note that a network that is not stable is not necessarily unstable as defined above, since the corresponding Markov process can be null recurrent.

The intent of this chapter is to provide an elementary introduction to the subject and at the same time impart some feeling for the development of this subject in the 1990's. An omission with regard to the latter is the interaction with contemporary developments for heavy traffic limits. This includes the example in [DaWa93], which highlighted the general lack of understanding of the asymptotics for even simple multiclass queueing networks. This material requires additional terminology and new concepts. Since it lies outside the scope of these lectures, we omit it with some regret.

There has also been interest in examples exhibiting instability with regard to certain questions arising in computer science. In this setting, the models that are investigated and the relevant questions can take on a somewhat different flavor. For one such topic, *adversarial queueing*, stability under a "worst case" scenario is examined, where an all-knowledgeable adversary is allowed to modify the precise timing of input into the system. Service times are deterministic and are most often assumed to be the same everywhere. We omit this topic and instead refer the reader to [BoKRSW96] and [AnAFKLL96].

As mentioned in Section 1.2, it is sometimes convenient to employ different notation for the classes of a network, depending on its routing. Most of the examples in this chapter will be reentrant lines; we will usually label the classes sequentially based on the order in which they appear along the route. When the network has more than one deterministic route, this route will be indicated by the first coordinate. In the first and last examples in Section 3.2, we will find it convenient to use different notation because of the structure of the networks, and we include the station as one of the coordinates.

#### 3.1. Basic examples of unstable networks

An elementary example of an unstable subcritical SBP network is given by the *Lu-Kumar network*. The example is presented in a deterministic setting in [LuK91]. Its proof there is short, and requires just a couple paragraphs. We present here both this version and the model in a random setting, whose proof is not difficult but requires a bit more work. We also mention a related earlier model from [KuS90] for a clearing policy.

Independently, an unstable static priority network, the *Rybko-Stolyar net*work, was analyzed in [RyS92]. Its behavior is almost identical to the Lu-Kumar network, with the routing differing in just one aspect. We discuss this network briefly. Although insufficiently appreciated at the time, both networks have since had a substantial effect on thinking in queueing theory.

#### The Lu-Kumar network

This network is a reentrant line consisting of two stations, with two classes at each station. Jobs following the deterministic route first visit station 1 after entering the network, next visit station 2 twice, and then visit station 1 a second time, before exiting the network. The route is depicted in Figure 3.1; as indicated at the beginning of the chapter, we order the classes according to their appearance along the route. The system evolves according to a preemptive SBP discipline, with jobs at class 4 having priority over those at class 1, and jobs at class 2 having priority over those at class 3.

In [LuK91], a deterministic version of this model is given, with jobs entering periodically at the times  $0, 1, 2, \ldots$  The deterministic service time at class k is given by  $m_k$ , with

$$m_2 = m_4 = 2/3$$
 (3.1a)

and

$$m_1 = m_3 = 0. \tag{3.1b}$$

This version of the model will be referred to as the *deterministic Lu-Kumar net*work. As we will see, even though the service at classes 1 and 3 is instantaneous, their presence affects the evolution of the network. This model is somewhat artificial, but is easy to analyze. (A modification with more general  $m_k$  will be discussed shortly.)

#### **Theorem 3.1.** The deterministic Lu-Kumar network is unstable.

Note that because of the presence of instantaneous events at classes 1 and 3, we need to specify an ordering of service in case of "ties". For this, we assume jobs at all classes complete service at times t— "just before" t; the assumption, in particular, allows jobs to be served at class 3 and move to class 4, before a new job enters the network at class 1 and is served there. The changes in the long-term evolution of the system induced by other orderings are minor.



FIG. 3.1.

Proof of Theorem 3.1. We assume that there are initially, at t = 0-,  $M \in \mathbb{Z}_+$  jobs at class 1 and no jobs elsewhere. Since  $m_1 = 0$ , the M jobs at class 1 immediately leave there and move to class 2, where they begin service. Because of the priority of class 2 over class 3, jobs at class 3 cannot be served until class 2 is empty, which means that class 4 must remain empty until then. Therefore, jobs at class 1 continue to be served until class 2 is empty. Since  $m_2 = 2/3$ , reasoning along these lines shows that at time 2M-, all classes are empty except for class 3, which has 3M jobs (the M original jobs plus 2M new jobs).

Since  $m_3 = 0$ , these 3M jobs are immediately served at class 3 and move to class 4, where they begin service. Because of the priority of class 4 over class 1, jobs entering the network at class 1 cannot be served until all of these jobs depart from class 4. Since  $m_4 = 2/3$ , this occurs at time 4M—. Over the elapsing time 2M, M' = 2M jobs have arrived at class 1; moreover, at time 4M—, there are no jobs elsewhere in the network. The state at time 4M— therefore has the same form as it had initially, but with twice as many jobs; moreover, over [0-, 4M-], there are never fewer than M jobs in the network. This cycle repeats itself indefinitely, resulting in always at least  $2^n M$  jobs in the  $n^{\text{th}}$  cycle, which goes to infinity as  $n \to \infty$ .

We note that for  $m_2 = m_4 = c$ , any choice of c > 1/2 suffices for the above instability of the network; c = 2/3 was chosen above for convenience. The key feature of the network is that the main body of jobs is forced to remain "clumped together" as it moves from class 1 to class 4 during each cycle. This induces underutilization of (or "starvation" at) both stations 1 and 2, and so is responsible for the instability of the network. The above argument does not rule out the possibility of the number of jobs in the network remaining bounded over time when starting from different initial states, since such states need not communicate.

The above deterministic setting for the Lu-Kumar network and the restrictions on  $m_k$ , in (3.1), are not essential features of the model. Assume instead that jobs enter the network according to a rate-1 Poisson process and are served at all classes according to independent exponentially distributed random variables with means  $m_k > 0$ . We refer to this model as the random Lu-Kumar network.

#### **Theorem 3.2.** The random Lu-Kumar network, with $m_2 + m_4 > 1$ , is unstable.

The proof of Theorem 3.2 is not difficult, but requires some preparation. Before proceeding, we first point out that if instead

$$m_1 + m_4 < 1, \quad m_2 + m_3 < 1$$
 (3.2)

(that is, the network is subcritical) and

$$m_2 + m_4 < 1,$$
 (3.3)

then the network is stable. This is shown in [DaWe96]. The argument employs the machinery of fluid models, which are discussed in detail in Chapter 4. The

exponential assumptions on the interarrival times and service times are important for neither direction, although in this framework, the Lu-Kumar network corresponds to a countable state Markov process. We recall the FBFS and LBFS reentrant lines, which were introduced in Chapter 1. It will be shown in Section 5.2 that when they are subcritical, such reentrant lines are always stable. The Lu-Kumar network, whose discipline is a mixture of these disciplines, is unstable.

We recall from Chapter 1 the following notation, which will reoccur more extensively later on in these lectures. Let  $Z_k(t)$  denote the number of jobs at class k at time t, with Z(t) being the corresponding vector. Since the interarrival and service times are all exponentially distributed, Z(t) is a Markov process. We set  $|Z(t)| = \sum_k Z_k(t)$ . Also, let  $W_j(t)$  denote the immediate workload at station j, that is,  $W_j(t)$  is the amount of time required to serve all jobs currently at j, if one excludes other jobs from entering the station.

Before giving the proof of Theorem 3.2, we make the following two observations. Suppose that

$$Z_2(t) > 0 \text{ and } Z_4(t) > 0 \text{ for all } t \in (t_1, t_2],$$
 (3.4)

for some  $t_1 \leq t_2$ . Then,

$$W_2(t_2) = W_2(t_1) - (t_2 - t_1).$$
(3.5)

The direction " $\leq$ " follows from the priority of class 4 over class 1, which prevents any jobs from entering station 2 when  $Z_4(t) > 0$ , and hence over  $(t_1, t_2]$ . The other direction is automatic. Since  $W_2(t_2) \geq 0$ , (3.5) gives an upper bound on  $\tau_0$ , the first time at which either  $Z_2(t) = 0$  or  $Z_4(t) = 0$ , in terms of  $W_2(0)$ . In particular,  $\tau_0 < \infty$  a.s.

Similarly, the priority of class 2 over class 3 prevents any jobs from entering class 4 as long as  $Z_2(t) > 0$ . Together with the sentence after (3.5), this implies that a.s.,

$$Z_2(t) = 0 \text{ or } Z_4(t) = 0 \text{ for all } t \ge \tau_0.$$
 (3.6)

As an immediate consequence, we have:

**Lemma 3.3.** For the random Lu-Kumar network, jobs in the classes 2 and 4 are a.s. not served simultaneously at any time  $t \ge \tau_0$ .

This simple observation is the basis for the proof of Theorem 3.2. It says, in essence, that service at these classes is restricted as if they belonged to the same station, with the condition  $m_2 + m_4 > 1$  implying that this "station" is supercritical. Consequently, the network is unstable. This observation was apparently first made in [BoZ92]. A more general version of it is used heavily in the work on global stability in [DaV00], which is discussed in Section 5.4.

Proof of Theorem 3.2. Set  $\sigma_i$  equal to the sum of the service times at classes 2 and 4 of the *i*<sup>th</sup> job entering the network after time 0. Then,  $\sigma_1, \sigma_2, \ldots$  are i.i.d. random variables with mean  $m_2 + m_4$ . The times at which these jobs are served are disjoint after time  $\tau_0$  because of Lemma 3.3. So, the time of departure from the network of the  $n^{\text{th}}$  of these jobs is at least  $S_n - \tau_0$ , where  $S_n = \sum_{i=1}^n \sigma_i$ . By the strong law of large numbers,

$$S_n/n \to m_2 + m_4$$
 as  $n \to \infty$ 

holds a.s. It follows from this limit and the preceding observation that

$$\limsup_{t \to \infty} D(t)/t \le 1/(m_2 + m_4) \quad \text{as } t \to \infty$$
(3.7)

holds a.s., where D(t) is the number of departures from the network over (0, t].

Let A(t) denote the number of arrivals in the network over (0, t]. Since A(t) is given by a rate-1 Poisson process, it also follows from the strong law that

$$A(t)/t \to 1 \quad \text{as } n \to \infty$$
 (3.8)

holds a.s. The difference A(t) - D(t) gives a lower bound on |Z(t)|. So, by (3.7) and (3.8),

$$\liminf_{t \to \infty} |Z(t)|/t \ge 1 - 1/(m_2 + m_4) > 0 \quad \text{a.s.},$$

with the last inequality holding since  $m_2 + m_4 > 1$ . Consequently,  $|Z(t)| \to \infty$  as  $t \to \infty$ .

#### The Kumar-Seidman network

A somewhat earlier example in [KuS90] exhibits behavior similar to the Lu-Kumar network. The model considered there consists of a reentrant line with two stations and four classes, with route again given by Figure 3.1. The model is again deterministic, with rate-1 arrivals and subcritical service times satisfying  $m_2 + m_4 > 0$ . The authors choose a continuous mass setting for their model, which is somewhat easier to work with. (This setting is also used for the model from [Se94] that is discussed in Section 3.2, where additional background is given.)

The discipline is a *clearing policy*. This requires each station to continue serving a class until there is no "job mass" left at that class, at which point the station begins service at one of the remaining nonempty classes, if there are any. (In the present example, each station has only two classes, and so there is only one such remaining class.) We refer to the network given in Figure 3.1 with this clearing policy as the *Kumar-Seidman network*. A clearing policy might be a practical choice for the discipline when there is a high start-up cost for switching the processing at a station from one task (i.e., class) to another.

The network was assumed to be subcritical, with  $m_2 + m_4 > 1$ . For a direct comparison with the deterministic Lu-Kumar network, we instead assume the more restrictive (3.1). Using the same initial state as in Theorem 3.1, with job mass M initially at class 1, it is not difficult to check that this model is unstable in the same way as the Lu-Kumar network, with the job mass going to infinity as  $t \to \infty$ . Namely, mass starting at and entering class 1 is served at classes 1 and 2. When the mass at class 2 is exhausted, service takes place at class 3 and



immediately afterwards at class 4, where all of the mass has moved. At this last step, service ceases at class 1, where mass now builds up while the mass at class 4 is served. By the time the last mass at class 4 has been served, the mass is 2M at class 1, which completes the cycle. Note that the clearing policy here plays the same role as the priority scheme in the Lu-Kumar network, with the main body of mass being forced to remain together as it moves from class 1 to class 4 during a cycle. This induces underutilization of both stations 1 and 2, and is responsible for the instability of the network.

The discipline of the Kumar-Seidman network depends on its earlier states rather than on the priorities of jobs currently in the network. So, as an example of instability, it is less convincing than the Lu-Kumar network. ([KuS90] also considered a "clear-a-fraction" discipline with the routing in Figure 3.2.) The paper pre-dates [LuK91]. For that reason, networks with the routing in Figure 3.1 (or Figure 3.2), under any discipline, are sometimes referred to as Kumar-Seidman networks.

#### The Rybko-Stolyar network

This network also consists of two stations, with two classes at each station. Jobs are assumed to follow one of two symmetric routes, visiting first one station and then the other, as indicated in Figure 3.2. The classes along the first route are labelled (1,1) and (1,2), in the order of their appearance, and the classes along the second route are labelled (2,1) and (2,2).

The system evolves according to a preemptive SBP discipline, with jobs at (2,2) having priority over jobs at (1,1) and jobs at (1,2) having priority over those at (2,1). That is, jobs at the last class along a route always have priority over jobs at the first class. Jobs are assumed to enter the network according to two independent rate-1 Poisson processes, and are served at the classes (i, k) according to independent exponentially distributed random variables, with rates  $m_{1,1} = m_{2,1}$  and  $m_{1,2} = m_{2,2}$ , and  $m_{i,k} > 0$ .

This model was examined in [RyS92], and is generally referred to as the *Rybko-Stolyar network*. Unlike the model in [LuK91], it is random. [RyS92] showed the following result.

**Theorem 3.4.** The Rybko-Stolyar network, with  $m_{1,2} > \frac{1}{2}$ , is unstable.



Fig. 3.3.

One can see, with some experimenting, that the Rybko-Stolyar network should evolve in the same basic manner as the random Lu-Kumar network, if one matches the classes (1,1), (1,2), (2,1), and (2,2) with the classes 1, 2, 3, and 4 of the latter network. Jobs entering the Rybko-Stolyar network at (1,1) have lower priority than jobs at (2,2) and jobs at (2,1) have lower priority than jobs at (1,2), and so, in either case, must wait until jobs at the latter classes are served. This is analogous to the relationship between classes 1 and 4, and 3 and 2 in the Lu-Kumar network. In fact, the Rybko-Stolyar network "becomes" the Lu-Kumar network if one connects the classes (1,2) and (2,1) as in Figure 3.3.

Both the Lu-Kumar and Rybko-Stolyar networks provide simple examples of unstable subcritical queueing networks, and can be used to motivate more complicated examples. The Lu-Kumar network has the advantage of being a reentrant line; the Rybko-Stolyar network is symmetric with respect to its two stations. A proof of Theorem 3.4 can be given along the same lines as that of Theorem 3.2, except that the step bounding  $\tau_0$  needs to be modified. (One can assume that station 2 is subcritical, from which it will follow that  $\tau_0 < \infty$ .) The proof in [RyS92] is different and does not rely directly on Lemma 3.3. We also note that the Rybko-Stolyar network is stable under the analogs of (3.2) and (3.3) for  $m_{i,k}$ . This was shown in [BoZ92].

#### 3.2. Examples of unstable FIFO networks

The examples given in Section 3.1 are for SBP disciplines that have been specifically designed to impede the even flow of jobs, and therefore "starve" their stations for work. These networks were initially regarded as artificial examples whose instability was not representative of "typical" disciplines, and so received insufficient attention. It turns out that even subcritical queueing networks with the FIFO discipline may be unstable. FIFO is a natural discipline and was a typical choice for the discipline of multiclass networks when multiclass networks started receiving attention. The demonstration of the existence of unstable subcritical FIFO queueing networks therefore had substantial influence on the stability theory of queueing networks.

Examples of subcritical FIFO queueing networks that are unstable were given in [Br94a], [Br94b], and [Se94]. We focus here primarily on the example in [Br94a]. It is the easiest to understand, and the mechanism that causes the uneven flow of jobs may be thought of as an extension of that in [LuK91].

#### An unstable FIFO example

The example in [Br94a] is a reentrant line consisting of two stations. Jobs following the deterministic route first visit station 1 after entering the network, next visit station 2 repeatedly for a total of K times (where K will be chosen large), and then visit station 1 a second time, before exiting the network. We employ the notation (j, k) here for a class, with j = 1, 2 denoting its station and k denoting the order this class is visited among classes of its station. In all, there are K + 2 classes in the network.

Jobs are assumed to enter the network according to a rate-1 Poisson process and have exponentially distributed service times with means  $m_{j,k}$  corresponding to the  $k^{\text{th}}$  visit to the  $j^{\text{th}}$  station, with

$$m_{1,2} = m_{2,1} = c \tag{3.9a}$$

and

$$m_{j,k} = \delta$$
 for  $(j,k) \neq (1,2)$  and  $(j,k) \neq (2,1)$ . (3.9b)

The route and mean service times can be depicted as in (3.10), with the mean service times being given above the arrows pointing from the corresponding classes:

$$\rightarrow (1,1) \xrightarrow{\delta} (2,1) \xrightarrow{c} (2,2) \xrightarrow{\delta} \cdots \xrightarrow{\delta} (2,K) \xrightarrow{\delta} (1,2) \xrightarrow{c} (3.10)$$

One requires c to be close to 1 and  $\delta$  to be small. For the computations in [Br94a],

$$\frac{399}{400} \le c < 1, \quad c^K \le \frac{1}{50}, \quad \delta \le \frac{1-c}{50K^2} \tag{3.11}$$

are used. For instance, one may choose

$$c = \frac{399}{400}, \quad K = 1,600, \quad \delta = 10^{-11}.$$
 (3.12)

With additional effort, less extreme values can be chosen. Note that on account of (3.11),

$$\rho_1 = m_{1,1} + m_{1,2} = \delta + c < 1, \quad \rho_2 = \sum_{k=1}^K m_{2,k} = c + (K-1)\delta < 1, \quad (3.13)$$

and so these networks are subcritical.

One can demonstrate the following result.

**Theorem 3.5.** FIFO queueing networks with the routing in (3.10) and mean service times in (3.9) and (3.11) are unstable.

According to a simulation in [Da95], instability already occurs at K = 4 for appropriate mean service times, although this is likely difficult to show analytically. (The simulation is actually for a variant with slightly different service times than those in (3.9), so that  $\rho_1 = \rho_2$  holds.) We also note that the network can be modified so that it remains unstable, without affecting the main structure of the corresponding proof. For instance, the long string of returns to station 2 in (3.10),

$$\rightarrow (2,1) \rightarrow (2,2) \rightarrow \cdots \rightarrow (2,K) \rightarrow$$

can be replaced by a route segment also involving a third station,

$$\rightarrow (2,1) \rightarrow (3,1) \rightarrow (2,2) \rightarrow (3,2) \rightarrow \cdots \rightarrow (2,K) \rightarrow (3,K) \rightarrow$$

where the service time at (3, k), for k = 1, ..., K, is the same as at (2, k), which is again chosen as in (3.9). So, consecutive returns to a station are not a crucial feature of the model. Similar modifications can be made to the SBP examples in Section 3.1 without affecting their instability.

In order to investigate the evolution of the queueing networks in Theorem 3.5, we employ the following notation. Here,  $Z_{j,k}(t)$  will denote the number of jobs at the class (j, k) at time t, with Z(t) denoting the corresponding vector and |Z(t)|being the total number of jobs. (Recall that since Z(t) does not reflect the order of jobs, more information is needed to specify the state of the corresponding Markov process.) By  $(j, k)^+$ , we will mean the set of classes occurring strictly after (j, k) along the route followed by jobs, and by  $Z_{j,k}^+(t)$  the number of jobs in  $(j, k)^+$ .

Most of the work in demonstrating Theorem 3.5 is for the following induction step.

**Proposition 3.6.** Consider a FIFO queueing network satisfying the routing in (3.10) and mean service times in (3.9) and (3.11), with

$$Z_{1,1}(0) = M, \quad Z_{1,1}^+(0) \le M/50.$$
 (3.14)

Then for some  $\epsilon > 0$ , large enough M, and appropriate T (depending on M),

$$P(Z_{1,1}(T) \ge 100M, \ Z_{1,1}^+(T) \le M) \ge 1 - e^{-\epsilon M}$$
 (3.15)

and

$$P(|Z(t)| \ge M/4 \text{ for all } t \in [0,T]) \ge 1 - e^{-\epsilon M}.$$
 (3.16)

We will later choose  $T \approx 2cM/(1-c)$ . Of course, the factor 50 in (3.14) is not special, although the ratio  $Z_{1,1}^+(0)/Z_{1,1}(0)$  should be small.

Once Proposition 3.6 has been established, the proof of Theorem 3.5 is quick. To see this, suppose Z(0) satisfies (3.14) for some large M. Repeated application of Proposition 3.6 implies that

$$P(|Z(t)| < M/4 \text{ for some } t \ge 0) \le 2\sum_{i=0}^{\infty} e^{-100^i \epsilon M},$$
 (3.17)

which  $\to 0$  as  $M \to \infty$ . All states of the Markov process Z(t) corresponding to the network communicate with one another, and so, by (3.17), no state is recurrent. Hence,  $|Z(t)| \to \infty$  a.s. as  $t \to \infty$ , for any Z(0). This implies Theorem 3.5. Since  $T \approx 2cM/(1-c)$ , one can also employ Proposition 3.6 to show that the number of jobs increases linearly in t, i.e.,

$$0 = \liminf_{t \to \infty} |Z(t)|/t \le \limsup_{t \to \infty} |Z(t)|/t < \infty.$$

Recall that the iterative procedure of viewing time intervals over which the number of jobs in a system grows geometrically, while the system returns to a "multiple" of its original state, was also employed in the proof of Theorem 3.1. Indeed, this is the most natural path to follow in attempting to analyze the asymptotic behavior of many unstable networks. One question one can ask for the FIFO queueing network in Theorem 3.5 is whether there are essentially different ways in which Z(t) can approach infinity. More generally, one can ask about the nature of the Martin boundary of the Markov process associated with this or other unstable queueing networks. These questions remain essentially uninvestigated.

#### Outline of the proof of Proposition 3.6

We outline here the proof of Proposition 3.6. We begin by introducing a sequence of stopping times  $S_1, S_2, \ldots, S_\ell, \ldots$  for the process Z(t). Jobs at either station at a given time t are ordered according to the times at which they are next served, so we can talk about a "first" or "last" job in this sense. (Due to the multiple classes at each station, jobs entering the network earlier may nevertheless be ordered behind more recent arrivals.) Let  $S_1$  denote the time at which the last of the original jobs (jobs at t = 0) at station 1 is served. Let  $S_2, S_3, \ldots, S_\ell, \ldots$  denote the successive times at which the last jobs at station 2 are served, where the ordering is made at  $t = S_{\ell-1}$ . Set  $S_L = S_{L+1} =$  $S_{L+2} = \cdots$ , where  $S_L$  is the time at which station 2 becomes empty. (On account of (3.13),  $\rho_2 < 1$  and so  $L < \infty$  a.s.) We can think of the intervals  $(S_{\ell}, S_{\ell+1}], \ell = 1, 2, \dots$ , as "cycles" at the end of which each job starting at (2, k), k < K, is at (2, k + 1). Note that no job can be served twice at station 2 before every other job there is served once, due to the FIFO discipline. We also let T (which appears in Proposition 3.6) denote the time at which the last job at (1,2) at time  $S_{2K}$  leaves the network.

We break the outline of the proof into four main steps, corresponding to the evolution of Z(t) over the intervals  $(0, S_1]$ ,  $(S_1, S_K]$ ,  $(S_K, S_{2K}]$  and  $(S_{2K}, T]$ . We present here the intuition for each step, with the reader being referred to [Br94a] for a rigorous analysis.

Step 1. Behavior on  $(0, S_1]$ . We assume, as in (3.14), that  $Z_{1,1}(0) = M$  with M large, and that there are few jobs elsewhere in the network. Since  $m_{1,1} = \delta \ll 1$ , one has  $S_1 \ll M$  except on a set of small probability. Also,  $m_{2,1} = c \gg \delta$ , and so at time  $S_1$ , nearly all of the original jobs in the network are still at (2, 1).

#### Table 3.1

This table gives the approximate number of jobs at each class of the network at the successive times  $0, S_1, \ldots, S_{K+1}, T$ . Classes marked with \* are classes having negligible numbers of jobs off sets of small probability. At  $t = S_{K+1}$ , the total number of jobs at (1,1) and (2,1) is approximately  $c^K M$ , which is itself negligible. Note that at t = T, the state is a "multiple" of that at time 0, with factor c/(1-c). Since  $S_K \approx cM/(1-c)$  and  $T - S_K \approx cM/(1-c)$ , one has  $T \approx 2cM/(1-c)$ 

$t \backslash (j,k)$	(1, 1)	(2, 1)	(2,2)	(2, 3)	 (2, K - 1)	(2, K)	(1,2)
0	M	*	*	*	*	*	*
$S_1$	*	M	*	*	*	*	*
$S_2$	*	cM	M	*	*	*	*
$S_3$	*	$c^2 M$	cM	M	*	*	*
÷		:	÷	÷			
$S_K$	*	$c^{K-1}M$	$c^{K-2}M$	$c^{K-3}M$	 cM	M	*
$S_{K+1}$	*	*	$c^{K-1}M$	$c^{K-2}M$	 $c^2 M$	cM	M
$S_{2K}$	*	*	*	*	*	*	M/(1-c)
Т	cM/(1-c)	*	*	*	*	*	*

Moreover, comparatively few new jobs have entered the network up to time  $S_1$ , and so there are few jobs at (1, 1). A schematic diagram for this and following steps is given in Table 3.1.

Of course, since we are working with random events here, the above behavior is sometimes violated. However, such exceptional events occur with probabilities that are exponentially small in M, and one can show they can be ignored without affecting the basic nature of the evolution of Z(t). Here and later on, we therefore neglect these exceptional probabilities. Needless to say, a rigorous proof requires accurate bookkeeping of such exceptional probabilities. We discuss this point after completing the description of Z(t) over [0, T].

Step 2. Behavior on  $(S_1, S_K]$ . We first consider the evolution of Z(t) over  $(S_1, S_2]$ . Over this time interval, the (approximately) M jobs at (2, 1) all move to (2, 2). Since  $m_{2,1} = c$ , the time it takes to serve these jobs is (approximately) cM. The time required to serve other jobs is minimal, so  $S_2 - S_1 \approx cM$ . During this time, (approximately) cM new jobs enter the system, which quickly move to (2,1). Thus, at  $t = S_2$ , there are (comparatively) few jobs in the system except at (2,2) and (2,1), where there are (approximately) M and cM jobs, respectively.

Continuing our reasoning along the same lines, we observe that over  $(S_2, S_3]$ , the jobs at (2,1) and (2,2) advance to (2,2) and (2,3), respectively. Since  $m_{2,2} = \delta \ll 1$ , the time required to serve the jobs at (2,2) is negligible; the time required for the jobs at (2,1) is  $c^2 M$ , so  $S_3 - S_2 \approx c^2 M$ . Over this time,  $c^2 M$  new jobs enter the system, which quickly move to (2,1). So, at time  $S_3$ , there are few jobs in the system except at (2,3), (2,2), and (2,1), where there are M, cMand  $c^2 M$  jobs, respectively. Proceeding inductively, we obtain that at time  $S_K$ , there are M jobs at (2, K), cM jobs at (2, K - 1), and so on down to (2,1), where there are  $c^{K-1}M$  jobs. At station 1, there are few jobs. The elapsed time  $S_K - S_{K-1} \approx c^{K-1} M$ . On account of (3.11),  $c^K$  is small, and so there are about

$$\sum_{\ell=0}^{K-1} c^{\ell} M \approx M/(1-c)$$
 (3.18)

jobs in the system. Likewise,  $S_K \approx cM/(1-c)$ .

We point out that the large number of "quick" classes for station 2, in conjunction with the FIFO discipline, serves to trap most jobs within station 2, and prevent them from reaching class (1,2), until there are few remaining jobs at the "slow" class (2,1). The behavior of this network thus mimics that of the Lu-Kumar network, with the role of the single low priority "quick" class of station 2 being played by these many "quick" classes.

Step 3. Behavior on  $(S_K, S_{2K}]$ . Over the short period of time  $(S_K, S_{K+1}]$ , the evolution of the system changes. The M jobs from (2, K) arrive at (1,2). Since  $m_{2,1} = c$ , these jobs require time cM to be served at station 1, during which time new arrivals at (1,1) will not be served. The cycles  $(S_{\ell}, S_{\ell+1}], \ell = K, \ldots, 2K-1$ , are all of much shorter duration than cM, because of (3.11), as is their union  $(S_K, S_{2K}]$ . By the end of this period, the jobs already at station 2 at time  $S_K$  have already arrived at (1,2); because of (3.18), there are essentially M/(1-c) such jobs. So, at time  $S_{2K}$ , there are essentially M/(1-c) jobs at (1,2) and no jobs elsewhere. Of course, here and elsewhere, we are taking liberties in ignoring "negligible" quantities of jobs and probabilities.

Step 4. Behavior on  $(S_{2K}, T]$ . During  $(S_{2K}, T]$ , the M/(1-c) jobs at (1,2) exit the system. The time required to serve these jobs is cM/(1-c). So,  $T - S_{2K} \approx cM/(1-c)$ . During this time, cM/(1-c) jobs enter the system. These new jobs are obliged to remain at (1,1) until time

$$T = S_{2K} + (T - S_{2K}) \approx 2cM/(1 - c).$$

At this time, there are few jobs elsewhere in the system. So at time T, the state of the system is a "multiple", by the factor c/(1-c), of the state at time 0. This is the type of bound needed in (3.15) of Proposition 3.6. The bound  $c \ge 399/400$ , that we are assuming in (3.11), will be sufficient to derive (3.15) when the above argument is carried out rigorously. (Presumably, the system exhibits the same behavior when c > 1/2.)

We still need to demonstrate (3.16) of Proposition 3.6, which gives a lower bound on |Z(t)| over [0, T]. The previous reasoning in fact shows that, except on a set of small probability, |Z(t)| will not drop much below M on [0, T]. This is because, up to time  $S_K$ , most of the original M jobs at (1,1) remain in the system, with there being approximately M/(1-c) jobs in the system at time  $S_K$ . Since  $m_{1,2} = c$ , before most of these jobs have left the system, an additional  $cM/2(1-c) \gg M$  jobs enter the system, which are trapped at (1,1) until time T. The bound in (3.16) follows from a rigorous version of this reasoning. We stated during the outline of the proof of Proposition 3.6 that the probabilities of the exceptional events we neglected were exponentially small in M. Here, we provide a short summary of the approach used in [Br94a] to show this, referring the reader there for more detail.

Let  $X_1, X_2, X_3, \ldots$  be i.i.d. mean-1 exponentially distributed random variables, with  $Y_n = X_1 + \cdots + X_n$ . Then, for each  $\alpha > 0$ , there exists  $\beta > 0$ , such that for  $n \ge 1$ ,

$$P(|Y_n - n|/n > \alpha) \le e^{-\beta n}.$$
(3.19)

This is a simple large deviations bound that can be demonstrated in the usual way, by applying Markov's Inequality to the moment generating function of  $Y_n$ . It extends immediately to i.i.d. exponentially distributed random variables with other means. The variables  $Y_n$  can also be inverted to obtain analogous exponential bounds on the number of exponentially distributed random variables occurring by a given time.

For a given class (j, k), one can write

$$Z_{j,k}(t) = Z_{j,k}(0) + A_{j,k}(t) - D_{j,k}(t), \qquad (3.20)$$

where  $A_{j,k}(t)$ , respectively  $D_{j,k}(t)$ , are the total number of jobs arriving at, respectively departing from, (j, k) over (0, t]. By applying (3.19) repeatedly, one can derive upper and lower bounds on  $A_{j,k}(t)$  and  $D_{j,k}(t)$ , and hence on  $Z_{j,k}(t)$ , over the times  $S_1, S_2, \ldots, S_{2K}, T$  defined earlier. For instance, replacing 1/50 by  $\eta$  for readability, the first bounds employed in [Br94a] are exponentially small upper bounds in M for the probabilities that

$$S_1 > 2\eta M$$
,  $A_{1,1}(S_1) > 3\eta M$ ,  $D_{2,1}(S_1) > 3\eta M$ ,

with the first of these bounds together with (3.19) being used for the last two bounds. The last two bounds are then applied to obtain exponentially small upper bounds on

$$Z_{2,1}(S_1) < (1 - 3\eta)M, \quad Z_{1,1}(S_1) + Z_{2,1}(S_1) > (1 + 4\eta)M,$$
  
 $Z_{2,1}^+(S_1) + D_{1,2}(S_1) > 4\eta M.$ 

Similar bounds, such as on

$$Z_{2,k}^+(S_k) + D_{1,2}(S_k) > 4\eta M, \quad k = 2, \dots, K,$$

are then obtained. These last bounds limit the rate at which jobs can move through the system. Together with further estimates, these bounds enable one to rigorously justify the reasoning employed in Steps 1-4.

#### Another unstable FIFO example

The following unstable FIFO reentrant line is given in [Se94]. It consists of four stations, each visited three times, with route given in Figure 3.4. As in Section 3.1, class k denotes the  $k^{\text{th}}$  class along the route.



FIG. 3.4. The service time for a class is given along the route immediately after the class.

The model is a continuous, deterministic analog of a queueing network, whose state at each class is given by a nonnegative real number representing the quantity of "mass" there. As time evolves, this mass is served in a continuous and deterministic manner. This model is an example of a *fluid network*. Fluid networks were mentioned briefly in Section 1.3 and are similar to the fluid models that are considered in detail in Chapter 4; we omit a systematic discussion here. We denote by  $Z_k(t)$  the amount of mass at class k at time t, by Z(t) the corresponding vector, and by |Z(t)| its magnitude. In analogy with our definition for queueing networks, we will say this fluid network is unstable if for some initial state,  $|Z(t)| \to \infty$  as  $t \to \infty$ .

As indicated in the figure, each station has one "slow" class and two "quick" classes. The "slow" classes have service times  $c_1$  and  $c_2$ , which are assumed to satisfy

$$2(c_1)^2 < c_2 < 1, \quad c_1 > 1/2;$$
 (3.21)

the "quick" classes have 0 service times. The service rates are given by the reciprocals  $1/c_i$  and  $\infty$ ; service at the "quick" classes is therefore instantaneous. The presence of the "quick" classes nonetheless affects the flow of mass through the system since, according to the FIFO discipline, mass at such a class must wait until earlier arriving mass at the station's "slow" class is served. As with previous reentrant lines, we assume mass enters the system (in this case, deterministically) at rate 1. These features are similar to those in [KuS90] and [LuK91].

It is easy to see that this network is subcritical, since the sum of service times at each station is less than 1. The amount of mass in the network goes to  $\infty$  as  $t \to \infty$ , however, for appropriate initial states. Such a state is given by

$$Z_{2}(0) = 2c_{1}M, \quad Z_{8}(0) = M, \quad Z_{10}(0) = \frac{c_{2} - 2(c_{1})^{2}}{2c_{1}c_{2}}M, \quad (3.22)$$
$$Z_{12}(0) = \frac{c_{2} - 2(c_{1})^{2}}{4(c_{1})^{2}c_{2}}M, \quad Z_{k}(0) = 0 \text{ elsewhere,}$$

where the mass at class 12 is understood to have arrived before that at class 10. (Since the service at both classes 2 and 8 is instantaneous, the ordering there is not important.) From this, one obtains:

#### **Theorem 3.7.** The FIFO fluid networks defined above are unstable.

As with the networks considered so far in this chapter, the proof consists of an iterative argument, with the state of the system returning to a geometrically growing "multiple" of the original state after each iteration. As before, |Z(t)|will grow linearly in t.

[Se94] also considers the variant of the above model with mass leaving the system after classes 3, 6, and 9 (as well as after class 12), and mass, at unit rate, entering the system at classes 4, 7, and 10 (as well as at class 1). Since the analog of Theorem 3.7 for this variant is somewhat easier to show, this is done in [Se94] and the proof of Theorem 3.7 is summarized. In both cases, replacement of "instantaneous" classes by "quick" classes, with service times  $\delta > 0$ , and the fluid network by the corresponding queueing network considerably complicates the bookkeeping necessary for keeping track of jobs. Although not published, this stochastic version is presumably doable.

#### An unstable FIFO network with quick service times

The examples of unstable subcritical networks given so far all have at least one class with mean service time greater than 1/2. In each of these examples, it follows that the traffic intensity  $\rho_j$  is greater than 1/2 at some station j. What happens when  $\rho_j$  is uniformly small at all stations? Must all such FIFO queueing networks be stable? The following example from [Br94b] shows this is not the case.

Jobs are assumed to follow one of two nearly identical routes, the "upper" and "lower" routes, at the end of which they exit from the system. As illustrated in (3.23), there are J stations along each route. (Because of space considerations, we label only the stations but not the classes along each route.)

Jobs are assumed to enter each route at rate 1/2. Along each route, the stations j = 2, ..., J are each visited seven times; at each such station, the first visit is "slow" and the remaining six visits are "quick". Only the visit to station 1 at the end of the lower route is "slow"; the three earlier visits to station 1 along both routes are "quick".

We now give a precise description of the network depicted in (3.23). We employ the notation (i, j, k) for a class, with  $i = u, \ell$  denoting its route,  $j = 1, \ldots, J$  denoting its station, and k denoting the order this class is visited, among classes of its station along this route. One has  $k = 1, \ldots, 7$  for  $j = 2, \ldots, J$ ; k = 1for i = u and j = 1; and k = 1, 2, 3 for  $i = \ell$  and j = 1. The two types of jobs are assumed to enter the system according to independent rate-1/2 Poisson processes. Service times of jobs are independent and exponentially distributed, with means

$$\begin{array}{ll} c & \text{at} & (i,j,1), & \text{for } i = u, \ell \text{ and } j = 2, \dots, J, \\ c & \text{at} & (\ell,1,3), & (3.24) \\ \delta & \text{at} & (i,j,k), & \text{for } i = u, \ell, \ j = 2, \dots, J \text{ and } k = 2, \dots, 7, \\ \delta & \text{at} & (u,1,1), (\ell,1,1) \text{ and } (\ell,1,2). \end{array}$$

We assume that

$$0 < c \le \frac{1}{100}, \quad 0 < \delta \le c^8, \quad J = \lfloor 2c^{-1}\log(c^{-1}) \rfloor.$$
 (3.25)

Each of the stations  $2, \ldots, J$  therefore has one comparatively slow and six (very) quick classes for each type of job; station 1 has only the single quick class for the upper jobs, and two quick classes and one slow class for lower jobs. The choice of parameters is made for technical reasons. (The coefficient 2 in the definition of J has been chosen so that  $(1 - c)^{-J} \sim c^{-2}$ . The bound  $c \leq 1/100$  is somewhat arbitrary.) One can think of this family of networks as being constructed by piecing together J copies of the middle section of networks of the type in (3.10). (We only need K = 7 here, although  $K \geq 7$  can instead be used.)

Under (3.25),

$$\rho_j \leq c + 6\delta \leq 2c$$
 for  $j = 1, \dots, J$ .

So, by choosing c small, the traffic intensity can be chosen as small as desired for each j. Nonetheless, the following is true.

**Theorem 3.8.** FIFO queueing networks with the routing in (3.23) and mean service times in (3.24)-(3.25) are unstable.

To demonstrate Theorem 3.8, one employs an appropriate analog of Proposition 3.6. One can then argue exactly the same way as immediately after Proposition 3.6 to finish the proof of Theorem 3.8. The spirit of the proof of this analog is similar to that of Proposition 3.6, although details are more involved. The purpose of the upper jobs is solely to restrict the flow of the lower jobs. Because of the multiple stations employed here, the same reasoning that shows, in the previous network, that the main body of jobs remains close together, cannot be applied directly. However, with the control resulting from the upper jobs, one can analyze the flow of lower jobs much as was done in Proposition 3.6. For more details, the reader is referred to [Br94b] or [Br95]. Presumably, the analog of Theorem 3.8 holds for an appropriate reentrant line, most likely with a route corresponding to that of the lower jobs in (3.23), although the reasoning given in [Br94b] no longer suffices.

Theorem 3.8 has the following interesting consequence. One can compare any FIFO queueing network satisfying (3.24)–(3.25) with the network that is obtained from it by replacing (3.24) with the assumption that the mean service time  $m_{i,j,k} = c$  at every class. The lengths of the mean service times for the new network are, of course, everywhere at least as great as those of the original network. This new network is a subcritical FIFO network of Kelly type with  $\rho_j \leq 7c$  for all *j*. Such a FIFO network has a stationary distribution that is given by (2.4) and (2.9). In particular, the stationary probability of there being *n* jobs at a given station is at most  $(1 - 7c)(7c)^n$ ,  $n \geq 1$ , which means that the network is in fact "very stable" for small *c*. This comparison shows that decreasing the mean service times within a queueing network may result in making it unstable.

At present, there is a lack of general criteria for the stability of FIFO queueing networks. The difficulties are illustrated by the previous examples. They do not, however, rule out two possible criteria for stability. Set  $m_j^{\min}$  ( $m_j^{\max}$ ) equal to the minimum (maximum) over all  $m_k$ ,  $k \in C(j)$ , where j is fixed, and set  $m_j^R = m_j^{\min}/m_j^{\max}$ .

The first criterion is that, for given  $m^R > 0$ , there exists an r > 0, so that if a FIFO network satisfies  $m_j^R \ge m^R$  and  $\rho_j \le r$ , for all j, then it is stable. The second criterion is that, for given r < 1, there exists an  $m^R < 1$ , so that if a FIFO network satisfies  $m_j^R \ge m^R$  and  $\rho_j \le r$ , for all j, then it is stable.

According to the first criterion, if the mean service times at a given station are not too different, then small enough traffic intensities suffice for stability independently of the specific structure of the network. Similarly, according to the second criterion, a subcritical network, with  $\rho_j \leq 1 - \epsilon$  for all j and given  $\epsilon > 0$ , is stable as long as the ratios  $m_j^R$  are close enough to 1. Networks of Kelly type, with  $m_j^R \equiv 1$ , make up the limiting case in the latter scenerio. Note that the first criterion becomes elementary if the routing of the network is instead specified before r is chosen, since the total number of classes is then fixed. In that setting, r can be chosen small enough so that  $\Sigma_j \rho_j < 1$ , which implies the network is stable by the example at the end of Section 4.4. No progress has been made toward justifying or disproving these criteria.

#### 3.3. Other examples of unstable networks

In the previous two sections, we have given a number of examples of subcritical queueing networks that are unstable. Here, we give several other examples of unstable queueing networks. In some of these cases, more can be said about the region of stability as the traffic intensity  $\rho$  varies.

The first example we consider is from [DaWe96]. It consists of a subcritical SBP reentrant line of Kelly type that is unstable. (Recall that a network is of Kelly type if all classes at a given station have the same mean service time.) This contrasts, of course, with the stability of FIFO networks of Kelly type that was shown in Chapter 2.

The second example is from [Du97]. It exhibits a family of SBP networks for which the region of stability is nonmonotone in the parameter  $\rho$ , behavior that is shared with the last example in the previous section. The region of stability is, moreover, explicitly calculated and is not convex.

The last example is from [BaB99]. The family of networks given there is FIFO, but with the added feature that the number of jobs on one of the given

routes in the network is bounded at any given time. The region of stability here is also calculated explicitly and is neither monotone nor convex. The boundary is, moreover, self-similar around one of its boundary points.

#### An unstable network of Kelly type

In contrast to FIFO networks, it is not sufficient for a subcritical queueing network to be of Kelly type for it to be stable. It is not difficult to produce a reentrant line, with an appropriate SBP discipline, that exhibits this behavior. Such an example is provided in [DaWe96].

This example consists of two stations, with each station being visited three times, and is depicted in Fig 3.5. The priority scheme is (6,5,1) at station 1 and (3,2,4) at station 2, e.g., the last class visited at station 1 has the highest priority and the first class visited there has the lowest priority. The discipline is preemptive. We assume, as usual, that jobs enter the network according to a rate-1 Poisson process. We also assume that the service times are exponentially distributed, with mean 0.3 at each class. The network is clearly subcritical, with  $\rho = (.9, .9)$ .

## **Theorem 3.9.** The static priority reentrant line of Kelly type in Figure 3.5 is unstable.

We motivate Theorem 3.9 by comparing the evolution of the reentrant line there with the evolution of the Lu-Kumar network; a rigorous argument can be given by mimicking the proof of Theorem 3.2. First note that if one both "combines" the last two classes of station 1, k = 5 and k = 6, into a single class and "combines" the first two classes of station 2, k = 2 and k = 3, into a single class, one obtains the four-class reentrant line whose route and mean service times are given in Figure 3.6.

We assign to the classes of the new reentrant line the same priority scheme as in the Lu-Kumar network, with k' = 4 having priority over k' = 1, and k' = 2 having priority over k' = 3. Adding the service times at the combined



FIG. 3.5. This reentrant line has priority scheme (6,5,1) at station 1 and (3,2,4) at station 2. All service times have mean 0.3.



FIG. 3.6. "Combining" classes k = 5 and k = 6 into a single class and k = 2 and k = 3 into a single class produces this reentrant line, with the above mean service times.

classes produces service times having gamma distributions with means 0.6, in both cases.

Some thought shows that this new network is a natural "projection" of that in Figure 3.5, in the sense that there is a pathwise correspondence between the evolution of jobs in the two networks, with the understanding that jobs at k = 5and k = 6, respectively at k = 2 and k = 3, in the old network are combined into k' = 4, respectively k' = 2, in the new network. This correspondence relies on the assigned priority scheme for the old network: both classes 5 and 6, respectively classes 2 and 3, have higher priority than class 1, respectively class 4, and so the information lost in the projection is immaterial in assigning the priority of service to jobs in the combined class with respect to the other remaining class. Moreover, since class 6 has higher priority than class 5 and class 3 has higher priority than class 2 in the old network, jobs currently in service in the new network at class 4 and at class 2 will not be preempted in the middle of their service by other jobs there. This allows us to maintain the pathwise correspondence between jobs in the two networks.

Because of the above relationship between the two networks, instability of one network implies instability of the other. The projected network is the same as the Lu-Kumar network in Theorem 3.2, except that the exponentially distributed service times at classes 2 and 4 are replaced by service times with gamma distributions there, each with mean 0.6. As mentioned after the statement of Theorem 3.2, the assumption that service times are exponential is not needed there, if one is willing to allow a more general state space. Since 0.6 > 0.5, as required in the theorem, it will follow that the network in Figure 3.6 is unstable. One can also show Theorem 3.9 without referencing Theorem 3.2, but instead by mimicking its proof; the setup there has some similarity with the "method of stages" employed in Section 2.4.

#### Two examples with nonconvex regions of stability

The first example is an SBP queueing network from [Du97]. As in the Rybko-Stolyar network of Section 3.1, jobs travel along one of two routes that are oriented in opposite directions. In the present setting, there are three stations having two classes each, with the classes labelled according to the route and position along the route, as in Figure 3.7. Jobs of the second route are assumed



FIG. 3.7. The priority scheme favors the classes (2,3), (2,2), and (1,3), respectively, at the stations 1, 2, and 3.

to have priority over jobs of the first route at each of the first two stations, with this being reversed at the third station. The discipline is preemptive. Jobs enter the routes according to independent rate- $\alpha_k$  Poisson processes, i = 1, 2, and have independent exponentially distributed service times with means  $m_{i,k}$ .

Consider the functions  $F_1(\rho)$ ,  $F_2(\rho)$ , and  $F_3(\rho)$  defined by

$$F_1(\rho) = (\rho_{1,1} + \rho_{2,3} - 1) \lor (\rho_{1,2} + \rho_{2,2} - 1) \lor (\rho_{1,3} + \rho_{2,1} - 1) \lor (\rho_{1,3} + \rho_{2,2} - 1),$$
  

$$F_2(\rho) = (\rho_{1,3} + \rho_{2,3} - 1)(1 - \rho_{1,2} - \rho_{2,2}) - (\rho_{1,2} + \rho_{2,3} - 1)(1 - \rho_{1,3} - \rho_{2,1}),$$
  

$$F_3(\rho) = (\rho_{1,3} + \rho_{2,3} - 1) \land [(\rho_{1,3} - \rho_{1,2}) \lor F_2(\rho)],$$

where  $\rho_{i,k} = \alpha_k m_{i,k}$ . (Recall that  $a \lor b = \max(a, b)$  and  $a \land b = \min(a, b)$ ). Also, set  $F(\rho) = F_1(\rho) \lor F_3(\rho)$ . The regions of stability/instability for the network can be explicitly written in terms of these functions.

**Theorem 3.10.** The SBP network in Figure 3.7 is stable if  $F(\rho) < 0$  and unstable if  $F(\rho) > 0$ .

The reader should not focus too much on the specifics of  $F(\rho)$ . For our purposes, it is enough to observe that the regions of stability/instability are explicit and are considerably more complicated for this relatively elementary network than for either the Lu-Kumar or Rybko-Stolyar network. Moreover, (a) the region of stability  $F(\rho) < 0$  is not convex and (b) stability is not monotone in  $\rho$ , i.e., one can find  $\rho < \rho'$  with  $F(\rho) > 0$  and  $F(\rho') < 0$ . Both properties follow from Theorem 3.10 by setting  $\rho_{1,1} = \rho_{2,2}$ ,  $\rho_{1,2} = \rho_{2,3}$ , and  $\rho_{1,3} = \rho_{2,1}$ , and restricting  $F(\rho) < 0$  to this three dimensional subspace. Slices of this region at fixed values of  $\rho_{1,1}$  can then be analyzed. As a quick check on the consistency of the instability condition  $F(\rho) > 0$ , note that  $F_1(\rho) > 0$  if any of the three stations is supercritical.

In order to show both directions of Theorem 3.10, [Du97] employs ergodic/ transience criteria developed in [MaM81] for reflected random walks on  $\mathbf{Z}_{+,0}^d$ . Motivation for the definition of  $F(\rho)$  is provided by fluid equations that correspond to the original network. These equations are related to fluid models, which will be studied in the next chapter.

The second example we discuss is from [BaB99] and examines a family of networks consisting of a single route with *controlled jobs* and multiple transverse


FIG. 3.8. The controlled jobs move horizontally and the cross jobs move vertically. The number of controlled jobs in the network at a given time is restricted, with excess controlled jobs required to wait at the outside buffer on the left. There are currently 3 jobs waiting at the outside buffer.

routes with cross jobs. ([BaB99] employs the terms controlled customers and cross customers.) Controlled jobs moving along their route visit each of the J stations in the network once; the cross jobs each visit only a single station before exiting the network. No more than L controlled jobs are allowed in the network at a given time; when there are more than L such jobs, the excess jobs wait at an outside buffer until a controlled job leaves the network, at which point the first such job enters the network. There is no such restriction on cross jobs. (See Figure 3.8.) All jobs are served according to the FIFO discipline at each station. These networks can be thought of as a simple model with window flow control for packet-switched communication networks.

Detailed analysis of such a system is possible under certain restrictive assumptions. Set J = 2 and L = 1, and assume that the service times for all jobs (both controlled and cross) are deterministic and take value 1. Also, assume that the interarrival times of the cross jobs at station 1 are deterministic and take value  $\tau$ , with  $\tau \ge 1$ , and that there are never any cross jobs at station 2. There are only minimal assumptions on the interarrival times for the controlled jobs, namely that they define a stationary and ergodic point process with some intensity  $\lambda$ .

On account of the outside control on jobs, this network differs from the other examples considered in this chapter. Also, because of the deterministic aspects of this model, it is more appropriate to weaken the definition of stable used elsewhere, and only require here that the model support a stationary process (that need not be ergodic). The definition of unstable remains the same as before.

This network is interesting because of how its region of stability depends on  $\tau$ . For fixed  $\tau$ , stability is monotone in the parameter  $\lambda$ , and does not otherwise depend on the arrival process for the controlled jobs; there is a  $\bar{\lambda}$  so that for  $\lambda < \bar{\lambda}$ , the network is stable, while for  $\lambda > \bar{\lambda}$ , it is unstable. In Figure 3.9,  $\bar{\lambda}$  is graphed as a function of  $\lambda_1 = \tau^{-1}$ , with the heavy line being both the graph of  $\bar{\lambda}$  and the boundary between the stable and unstable regions.

As in the example in Figure 3.7, the region of stability of the network is not convex. Also, as in previous examples, it is not monotone. (This nonmonotonic-



FIG. 3.9. This is the graph of  $\bar{\lambda}$  as a function of  $\lambda_1 = \tau^{-1}$ , which is the boundary between the stable and unstable regions. (The heavy line is the graph of  $\bar{\lambda}$ ; the dotted lines help one track the piecewise linear increments of  $\bar{\lambda}$ .) The graph of  $\bar{\lambda}$  is self-similar around  $(0, \frac{1}{2})$ . (Example reprinted from [BaB99] with permission from Baltzer.)

ity is in terms of the intensity of the cross traffic, though, instead of in terms of the mean service times, as in the previous examples.) Moreover, the graph of  $\bar{\lambda}$  is self-similar around  $(0, \frac{1}{2})$ , as is indicated in Figure 3.9. An open question is what part of this behavior remains when the deterministic interarrival and service times in the model are randomized.

Two other families of networks with nonmonotone regions of stability are given in [Br98a] and [DaHV99]. In both cases, decreasing the service time distributions can destabilize the network. In [Br98a], this is done by showing a network can be stabilized by inserting a single class station immediately before the visit to each class. (Related deterministic work was done in [Hu94].) In [DaHV99], this is done in the context of global stability, which is discussed in Section 5.4.

# Chapter 4

# Stability of queueing networks

In Chapter 2, we demonstrated the stability of several families of queueing networks by explicitly computing their stationary distributions. For queueing networks with other disciplines, or with nonexponential interarrival and service times, one cannot expect such explicit expressions. So, in order to investigate the stability of more general queueing networks, another approach is needed. Such an approach should be more qualitative and less computational in nature.

The approach we employ in this chapter to study the stability of queueing networks employs fluid limits and fluid models. Fluid models were discussed briefly in Section 1.3; we will examine both fluid limits and fluid models in detail here. Employing these tools, one can reduce the study of queueing networks to their simpler deterministic analogs. The basic theory is given here; applications will be given in Chapter 5.

In the next several paragraphs, we give some background on previous results on the stability of queueing networks. We recall that a queueing network is defined to be stable if its underlying Markov process is positive Harris recurrent. As mentioned earlier, when the state space is countable and all states communicate, this definition reduces to the usual definition of positive recurrence. Results in earlier works are typically stated within this more restrictive framework.

Interest in whether general families of queueing networks are stable has developed since the 1980's. Most early work was restricted to single class networks (see, e.g., [Bo86], [Si90], [BaF94], [ChTK94], and [MeD94]). Work on multiclass networks usually dealt with deterministic systems, with either discrete or continuous job mass (see, e.g., [PeK89], [KuS90], [LuK91], and [Ku93]); examples of both types of systems were given in Chapter 3. "Stability" for such deterministic systems was shown in various cases in the above literature, with stability, in this context, typically meaning that the quantity of job mass in the system

converges to 0 or remains bounded over time. As in the stochastic setting,  $\rho < e$  is the natural requirement for stability.

Examples of unstable deterministic networks, with  $\rho < e$ , were given in [KuS90] and [LuK91]. An example of an unstable network in the stochastic setting was given in [RyS92]; these and other examples were discussed in Chapter 3. Such examples illustrate the importance of the discipline in determining whether a queueing network is stable.

It has long been believed that queueing networks should be stable under general assumptions. Inclusive conditions for stability are not known (and likely do not exist). Instead, stability is typically shown in the context of a specific discipline, under  $\rho < e$  and perhaps other constraints. The approach we present in this chapter, using fluid limits and fluid models, reduces the study of such problems to a simpler, deterministic setting.

The foundation for this approach was given in [RyS92]. For a two station FIFO queueing network, with the same routing as in Figure 3.2 and with exponential interarrival and service times, the authors showed stability when  $\rho < e$ . Their argument involved showing the "stability" of solutions of a related deterministic system, and showing that rescaled solutions of the random system remain close to those of the deterministic system. As pointed out in [RyS92], this procedure is, in principle, quite general in nature. However, for all but the simplest systems, technical problems arise when comparing solutions of the random and deterministic systems.

Independently, [St95] and [Da95] developed criteria for the stability of queueing networks, in terms of the stability of limits of rescaled solutions of the network. (In the terminology of this chapter, [St95] used fluid limits and [Da95] used both fluid limits and fluid models.) [St95] assumed exponential distributions for the interarrival and service times; [Da95] considered more general distributions, but at the price of requiring the use of Markov processes with general state space. Applications illustrating this approach are given in [Da95].

The material in this chapter is based on the approach taken in [Da95]. The main theorem of the chapter, Theorem 4.16, corresponds to Theorem 4.2 in [Da95]. Our approach here is a modification of that in [Br98a]. Care has been taken to give a detailed presentation of the material, including that cited in [Da95]. As a consequence, the chapter is quite long; in the remainder of the introduction, we summarize its contents.

#### Summary of chapter

Chapter 4 consists of five sections. In Section 4.1, we present the foundations for the Markov processes we will need. We first give a detailed construction of the underlying Markov process for an HL queueing network. We next summarize relevant results from general Markov process theory. The third part of the section defines Harris recurrence and positive Harris recurrence, and gives an alternative formulation, in Theorem 4.1, that we will use.

We recall that a queueing network is e-stable, if its underlying Markov process is ergodic. In the last part of Section 4.1, we give general conditions under which ergodicity holds. At the end of Section 4.4, we will use this to give criteria under which a queueing network is *e*-stable. In the continuous time, countable state space setting, positive Harris recurrence and ergodicity are equivalent.

Much of the material in Section 4.1 may be unfamiliar to readers. We point out that concepts such as positive Harris recurrent and petite are motivated by similar concepts in both the discrete time and countable state space settings. Various proofs either rely on, or are motivated by, similar results in these settings. Those readers who are interested in further background can refer to Section 4.5, which serves as an appendix to this section.

When the interarrival and service times are exponentially distributed, the underlying Markov process of the queueing network can, for many disciplines, be constructed on a countable state space. This considerably simplifies the preparation that is required to derive Theorem 4.16. In particular, one does not require the general machinery that is introduced in Section 4.1. For readers wishing such a "shortcut", we present a summary, at the end of the different sections, saying how the general approach presented here can be modified. (In Section 4.4, the summary is instead presented after Theorem 4.16.) Here in the introduction, we will also point out these shortcuts.

In Section 4.2, we present two results on bounded sets that we will need later on. The first, Proposition 4.6, is a variant of Foster's Criterion, which we refer to as the Multiplicative Foster's Criterion. The main condition is that, off of a bounded set, the Markov process  $X(\cdot)$  have a uniformly negative drift on an appropriate time scale. One also requires that the bounded set be either petite or uniformly small, which are defined in Section 4.1. (In essence, petite means that all sets, weighted according to some measure, are "equally accessible" from any point in the petite set. Uniformly small is a somewhat stronger concept that is defined similarly.) One concludes that  $X(\cdot)$  is positive Harris recurrent when the condition petite is assumed and ergodic when uniformly small is assumed. In the countable state space setting, the petite and uniformly small conditions can be dropped, since the empty state will be equally accessible from all points in the bounded set. Moreover, Theorem 4.1, from Section 4.1, is not needed for the proof of the Multiplicative Foster's Criterion in the countable state space setting.

The other result from Section 4.2, that we will need later, is Proposition 4.7. It says that when the interarrival times of a queueing network satisfy certain conditions, bounded sets will be uniformly small, and hence also petite; this enables us to employ the Multiplicative Foster's Criterion. The proposition is not needed in the countable state space setting.

Section 4.3 introduces fluid models and fluid limits. The first part of the section recalls the queueing network equations that were discussed briefly in Section 1.3; we provide more detail here. Fluid model equations are the deterministic analogs of the queueing network equations. They were also discussed briefly in Section 1.3; we provide further detail in the second part of Section 4.3. There, emphasis is placed on the basic fluid model equations, which are the fluid model equations that do not depend on a specific discipline. Proposition 4.11, in the subsection, presents several elementary results on fluid models that will be

used later. Two examples are given that illustrate nonuniqueness of fluid model solutions and instability when the system is subcritical.

Fluid limits are introduced in the last part of Section 4.3. They provide a rigorous connection between the queueing network and fluid model equations, with the latter being satisfied by limits of solutions of the former, under a "law of large numbers" scaling. Variants of fluid limits have been present in the literature at least since [Ne82] (see also [ChM91]).

The use of fluid limits, as in Proposition 4.12 of this section, makes the relationship between queueing network and fluid model equations precise. The connection between queueing network and fluid model equations will be important in Section 4.4, where we use the stability of fluid models (and, indirectly, of fluid limits) to show the stability of queueing networks. The approach taken in this section remains essentially the same when employed in the countable state space setting.

In Section 4.4, we demonstrate Theorem 4.16, which is the main result on the stability of queueing networks. As assumptions, one needs bounded sets to be petite and the fluid limits to be stable. Proposition 4.7, from Section 4.2, gives sufficient conditions for the former property to hold. The Multiplicative Foster's Criterion will be used, in conjunction with Proposition 4.7 and the bounds given in Section 4.4, to demonstrate Theorem 4.16. The work required for this simplifies considerably in the countable state space setting.

After the theorem, we provide various commentary, such as on modifications of its assumptions. One such modification, substituting uniformly small for petite, suffices for the stronger conclusion that the network is e-stable. The result, Theorem 4.17, follows by applying Theorem 4.3 from Section 4.1, and otherwise reasoning as in the proof of Theorem 4.16. As mentioned earlier, in the continuous time, countable state space setting, stability and e-stability are equivalent.

# 4.1. Some Markov process background

In this section, we first introduce the Markov processes that are associated with HL queueing networks. We next consider these Markov processes in an abstract setting, in which we define positive Harris recurrence. We then present a useful alternative characterization of positive Harris recurrence which will be applied to queueing networks in the following sections. Some of the background for this material is relegated to Section 4.5, which serves as an appendix to this section.

We note that in the countable state space setting, the material that is needed from this section is minimal. The construction of the Markov process simplifies, since sample paths are piecewise constant with finite jump rates. Moreover, standard recurrence concepts from Markov chain theory apply, and so the discussion of Harris recurrence that is given here is not needed. More detail is given at the end of the section.

#### Definition of the Markov process

In this subsection, we give a careful construction of the Markov process that is associated with a given HL queueing network. Readers are encouraged to consider this material, but those who are not interested in the technical details should feel free to skip ahead to the next subsection. If the reader does so, he/she should keep in mind that we are defining continuous time Markov processes, on an appropriate space and with dynamics that correspond to the queueing networks of interest. One should note the "norm" |x| of a state x, which is given in (4.3): it is the sum of components corresponding to the queue lengths, and residual interarrival and service times of the state. It will be employed later on in the chapter.

We begin by defining the state space  $(S, \mathscr{S})$  of the Markov process. In order to motivate the different components of states of the space, we will repeatedly allude to the various queueing network quantities they will correspond to.

The state of the Markov process, at any time, will be given by a point x = (y, r), where

$$y \in (\mathbf{Z} \times \mathbf{R})^{\infty} \times \mathbf{R}^{|\mathcal{A}|} \times \mathbf{R}^{K} \quad \text{and} \quad r \in \mathbf{R}^{K},$$
(4.1)

and the coordinates are subject to appropriate positivity restrictions. Recall that  $\mathcal{A}$  is the set of classes at which external arrivals are allowed. It is assumed that only a finite number of the pairs of coordinates of  $(\mathbf{Z} \times \mathbf{R})^{\infty}$ , indexed by i, are nonzero. For such a pair, the first coordinate  $k_i, k_i = 1, \ldots, K$ , is to be interpreted as the current class of a job in the network, with the second coordinate  $s_i, s_i \geq 0$ , measuring how long ago the job entered this class, where  $s_i$  is given in descending order. The job with the largest second coordinate, among those with first coordinate k, is thus the first or "oldest" job of class k. (For "ties" where two or more pairs have the same second coordinate, order the job with the smaller k coordinate first.)

We denote by  $z = (z_1, \ldots, z_K)$  the number of jobs in each class, and set  $|z| = \sum_{k=1}^{K} z_k$ . The vector r is assumed to have coordinates  $r_k \in [0, 1]$ ,  $k = 1, \ldots, K$ , with  $r_k = 0$ , for  $z_k = 0$ , and which sum to 1 over each nonempty station j. For each nonempty class k, the coordinate  $r_k$  is to be interpreted as the service rate of the oldest job of this class, with other jobs receiving no service. (This is the HL property.) The coordinates  $u_k$ ,  $u_k > 0$ , of  $\mathbf{R}^{|\mathcal{A}|}$  are to be interpreted as the remaining time before the next arrival at k of a job from outside the network.) The coordinates  $v_k$  of  $\mathbf{R}^K$ , for the y component, are the residual service times for the oldest job at each class  $k, k = 1, \ldots, K$ , with  $v_k > 0$  except when  $z_k = 0$ , in which case we set  $v_k = 0$ . We denote by u and v the corresponding vectors, and set  $|u| = \sum_{k \in \mathcal{A}} |u_k|$  and  $|v| = \sum_{k=1}^{K} |v_k|$ . We denote by S the space given by (4.1) and the above restrictions on the

We denote by S the space given by (4.1) and the above restrictions on the coordinates. We wish to specify a metric on S. For this, it is convenient to denote by  $\tilde{r}_i$  the service rate assigned to the  $i^{\text{th}}$  job,  $i = 1, \ldots, |z|$ . The metric is defined by adding up the contribution of each of the coordinates in (4.1), after taking differences for individual terms. Specifically, for  $x, x' \in S$ , with corresponding

coordinates as denoted above, we set

$$d(x, x') = \sum_{i=1}^{\infty} ((|k_i - k'_i| + |s_i - s'_i| + |\tilde{r}_i - \tilde{r}'_i|) \wedge 1) + \sum_{k \in \mathcal{A}} |u_k - u'_k| + \sum_{k=1}^{K} |v_k - v'_k|.$$
(4.2)

For most purposes, we will not require the full metric, but just the associated "norm"  $|\cdot|$ , with

$$|x| = |z| + |u| + |v|.$$
(4.3)

 $(|x| \text{ can be interpreted as the distance from } x \text{ to the "origin", which is not in the space, however.) Note that <math>|x|$  is continuous as a function of x. We also equip S with the standard Borel  $\sigma$ -algebra inherited from the metric, which we denote by  $\mathscr{S}$ .

Since only a finite number of the coordinate pairs in  $(\mathbf{Z} \times \mathbf{R})^{\infty}$  are nonzero, it is not difficult to see that the metric  $d(\cdot, \cdot)$  is separable. It is also locally compact; with a bit of work, one can show this by showing that an open ball around a point x is homeomorphic to a finite product of intervals of the form (0, 1) and [0, 1). (Half-closed intervals are needed when either  $s_i = 0$ ,  $s_i = s_{i'}$ for some  $i \neq i'$ , or the coordinates  $r_k$  of r corresponding to a given station j are on the boundary of the simplex  $\sum_{k \in \mathcal{C}(j)} r_k = 1$ .) The metric is not complete, since one can choose Cauchy sequences  $x_n$ , with  $u_n \to 0$  or  $v_n \to 0$  as  $n \to \infty$ , that have no limit in S. It is, however, homeomorphic to the complete metric  $d'(\cdot, \cdot)$  obtained by replacing the term  $\sum_{k \in \mathcal{A}} |u_k - u'_k|$  in (4.2) by

$$\sum_{k \in \mathcal{A}} \left( |u_k - u'_k| + \left| \frac{1}{u_k} - \frac{1}{u'_k} \right| \wedge 1 \right),$$

and  $\sum_{k=1}^{k} |v_k - v'_k|$  by the analogous term (where we set  $\frac{1}{0} - \frac{1}{0} = 0$ ). We prefer to work with the simpler metric  $d(\cdot, \cdot)$ , since we will not require S to be complete.

We now formally define the Markov process *underlying* an HL queueing network as the stochastic process X(t),  $t \ge 0$ , on S that undergoes the following evolution. We continue to use the same suggestive queueing vocabulary that was employed in motivating the construction of S.

We set X(t) = (Y(t), R(t)), with Y(t) and R(t) taking values y and r as in (4.1). (The coordinates  $s_i$  of y are allowed to exceed t, and so jobs may, in effect, arrive before time 0.) The evolution of X(t), in between arrivals and departures of jobs at classes, is given by the service rates  $R_k(t)$ , which are constant over such intervals. Upon an arrival or departure somewhere in the network at time t, the stochastic process is continued by assigning new service rates R(t) = f(Y(t)), where f is a measurable function. The choice of fcorresponds to the discipline of the corresponding HL queueing network.

In order to describe the transition of X(t) upon the arrival or departure of a job at a class, we introduce the sequences of positive i.i.d. random variables  $\xi_k(i), k \in \mathcal{A}$ , and  $\gamma_k(i), k = 1, \ldots, K$ , with  $i = 1, 2, 3, \ldots$ , which correspond to the interarrival and service times of the queueing network. We will also need the sequence of i.i.d. random vectors  $\phi^k(i), i = 1, 2, 3, \ldots$ , with  $\phi^k(i) = e_\ell$  for some  $\ell = 1, \ldots, K$  or  $\phi^k(i) = 0$ , which give the routing of a job upon completion of its service at a class. (Here,  $e_\ell \in \mathbf{R}^K$  is the unit vector in the positive  $\ell$  direction.) We assume the sequences  $\xi, \gamma$ , and  $\phi$  are mutually independent. These sequences, together with the function f and the initial state x, will determine the evolution of  $X(\cdot)$  for all times along each sample path.

The process X(t) can be constructed inductively as follows. At times t between arrivals and departures, R(t) = r gives the rate of decrease of each component of the residual service time vector V(t) = v. When  $V_k(t-) = 0$  occurs for some k, service at k for the oldest job of that class is assumed to be completed, with the job being routed to another class  $\ell$ , if  $\phi^k(i) = e_\ell$ , and leaving the network, if  $\phi^k(i) = 0$ , where  $\phi^k(i-1)$  is the previous routing vector applied at k. When this occurs, one sets  $V_k(t) = \gamma_k(i)$  if class k is then nonempty, and sets  $V_k(t) = 0$ , if k is empty. Until a job leaves its class, its age  $S_i(t) = s_i$  continues to increase at rate 1. (The label i of a job will typically change as it moves from one class to another.)

The components of the residual interarrival time vector U(t) = u always decrease at rate 1 until hitting 0. At the time t for which this occurs at a given k, one includes the pair (k, 0) in the state Y(t) and sets  $U(t) = \xi_k(i)$ , where i is the index of the first unused interarrival time at k at time t. If the class k is empty at time t-, and hence  $V_k(t-) = 0$ , one sets  $V_k(t) = \gamma_k(i')$ , where i' is the index of the first unused service time at time t.

In various cases, the state space S can be simplified for queueing networks with specific HL disciplines, or specific interarrival and service time distributions. For instance, when the interarrival and service times are exponentially distributed, one can drop the u and v coordinates from the description of the state (unless r is chosen to depend on them).

Preemptive static buffer priority disciplines were introduced in Chapter 1. Such networks are HL, with  $\tilde{r}_i = 1$  automatically holding for the oldest job of the highest ranked nonempty class at each station. One can therefore drop the age of jobs from the state space descriptor. As mentioned above, the residual interarrival and service times can also be dropped from the descriptor when the corresponding variables are exponentially distributed. The state space can then be reduced to points in  $\mathbf{Z}^K$  with nonnegative coordinates, if one implicitly assumes the given priority scheme that orders classes at each station, since one is able to drop the coordinate r that governs service.

Networks with the FIFO discipline are also HL, since the oldest job at a station is always served first. If one chooses, one can drop the age of jobs from the state space descriptor, by instead ordering the ages at each station; one can also drop the coordinate r. As before, coordinates can also be eliminated when the interarrival and service times are exponentially distributed. The resulting state space will then be countable.

The state space S is large enough to contain the information needed for the queueing networks we will investigate. It can, however, be modified if one has other applications in mind. For instance, a state space descriptor for how long ago a job entered the network can be appended. This is needed for the FISFO networks mentioned briefly in Section 1.2 (see, e.g., [Br01]). Also, one can associate with each job in a class a residual service time, instead of with just the class itself (see, e.g. [Wi98]). The definitions (4.2) and (4.3) then need to be modified accordingly.

#### Foundations and terminology

It is not difficult to see that the process  $X(\cdot)$  just defined is time homogeneous and Markov, and that its sample paths are right continuous. Although  $X(\cdot)$  is not a jump process, it evolves in a simple manner, having only isolated discontinuities and evolving deterministically in between. In particular, after a jump, the state of  $X(\cdot)$  is explicitly known until its next jump, with its evolution being linear in its coordinates.

The process  $X(\cdot)$  is an example of a *piecewise-deterministic Markov process* (PDP). Such processes are discussed in [Da84] and [Da93] in detail; we will rely on the latter in our discussion. In [Da93], PDPs are the more general family of processes whose evolution in between jumps, rather than being linear, is determined by a locally Lipschitz continuous vector field. Also, "killing" at a rate dependent on the position is allowed in [Da93]; after such killing, the process jumps according to a given random rule. In our setting, there is no such killing. Since, in between jumps,  $X(\cdot)$  lives in a subset of S which is homeomorphic to an open ball in  $\mathbf{R}^d$ , for some d, one can append a "boundary"  $\partial S$  to S, which  $X(\cdot)$  hits at a time t- immediately before a jump. For the setting in [Da93], this approach is useful in assigning the jump rule for the process, and it also allows one to define the process on the space  $D_S = D_S[0,\infty)$  of right continuous paths on S with left limits. However, since the process jumps instantly upon hitting  $\partial S$ , the introduction of  $\partial S$  has its own inconveniences. We avoid these complications and just stick with the space S, referring the reader to Sections 24 and 25 of [Da93] for the technical details in the more general setting.

So far, we have not defined the filtration for the process  $X(\cdot)$ . We let  $\mathcal{F}_t^0$  denote the natural filtration

$$\mathcal{F}_t^0 = \sigma(X(s), \ 0 \le s \le t) \tag{4.4}$$

and let  $\mathcal{F}^0_{\infty}$  be the  $\sigma$ -algebra generated by  $\mathcal{F}^0_t$ , for  $t \geq 0$ . Rather than employing  $\mathcal{F}^0_t$  and  $\mathcal{F}^0_{\infty}$  directly, we will use appropriate completions. As in [Da93], for each initial probability measure  $\mu$  on S, one can define a measure  $P_{\mu}$  on the sample space corresponding to the process. Letting  $\mathcal{F}^{\mu}_t$  be the completion of  $\mathcal{F}^0_t$  obtained by including all  $P_{\mu}$ -null sets of  $\mathcal{F}^0_{\infty}$ , one sets

$$\mathcal{F}_t = \bigcap_{\mu} \mathcal{F}_t^{\mu}, \tag{4.5}$$

and denotes by  $\mathcal{F}_{\infty}$  the  $\sigma$ -algebra generated by  $\mathcal{F}_t$ , for  $t \geq 0$ . We henceforth employ  $\{\mathcal{F}_t, t \geq 0\}$  as the filtration for the process  $X(\cdot)$ , and use  $P_{\mu}$  to denote

the corresponding probability measures. This setup is typical of general Markov process theory.

In addition to being complete in the sense of (4.5), the family  $\{\mathcal{F}_t, t \geq 0\}$  is right continuous, that is,

$$\mathcal{F}_t = \mathcal{F}_{t+} \stackrel{\text{def}}{=} \bigcap_{\epsilon > 0} \mathcal{F}_{t+\epsilon}.$$
(4.6)

Because  $X(\cdot)$  is a PDP, this is not difficult to show. (See Theorem 25.3 in [Da93].) The properties (4.5) and (4.6) will be needed shortly for the Debut Theorem and the strong Markov property.

For  $t \geq 0, x \in S$ , and  $A \in \mathscr{S}$ , set

$$P^t(x,A) = P_x(X(t) \in A).$$

It is shown in [Da93] that  $P^{\cdot}(\cdot, \cdot)$  is a probability transition kernel on  $(S, \mathscr{S})$ . That is, for fixed t and A,  $P^t(\cdot, A)$  is  $\mathscr{S}$ -measurable; for fixed t and x,  $P^t(x, \cdot)$  defines a probability measure on S; and  $P^{s+t} = P^s \circ P^t$ , for  $s, t \geq 0$ , defines a semigroup. However,  $X(\cdot)$  need not be a Feller process. (Recall that  $X(\cdot)$  is a Feller process if, in addition,  $P^t : C(S) \to C(S)$ , where C(S) denotes the continuous bounded functions on S.) This may be the case even when  $X(\cdot)$  corresponds to a queueing network with a FIFO or SBP discipline, since even a small change in the future arrival time of a job may cause it to be served after another job instead of before. This can induce a major change in the future evolution of  $X(\cdot)$ , which will imply that for  $f \in C(S)$ ,  $P^t f$  need not be continuous. Nevertheless, as shown in [Da93],  $X(\cdot)$  is strong Markov.

In the next subsection, we will summarize the recurrence theory for Markov processes that we will use for queueing networks. Part of the appropriate literature assumes that these processes are *Borel right*. The intent there is to employ a well-studied general framework, but this comes at the cost of implicitly assuming some familiarity with a technical theory. Such knowledge will not be relevant for our applications, and so we relegate to Section 4.5 a brief discussion of the material. We summarize it here, noting that for a Borel right process, the  $\sigma$ -algebras satisfy (4.5) and (4.6), the process  $X(\cdot)$  is defined on a "reasonable" state space, has a transition semigroup, is right continuous, and the process  $(f \circ X)(\cdot)$  is right continuous when f is an  $\alpha$ -excessive function. The last assumption can be replaced by the strong Markov property.

#### Harris recurrence of Markov processes

We discuss here basic criteria for positive recurrence of Markov processes on general state spaces. Since little is a priori assumed about either the state space or the process itself, one must provide some structure to be able to say anything of interest. There is a developed theory for the corresponding discrete time Markov processes that goes back to Doeblin (see [Do53] for an account), was developed in [Ha56], and includes contributions by [Or71], [AtN78], and [Nu78] among others. Standard references are [Nu84] and [MeT93d]. The basic approach in the discrete time setting is to formulate conditions that are general, but nonetheless enable one to mimic the machinery for discrete time Markov chains. With the aid of resolvents, problems in the continuous time setting can be reformulated in discrete time, which is the approach we summarize here.

Following [MeT93a; MeT93b; MeT93c] and [KaM94], we will assume that the Markov processes  $X(\cdot)$  are Borel right processes on a locally compact and separable metric state space  $(S, \mathscr{S})$ , where  $\mathscr{S}$  is the Borel  $\sigma$ -algebra generated by the metric. The process  $X(\cdot)$  will have a transition semigroup  $P^t$  acting on bounded measurable functions, its paths will be right continuous, and the process will be strong Markov. (These properties follow from the assumption that the processes are Borel right.) When working with discrete time Markov processes, we will assume the state space satisfies the same properties.

One wishes to formulate concepts that are the analogs of those for Markov chains, although there will be aspects not present in the countable state space theory. For  $A \in \mathscr{S}$ , let

$$\tau_A = \inf\{t \ge 0 : X(t) \in A\}, \quad \eta_A = \int_0^\infty \mathbf{1}\{X(t) \in A\} \, dt. \tag{4.7}$$

By the Debut Theorem (see e.g., [Sh88]),  $\tau_A$  is a stopping time. A Markov process is said to be  $\varphi$ -irreducible, for a nontrivial  $\sigma$ -finite measure  $\varphi$  on  $(S, \mathscr{S})$ , if

$$\varphi(A) > 0$$
 implies  $E_x[\eta_A] > 0$  for all  $x \in S$ ; (4.8)

 $\varphi$  is called an *irreducibility measure*.

If for some nontrivial  $\sigma$ -finite measure  $\varphi$ ,

$$\varphi(A) > 0$$
 implies  $P_x(\eta_A = \infty) = 1$  for all  $x \in S$ , (4.9)

then  $X(\cdot)$  is *Harris recurrent*. (This definition goes back to [AzKR67].) It is not difficult to show that it is equivalent to the condition that

$$\varphi(A) > 0$$
 implies  $P_x(\tau_A < \infty) = 1$  for all  $x \in S$  (4.10)

(see [KaM94] or [MeT93a]), although the choice of  $\varphi$  satisfying (4.10) need not satisfy (4.9). (Consider, for example, the process  $X(t) = e^{it}$  on the unit circle, where  $\varphi$  is concentrated at a point.) Both formulations are useful in practice.

A  $\sigma$ -finite measure  $\pi$  on  $(S, \mathscr{S})$  satisfying

$$\pi(A) = \pi P^t(A) \stackrel{\text{def}}{=} \int P^t(x, A) \,\pi(dx) \quad \text{for } A \in \mathscr{S}, \ t \ge 0,$$
(4.11)

is stationary (or invariant). (Note that the definition does not involve  $\varphi$ .) It was shown in [Ge79] that if  $X(\cdot)$  is Harris recurrent, then there is a unique stationary measure, up to a constant multiple. (We will discuss this result in Section 4.5.) If the stationary measure  $\pi$  is finite, it may be normalized to a probability measure. Harris recurrent processes with such  $\pi$  are positive Harris recurrent.

The reader should be aware that Harris recurrence and positive Harris recurrence have somewhat different implications than recurrence and positive recurrence, in the countable state space setting. For instance, if  $\varphi$  is concentrated at a point x, then a Markov chain can have x as an absorbing point and still be positive Harris recurrent. When all states communicate, the definitions are equivalent.

For discrete time Markov processes,  $\varphi$ -irreducibility, Harris recurrence, and positive Harris recurrence are defined by the analogs of (4.8), (4.9) and (4.11). (In this setting, (4.9) and (4.10) are clearly equivalent by the strong Markov property.) Since there is a wealth of theory available for such Markov processes, it is fruitful to be able to translate continuous time problems into the discrete time setting. This can be done by using the *resolvent* of the continuous time Markov process,

$$R(x,A) \stackrel{\text{def}}{=} \int_0^\infty e^{-t} P^t(x,A) \, dt \quad \text{for } x \in S, \ A \in \mathscr{S}.$$
(4.12)

The Markov process  $\tilde{X}(n)$ , n = 0, 1, 2, ..., with one-step transition probability given by  $R(\cdot, \cdot)$ , is known as an *R*-chain. The *R*-chain  $\tilde{X}(\cdot)$  can also be constructed directly from  $X(\cdot)$  by setting

$$X(n) = X(\sigma_n), \quad n = 0, 1, 2, \dots,$$
 (4.13)

where the sequence  $\sigma_0, \sigma_1, \sigma_2, \ldots$  of random variables, with  $\sigma_0 = 0$ , is independent of  $X(\cdot)$  and has i.i.d. mean-1 exponentially distributed increments. (One enriches the sample space so as to include such a sequence.) It is easy to check that  $\tilde{X}(\cdot)$  is  $\varphi$ -irreducible if and only if  $X(\cdot)$  is. One can also show that the same is true for Harris recurrence and positive Harris recurrence, and that the same stationary measure is shared by both processes. The arguments are fairly quick although not immediate; they are given in Section 4.5.

Although the irreducibility measure for a given process  $X(\cdot)$  is not unique, there exists a maximal irreducibility measure  $\psi$ , i.e., an irreducibility measure for the process such that  $\varphi \ll \psi$  for any other irreducibility measure  $\varphi$ , and such that

$$\psi(\{x : P_x(\eta_A \neq 0) > 0\}) = 0 \tag{4.14}$$

for  $\psi(A) = 0$  and  $A \in \mathscr{S}$ . ( $\varphi \ll \psi$  means that  $\varphi$  is absolutely continuous with respect to  $\psi$ .) The measure  $\psi$  is equivalent to

$$\psi'(A) \stackrel{\text{def}}{=} \int_{S} R(x, A)\varphi'(dx) \quad \text{for } x \in S, \ A \in \mathscr{S},$$
 (4.15)

for any irreducibility measure  $\varphi'$ . (That is,  $\psi \ll \psi'$  and  $\psi' \ll \psi$ .) Since the existence of a stationary measure  $\pi$  does not depend on the choice of  $\varphi$ , one is free to assume that  $\varphi$  is maximal when addressing such questions.

Maximal irreducibility measures are frequently used in discrete time, where (4.14) and (4.15) are replaced by

$$\psi(\{x : P_x(\tau_A < \infty) > 0\}) = 0 \tag{4.16}$$

$$\psi'(A) \stackrel{\text{def}}{=} \sum_{n=0}^{\infty} 2^{-(n+1)} \int_{S} P^{n}(x, A) \varphi'(dx) \quad \text{for } x \in S, \ A \in \mathscr{S}.$$
(4.17)

The existence of such a measure  $\psi$  and its equivalence to  $\psi'$  in the latter setting can be found on, e.g., page 88 of [MeT93d]. One can check that (4.14) and (4.15) hold for the process  $X(\cdot)$  for a given irreducibility measure  $\psi$  if and only if (4.16) and (4.17) hold for its *R*-chain  $\tilde{X}(\cdot)$ , since the right side of (4.17) is equivalent to  $\psi''(A) = \int_{S} P(x, A)\varphi'(dx)$  in this case.

The above definitions of Harris recurrent and positive Harris recurrent, while elegant, can be difficult to apply in practice. For applications, the following alternative formulation involving petite sets is very useful. A nonempty set  $A \in \mathscr{S}$  is said to be *petite* if for some fixed probability measure a on  $(0, \infty)$  and some nontrivial measure  $\nu$  on  $(S, \mathscr{S})$ ,

$$\nu(B) \le \int_0^\infty P^t(x, B) a(dt) \tag{4.18}$$

for all  $x \in A$  and all  $B \in \mathscr{S}$ ;  $\nu$  is then called a *petite measure*. A petite set A has the property that each set B is "equally accessible" from all points  $x \in A$  with respect to the measure  $\nu$ . Note that any nonempty measurable subset of a petite set is also petite. When (4.18) holds, with a being concentrated at a single point  $m_0$ , A is said to be *small*, and  $\nu$  is called a *small measure*. Petite and small sets are defined analogously in the discrete time setting.

Let

$$\tau_A(\delta) = \inf\{t \ge \delta : X(t) \in A\}.$$

Theorem 4.1 below gives practical alternative characterizations of Harris recurrence and positive Harris recurrence in terms of petite sets. Versions of Theorem 4.1 are stated in [MeT93a; MeT93b; MeT93c], with that in [MeT93a] being used here. Discrete time analogs of the different parts of Theorem 4.1 are known. (See, e.g., [Or71], [Nu84], and [MeT93d].)

**Theorem 4.1.** (a) A Markov process  $X(\cdot)$  is Harris recurrent if and only if there exists a closed petite set A with

$$P_x(\tau_A < \infty) = 1 \quad for \ all \ x \in S. \tag{4.19}$$

(b) Suppose the Markov process  $X(\cdot)$  is Harris recurrent. Then,  $X(\cdot)$  is positive Harris recurrent if and only if there exists a closed petite set A such that for some  $\delta > 0$  (or, equivalently, for any  $\delta > 0$ ),

$$\sup_{x \in A} E_x[\tau_A(\delta)] < \infty. \tag{4.20}$$

We next make some general comments about Theorem 4.1. We then indicate how the theorem will be applied and discuss its proof. More detail on the proof will be supplied in Section 4.5.

and

We note that the irreducibility measure  $\varphi$  in (4.8) and the measure  $\nu$  in (4.18) employed in the definitions of Harris recurrence and petite sets are different in general. In [MeT93a], petite sets rather than closed petite sets are employed for Harris recurrence, although closed petite sets are needed for positive Harris recurrence. We assume the sets are closed in both cases; this simplifies the proof of one of the steps. We note that the more useful direction (and the only one used in these lectures) is that Harris recurrence and positive Harris recurrence follow from (4.19) and (4.20), respectively. If one wishes, one can base the definition of Harris recurrence on (4.19), rather than on the irreducibility measure as in (4.9); this is done, for instance, in [As03] (and in [Dur96], in discrete time). This will simplify the work in showing the existence of a stationary measure.

Theorem 4.1 will prove very useful in conjunction with Section 4.2. There, we will show that for the underlying Markov process  $X(\cdot)$  of a queueing network,

$$A = \{x : |x| \le \kappa\} \quad \text{is a closed small set for each } \kappa > 0, \tag{4.21}$$

where |x| is given by (4.3), if appropriate conditions hold for the distributions of the interarrival times for the queueing network. In Section 4.4, we will show that the conditions (4.19) and (4.20) in the theorem will follow from the stability of the associated fluid limits, which are introduced in Section 4.3.

We next briefly discuss the proof of Theorem 4.1. The demonstration that  $X(\cdot)$  is Harris recurrent if (4.19) holds is elementary, if one sets  $\varphi = \nu$ . We summarize the argument here. Note that  $X(\tau_A) \in A$ , since A is closed. Starting from any  $x \in S$  and applying the strong Markov property, one can therefore show that by a large enough fixed time T (depending on x),  $X(\cdot)$  will, with at least a given positive probability that depends on A, hit a specified set B with  $\nu(B) > 0$ . Repetition of this reasoning, using appropriate random times  $T_n$  which depend on  $X(T_{n-1})$ , will imply that the probability B has not been hit after n iterations decreases exponentially quickly in n. This implies (4.10), and hence that  $X(\cdot)$  is Harris recurrent. ([MeT93a] gives a different argument that does not assume A is closed.)

The other direction in Part (a), and both directions in Part (b) of Theorem 4.1, require work. In Section 4.5, we will present a summary of the proofs. We will state there a discrete time analog of Theorem 4.1 and indicate how Theorem 4.1 can be shown using this, and the correspondence mentioned earlier between Harris recurrence, positive Harris recurrence, and the stationary measures for the Markov process and its R-chain. We will also provide a summary of the proof of the existence of a stationary measure for a discrete time recurrent Markov process, since it helps illustrate the nature of the discrete time theory.

# Ergodicity

A continuous time Markov process  $X(\cdot)$  is said to be *ergodic* if it possesses a stationary probability measure  $\pi$  for which

$$\lim_{t \to \infty} ||P^t(x, \cdot) - \pi(\cdot)|| = 0 \quad \text{for all } x \in S.$$
(4.22)

(Here,  $|| \cdot ||$  denotes the total variation norm.) We recall from Section 1.2 that, when the underlying Markov process of a queueing network is ergodic, the queueing network is said to be *e-stable*. It is not difficult to see that positive Harris recurrence follows from ergodicity. Frequently, because of the following results, ergodicity of the Markov process also follows from positive Harris recurrence without much additional effort.

The first result is from Theorem 6.1 of [MeT93b].

**Theorem 4.2.** Suppose that a Markov process  $X(\cdot)$  is positive Harris recurrent. Then,  $X(\cdot)$  is ergodic if and only if some skeleton chain is  $\varphi$ -irreducible for some measure  $\varphi$ .

By a skeleton chain, we mean the Markov process defined by restricting  $X(\cdot)$  to the times  $n\Delta$ ,  $n = 0, 1, 2, \ldots$ , for some  $\Delta > 0$ . The necessity of irreducibility on some skeleton chain is clear. To show that this is also sufficient, one can employ the corresponding result for discrete time Markov processes and the fact that the norm of the difference in (4.22) is decreasing in t; we do not go into details here.

For our purposes, it will be more useful to have a sufficient condition for ergodicity in terms of small sets. When a set  $A \in \mathscr{S}$  is small with respect to the same measure  $\nu$  at each  $m_0 \in [s_1, s_2]$ , for some  $0 < s_1 < s_2$ , we will say that A is uniformly small on  $[s_1, s_2]$ , or, more briefly, uniformly small.

**Theorem 4.3.** Suppose that a Markov process  $X(\cdot)$  is positive Harris recurrent, and that some closed set A satisfies (4.19) and is uniformly small on  $[s_1, s_2]$ , for some  $0 < s_1 < s_2$ . Then,  $X(\cdot)$  is ergodic.

Theorem 4.3 follows from Theorem 4.2. One can show this by restarting  $X(\cdot)$  at  $\tau_A$  and applying the strong Markov property. Note that since A is closed,  $X(\tau_A) \in A$ . Theorem 4.3 is also proved in [As03] as Part (iii) of Proposition 3.8 on page 203, although in a somewhat different setting. Theorem 4.3 will be employed in Section 4.2.

### Countable state space setting

As mentioned at the beginning of the section, the material that is needed from this section simplifies enormously in the countable state space setting. Since the interarrival and service times are exponentially distributed, one can drop the u and v coordinates from the description of the state of the process in the first subsection. The underlying jump process  $X(\cdot)$  of the queueing network can be defined in the natural way. It will be constant in between arrivals and departures of jobs at the different classes, with arrivals occurring at rate  $\alpha_k$ and departures at rate  $R_k(t)\mu_k$  at class k. As before, the service rate vector  $R(\cdot)$  remains constant until the next arrival or departure, at which time the new value R(t) = f(X(t)) is assigned, where the choice of f corresponds to the discipline. (The service rate vector is not part of the state space here.) In this setting, the norm in (4.3) is replaced by the simpler |x| = |z|, where z is the queue length vector. The state space is assigned the discrete topology. The second subsection will no longer be needed, since  $X(\cdot)$  is now a Markov chain, with bounded jump rates, and so standard Markov chain theory applies (see, e.g. [Re92]). For consistency with the general case, we choose its filtration to be  $\{\mathcal{F}_t, t \ge 0\}$  as in (4.5), although (4.4) could also be used. In either case, the filtration will be right continuous and  $X(\cdot)$  will be strong Markov.

The subsection on Harris recurrence can also be omitted. In the countable state space setting, Harris recurrence is equivalent to the existence of a recurrent state x that is accessible from all other states, i.e.,  $P_y(\tau_x < \infty) = 1$  for each  $y \in S$  (where  $\tau_x \stackrel{\text{def}}{=} \tau_{\{x\}}$ ). Positive Harris recurrence corresponds to  $E_x[\tau_x(\delta)] < \infty$ , for some  $\delta > 0$ . So, standard Markov chain theory can be applied. The concepts petite and small are no longer needed. They will be used later, in the general setting, only in the context of Proposition 4.6 for the sets  $A = \{x : |x| \le \kappa\}, \kappa > 0$ . These concepts are not needed in the countable state space setting since the state 0 will be uniformly accessible from A. We note that all points in a state space are small sets, and that Theorem 4.1 is elementary in the countable state space setting (in addition to no longer being needed).

In the countable state space, continuous time setting, it is routine to show that positive Harris recurrence implies ergodicity. So, Theorems 4.2 and 4.3 are no longer needed. (One can apply discrete time theory to any skeleton chain, which will be aperiodic, and then apply the monotonicity of  $||P^t(x, \cdot) - \pi(\cdot)||$  in t.)

## 4.2. Results for bounded sets

Bounded sets will play an important role in Section 4.4 in showing the stability of queueing networks. In Section 4.2, we show two results involving bounded sets we will need there. Here and elsewhere in these lectures, a *norm*  $|\cdot|$  denotes a nonnegative function on a state space S. A set  $B \subseteq S$  is said to be *bounded* if  $\sup_{x \in B} |x| < \infty$ .

The first of these two results, Proposition 4.6, is a generalization of Foster's Criterion. As background, we first state Foster's Criterion, along with its proof. Both versions give useful criteria for positive Harris recurrence when the Markov process under consideration has a uniformly negative drift off of a bounded subset of the state space. In Proposition 4.6, we also give an analogous condition for ergodicity of the Markov process.

The second result, Proposition 4.7, gives criteria under which the bounded sets are uniformly small for the Markov process X underlying a queueing network. This condition is used in conjunction with Theorem 4.3 and Proposition 4.6 to show the Markov process is ergodic.

# Foster's Criterion

Foster's Criterion is a simple, but very useful, criterion for demonstrating positive recurrence of Markov chains with a norm. We state it in the original discrete time context. **Proposition 4.4 (Foster's Criterion).** Let X(n), n = 0, 1, 2, ..., be a Markov chain on which all states communicate. Suppose that

$$E_x|X(1)| < \infty \quad for \ all \ x \in A, \tag{4.23}$$

where  $A = \{x : |x| \le \kappa\}$  for some  $\kappa \ge 0$ , and  $|A| < \infty$ . Also, suppose that for some  $\epsilon > 0$ ,

$$E_x|X(1)| \le |x| - \epsilon \quad \text{for all } x \notin A. \tag{4.24}$$

Then,  $X(\cdot)$  is positive recurrent.

Foster's Criterion states, in essence, that if a Markov chain has a uniformly negative drift off of a bounded set, and if its behavior on the bounded set is not too bad, then the Markov chain will be positive recurrent. [Fo53] gave a computational proof when the state space is the nonnegative integers with the corresponding norm and  $\kappa = 0$ . Foster's Criterion is often employed in the above slightly more general setting of Proposition 4.4.

Proof of Proposition 4.4. Since all states communicate and  $|A| < \infty$ , in order to show that  $X(\cdot)$  is positive recurrent, we claim it suffices to show  $E_x[\tau_A] < \infty$ for  $x \in A$ . The claim is clear when A is a singleton. When A has more states, one can show it by noting that, by the strong Markov property, the expected number of returns to A grows linearly in n. The same is therefore true for at least one state in A, which must be positive recurrent.

To show that  $E_x[\tau_A] < \infty$  for  $x \in A$ , we set

$$M(n) = |X(n)| + \epsilon n \quad \text{for all } n. \tag{4.25}$$

On account of (4.24),  $M(n \wedge \tau_A)$  is a nonnegative supermartingale on the natural filtration of  $X(\cdot)$ . It follows by the Optional Sampling Theorem that

$$E_y[M(\tau_A)] \leq |y|$$
 for all  $y \notin A$ .

It follows from this and (4.25) that

$$E_y[\tau_A] \le \frac{1}{\epsilon} E_y[M(\tau_A)] \le |y|/\epsilon \quad \text{for } y \notin A.$$

Together with (4.23), this implies that for  $x \in A$ ,

$$E_x[\tau_A] \le 1 + \sum_{y \notin A} p(x, y) E_y[\tau_A] \le 1 + \frac{1}{\epsilon} \sum_{y \notin A} |y| p(x, y) < \infty,$$

where  $p(\cdot, \cdot)$  denotes the one-step transitional probabilities. So,  $E_x[\tau_A] < \infty$  for  $x \in A$ , and  $X(\cdot)$  is positive recurrent.

In various cases, one might not know that (4.24) holds for a given Markov chain or Markov process, but rather that  $|X(\cdot)|$  decreases linearly "on the average", over a much longer time interval. In our applications in Section 4.4, this will occur over time intervals of the form [t, t + c|X(t)|], for large |X(t)|, where

c > 0. To accommodate such a setting, we employ the following generalization of Foster's Criterion. Here, time is chosen to be continuous, although time could also be chosen to be discrete, if one omits the conclusion on ergodicity. If time is continuous and the state space is general, then  $X(\cdot)$  is assumed to satisfy the regularity conditions for Markov processes given in Section 4.1, in the second paragraph of the subsection on Harris recurrence. Versions of Proposition 4.5 are given in [MaM81], [Fi89], [MeT94], and [FoK04].

**Proposition 4.5 (Generalized Foster's Criterion).** Suppose that  $X(\cdot)$  is a continuous time Markov process, such that for some  $\epsilon > 0$ ,  $\kappa > 0$ , and measurable function  $g: S \to \mathbf{R}$  with  $g(x) \ge \delta > 0$ ,

$$|E_x|X(g(x))| \le (|x| \lor \kappa) - \epsilon g(x) \quad \text{for all } x.$$

$$(4.26)$$

Then,

$$E_x[\tau_A(\delta)] \le \frac{1}{\epsilon} (|x| \lor \kappa) \quad \text{for all } x, \tag{4.27}$$

where  $A = \{x : |x| \le \kappa\}$ . In particular, if A is a closed petite set, then  $X(\cdot)$  is positive Harris recurrent. If A is closed and is uniformly small on  $[s_1, s_2]$ , for some  $0 < s_1 < s_2$ , then  $X(\cdot)$  is ergodic.

One can check that Proposition 4.4 is a special case of the discrete time version of Proposition 4.5, with  $g \equiv 1$ , since (4.23) and (4.24) follow from (4.26) after a new choice of  $\kappa$ , and since any finite set will be petite if all states communicate.

In these lectures, we will employ the following case of Proposition 4.5.

**Proposition 4.6 (Multiplicative Foster's Criterion).** Suppose that  $X(\cdot)$  is a continuous time Markov process, such that for some c > 0,  $\epsilon > 0$ , and  $\kappa > 0$ ,

$$E_x|X(c(|x|\vee\kappa))| \le (1-\epsilon)(|x|\vee\kappa) \quad \text{for all } x. \tag{4.28}$$

If  $A = \{x : |x| \le \kappa\}$  is a closed petite set, then  $X(\cdot)$  is positive Harris recurrent. If A is closed and is uniformly small on  $[s_1, s_2]$ , for some  $0 < s_1 < s_2$ , then  $X(\cdot)$  is ergodic.

Setting  $g(x) = c(|x| \lor \kappa)$ , Proposition 4.6 follows immediately from Proposition 4.5. The proof of Proposition 4.5 uses an elementary martingale argument together with Theorems 4.1 and 4.3 of the previous section. (This is the only place in these lectures where we will use Theorems 4.1 and 4.3.)

*Proof of Proposition* 4.5. Suppose that (4.27) holds. Then, clearly so does (4.19) of Theorem 4.1. Also by (4.27),

$$\sup_{x \in A} E_x[\tau_A(\delta)] \le \kappa/\epsilon$$

must hold, and therefore so does (4.20). If A is assumed to be closed and petite, it therefore follows from both halves of Theorem 4.1 that  $X(\cdot)$  is positive Harris recurrent. If A is assumed to be closed and uniformly small on some  $[s_1, s_2], 0 < \infty$ 

 $s_1 < s_2$ , it follows from this and Theorem 4.3 that  $X(\cdot)$  is ergodic. So, it suffices to show that (4.27) holds.

Set  $\sigma_0 = 0$ , and let  $\sigma_1, \sigma_2, \ldots$  denote the stopping times defined inductively by

$$\sigma_n = \sigma_{n-1} + g(X(\sigma_{n-1})).$$

By (4.26) and the strong Markov property,

$$E_x[|X(\sigma_n)| \mid \mathcal{F}(\sigma_{n-1})] \le (|X(\sigma_{n-1})| \lor \kappa) - \epsilon g(X(\sigma_{n-1}))$$

for all x, where  $\mathcal{F}(T) \stackrel{\text{def}}{=} \mathcal{F}_T$  is the  $\sigma$ -algebra corresponding to the stopping time T. Set  $M(0) = |x| \lor \kappa$  and

$$M(n) = |X(\sigma_n)| + \epsilon \sigma_n \quad \text{for } n \ge 1.$$
(4.29)

Also, set  $\mathcal{G}_n = \mathcal{F}(\sigma_n)$  and note that  $\sigma_n \in \mathcal{G}_{n-1}$ . One can check that

$$E_x[M(n) \mid \mathcal{G}_{n-1}] \le M(n-1) \text{ for } n \le \rho,$$

where  $\rho$  is the first time n > 0 at which  $M(n) \in A$ . So,  $M(n \wedge \rho)$  is a nonnegative supermartingale on  $\mathcal{G}_n$ .

It follows by the Optional Sampling Theorem that

$$E_x[M(\rho)] \le |x| \lor \kappa.$$

Note that  $\tau_A(\delta) \leq \sigma_{\rho}$ . Therefore, by (4.29) and the above inequality,

$$\epsilon E_x[\tau_A(\delta)] \le E_x[M(\rho)] \le |x| \lor \kappa,$$

which implies (4.27), as desired.

Criteria for bounded sets to be petite or uniformly small

As mentioned earlier, we will employ Proposition 4.6 in Section 4.4 to establish criteria for when the Markov process  $X(\cdot)$  underlying a queueing network is positive Harris recurrent or is ergodic. In order to cite Proposition 4.6, we need to be able to show bounded sets are petite or uniformly small. These conditions will not automatically hold, since states need not "communicate" in general. For instance, when the distributions of the interarrival times, at two classes  $k_1$ and  $k_2$  of a queueing network, are both integer valued, states x and x' for which the residual interarrival times satisfy

$$u_{k_2} - u_{k_1} \neq (u'_{k_2} - u'_{k_1}) \mod 1$$

cannot both be visited along the same sample path.

In order to rule out such behavior, the following two conditions on the distributions of the interarrival times  $\xi_k(1)$ ,  $k \in \mathcal{A}$ , are often assumed. The first is that  $\xi_k(1)$  is unbounded, that is, for each  $k \in \mathcal{A}$ ,

$$P(\xi_k(1) \ge t) > 0 \quad \text{for all } t. \tag{4.30}$$

The second is that for some  $\ell_k \in \mathbf{Z}_+$ , the  $\ell_k$ -fold convolution of  $\xi_k(1)$  and Lebesgue measure are not mutually singular. That is, for  $k \in \mathcal{A}$  and some nonnegative  $q_k(\cdot)$  with  $\int_0^\infty q_k(t) dt > 0$ ,

$$P(\xi_k(1) + \dots + \xi_k(\ell_k) \in [c, d]) \ge \int_c^d q_k(t) dt$$
(4.31)

for all c < d. When arrivals in the network are permitted at only one class, e.g., as in reentrant lines, it is enough to assume just (4.30) to show bounded sets are petite.

It is annoying to need to assume either condition, especially the first, since they rule out reasonable distributions for which one should expect the underlying Markov process to be positive Harris recurrent. It appears difficult, however, to formulate simple criteria that are robust over networks with general routing structures and disciplines. For interarrival times not satisfying (4.30) and (4.31), one needs to show the existence of a petite or uniformly small set directly.

Proposition 4.7 is the main result in this subsection. It states that when (4.30) and (4.31) are satisfied for the interarrival times of a queueing network, then bounded sets will be uniformly small. It follows immediately from this that bounded sets are also petite. (No requirements are made on the service times.) A related result is given in Lemma 3.7 of [MeD94].

**Proposition 4.7.** Assume that the interarrival times of an HL queueing network satisfy (4.30) and (4.31). Then, for each  $\kappa > 0$ , the set  $A = \{x : |x| \le \kappa\}$  is uniformly small on  $[s_1, s_2]$  for some  $0 < s_1 < s_2$ .

When arrivals in the network are permitted at only one class, one can instead use the following result to show bounded sets are petite.

**Proposition 4.8.** Assume that the interarrival times of an HL queueing network, with  $|\mathcal{A}| = 1$ , satisfy (4.30). Then, for each  $\kappa > 0$ , the set  $A = \{x : |x| \le \kappa\}$  is petite.

Since |x| is continuous in x, the above sets A are closed. Once (4.28) has been verified, Proposition 4.8 can therefore be used in conjunction with Proposition 4.6 to show that the underlying Markov process  $X(\cdot)$  of the queueing network is positive Harris recurrent, and hence that the queueing network is stable. Similarly, Proposition 4.7 can be used with the proposition to show  $X(\cdot)$ is ergodic, and hence that the queueing network is e-stable.

We will prove Proposition 4.7, which requires some effort. The argument for Proposition 4.8 is simpler, an outline of which goes as follows. Choose  $t_1 > 0$ and  $\epsilon > 0$ , so that for all  $|x| \leq \kappa$ , with  $\kappa$  fixed, the probability is at least  $\epsilon$ that the queueing network will be empty over some interval [t(x) - 1, t(x)) but not remain empty over the entire interval  $[t(x), t_1)$ , where  $t(x) \in [1, t_1]$ . It is possible to do this because of (4.30). One can then show that the bounded set A will be petite, by choosing the probability measure a in (4.18) to be uniform over  $(0, t_1)$ , and choosing the petite measure  $\nu$  to be uniform over the empty states of the queueing network with residual interarrival times in (0, 1), so that its density is  $\epsilon/t_1$  with respect to this set. Details are similar to parts of the argument for the more complicated construction in the proof of Proposition 4.7.

Before beginning the proof of Proposition 4.7, we provide a summary of the argument and introduce some notation. Let  $L_0 = \max_{k \in \mathcal{A}} \ell_k$ , where  $\ell_k$  is as in (4.31). Then, the distribution of  $\Xi_k(L_0) = \sum_{i=1}^{L_0} \xi_k(i)$  has an absolutely continuous component for all k. If L is chosen large enough, then, for each k, the distribution of  $\Xi_k(L)$  will uniformly cover some interval of length  $\kappa + 3$ , say  $[a_k - \kappa, a_k + 3]$ , where  $\kappa$  is chosen as in the proposition. On account of (4.30),  $\xi_k(L+1)$  can take arbitrarily large values, which we specify to be in  $[b_k, b_k + 1]$  for some  $b_k$ , with  $b_k \geq N$  for some large N. It follows that, for  $u_k \leq \kappa$ , the distribution of  $u_k + \Xi_k(L+1)$  uniformly covers  $[a_k + b_k + 1, a_k + b_k + 3]$ , with no external arrivals at k occurring over the long time period  $[\bar{a}, N)$ , where  $u_k$  is the initial residual interarrival time and  $\bar{a} = \max_k \{a_k + 3\}$ .

For large enough N, the network will be empty with positive probability by time N/2, and hence remain empty until time N, with the probability not depending on the initial state x, for  $|x| \leq \kappa$ . When a state y is empty, it is specified by its residual interarrival time vector u. The state is empty, with positive probability, at the times  $s \in [N/2, N/2 + 1]$ , and at these times, each coordinate  $u_k$  will have an absolutely continuous component covering

$$\mathcal{J}_k = [a_k + b_k + 1 - N/2, a_k + b_k + 2 - N/2].$$

These bounds do not depend on the initial state x, for  $|x| \leq \kappa$ . The set  $A = \{x : |x| \leq \kappa\}$  will therefore be uniformly small on [N/2, N/2 + 1], with the small measure  $\nu$  in (4.18) being uniformly distributed over the empty states y with residual service times in the Cartesian product of  $\mathcal{J}_k$ , for  $k \in \mathcal{A}$ .

We proceed to demonstrate Proposition 4.7 along the lines outlined in the last two paragraphs. On account of (4.31) we may choose L large enough so that for some  $\epsilon_1 > 0$  and  $a_k > \kappa$ ,

$$P(\Xi_k(L) \in [t_1, t_2]) \ge \epsilon_1(t_2 - t_1) \text{ for } [t_1, t_2] \subseteq [a_k - \kappa, a_k + 3],$$

for all  $k \in \mathcal{A}$ . That is,  $\Xi_k(L)$  has density at least  $\epsilon_1$  at all times in the interval  $[a_k - \kappa, a_k + 3]$ . For  $|x| \leq \kappa$  (and hence  $u_k \leq \kappa$ ), this implies

$$P(u_k + \Xi_k(L) \in [t_1, t_2]) \ge \epsilon_1(t_2 - t_1) \quad \text{for } [t_1, t_2] \subseteq [a_k, a_k + 3].$$
(4.32)

Also, by (4.30), for any N, there exist times  $b_k \ge N$  so that

$$P(\xi_k(L+1) \in [b_k, b_k+1]) \ge \epsilon_2$$
(4.33)

for some  $\epsilon_2 > 0$ ; we will specify N later. We introduce the following terminology, setting

$$G_{1,k} = \{ \omega : u_k + \Xi_k(L) \in [a_k, a_k + 3] \}, G_{2,k}(t_{1,k}, t_{2,k}) = \{ \omega : u_k + \Xi_k(L+1) \in [t_{1,k}, t_{2,k}] \}, G_1 = \bigcap_{k \in \mathcal{A}} G_{1,k}, \quad G_2(\mathbf{t}_1, \mathbf{t}_2) = \bigcap_{k \in \mathcal{A}} G_{2,k}(t_{1,k}, t_{2,k}), G = G_1 \cap G_2(\mathbf{t}_1, \mathbf{t}_2),$$

where  $\mathbf{t}_i = (t_{i,k}, k \in \mathcal{A})$ . Also, set  $\mathcal{I}_k = [a_k + b_k + 1, a_k + b_k + 3]$ .

We break most of the work in proving Proposition 4.7 into two lemmas. The first gives the following lower bound on P(G), for  $\mathbf{t}_i$  having coordinates  $t_{i,k} \in \mathcal{I}_k$ .

**Lemma 4.9.** For given  $\mathbf{t}_i$ , i = 1, 2, with  $t_{1,k} \leq t_{2,k}$  and  $t_{i,k} \in \mathcal{I}_k$ , for  $k \in \mathcal{A}$ ,

$$P(G) \ge (\epsilon_1 \epsilon_2)^{|\mathcal{A}|} \prod_{k \in \mathcal{A}} (t_{2,k} - t_{1,k}).$$

$$(4.34)$$

Proof. One has

$$\begin{split} P(G) &= P(G_1 \cap G_2(\mathbf{t}_1, \mathbf{t}_2)) = \prod_{k \in \mathcal{A}} P(G_{1,k} \cap G_{2,k}(t_{1,k}, t_{2,k})) \\ &\geq \prod_{k \in \mathcal{A}} \int_{b_k}^{b_k + 1} P(u_k + \Xi_k(L) \in [t_{1,k} - s, t_{2,k} - s]) P(\xi_k(L+1) \in ds) \\ &\geq (\epsilon_1 \epsilon_2)^{|\mathcal{A}|} \prod_{k \in \mathcal{A}} (t_{2,k} - t_{1,k}). \end{split}$$

The second equality follows from the independence of the interarrival times  $\xi_k(i)$  for different k. The first inequality is gotten by writing  $\Xi_k(L+1)$  as a convolution of  $\Xi_k(L)$  with  $\xi_k(L+1)$ , and noting that for  $s \in [b_k, b_k+1]$ ,  $G_{1,k}$  occurs when  $u_k + \Xi_k(L)$  is contained in  $[t_{1,k} - s, t_{2,k} - s]$ . The second inequality follows from the bounds in (4.32) and (4.33).

Let

$$\sigma = \inf\{t \ge \bar{a} : Z(t) = 0\},\$$

where  $\bar{a} = \max_k \{a_k + 3\}$  and  $Z_k(t)$  is the number of jobs at class k at time t. The next result says that, given the event G, there is a uniform upper bound on  $\sigma$  that does not depend on the initial state x.

**Lemma 4.10.** For given L and  $\kappa$ , and large enough N, there exists  $\epsilon_3 > 0$  so that

$$P_x(\sigma \le N/2 \mid G) \ge \epsilon_3 \tag{4.35}$$

for all  $|x| \leq \kappa$ , and  $\mathbf{t}_i$ , i = 1, 2, with  $t_{1,k} < t_{2,k}$  and  $t_{i,k} \in \mathcal{I}_k$ .

*Proof.* The reasoning behind (4.35) is not difficult. Since the notation that is involved can become cumbersome, we avoid it as much as possible, and argue in terms of basic queueing quantities.

We first note that for large enough M and small enough  $\delta > 0$ , for any class k, (a) There exist classes  $k_1, k_2, \ldots, k_n$ , with  $k_1 = k, n \leq M$ , and

$$\left(1-\sum_{\ell} P_{k_n,\ell}\right)\prod_{i=1}^{n-1} P_{k_i,k_{i+1}} \ge \delta.$$

That is, a job starting at any k has positive probability  $\delta$  of following a designated route and leaving the network in at most M steps. (b)  $P(\gamma_k(1) \leq M) \geq 1/2$  for all k. That is, there is a uniform bound on the service time distributions.

Let  $\zeta$  be the total service time required by an arbitrary job that is either initially in the network or later enters it. In the former case, we know that its residual service time is at most  $\kappa$ . Therefore, by (a) and (b), and the independence of the corresponding events,

$$P(\zeta \le M^2 + \kappa) \ge \delta 2^{-M}. \tag{4.36}$$

On the event G, (c) No jobs enter the network over  $(\bar{a}, N)$ . (d) At most  $|\mathcal{A}|L$  customers enter the network over (0, N), for a total of  $\Lambda$  jobs in the network up to time N, with

$$\Lambda \le |\mathcal{A}|L + |z| \le |\mathcal{A}|L + \kappa \stackrel{\text{def}}{=} L'.$$

(c) and (d) follow from the definitions of  $G_1, b_k, \mathcal{I}_k$ , and  $G_2$ . Let  $\zeta_1, \zeta_2, \ldots, \zeta_A$  denote the total service times of these  $\Lambda$  jobs. By (4.36),

$$P_x\left(\sum_{\ell=1}^{\Lambda}\zeta_\ell \le L'(M^2 + \kappa)\right) \ge (\delta 2^{-M})^{L'}.$$
(4.37)

We now set

$$N = 4(L'(M^2 + \kappa) \vee \bar{a})$$
 and  $\epsilon_3 = (\delta 2^{-M})^{L'}$ .

Then,  $(N/4, N/2] \subseteq (\bar{a}, N/2]$ , and so under the event in (4.37) and (c),  $\sigma \leq N/2$ . Together with (4.37), this implies (4.35).

Proposition 4.7 follows from Lemmas 4.9 and 4.10.

Proof of Proposition 4.7. The bounds in (4.34) and (4.35) imply that

$$P_x(\sigma \le N/2; \ G) \ge (\epsilon_1 \epsilon_2)^{|\mathcal{A}|} \epsilon_3 \prod_{k \in \mathcal{A}} (t_{2,k} - t_{1,k})$$

$$(4.38)$$

for  $|x| \leq \kappa$ , and  $\mathbf{t}_i$ , i = 1, 2, chosen so that  $t_{1,k} < t_{2,k}$  and  $t_{i,k} \in \mathcal{I}_k$ . For  $s \in [N/2, N/2 + 1]$ , it follows that

$$P_{x}(Z(s) = 0 \text{ and } U_{k}(s) \in [t_{1,k} - s, t_{2,k} - s], \ k \in \mathcal{A})$$
  

$$\geq (\epsilon_{1}\epsilon_{2})^{|\mathcal{A}|}\epsilon_{3} \prod_{k \in \mathcal{A}} (t_{2,k} - t_{1,k}), \qquad (4.39)$$

since the event in (4.39) contains the event (4.38). To see this, note that if the network is empty at time  $\sigma \leq N/2$ , it will, on G, remain empty until at least time  $N \leq \min_{k \in \mathcal{A}} \{a_k + b_k\}$ . At the intermediate times s, the residual interarrival times will be the shifts, by s, of the times at which the events  $u_k + \Xi_k(L+1)$ , given in  $G_{2,k}(t_{1,k}, t_{2,k})$ , occur.

When a state y is empty, it is specified by its residual interarrival time vector  $u = \{u_k, k \in \mathcal{A}\}$ . This is the case at time s for the event on the left side of (4.39). The inequality (4.39) states that, for  $|x| \leq \kappa$ , the distribution of the residual time U(s), for  $s \in [N/2, N/2+1]$ , has a component that is absolutely continuous,

with density  $\epsilon = (\epsilon_1 \epsilon_2)^{|\mathcal{A}|} \epsilon_3$ , with respect to  $|\mathcal{A}|$ -dimensional Lebesque measure  $\lambda$  that is restricted to the rectangle

$$\prod_{k \in \mathcal{A}} [a_k + b_k + 1 - N/2, a_k + b_k + 2 - N/2] \subseteq \prod_{k \in \mathcal{A}} [a_k + b_k + 1 - s, a_k + b_k + 3 - s].$$

It follows that, for all  $s \in [N/2, N/2 + 1]$ , the set A is small with respect to the measure  $\nu = \epsilon \lambda$ .

#### *Countable state space setting*

The work required in this section simplifies considerably in the countable state space setting. One still needs to demonstrate Proposition 4.5. The proof of (4.27) proceeds as before. In this setting, one can conclude directly from (4.27) that  $X(\cdot)$  is ergodic (or is positive Harris recurrent), if  $|A| < \infty$ , by using

$$\inf_{x \in A} P_x(\tau_0 \le 1) > 0;$$

the expected number of returns to the empty state 0 therefore grows linearly in t, which implies 0 is positive recurrent. (As before,  $A = \{x : |x| \le \kappa\}$ .) So, neither petite nor uniformly small is needed as an assumption for the proposition. Since Proposition 4.6 is a direct consequence of Proposition 4.5, the same is also true there.

The concepts petite and uniformly small will be used later only in the context of Proposition 4.6. This, in particular, makes Propositions 4.7 and 4.8 unnecessary. The propositions are easy to show, however, since

$$\inf_{x \in A} P_x(X(s) = 0) \ge \epsilon(s)$$

for appropriate  $\epsilon(s) > 0$ , where  $\epsilon(s)$  is uniformly bounded away from 0 on  $[s_1, s_2]$ , for  $0 < s_1 < s_2$ .

#### 4.3. Fluid models and fluid limits

In Section 1.3, we discussed fluid models and their connection with queueing network equations. The purpose there was to give a preview of these concepts. We return now to this material, this time giving a thorough presentation. The section consists of three subsections, covering queueing network equations, fluid models, and fluid limits, as well as a short comment on the countable state space setting.

Fluid models and fluid limits are studied in [Da95]. Modifications are given in [Ch95], [DaM95], [Br98a], and [Br98b] among other places. The approach taken here is closest to [Br98a], but with some further modification in the approach and in some of the definitions.

The fluid models of main interest to us will be subcritical. Fluid models are also an important tool in the study of heavy traffic limits, where the fluid models that are employed are critical; we will not discuss fluid models in the latter context here. (See [Br98b], [Wi98], and [BrD01] for background and further references.)

#### Queueing network equations

In the construction, in Section 4.1, of the Markov process  $X(\cdot)$  underlying an HL queueing network, we introduced sequences of positive i.i.d. random variables  $\xi_k(i), k \in \mathcal{A}$ , and  $\gamma_k(i), k = 1, \ldots, K$ , with  $i = 1, 2, 3, \ldots$ , which correspond to the interarrival and service times of the queueing network. We also introduced the sequence of i.i.d. random vectors  $\phi^k(i), i = 1, 2, 3, \ldots$ , which give the routing of a job upon completion of its service at a class. The corresponding sequences  $\xi$ ,  $\gamma$ , and  $\phi$  were assumed to be mutually independent. Here, we will find it convenient to also denote by  $\xi_k(0)$  and  $\gamma_k(0)$  the initial residual interarrival and service times of the queueing network; they are included in the initial state X(0) = x.

We will employ the random quantities  $E(\cdot)$ ,  $\Gamma(\cdot)$ , and  $\Phi(\cdot)$ , which are defined in terms of the partial sums of  $\xi$ ,  $\gamma$ , and  $\phi$ . The *external arrival process*  $E(t) = \{E_k(t), k = 1, \ldots, K\}, t \ge 0$ , counts the number of arrivals at each class from outside the network. That is, for  $k \in \mathcal{A}$ ,

$$E_k(t) = \max\{n : \Xi_k(n) \le t\},\$$

where

$$\Xi_k(n) = \sum_{i=0}^{n-1} \xi_k(i)$$

The cumulative service time process  $\Gamma(n) = \{\Gamma_k(n_k), k = 1, ..., K\}, n = (n_1, ..., n_K)$  with  $n_k = 1, 2, ...,$  is given by

$$\Gamma_k(n_k) = \sum_{i=0}^{n_k-1} \gamma_k(i).$$

The routing process  $\Phi(n) = \{\Phi^k(n), k = 1, \dots, K\}, n = 1, 2, \dots$ , is given by

$$\Phi^k(n) = \sum_{i=1}^n \phi^k(i)$$

As mentioned in Section 4.1, the sequences  $\xi, \gamma$ , and  $\phi$ , together with the initial state x and the discipline rule, determine the evolution of the process  $X(\cdot)$  for all times along each sample path. The same is therefore true for  $(E(\cdot), \Gamma(\cdot), \Phi(\cdot))$ , which is referred to as the *primitive triple* of the queueing network.

As in Section 1.2, we will employ the means  $\alpha_k m_k$ , and  $P_{k,\ell}$  that are defined from  $\xi, \gamma$ , and  $\phi$ . They are given by

$$\alpha_k = 1/E[\xi_k(1)] \text{ for } k \in \mathcal{A}, \quad m_k = E[\gamma_k(1)] \text{ for } k = 1, \dots, K$$
$$P_{k,\ell} = P(\phi^k(1) = e_\ell),$$

with  $\alpha = \{\alpha_k, k = 1, \ldots, K\}$  being the external arrival rate, M being the diagonal matrix having the mean service times  $m_k$  at its diagonal entries, and  $P = \{P_{k,\ell}, k, \ell = 1, \ldots, K\}$  being the mean transition matrix (or mean routing matrix). As before,  $\mu_k = 1/m_k$  is the service rate. Throughout these lectures, we will implicitly assume that  $E[\xi_k(1)] < \infty$  for  $k \in \mathcal{A}$  and  $E[\gamma_k(1)] < \infty$  for  $k = 1, \ldots, K$ , and so  $\alpha_k, m_k, \mu_k \in (0, \infty)$ . As in (1.2),  $Q \stackrel{\text{def}}{=} (I - P^T)^{-1} = \sum_{n=0}^{\infty} (P^T)^n$ , which is finite since the network is open. Also, the total arrival rate  $\lambda = Q\alpha$  and the traffic intensity  $\rho$ , with  $\rho_j = \sum_{k \in \mathcal{C}(j)} m_k \lambda_k$ , are defined as in (1.5) and (1.7).

Queueing network equations tie together random vectors that describe the evolution of a given queueing network. Examples of such equations, with the vectors A(t), D(t), T(t), and Z(t), were given in Section 1.3. In the present more general setting, it will be more convenient to employ the 6-tuple

$$\mathfrak{X}(t) = (A(t), D(t), T(t), W(t), Y(t), Z(t)).$$
(4.40)

Here, the vector  $W(t) = (W_1(t), \ldots, W_J(t))$  is the *immediate workload*. That is,  $W_j(t)$  is the amount of time required to serve all jobs currently at station j,  $j = 1, \ldots, J$ , if all jobs arriving after time t are ignored. The vector  $Y(t) = (Y_1(t), \ldots, Y_J(t))$  is the *cumulative idle time*, that is, the cumulative time that each of the stations  $j = 1, \ldots, J$  is not working. Note that A(t), D(t), T(t), and Z(t) are class-level vectors, whereas W(t) and Y(t) are station-level vectors. From our perspective,  $\mathfrak{X}(\cdot)$  contains all of the essential information on the evolution of the queueing network; it will be used as the starting point for our computations. With a slight abuse of notation, we will refer to  $\mathfrak{X}(\cdot)$  as the *queueing network process*.

We note that  $T(\cdot)$  and  $Y(\cdot)$  are continuous and that  $A(\cdot)$ ,  $D(\cdot)$ ,  $W(\cdot)$ , and  $Z(\cdot)$  are right continuous with left limits. All of the variables are nonnegative in each component, with  $A(\cdot)$ ,  $D(\cdot)$ ,  $T(\cdot)$ , and  $Y(\cdot)$  being nondecreasing. By assumption, one has

$$A(0) = D(0) = T(0) = 0$$
 and  $Y(0) = 0.$  (4.41)

One can check that the components of  $\mathfrak{X}(\cdot)$  satisfy the queueing network equations

$$A(t) = E(t) + \sum_{k} \Phi^{k}(D_{k}(t)), \qquad (4.42)$$

$$Z(t) = Z(0) + A(t) - D(t),$$

$$(4.43)$$

$$W(t) = CP(A(t)) + Z(0)) = CP(t)$$

$$(4.44)$$

$$W(t) = C\Gamma(A(t) + Z(0)) - CT(t), \qquad (4.44)$$

$$CT(t) + Y(t) = et, (4.45)$$

$$Y_j(t)$$
 can only increase when  $W_j(t) = 0$ ,  $j = 1, \dots, J$ , (4.46)

for all  $t \ge 0$ . Here, C is the constituency matrix,

$$C_{j,k} = \begin{cases} 1 & \text{if } k \in \mathcal{C}(j), \\ 0 & \text{otherwise,} \end{cases}$$

and  $e = (1, ..., 1)^T$ .

The equations (4.42)–(4.46) are not difficult to verify. Equations (4.42) and (4.43) are the same as (1.8) and (1.9), and hold for the same reasons as before. The equality (4.44) states that the amount of current work at each station is equal to the sum of the cumulative amount of work having arrived at all of its classes less the sum of the cumulative service rendered at these classes. Equation (4.45) can be taken as the defining relation for the idletime Y(t). In (4.46), we mean that  $Y_j(t_2) > Y_j(t_1)$  implies  $W_j(t) = 0$  for some  $t \in [t_1, t_2]$ , which reflects the nonidling property. Since  $Y(\cdot)$  is continuous, it can also be written as

$$\int_0^\infty W_j(t)dY_j(t) = 0, \quad j = 1, \dots, J.$$

The equations (4.42)-(4.46) hold for all disciplines. One can check that HL queueing networks also satisfy

$$\Gamma(D(t)) \le T(t) < \Gamma(D(t) + e), \tag{4.47}$$

where the inequalities are componentwise and e denotes the K-vector of all 1's. (Whether e denotes a K-vector or J-vector will be clear from the context.) The equations (4.42)–(4.47) will be referred to as the *basic queueing network* equations.

Equations (4.42)–(4.47) do not specify the discipline of the queueing network. For multiclass queueing networks, there is consequently not enough information to solve for  $\mathfrak{X}(\cdot)$ . Later, when working with specific examples, an additional equation (or equations) will be introduced that correspond to the discipline. Such an equation will be referred to as an *auxiliary queueing network equation*. For example, for FIFO networks, this additional equation is

$$D_k(t + W_j(t)) = Z_k(0) + A_k(t)$$
 for  $k = 1, \dots, K.$  (4.48)

For SBP networks with preemption, the equation is

$$t - T_k^+(t)$$
 can only increase when  $Z_k^+(t) = 0$  for  $k = 1, \dots, K$ . (4.49)

Here,  $Z_k^+(t)$  denotes the sum of the queue lengths at the station j = s(k) of classes having priority at least as great as k, and  $T_k^+(t)$  denotes the corresponding sum of cumulative service times. The order of the priorities is given by the specific SBP discipline.

As mentioned in Section 1.3, there is some flexibility in the choice of the components of  $\mathfrak{X}(\cdot)$  and the corresponding queueing network equations. For specific disciplines, one typically eliminates one or more of these components. For instance, for HL queueing networks, it is generally not necessary to employ both  $D(\cdot)$  and  $T(\cdot)$ . Since different variables will be more natural in different settings, we employ the flexible formulation given above.

The discerning reader might note that in Section 1.3, we employed equation (1.10) rather than (4.47) to relate D(t) and T(t). This has the advantage of leading to the formula in (1.11), but does not incorporate the HL property.

Both formats lead to the same fluid model equation (4.55) given below, if the HL property is implicitly assumed in conjunction with (1.10).

#### Fluid models

Fluid model equations were discussed in Section 1.3. They are the deterministic analog of queueing network equations, with the random quantities  $E(\cdot)$ ,  $\Gamma(\cdot)$ , and  $\Phi(\cdot)$  being replaced by their respective means  $\alpha$ , M, and P. The fluid model equations corresponding to (4.42)-(4.46) are

$$A(t) = \alpha t + P^T D(t), \qquad (4.50)$$

$$Z(t) = Z(0) + A(t) - D(t)$$
(4.51)

$$W(t) = CM(A(t) + Z(0)) - CT(t), \qquad (4.52)$$

$$CT(t) + Y(t) = et, \tag{4.53}$$

 $Y_j(t)$  can only increase when  $W_j(t) = 0, \ j = 1, \dots, J,$  (4.54)

for all  $t \geq 0$ . In the HL setting, one includes

$$T(t) = MD(t), \tag{4.55}$$

which corresponds to (4.47). For a given choice of  $\alpha$ , M, and P, the fluid model equations (4.50)-(4.55) will be referred to as the *basic fluid model equations*. Similarly, the fluid model consisting of the equations (4.50)-(4.55) will be referred to as the *basic fluid model*.

Equations (4.50)–(4.55) do not specify the discipline of the corresponding queueing network. So, as was the case for the queueing network equations, an additional fluid model equation (or equations) still needs to be added. Such an equation will be a deterministic expression involving  $A(\cdot)$ ,  $D(\cdot)$ ,  $T(\cdot)$ ,  $W(\cdot)$ ,  $Y(\cdot)$ , and  $Z(\cdot)$ , and will be referred to as an *auxiliary fluid model equation*. For networks with FIFO and SBP disciplines, the auxiliary fluid model equations are given by (4.48) and (4.49); two examples involving particular SBP networks will be given shortly. In these lectures, a *fluid model* will be a set of fluid model equations that includes the basic fluid model equations (4.50)–(4.55). Solutions of such equations are *fluid model solutions*. In Section 1.3, we were a bit vague on what is meant by a fluid model corresponding to a queueing network. We will make this precise in the next subsection where fluid limits are introduced.

The same notation was used in equations (4.50)-(4.55) as in (4.42)-(4.47) for the unknown variables  $A(\cdot)$ ,  $D(\cdot)$ ,  $T(\cdot)$ ,  $W(\cdot)$ ,  $Y(\cdot)$ , and  $Z(\cdot)$ . When convenient, we will employ the same vocabulary for the fluid model analogs of queueing network quantities, such as the immediate workload  $W(\cdot)$  and the queue length  $Z(\cdot)$ . We will employ  $\mathfrak{X}(t)$ , given in (4.40), for solutions of fluid models of (4.50)-(4.55) and the auxiliary equations that may be added. The use of the same notation for the queueing network and fluid model variables is in general helpful; the one that is meant will be clear from the context.

Equations such as (4.50)-(4.55) are also referred to as *fluid model equations* without delay, since one is, in effect, setting u = v = 0 here, where u and v are

the residual interarrival and service times of the initial state x that were introduced in Section 4.1. For general residual times, the corresponding equations are referred to as *fluid model equations with delay*. In that setting, one needs to modify (4.50), (4.52) and (4.55). The resulting equations are a bit awkward to work with. We will not require these more general equations here, and unless indicated to the contrary, the fluid model equations considered here will always be assumed to be without delay.

We will assume that all of the components of  $\mathfrak{X}(\cdot)$  are nonnegative, with  $A(\cdot)$ ,  $D(\cdot)$ ,  $T(\cdot)$ , and  $Y(\cdot)$  being nondecreasing. Using (4.50)–(4.53), one can check that

$$A(0) = D(0) = T(0) = 0 \quad \text{and} \quad Y(0) = 0, \tag{4.56}$$

which is the analog of (4.41). Employing (4.51), (4.52), and (4.55), one can also show the useful relationship between the queue length and immediate workload vectors,

$$W(t) = CMZ(t) \quad \text{for all } t. \tag{4.57}$$

Using the basic fluid model equations, it is not difficult to check that knowledge of any of D(t), T(t), or Z(t), at a given t, and knowledge of Z(0) are enough to determine all of the components of  $\mathfrak{X}(t)$ . Simple examples show this is not true for either A(t), W(t), or Y(t).

Starting first with  $T(\cdot)$  and  $Y(\cdot)$  in (4.53), it is easy to show that  $A(\cdot), D(\cdot), T(\cdot), W(\cdot), Y(\cdot)$ , and  $Z(\cdot)$  are all Lipschitz continuous. That is, for some N > 0 (depending on the triple  $(\alpha, M, P)$ ),

$$|f(t_2) - f(t_1)| \le N|t_2 - t_1| \quad \text{for all } t_1, t_2 \ge 0, \tag{4.58}$$

if  $f(\cdot)$  is any of the above functions. (Recall that, when dealing with vectors, we always employ the sum norm, although this is a matter of convenience.) Consequently, these functions are absolutely continuous, and so f'(t) exists a.e., with

$$f(b) - f(a) = \int_{a}^{b} f'(t)dt$$
 for all  $a, b.$  (4.59)

Times at which the derivative exists for all of the components of  $\mathfrak{X}(\cdot)$  will be referred to as *regular points*.

The representation in (4.59) will be quite useful later on. Assume, for instance, that the dot product  $(Z'(t), w) \leq -\epsilon$ , for some fixed  $\epsilon > 0$  and fixed vector w with nonnegative coordinates, whenever  $Z(t) \neq 0$  and t is a regular point. Then, it is not difficult to see, using (4.59), that

$$Z(t) = 0 \text{ for } t \ge (Z(0), w)/\epsilon.$$
 (4.60)

In analogy with queueing networks, one can envision the components of  $\mathfrak{X}(\cdot)$  for fluid models in terms of continuous "job mass" flowing through the system. Also in analogy with queueing networks, for the prescribed triple  $(\alpha, M, P)$ , stations are defined as *subcritical* or *critical*, if  $\rho_j < 1$  or  $\rho_j = 1$ , where  $\rho_j$  is given by (1.7), for Q and  $\lambda$  defined as in (1.2) and (1.5). The fluid model is labelled correspondingly if all stations are of the same type. We recall from (1.18) that a fluid model is *stable* if there exists an N > 0, so that for any solution of its fluid model equations, the  $Z(\cdot)$  component satisfies

$$Z(t) = 0 \quad \text{for } t \ge N |Z(0)|. \tag{4.61}$$

The main result in Section 4.4, Theorem 4.16, gives general criteria for the stability of the corresponding queueing network. An important condition is that its fluid model be stable. (Other conditions involve the interarrival time distributions of the queueing network.) Ascertaining whether a fluid model is stable is itself not an elementary problem in general, and will be discussed in Chapter 5.

The following results are easy to derive using our present machinery, and will be useful for showing analogous results for queueing networks. Parts (a), (b), (c) and (d) of Proposition 4.11 will be used, respectively, for Example 1 of Section 4.4, Corollaries 1 and 2 of Proposition 4.12 of this section, and Proposition 5.21 on Section 5.5. As usual, the inequalities in Part (b) are to be interpreted componentwise.

**Proposition 4.11.** (a) Any fluid model with  $\sum_{j} \rho_{j} < 1$  is stable. (b) For any solution of a fluid model,

$$\liminf_{t \to \infty} Y(t)/t \ge e - \rho.$$

The rate of convergence is uniform over bounded |Z(0)| for these solutions. When Z(0) = 0,  $Y(t) \ge (e - \rho)t$  for all t. (c) Suppose that for some solution of a fluid model,  $Z_k(t) < Z_k(0)$  for some t and all k. Then,  $\rho < e$ . (d) Suppose that for a fluid model,  $\rho_j > 1$  for some j. Then, for some  $\epsilon > 0$ ,  $|Z(t)| \ge \epsilon t$  for all t and all fluid model solutions.

*Proof.* By (4.50) and (4.51),

$$Z(t) - Z(0) = \alpha t - (I - P^T)D(t).$$

Multiplying both sides by CMQ gives

$$CMQ(Z(t) - Z(0)) = \rho t - CMD(t) = \rho t - CT(t).$$
 (4.62)

The first equality employs  $\rho = CM\lambda = CMQ\alpha$ , which follows from (1.5) and (1.7), and the second equality follows from (4.55). By (4.53), (4.62) can be rewritten as

$$CMQ(Z(t) - Z(0)) = (\rho - e)t + Y(t).$$
(4.63)

Parts (c) and (d) follow quickly from the resulting inequality

$$CMQ(Z(t) - Z(0)) \ge (\rho - e)t.$$
 (4.64)

Under the assumption in (c), the left side of (4.64) is negative in each coordinate for that t, and so  $\rho < e$ . (Note that  $Q \ge I$ .) Let  $\rho_j > 1$  for a given j, as in (d). The j coordinate of the left side of (4.64) is bounded below by  $(\rho_j - 1)t$ , and the matrix CMQ is constant. So, (d) holds for an appropriate choice of  $\epsilon > 0$ . The display in Part (b) follows from (4.63), after dividing both sides by t and taking limits. Since  $CMQZ(0)/t \to 0$  as  $t \to \infty$ , the rate of convergence for the lim inf in (b) is also uniform over bounded |Z(0)|. The case where Z(0) = 0 is an immediate consequence of (4.63).

For Part (a), multiplication of both sides of (4.62) by  $e^T$  and taking derivatives gives

$$e^T CMQZ'(t) = e^T (\rho - CT'(t))$$

at all regular points. By (4.53), for each choice of j, this is

$$\leq \sum_{j'} \rho_{j'} - \sum_{k \in \mathcal{C}(j)} T'_k(t) = \sum_{j'} \rho_{j'} - 1 + Y'_j(t).$$

By (4.54) and (4.57),  $Y'_j(t) = 0$  for at least one choice of j when  $Z(t) \neq 0$ . Setting  $\epsilon = 1 - \sum_{j'} \rho_{j'} > 0$ , it follows that

$$e^T CMQZ'(t) \le -\epsilon$$

at such points. By the bound in (4.60),

$$Z(t) = 0 \quad \text{for } t \ge e^T C M Q Z(0) / \epsilon,$$

and so the fluid model is stable.

Even though fluid model equations are simplifications of the corresponding queueing network equations, we will need to exercise some caution in their application. In particular, a fluid model need not have a unique solution for given initial data. This is not surprising for the basic fluid model equations, since the solutions may depend on the discipline, which is not included in these equations. However, it is not difficult to show this is also sometimes the case when either equation (4.48) (corresponding to the FIFO discipline) or equation (4.49) (corresponding to an SBP discipline) is added to the basic queueing network equations. So, nonuniqueness can persist even when the discipline has been specified.

We conclude this subsection with two examples of fluid models which exhibit this nonuniqueness. Both examples employ fluid models that correspond to certain parameter values for the Rybko-Stolyar network that was introduced in Section 3.1. The routes, which are given in Figure 3.2, each possess two stations, which are each visited once along the route. The discipline is preemptive SBP, with priority at each station given to the class of jobs that are about to leave the network. Classes are labelled by (i, k), where i = 1, 2 denotes the route and k = 1, 2 denotes the sequential ordering of the class along the route. The fluid model is assumed to consist of the basic fluid model equations (4.50)–(4.55) together with (4.49). The routing matrix P here is given by the above routing. The external arrival rates are given by

$$\alpha_{1,1} = \alpha_{2,1} = 1, \quad \alpha_{1,2} = \alpha_{2,2} = 0;$$

and the service times are assumed to satisfy

$$m_1 \stackrel{\text{def}}{=} m_{1,1} = m_{2,1} > 0, \quad m_2 \stackrel{\text{def}}{=} m_{1,2} = m_{2,2} > 0,$$

which will be further specified in the examples. The SBP equation (4.49) is assumed to incorporate the above priority scheme.

**Example 1.** A fluid model with nonunique solutions. In this example, we assume that  $m_1 < m_2$  and assign the initial data

$$Z_{1,1}(0) = Z_{2,1}(0) = 1, \quad Z_{1,2}(0) = Z_{2,2}(0) = 0.$$

On account of the discipline, when either  $Z_{1,2}(t_0) \neq 0$  or  $Z_{2,2}(t_0) \neq 0$ , this is enough to uniquely determine the evolution of  $\mathfrak{X}(t)$  for small times after  $t_0$ . This is not the case for the given initial data since, as one can check, there exist distinct solutions over  $t \in [0, 1/(1 - \mu_1)]$ , with

$$Z_{i,1}(t) = 1 + (1 - \mu_1)t, \quad Z_{i,2}(t) = (\mu_1 - \mu_2)t, \quad (4.65)$$
$$Z_{i',1}(t) = 1 + t, \quad Z_{i',2}(t) = 0,$$

for either i = 1 or i = 2, where i' denotes the other route. (As usual,  $\mu_k = 1/m_k$ .)

One can interpret the above evolution of the fluid model as follows. The initial job mass in route i "gets an infinitesimal lead" over that in route i', with job mass starting to flow into the class k = 2 along the route before this starts to occur along route i'. Once this flow begins, since  $\mu_2 < \mu_1$ , there will be mass at class (i, 2). The mass at (i, 2) has priority of service over that at (i', 1). This prevents service at (i', 1), and so the class (i', 2) remains empty. Mass from (i, 1) continues to flow to (i, 2) until at least time  $1/(\mu_1 - 1)$ , after which class (i, 1) is empty, and so the rate at which mass enters (i, 2) slows to  $\alpha_{i,1} = 1$ . Over these times, there will be mass at (i, 2) and no service at (i', 1).

In addition to the solutions of the fluid model with Z(t) given by (4.65), there is the symmetric solution over  $t \in [0, (\mu_1 + \mu_2)/(\mu_1\mu_2 - \mu_1 - \mu_2)]$ , with

$$Z_{i,1}(t) = 1 + (1 - \mu_1 \mu_2 / (\mu_1 + \mu_2))t, \quad Z_{i,2}(t) = 0, \tag{4.66}$$

for i = 1, 2. Here, the fraction of effort allocated to the classes with k = 1 is  $\mu_2/(\mu_1 + \mu_2)$ , with the fraction allocated to k = 2 being  $\mu_1/(\mu_1 + \mu_2)$ . This allocation of effort keeps the high priority classes with k = 2 empty, and so allows continual service at both classes with k = 1.

The solution given by (4.66) is unstable in the sense that, at any time  $t_0$ , there are other solutions emanating from it. This occurs when the classes along one of the routes start to "monopolize" the service at their stations. After time  $t_0$ , these solutions evolve as in (4.65), except for the lag in time and the different starting mass at the classes with k = 1. With a bit of effort, one can check that all of the solutions of the fluid model in a neighborhood of time 0 and with the assigned initial data are given by the above solutions. That is, the solution in (4.66) is followed until some assigned time, after which the solutions evolve as in (4.65).

We point out that for  $m_1 + m_2 < 1/2$ , one has  $\rho_1 + \rho_2 < 1$ , and so the assumptions of Part (a) of Proposition 4.11 are satisfied. Therefore, under this additional condition, the above fluid model is stable. As mentioned above the

proposition, this is sufficient for the corresponding queueing networks to be stable (under appropriate conditions on the interarrival time distributions). So, the nonuniqueness of solutions for the fluid model is not connected with the stability of either the fluid model or the corresponding queueing network.

We also point out that, although we have not bothered to construct the above fluid model solutions for all time, these solutions can always be extended past the times that are given. One way of doing this is to employ fluid limits, which are introduced in the next section.  $\hfill \Box$ 

**Example 2.** A fluid model that has a nonzero solution with Z(0) = 0. In this example, we assume that  $m_1 < 1/3$  and  $m_2 = 2/3$ . (We fix  $m_2$  in order to simplify the coefficients in our computations.) By Theorem 3.4, the corresponding queueing network is unstable, if the interarrival and service times are exponentially distributed. The fluid model exhibits similar behavior, which we show for Z(0) = 0. We do this by constructing a self-similar solution, in the spirit of the proof of Theorem 3.1. We give the values of  $T(\cdot)$  and  $Z(\cdot)$ ; using (4.50)–(4.55), the other components of  $\mathfrak{X}(\cdot)$  can be calculated from these.  $(T(\cdot) \text{ and } Z(\cdot) \text{ can also be calculated from each other.)}$ 

We proceed in two steps. We first construct a solution  $\hat{\mathfrak{X}}(\cdot)$  of the fluid model on [0,2], with  $|\tilde{Z}(0)| = 1$  and  $|\tilde{Z}(2)| = 2$ , where all the mass at t = 0 is at class (1,1) and that at t = 2 is at class (2,1). Because of the symmetry of the network and the doubling of mass by t = 2, the same reasoning allows us to extend the solution up until t = 6, when the total mass is 4 and all the mass is again at class (1,1). We then piece together scaled versions of this solution so that the resulting solution is defined over  $[0, \infty)$ , and grows linearly starting at Z(0) = 0. We note that since  $\rho_1, \rho_2 < 1, Z(t) \equiv 0$  gives another solution of the fluid model. So, this construction gives another example of the nonuniqueness of fluid model solutions under an SBP discipline.

Set  $b_1 = 1/(\mu_1 - 1) = m_1/(1 - m_1)$ . We construct  $\tilde{T}(t)$  and  $\tilde{Z}(t)$ , with  $t \in [0, 2]$ , piecewise over the time intervals  $[0, b_1]$  and  $[b_1, 2]$ . We choose  $\tilde{T}'(t)$  so that it is constant over each interval, with

$$\tilde{T}'_{1,1}(t) = \begin{cases}
1 & \text{for } t \in (0, b_1), \\
m_1 & \text{for } t \in (b_1, 2), \\
\tilde{T}'_{1,2}(t) = 1 & \text{for } t \in (0, 2), \\
\tilde{T}'_{2,1}(t) = \tilde{T}'_{2,2}(t) = 0 & \text{for } t \in (0, 2).
\end{cases}$$
(4.67)

Integration then gives  $\tilde{T}(t)$ . The function  $\tilde{Z}(t)$  will be linear over these intervals, with values at the endpoints given by

$$\tilde{Z}_{1,1}(0) = 1, \quad \tilde{Z}_{1,1}(b_1) = \tilde{Z}_{1,1}(2) = 0,$$

$$\tilde{Z}_{1,2}(0) = 0, \quad \tilde{Z}_{1,2}(b_1) = b_1(\mu_1 - 3/2), \quad \tilde{Z}_{1,2}(2) = 0,$$

$$\tilde{Z}_{2,1}(0) = 0, \quad \tilde{Z}_{2,1}(2) = 2, \quad \tilde{Z}_{2,2}(0) = \tilde{Z}_{2,2}(2) = 0.$$
(4.68)

Note that the value of  $\mu_1$ , for  $\mu_1 > 3/2$ , does not affect the value of  $\hat{Z}(2)$ .

The choice of  $\tilde{T}(\cdot)$  and  $\tilde{Z}(\cdot)$  as in (4.67)–(4.68) corresponds to the assignment of all effort at both stations to the classes along the first route. This is consistent with the discipline, because class (1, 2) has priority over (2, 1), and there is never any mass at (2, 2) to impede service at (1, 1). The time interval is divided into two parts, with the latter beginning when class (1, 1) first becomes empty. As claimed earlier, all of the 2 units of mass at t = 2 is at class (2, 1). A repetition of this reasoning, this time over the interval [2, 6], shows that at t = 6, all of the mass is again at class (1, 1), with

$$\tilde{Z}_{1,1}(6) = 4.$$

The second step consists of piecing together scaled versions of the above construction. For  $t \in [\frac{1}{2}4^{i+1}, \frac{1}{2}4^{i+2}]$ , set

$$T(t) = 4^{i} (\tilde{T}(4^{-i}t - 2) + U), \quad Z(t) = 4^{i} \tilde{Z}(4^{-i}t - 2), \tag{4.69}$$

where U is the constant vector with

$$U_{1,1} = m_1, \quad U_{2,1} = 2m_1, \quad U_{1,2} = U_{2,2} = 4/3,$$

and set T(0) = Z(0) = 0. It is straightforward to check that over  $[\frac{1}{2}4^{i+1}, \frac{1}{2}4^{i+2}]$ , with  $i \in \mathbb{Z}$ ,  $T(\cdot)$  and  $Z(\cdot)$  give a solution of the fluid model, since the effect of the scaling terms  $4^i$  and  $4^{-i}$  cancel each other out and since the translation terms do not affect the evolution of solutions. (i = 0 corresponds to the solution $\tilde{\mathfrak{X}}(\cdot)$ , after a time shift.) By (4.68),  $\tilde{Z}(6) = 4\tilde{Z}(0)$ ; therefore,  $Z(\cdot)$  is consistently defined at the endpoints  $\frac{1}{2}4^i$ . One can also check that the same is true for  $T(\cdot)$ . So,  $T(\cdot)$  and  $Z(\cdot)$  define a solution over  $(0, \infty)$ . Since

$$\lim_{t\downarrow 0} T(t) = \lim_{t\downarrow 0} Z(t) = 0,$$

 $T(\cdot)$  and  $Z(\cdot)$  are continuous at 0, which implies that  $T(\cdot)$  and  $Z(\cdot)$  define a solution over  $[0, \infty)$ , as desired.

It follows from  $|Z(2)| = |\tilde{Z}(0)| = 1$  and (4.69), evaluated  $t = \frac{1}{2}4^i$ ,  $i \in \mathbb{Z}_+$ , that

$$\limsup_{t \to \infty} |Z(t)|/t \ge 1/2.$$

One can, in fact, check with a little more work that  $\lim_{t\to\infty} |Z(t)|/t = 1/2$ .

#### Fluid limits

The basic fluid model equations (4.50)-(4.55) are obtained from the corresponding queueing network equations (4.42)-(4.47) by substituting the means  $\alpha, M$ , and P for  $E, \Gamma$ , and  $\Phi$ . Auxiliary equations that specify the discipline, such as (4.48) and (4.49), are obtained similarly from the corresponding queueing network equations. In addition to being obtained by this formal substitution, fluid model equations can be derived from a "law of large numbers" scaling of the form  $\mathfrak{X}(st)/s$  as  $s \to \infty$ . This interpretation will be important in Section 4.4 when we use the stability of fluid models to show the stability of the corresponding queueing networks. Fluid limits make this limiting procedure precise.

We first define the set G on which we will take fluid limits. Let G be the set on which the strong law of large numbers holds for the external arrivals, service times, and routing of a queueing network. That is, for  $\omega \in G$ ,

$$\lim_{n \to \infty} \frac{1}{n} \sum_{i=1}^{n} \xi_k(i) = 1/\alpha_k, \quad \lim_{n \to \infty} \frac{1}{n} \sum_{i=1}^{n} \gamma_k(i) = m_k, \quad (4.70)$$
$$\lim_{n \to \infty} \frac{1}{n} (\Phi^k(n))_\ell = P_{k,\ell},$$

where  $k \in \mathcal{A}$  in the first term and k = 1, ..., K elsewhere, and  $\ell = 1, ..., K$ in the last term. One has P(G) = 1. (Other sets G', with P(G') = 1 and satisfying (4.70), can also be used.)

We define fluid limits as follows. Let  $(a_n, x_n)$  be a sequence of pairs, with

$$\lim_{n \to \infty} a_n = \infty, \quad \limsup_{n \to \infty} |z_n|/a_n < \infty, \tag{4.71}$$
$$\lim_{n \to \infty} |u_n|/a_n = \lim_{n \to \infty} |v_n|/a_n = 0.$$

Here,  $a_n \in \mathbf{R}_+$  and  $x_n \in S$ , with  $z_n, u_n$ , and  $v_n$  denoting the queue length, residual interarrival time, and residual service time vectors, with  $|x_n| = |z_n| + |u_n| + |v_n|$  as in (4.3). (Throughout the remainder of this section and the next, we will employ subscripts for terms in sequences as well as for coordinates of vectors; which one is meant will be clear from the context.) A *fluid limit* of the queueing network, with queueing network process  $\mathfrak{X}(\cdot)$ , is any limit

$$\bar{\mathfrak{X}}(t) = \lim_{n \to \infty} \frac{1}{a_n} \mathfrak{X}^{x_n}(a_n t), \qquad (4.72)$$

for any choice of  $\omega \in G$  and any sequence  $(a_n, x_n)$  satisfying (4.71). Convergence is required to be uniform on compact sets (u.o.c.), and is to be interpreted componentwise with respect to each of the six components of  $\overline{\mathfrak{X}}(\cdot)$ . Here and later on, the superscript gives the initial state of a process, e.g.,  $\mathfrak{X}^x(\cdot)$  indicates that  $\mathfrak{X}^x(0) = x$ .

We will say that the family of these fluid limits is associated with the queueing network. Such a family will be stable if there exists an N > 0, so that for any of its fluid limits, the  $\overline{Z}(\cdot)$  component satisfies

$$\bar{Z}(t) = 0 \quad \text{for } t \ge N |\bar{Z}(0)|.$$
 (4.73)

This definition of stability is the analog of that for fluid models in (4.61). (The scaled sequences  $\{\mathfrak{X}^{x_n}(a_n t)/a_n, n \in \mathbb{Z}_+\}$  that are obtained from a queueing network process, and their limits are frequently referred to as a fluid limit model. We do not use that terminology here.)

We next employ fluid limits to specify the relationship between queueing networks and fluid models that we will use in the succeeding sections. Let  $\mathcal{M}$
be a fluid model, that is, a set of fluid model equations including the basic fluid model equations (4.50)–(4.55). Then,  $\mathcal{M}$  is associated with a queueing network if, for  $\omega \in G$ , each sequence  $(a_n, x_n)$  satisfying (4.71) possesses a subsequence  $(a_{i_n}, x_{i_n})$  (depending on  $\omega$ ), on which the components of the scaled sequences  $\mathfrak{X}^{x_{i_n}}(a_{i_n}t)/a_{i_n}$  obtained from the queueing network process converge u.o.c., and this limit satisfies the fluid model equations of  $\mathcal{M}$ . Such a limit will automatically satisfy (4.56) and the positivity and monotonicity properties given immediately before (4.56).

We note that these limits are, by definition, fluid limits of the queueing network. We are thus requiring here that each sequence possess a subsequence with a fluid limit that satisfies the fluid model equations. It also follows from this definition that every fluid limit must satisfy the fluid model equations. (The definitions given here for fluid limits and associated fluid models differ somewhat from those in the literature.)

Consider a fluid model whose triple  $(\alpha, M, P)$  corresponds to the triple  $(E(\cdot), \Gamma(\cdot), \Phi(\cdot))$  of a queueing network. (That is  $\alpha, M$ , and P are the means corresponding to  $E(\cdot), \Gamma(\cdot)$ , and  $\Phi(\cdot)$ .) In Proposition 4.12, we will show that, for HL queueing networks, a converging subsequence always exists that satisfies the basic fluid model equations. So, this basic fluid model is always associated with the queueing network. Consequently, in order to show that a fluid model (with corresponding triple  $(\alpha, M, P)$ ) is associated with a given queueing network, it suffices to show that its auxiliary fluid model equations are also satisfied by all fluid limits.

The basic fluid model for a given queueing network is uniquely specified. Typically, there will be a particular "canonical" fluid model that is associated with the queueing network, which includes the basic fluid model equations, together with an appropriate equation (or equations) that describe the discipline. The choice of these auxiliary equations is usually fairly natural. We will, for example, employ (4.48) as the auxiliary fluid model equation for the canonical fluid model for FIFO queueing networks and (4.49) for SBP queueing networks. Note that there exist other associated fluid models; for instance, the basic fluid model for a queueing network is always associated with it. We will also use associated in the opposite direction and say that a queueing network is associated with a fluid model, although there will be many such queueing networks for a given fluid model, since different choices of the distributions of  $\xi(1)$  and  $\gamma(1)$ , with the same means, are possible.

We now state Proposition 4.12, which was cited above.

**Proposition 4.12.** For each HL queueing network and  $\omega \in G$ , every sequence of pairs  $(a_n, x_n)$  satisfying (4.71) possesses a subsequence  $(a_{i_n}, x_{i_n})$  on which the limit in (4.72) exists and is u.o.c. This limit satisfies the basic fluid model equations (4.50)–(4.55) with the corresponding triple. Hence, each HL queueing network is associated with its basic fluid model.

*Proof.* We need to show the existence of a limit  $\overline{\mathfrak{X}}(\cdot)$  that satisfies (4.50)–(4.55) and is u.o.c. These properties will follow from the queueing network equations (4.42)–(4.47) and (4.70). First note that for a given  $\omega \in G$ , one can restrict

the sequence  $x_n$  to a subsequence  $x_{i_n}$ , so that for some  $\overline{T}(\cdot)$ ,

$$\frac{1}{a_{i_n}}T^{x_{i_n}}(a_{i_n}t) \to \bar{T}(t) \quad \text{as } n \to \infty, \tag{4.74}$$

on a dense set of t, say, the nonnegative rationals. On account of (4.45), the terms on the left side of (4.74) are all Lipschitz continuous with coefficient 1, and so the sequence is uniformly equicontinuous. The limit in (4.74) therefore holds u.o.c. for all t, if  $\bar{T}(\cdot)$  is replaced by its continuous extension to all of  $\mathbf{R}_{+,0}$ . Applying (4.45) again, it follows that

$$\bar{Y}(t) = \lim_{n \to \infty} \frac{1}{a_{i_n}} Y^{x_{i_n}}(a_{i_n}t)$$

also exists, with convergence being u.o.c. Moreover, (4.53) is satisfied and both  $\overline{T}(\cdot)$  and  $\overline{Y}(\cdot)$  are Lipschitz continuous.

To show (4.55), note that on account of (4.47),

$$\frac{1}{a_{i_n}}\gamma_k(0) + \frac{1}{a_{i_n}}\sum_{i=1}^{D_k^{x_{i_n}}(a_{i_n}t)-1}\gamma_k(i) \le \frac{1}{a_{i_n}}T_k^{x_{i_n}}(a_{i_n}t) 
< \frac{1}{a_{i_n}}\gamma_k(0) + \frac{1}{a_{i_n}}\sum_{i=1}^{D_k^{x_{i_n}}(a_{i_n}t)}\gamma_k(i), \quad (4.75)$$

for given k, t, and n. Since  $\omega \in G$ , it follows from (4.70) that, for given  $\epsilon > 0$  and large n, the quantity to the right of the strict inequality is at most

$$\frac{1}{a_{i_n}}(m_k+\epsilon)D_k^{x_{i_n}}(a_{i_n}t)+\epsilon.$$

A similar lower bound, with  $\epsilon$  replaced by  $-\epsilon$ , holds for the quantity to the left of the other inequality. Letting  $n \to \infty$ , it follows that  $m_k D_k^{x_{i_n}}(a_{i_n}t)/a_{i_n}$  converges to the limit of  $T_k^{x_{i_n}}(a_{i_n}t)/a_{i_n}$ , which is  $\overline{T}_k(t)$ . Since  $\overline{T}(\cdot)$  is continuous and  $D^{x_{i_n}}(\cdot)$  is nondecreasing, this convergence is u.o.c. So,

$$\bar{D}(t) = \lim_{n \to \infty} \frac{1}{a_{i_n}} D^{x_{i_n}}(a_{i_n}t)$$

also exists, with convergence being u.o.c. and (4.55) holding. Since  $\overline{T}(\cdot)$  is Lipschitz continuous, so is  $\overline{D}(\cdot)$ .

We next show (4.50). Here, one uses the limits involving  $\sum_{i=1}^{n} \xi(i)$  and  $\Phi(n)$  in (4.70), and  $|u_n|$  in (4.71), as  $n \to \infty$ . Together with the above limit  $\overline{D}(\cdot)$ , they imply that

$$\frac{1}{a_{i_n}} E^{x_{i_n}}(a_{i_n}t) \to \alpha t, \quad \frac{1}{a_{i_n}} \sum_k \Phi(D_k^{x_{i_n}}(a_{i_n}t)) \to P^T \bar{D}(t)$$
(4.76)

for all t, as  $n \to \infty$ . Consequently, the limit  $\bar{A}(t)$  exists and (4.50) holds. Since  $E^{x_{i_n}}(\cdot)$  and  $\Phi^k(\cdot)$  are monotone and convergence to  $\bar{D}(\cdot)$  is u.o.c., convergence to  $\bar{A}(\cdot)$  is also u.o.c. Since  $\bar{D}(\cdot)$  is Lipschitz continuous, so is  $\bar{A}(\cdot)$ .

Equation (4.51) and convergence to  $\bar{Z}(\cdot)$  follow quickly from (4.43) and the previous limits, on a further subsequence that is chosen so that  $\bar{Z}(0)$  exists. The derivation of (4.52) from (4.44) and convergence to  $\bar{W}(\cdot)$  are similar to the previous steps.

We still need to show (4.54). Suppose that for some j and all  $t \in [t_1, t_2]$ ,  $\overline{W}_j(t) > 0$ . Since  $\overline{W}_j(\cdot)$  is continuous, it is bounded away from 0 on the interval. Convergence to  $\overline{W}(\cdot)$  is u.o.c., and so for sufficiently large n,  $W_j^{x_{i_n}}(a_{i_n}t) > 0$  on  $[t_1, t_2]$  as well. Because of (4.46), one must have  $Y_j^{x_{i_n}}(t_1) = Y_j^{x_{i_n}}(t_2)$ . Since the same equality also holds in the limit as  $n \to \infty$ , one obtains (4.54), as desired.

Proposition 4.12 will be applied to Theorem 4.16 in Section 4.4. We mention here the following quick applications of the proposition and Part (b) of Proposition 4.11. The first application says that for a queueing network, the limiting proportion of time is positive that a given subcritical station j is empty. We note that this, by itself, does not imply the queueing network is stable. As the examples in Chapter 3 show, different stations can be empty at nonoverlapping times, with  $|Z(t)| \to \infty$  as  $t \to \infty$  nevertheless holding. The second application is a modification of the one just described, and will be employed in Proposition 4.15 of Section 4.4.

**Corollary 1.** For each HL queueing network and any initial state x,

$$\liminf_{t \to \infty} Y^x(t)/t \ge e - \rho \quad on \ G. \tag{4.77}$$

Moreover, for each  $\epsilon > 0$ , there exists a  $c_0$ , so that

$$\liminf_{|x| \to \infty} Y^x(c|x|)/c|x| \ge (1-\epsilon)e - \rho \quad on \ G, \tag{4.78}$$

for all  $c \geq c_0$ .

*Proof.* We demonstrate (4.77). Suppose that for some  $\omega \in G$ , the above limit is false. That is, along some sequence of times  $a_n$ , with  $a_n \to \infty$  as  $n \to \infty$ ,

$$\lim_{n \to \infty} Y_j^x(a_n)/a_n < 1 - \rho_j \quad \text{for some } j.$$
(4.79)

By Proposition 4.12,

$$\bar{\mathfrak{X}}(t) = \lim_{n \to \infty} \mathfrak{X}^x(a_{i_n} t) / a_{i_n}$$

exists along some subsequence  $a_{i_n}$ , and satisfies the basic fluid model equations. Since  $\bar{Z}(0) = 0$ , it follows from Part (b) of Proposition 4.11 that  $\bar{Y}(t) \ge (e - \rho)t$  for all t. Therefore,

$$\liminf_{n \to \infty} Y^x(a_{i_n}t)/a_{i_n} \ge (e-\rho)t, \tag{4.80}$$

which contradicts (4.79).

The argument for (4.78) is almost the same. In (4.79), one replaces  $a_n$  by  $t|x_n|$  and  $1 - \rho_j$  by  $1 - \epsilon - \rho_j$ , for given  $\epsilon > 0$ . The fluid limit will now satisfy  $|\bar{Z}(0)| \leq 1$ . One obtains from Proposition 4.11 that  $\bar{Y}(t) \geq ((1 - \epsilon)e - \rho)t$  for given  $\epsilon$  and  $c \geq c_0$ , for large enough  $c_0$ . Substitution of the limiting sequence gives the analog of (4.80), which contradicts the analog of (4.79).

We also note that Proposition 4.12 provides a means for showing the existence of fluid model solutions with given initial data  $\mathfrak{X}(0) = x$ . Suppose that  $(a_n, x_n)$ satisfies (4.71) and that  $z_n/a_n \to z$  as  $n \to \infty$ , where z denotes the queue length vector for x. Then, for  $\omega \in G$ , any fluid limit  $\overline{\mathfrak{X}}(\cdot)$  will satisfy the basic fluid model equations (4.50)–(4.55), with  $\overline{Z}(0) = z$ . The same conclusion will hold for more general fluid models that are associated with some queueing network.

Similar reasoning, together with Part (c) of Proposition 4.11, leads to the following elementary result, which will be used in the proof of Proposition 4.15.

## **Corollary 2.** Suppose that the family of fluid limits associated with an HL queueing network is stable. Then, the queueing network is subcritical.

Proof. Consider a sequence of pairs  $(a_n, x_n)$  satisfying (4.71), where for each coordinate  $(z_n)_k$  of  $z_n$ ,  $\liminf_{n\to\infty}(z_n)_k/a_n > 0$ . For any fluid limit  $\overline{\mathfrak{X}}(\cdot)$  of this sequence,  $\overline{Z}(0) > 0$ . Also, since the family of fluid limits is stable,  $\overline{Z}(N|\overline{Z}(0)|) = 0$  for N chosen as in (4.73). By Proposition 4.12, such a fluid limit will exist and will satisfy the basic fluid model equations. It therefore follows from Part (c) of Proposition 4.11 that the fluid model is subcritical, and hence so is the queueing network.

#### Countable state space setting

There are only minor simplifications in this section when the state space S of a queueing network is countable. There are no changes in the approach to queueing network equations and fluid models. The comment on fluid models with delay is now vacuous, since the residual times u and v are no longer part of the state space descriptor. A few statements in the subsection on fluid limits simplify slightly. In (4.71), one omits the conditions on  $u_n$  and  $v_n$ . Also, in the proof of Proposition 4.12, the limits obtained from (4.75) and (4.76) no longer depend on assumptions on the residual times.

#### 4.4. Demonstration of stability

In this chapter, we have stated a number of results for Markov processes and queueing networks. Here, we employ these results to demonstrate the stability and *e*-stability of queueing networks in Theorems 4.16 and 4.17. Proposition 4.6 of Section 4.2 and Proposition 4.12 of Section 4.3 will be the main ingredients for this. In addition, we will need two further results, Propositions 4.14 and 4.15, which we provide in this section. Our approach here is a modification of that in [Br98a], which is based on that in [Da95].

Bounds on |Z(t)|, |U(t)|, and |V(t)|

Proposition 4.6 of Section 4.2 is the Multiplicative Foster's Criterion; it requires bounds on the expectation of  $E_x|X(c(|x|\vee\kappa))|$ , which are given in (4.28). In order to obtain these bounds, we will need bounds on |Z(t)|, |U(t)|, and |V(t)|, and their expectations. These bounds will be provided in Propositions 4.14 and 4.15. To obtain them, we will use the following lemma.

**Lemma 4.13.** Let  $\beta(1), \beta(2), \beta(3), \ldots$  be i.i.d. positive random variables with finite mean. Set  $B_1(t) = \frac{1}{t} \max\{n : \sum_{i=1}^{n-1} \beta(i) \le t\}$ . Then,

$$\sup_{t \ge 1} E[B_1(t); B_1(t) \ge M] \to 0 \quad as \ M \to \infty.$$
(4.81)

Set  $B_2(t) = \frac{1}{t} \max\{\beta(n) : \sum_{i=1}^{n-1} \beta(i) \le t\}$ , and let  $G_\beta$  denote the set on which  $\frac{1}{n} \sum_{i=1}^n \beta(i) \to m$  as  $n \to \infty$ . Then,

$$B_2(t) \to 0 \quad on \ G_\beta \ as \ t \to \infty,$$
 (4.82)

and

$$E[B_2(t)] \to 0 \quad as \ t \to \infty. \tag{4.83}$$

*Proof.* We first show (4.81). For i = 1, 2, 3, ..., set

$$\tilde{\beta}(i) = \begin{cases} \delta & \text{for } \beta(i) \ge \delta, \\ 0 & \text{for } \beta(i) < \delta, \end{cases}$$

where  $\delta \in (0, 1]$  is chosen small enough so that  $P(\beta(1) > \delta) > 0$ . Define  $\tilde{B}_1(t)$ analogously to  $B_1(t)$ , but with respect to  $\tilde{\beta}(i)$  instead of  $\beta(i)$ . Since  $\tilde{B}_1(t) \ge B_1(t)$ , to show (4.81), it suffices to show the analogous limit for  $\tilde{B}_1(t)$ .

For  $\ell = 0, 1, 2, ..., \text{let } \zeta(\ell)$  denote the number of indices n, n = 0, 1, 2, ..., for which  $\sum_{i=1}^{n} \tilde{\beta}(i) = \delta \ell$ . Then,  $\zeta(0), \zeta(1), \zeta(2), ...$  are i.i.d. random variables with finite means and

$$\tilde{B}_{1}(t) \leq \frac{1}{t} \sum_{\ell=0}^{|t/\sigma|} \zeta(\ell).$$
(4.84)

Using Markov's Inequality, one can check that

$$\sup_{t \ge 1} P\left(\frac{1}{t} \sum_{\ell=0}^{\lceil t/\delta \rceil} \zeta(\ell) \ge M\right) \le \frac{3}{\delta M} E[\zeta(0)] \to 0 \quad \text{as } M \to \infty.$$
(4.85)

By (4.84) and the exchangeability of  $\zeta(\ell)$ ,

$$\sup_{t\geq 1} E[\tilde{B}_1(t); \tilde{B}_1(t) \geq M] \leq \sup_{t\geq 1} \frac{3}{\delta} E\left[\zeta(0); \frac{1}{t} \sum_{\ell=0}^{\lceil t/\delta \rceil} \zeta(\ell) \geq M\right].$$

Because of (4.85), this  $\rightarrow 0$  as  $M \rightarrow \infty$ , which is the desired limit.

To show (4.82), note that on  $G_{\beta}$ ,

$$\frac{1}{n}\sum_{i=1}^{\lfloor cn\rfloor}\beta(i)\rightarrow cm \quad \text{as }n\rightarrow\infty,$$

where convergence is uniform on  $c \in [0, 1]$ . Therefore, for given  $\epsilon > 0$  and large enough n,

$$\max_{i \in [c_1 n, c_2 n)} \beta(i) \le \sum_{i = \lfloor c_1 n \rfloor}^{\lfloor c_1 n \rfloor} \beta(i) \le (c_2 - c_1 + \epsilon) mn$$

for all  $0 \le c_1 \le c_2 \le 1$ . Setting  $c_2 = c_1 + \epsilon$  and  $n = \lfloor 2t/m \rfloor$ , one has, for large enough t, that

$$B_2(t) \le \frac{1}{t} \sup_{c_1 \in [0,1]} \max_{i \in [c_1 n, (c_1 + \epsilon)n)} \beta(i) \le 2\epsilon mn/t \le 4\epsilon.$$

Since  $\epsilon > 0$  is arbitrary, (4.82) follows.

We will show that

$$\sup_{t \ge 1} E[B_2(t); B_2(t) \ge M] \to 0 \quad \text{as } M \to \infty.$$
(4.86)

Together with (4.82), this will imply (4.83) since  $P(G_{\beta}) = 1$ . For this, we again choose  $\delta \in (0, 1]$  small enough so  $P(\beta(1) > \delta) > 0$ , and let  $\beta'(1)$ ,  $\beta'(2)$ ,  $\beta'(3), \ldots$  denote the subsequence obtained by restricting the original sequence to terms with  $\beta(i) \ge \delta$ . Clearly,  $B_2(t) \le \frac{1}{t} \max_{1 \le i \le \lceil t/\delta \rceil} \beta'(i)$ . Also, by Markov's Inequality,

$$\sup_{t \ge 1} P\left(\frac{1}{t} \max_{1 \le i \le \lceil t/\delta \rceil} \beta'(i) \ge M\right) \le \sup_{t \ge 1} \frac{1}{Mt} E\left[\max_{1 \le i \le \lceil t/\delta \rceil} \beta'(i)\right]$$
$$\le \frac{2}{\delta M} E[\beta'(1)],$$

which  $\rightarrow 0$  as  $M \rightarrow \infty$ , since  $\beta'(1)$  has finite mean. It follows that the left side of (4.86) is at most

$$\frac{2}{\delta} \sup_{t \ge 1} E\left[\beta'(1); \ \frac{1}{t} \max_{1 \le i \le \lceil t/\delta \rceil} \beta'(i) \ge M\right],$$

which  $\rightarrow 0$  as  $M \rightarrow \infty$ . This shows (4.86) and hence (4.83).

By applying the first part of Lemma 4.13, it is easy to obtain uniform bounds on the growth of the queue length  $Z^x(t)$  of an HL queueing network. (The result is not dependent on the HL property and so holds for more general networks, once defined.) Recall that the set G is given in (4.70).

282

**Proposition 4.14.** For each HL queueing network, c > 0, and  $\omega \in G$ ,

$$\limsup_{|x| \to \infty} |Z^{x}(c|x|)| / |x| \le c|\alpha| + 1.$$
(4.87)

Moreover,

$$\sup_{t \ge 1} \sup_{|x| \le t} E[|Z^x(t)|/t; \ |Z^x(t)| \ge Mt] \to 0 \quad as \ M \to \infty;$$
(4.88)

in particular,

$$\sup_{t \ge 1} \sup_{|x| \le t} E[|Z^x(t)|]/t < \infty.$$
(4.89)

*Proof.* At any time t,

$$|Z^{x}(t)| \le |E^{x}(t)| + |z|, \qquad (4.90)$$

that is, the sum of the queue lengths is bounded by the number of external arrivals plus the sum of the original queue lengths. Also, on G,

$$\limsup_{|x|\to\infty} E_k^x(c|x|)/|x| \le c\alpha_k,$$

for any k, since the number of external arrivals will only be reduced, at  $k \in \mathcal{A}$ , as the initial residual interarrival time  $u_k$  of x increases. Summation of the components in this inequality and application of (4.90) implies (4.87), since  $|z| \leq |x|$ .

To obtain the uniform integrability bound in (4.88), set  $\beta(i) = \xi_k(i)$  in Lemma 4.13, where  $\xi_k(1), \xi_k(2), \ldots$  is the sequence of interarrival times at  $k \in \mathcal{A}$  defined earlier in the chapter. Since

$$E_k^x(t)/t \le B_1(t)$$

for all t and x, it follows from (4.81) that

$$\sup_{t \ge 1} \sup_{x} E[E_k^x(t)/t; \ E_k^x(t) \ge Mt] \to 0 \quad \text{as } M \to \infty.$$

This limit holds trivially for  $k \notin A$ . Since the sum of uniformly integrable random variables is uniformly integrable,

$$\sup_{t\geq 1} \sup_{x} E[|E^{x}(t)|/t; \ |E^{x}(t)|\geq Mt] \to 0 \quad \text{as } M\to\infty.$$

Together with (4.90), this implies (4.88). The limit in (4.89) is an immediate consequence of (4.88).

By applying the last two limits in Lemma 4.13, one can obtain the following bounds on the growth of the residual times  $U^x(t)$  and  $V^x(t)$ , as  $|x| \to \infty$ .

**Proposition 4.15.** For each subcritical HL queueing network, there exists a  $c_0 \ge 1$ , so that for  $c \ge c_0$ ,

$$\frac{1}{|x|}|U^x(c|x|)| \to 0, \ \frac{1}{|x|}|V^x(c|x|)| \to 0 \quad on \ G \ as \ |x| \to \infty.$$
(4.91)

Moreover,

$$\sup_{t \ge 1} \sup_{|x| \le t} \frac{1}{t} E|U^x(t)| < \infty, \quad \sup_{t \ge 1} \sup_{|x| \le t} \frac{1}{t} E|V^x(t)| < \infty$$
(4.92)

and

$$\frac{1}{|x|}E|U^{x}(c|x|)| \to 0, \ \frac{1}{x}E|V^{x}(c|x|)| \to 0 \quad as \ |x| \to \infty.$$
(4.93)

In particular, (4.91)-(4.93) all hold when the family of fluid limits that is associated with the queueing network is stable.

*Proof.* The last assertion is an immediate consequence of Corollary 2 to Proposition 4.12, which asserts that the queueing network is subcritical when the fluid limits are stable.

For the first half of (4.91), we set  $\beta(i) = \xi_k(i)$  in Lemma 4.13, for a given  $k \in \mathcal{A}$ . Clearly, for all t > 0, x, and  $\omega$ ,

$$\frac{1}{t}U_k^x(t) \le B_2(t) \lor \frac{1}{t}u_k.$$
(4.94)

For  $c \ge 1$  and t = c|x|, one has

$$\frac{1}{|x|} U_k^x(c|x|) \le cB_2(c|x|), \tag{4.95}$$

since  $u_k \leq |x| \leq c|x|$ , and so the initial interarrival residual times have expired by then. The first half of (4.91) follows from (4.82) and (4.95), since  $G \subseteq G_\beta$ .

Taking expectations in (4.94) implies

$$\frac{1}{t}E[U_k^x(t)] \le E[B_2(t)] + \frac{1}{t}u_k.$$

Summing over  $k \in \mathcal{A}$  implies the first half of (4.92), since  $\sum_{k \in \mathcal{A}} u_k = |u| \le x \le t$ , for  $|x| \le t$ , and  $\sup_{t \ge 1} E[B_2(t)] < \infty$  because of (4.83). The first half of (4.93) follows from (4.83) and (4.95).

The argument for the second half of (4.91) is similar to that just given for the first half, but with  $\beta(i) = \gamma_k(i)$  in Lemma 4.13, for any k. For all x and  $\omega$ ,

$$\frac{1}{t}V_k^x(t) \le B_2(t) \lor \frac{1}{t}v_k.$$
(4.96)

Setting t = c|x|, for given c, this implies that

$$\frac{1}{|x|} V_k^x(c|x|) \le cB_2(c|x|) \quad \text{on } \sigma_k^x \le c|x|, \tag{4.97}$$

where  $\sigma_k^x$  is the time at which the initial service residual time expires.

The time at which this occurs is not obvious, since the amount of service that class k receives will depend on the discipline. Note, however, that for large enough c and |x| (where the latter depends on  $\omega$ ), and for  $\omega \in G$ , (4.78) of the first corollary to Proposition 4.12 implies that the station j = s(k) is idle, and hence empty, at some time before c|x|. Here,  $\epsilon > 0$  is chosen small enough so that  $\rho_i < 1 - \epsilon$ , which is possible since the network is subcritical. So,

$$\sigma_k^x \leq c|x|$$
 for large enough  $|x|$ ,

for  $\omega \in G$ . Consequently, (4.97) will eventually hold on G, without the restriction that  $\sigma_k^x \leq c|x|$ . The second half of (4.91) follows from this and (4.82), since  $G \subseteq G_\beta$ .

The argument for the second half of (4.92) is the same as that for the first half of (4.92), where one uses (4.96) instead of (4.94). We still need to demonstrate the second half of (4.93). Taking expectations in (4.97), after restricting to the set  $\sigma_k^x \leq c|x|$ , and applying (4.83) gives

$$\frac{1}{|x|} E[V_k^x(c|x|) \mathbf{1}\{\sigma_k^x \le c|x|\}] \to 0 \quad \text{as } |x| \to \infty.$$

On the other hand,  $V_k^x(c|x|) \le v_k \le |x|$  on  $\sigma_k^x > c|x|$ . So, by (4.91) and bounded convergence, one also has

$$\frac{1}{|x|} E[V_k^x(c|x|) \mathbf{1}\{\sigma_k^x > c|x|\}] \to 0 \quad \text{as } |x| \to \infty.$$

Together, these limits imply that  $E[V_k^x(c|x|)]/|x| \to 0$  as  $|x| \to \infty$ , which implies (4.93).

#### The main theorem

We now have the needed background to show the main result on stability for queueing networks, Theorem 4.16. A similar result for *e*-stability will be shown at the end of the section.

**Theorem 4.16.** For a given HL queueing network, suppose that  $A = \{x : |x| \le \kappa\}$  is petite for each  $\kappa > 0$ . If the family of fluid limits that is associated with the queueing network is stable, then the queueing network is stable. In particular, the queueing network is stable whenever an associated fluid model is stable.

Before proving the theorem, we first provide some motivation for its assumptions. After the proof, we discuss the theorem further and provide some extensions.

Petite was defined in the last part of Section 4.1, and says, in essence, that each set, after being weighted according to some measure  $\nu$ , is "equally accessible" from all states in a petite set A. We require the above sets  $A = \{x : |x| \le \kappa\}$ to be petite in order to be able to apply Proposition 4.6 in the proof of the theorem. On account of Proposition 4.7, the assumptions (4.30) and (4.31) on the distributions of the interarrival times suffice for these sets to be petite. By Proposition 4.8, (4.30) suffices when  $|\mathcal{A}| = 1$ . For the sake of concreteness, results on the stability of queueing networks in the literature have often made these assumptions, rather than assuming the sets A are petite. It is not clear what broader conditions might replace them.

In the statement of the theorem, one also needs to assume that either the associated fluid limits or associated fluid model is stable. Demonstration of either of these conditions requires most of the effort, in practice, when applying the theorem. The assumption on the stability of fluid limits is more general than the corresponding assumption on fluid models.

The latter assumption is the more tractable version, and is the one typically used in applications. It reduces a random problem, the stability of a queueing network, to a (presumably simpler) deterministic problem, the stability of a fluid model. This problem consists of analyzing the solutions of the deterministic equations that constitute the fluid model. In the first three sections of Chapter 5, we will present several examples that illustrate the power of this approach. In such applications, one also needs to show that a specific fluid model is associated with a given queueing network. This is typically routine, with the argument for justifying the auxiliary fluid model equations proceeding along the same lines as the proof of Proposition 4.12.

Unfortunately, there is no satisfactory converse to Theorem 4.16, where the stability of an appropriately chosen fluid model follows from the stability of the queueing network. This will be discussed in Section 5.5.

*Proof of Theorem* **4.16**. We first note that the family of fluid limits that is associated with a queueing network is a subset of the solutions of any associated fluid model. This is an immediate consequence of the definition of an associated fluid model. Stability of an associated fluid model therefore implies stability of the associated fluid limits. So, the second claim in the statement of the theorem follows from the first.

Recall that for any x in the state space, |x| = |z| + |u| + |v|. Setting  $t = c(|x| \vee 1)$ , with  $c \ge 1$ , in (4.89) of Proposition 4.14 and (4.92) of Proposition 4.15, one has

$$E_x|X(c(|x| \lor 1))| \le bc(|x| \lor 1) \tag{4.98}$$

for some b and any x. We will demonstrate here that for large enough c,

$$\frac{1}{|x_n|} E_{x_n} |X(c|x_n|)| \to 0 \quad \text{as } n \to \infty,$$
(4.99)

for any sequence  $x_n$ , with  $x_n \to \infty$  as  $n \to \infty$ . This is equivalent to

$$\frac{1}{|x|}E_x|X(c|x|)| \to 0 \quad \text{as } |x| \to \infty.$$
(4.100)

Together, (4.98) and (4.100) imply (4.28) for an appropriate choice of  $\kappa$ . This is the basic assumption for the Multiplicative Foster's Criterion (Proposition 4.6).

Since |x| is continuous in x, the set  $A = \{x : |x| \le \kappa\}$  is closed. By assumption, A is petite. So, it will follow from (4.28) that the underlying Markov process

 $X(\cdot)$  is positive Harris recurrent, and hence the queueing network is stable. The remainder of the proof is devoted to the demonstration of (4.99).

We break the demonstration of (4.99) into two main steps. Step 1 consists of translating the condition (4.73), for fluid limit stability, into a related limit (4.104) on the expected value of  $|Z(\cdot)|$  for the queueing network. We restrict our attention here to sequences of pairs  $(a_n, x_n)$  as in (4.71), since this condition is assumed for fluid limits. In Step 2, we start the process at general  $x_n$ . After restarting it at times  $c_1|x_n|$  where (4.71) is satisfied, we employ (4.104) from Step 1 to get (4.110). Using Proposition 4.15, we finish the proof by showing the corresponding limit, but with  $|X(\cdot)|$  replacing  $|Z(\cdot)|$ , which gives us (4.99).

Step 1. Assume that the sequence of pairs  $(a_n, x_n)$  satisfies (4.71), and choose  $\hat{z}$  so that  $\hat{z} \geq \limsup_{n \to \infty} |z_n|/a_n$ . We first show that

$$\frac{1}{a_n} |Z^{x_n}(N\hat{z}a_n)| \to 0 \quad \text{on } G \text{ as } n \to \infty, \tag{4.101}$$

for appropriate N > 1 not depending on the sequence.

By Proposition 4.12, any subsequence  $\mathfrak{X}^{x_{i_n}}(\cdot)$  of  $\mathfrak{X}^{x_n}(\cdot)$  has a fluid limit  $\overline{\mathfrak{X}}(\cdot)$ along some further subsequence  $(a_{\ell_n}, x_{\ell_n})$ , as in (4.72). Since the family of fluid limits that is associated with the queueing network is assumed to be stable and  $\hat{z} \geq |\overline{Z}(0)|$ , this implies

$$\overline{Z}(t) = 0 \quad \text{for } t \ge N\hat{z} \tag{4.102}$$

and appropriate N, where we can assume N > 1. Restating the limit in terms of the original process, after setting  $t = N\hat{z}$ , one obtains

$$\frac{1}{a_{\ell_n}} |Z^{x_{\ell_n}}(N\hat{z}a_{\ell_n})| \to 0 \quad \text{as } n \to \infty.$$
(4.103)

Since a subsequence  $(a_{\ell_n}, x_{\ell_n})$  satisfying (4.103) exists for every subsequence  $(a_{i_n}, x_{i_n})$  of  $(a_n, x_n)$ , the limit holds along the entire sequence as in (4.101).

We claim that

$$\frac{1}{a_n} E|Z^{x_n}(N\hat{z}a_n)| \to 0 \quad \text{as } n \to \infty.$$
(4.104)

On account of (4.101), it suffices to show that the sequence there is uniformly integrable for large enough n. This follows from (4.88) of Proposition 4.14, since  $\limsup_{n\to\infty} |x_n|/a_n < N\hat{z}$ .

Step 2. We now allow  $x_n$  to be arbitrary, assuming only that  $|x_n| \to \infty$  as  $n \to \infty$ . We wish to restart the processes  $X^{x_n}(\cdot)$  at times  $c_1|x_n|$ , where  $c_1 \ge 1$  is nonrandom, and at which the following three limits hold for  $\omega \in G$ :

$$\limsup_{n \to \infty} \frac{1}{|x_n|} |Z^{x_n}(c_1|x_n|)| \le c_1 |\alpha| + 1, \tag{4.105}$$

$$\frac{1}{|x_n|}|U^{x_n}(c_1|x_n|)| \to 0 \quad \text{as } n \to \infty,$$
(4.106)

$$\frac{1}{|x_n|} |V^{x_n}(c_1|x_n|)| \to 0 \quad \text{as } n \to \infty.$$
(4.107)

The first limit follows from (4.87) of Proposition 4.14, with arbitrary  $c_1$ , and the last two limits follow from (4.91) of Proposition 4.15, with  $c_1 \ge c_0$ , where  $c_0$  is given in Proposition 4.15.

We set

$$a'_{n} = |x_{n}|, \quad x'_{n} = X^{x_{n}}(c_{1}|x_{n}|), \quad z'_{n} = Z^{x_{n}}(c_{1}|x_{n}|),$$
  
 $u'_{n} = U^{x_{n}}(c_{1}|x_{n}|), \quad v'_{n} = V^{x_{n}}(c_{1}|x_{n}|).$ 

On account of (4.105)–(4.107), the random sequence  $(a'_n, x'_n)$  satisfies (4.71) on the set G. For  $\hat{z}' \stackrel{\text{def}}{=} c_1 |\alpha| + 1$ , one has  $\hat{z}' \geq \limsup_{n \to \infty} |z'_n| / a'_n$  on all such sequences. It follows from (4.104) that

$$\frac{1}{|x_n|} E_{x'_n} |Z(N\hat{z}'|x_n|)| \to 0 \quad \text{on } G \text{ as } n \to \infty.$$

$$(4.108)$$

(In (4.108), we write the initial state  $x'_n$  outside the expectation since  $x'_n$  is random; for nonrandom initial states as in (4.104), the two formulations are of course equivalent.)

Set  $c = c_1 + N\hat{z}' = c_1 + N(c_1|\alpha| + 1)$ . Since  $X(\cdot)$  is Markov and P(G) = 1, it follows from (4.108) that

$$\frac{1}{|x_n|} E_{x_n}[Z(c|x_n|) \mid \sigma(X(c_1|x_n|))] \to 0 \quad \text{as } n \to \infty$$
(4.109)

holds a.s. It is not difficult to check that, because of (4.109),

$$\frac{1}{|x_n|}|Z^{x_n}(c|x_n|)| \to 0 \quad \text{as } n \to \infty$$

holds in probability. On account of (4.88) of Proposition 4.14, the sequence is uniformly integrable. So, in fact,

$$\frac{1}{|x_n|} E|Z^{x_n}(c|x_n|)| \to 0 \quad \text{as } n \to \infty.$$
(4.110)

On the other hand, by (4.93) of Proposition 4.15,

$$\frac{1}{|x_n|} E|U^{x_n}(c|x_n|)| \to 0, \ \frac{1}{|x_n|} E|V^{x_n}(c|x_n|)| \to 0 \quad \text{as } n \to \infty.$$
(4.111)

Together with (4.110), these two limits imply that

$$\frac{1}{|x_n|} E|X^{x_n}(c|x_n|)| \to 0 \quad \text{as } n \to \infty,$$

which is (4.99). This completes the proof of Theorem 4.16.

The reader should note that, in the proof of Theorem 4.16, the only places where the stability of the fluid limits is used are (a) in (4.102), and (b) where Proposition 4.15 is applied. Stability is used in (b) only to conclude the queueing network is subcritical. The latter use of stability can of course be avoided by directly assuming the queueing network is subcritical. The condition in (a) is the major "given" in the theorem, with verification of the stability of the fluid limits, as needed in (4.102), being left to specific applications.

The definition of fluid limit, that is given in Section 4.3, differs somewhat from that given in [Da95]. There, one sets  $a_n = |x_n|$ , and, unlike (4.71), makes no assumptions on  $u_n$  and  $v_n$ . This permits the limiting residual times  $\bar{u}$  and  $\bar{v}$ to be different than 0. The argument for deriving (4.99) is then essentially that given in Step 1 of the above proof together with the limits in (4.111), except that one now requires the stability of the corresponding fluid limits with delay (which will satisfy the corresponding fluid model equations with delay). The approach used here and in [Br98a] allows one to avoid dealing with fluid limits and fluid models with delay altogether. (An intermediate approach for fluid models was given in [Ch95].)

As noted at the end of the preceding sections, analyzing the Markov process  $X(\cdot)$  becomes easier when the state space is countable. We point out that the same is true here, although in this context, the absence of coordinates corresponding to the exponential residual times, rather than countability, is what is important. In this simpler setting, the last two displays in Lemma 4.13 are not needed, nor is Proposition 4.15. The proof of Theorem 4.16 simplifies, with Step 2 no longer being needed. In Step 1, one can instead set  $a_n = |x_n| = |z_n|$ . Then, (4.104) immediately implies (4.99), with c = N.

The stability assumptions in Theorem 4.16 are sufficient for most applications. On occasion, somewhat greater flexibility in the definition of fluid limits is helpful. One can, for instance, replace the assumption of fluid limit stability with

$$\frac{1}{a_n} |Z^{x_n}(N\hat{z}a_n)| \to 0 \quad \text{in probability as } n \to \infty, \tag{4.112}$$

for all pairs of sequences  $(a_n, x_n)$  satisfying (4.71). This generalizes pointwise convergence in (4.101), at the beginning of Step 1 in the proof of the theorem, to convergence in probability. The conclusion (4.104), in Step 1, follows as before from the uniform integrability condition on |Z(t)| in (4.88). In [Br99], the concept *asymptotic stability* is used; this is a weaker variant of fluid limit stability that implies (4.112). As in the previous paragraph, matters simplify in the exponential setting when employing (4.112).

Throughout this chapter, we have assumed that queueing networks are headof-the-line. If one wishes, one can replace this assumption by the assumption that only a bounded number of jobs at any class and time have already received some service there. The approach is then essentially the same as before, with only the obvious modifications being made at certain points. One needs to replace the residual service time vector, in the construction of the state space and the Markov process in the first part of Section 4.1, by a vector with correspondingly more components. Under the corresponding new definition of the norm |x|, the same results leading up to Theorem 4.16 hold as before. The basic fluid model equations (4.50)–(4.55) are still valid. (The queueing network inequality (4.47) leading to (4.55) needs to be modified, though.) Other results where the reasoning proceeds as before include Proposition 4.12, where the fluid limits satisfy the basic fluid model equations, and Proposition 4.15, where the bounds on  $|V^x(\cdot)|$  continue to hold.

This approach fails when the number of jobs that have received partial service is allowed to become unbounded. In particular, the bounds on  $|V^x(\cdot)|$  in Proposition 4.15 can fail, and  $|V^x(c|x|)|$  can be of the same order of magnitude as |x|. Changing the sum norm on |v| to the max norm does not help, since the relationship between fluid limits and the basic fluid model equations in Proposition 4.12 need not continue to hold.

#### e-stability

By strengthening the assumptions in Theorem 4.16 so that the sets A there are uniformly small, one can show, in Theorem 4.17, that the queueing network is e-stable. Theorem 4.17 follows by using the same argument as in Theorem 4.16, but instead applying the last part of Proposition 4.6 to it. A similar result is given in [Du96].

**Theorem 4.17.** Assume that an HL queueing network satisfies the same conditions as in Theorem 4.16, except that the condition that  $A = \{x : |x| \le \kappa\}$  be petite is replaced by the condition that, for each  $\kappa > 0$ , A be uniformly small on an interval  $[s_1, s_2]$ , with  $0 < s_1 < s_2$ . Then, the queueing network is e-stable.

*Proof.* Whenever a set is small, it is also petite. So, all of the assumptions in the statement of Theorem 4.16 are satisfied. Consequently, all steps in the proof remain valid, which include the bound and limit given in (4.98) and (4.100). As before, they together imply the basic assumption (4.28) for the Multiplicative Foster's Criterion (Proposition 4.6), for some choice of  $\kappa$ .

Since |x| is continuous in x, the set A is closed. By assumption, it is uniformly small. Employing the last part of the Multiplicative Foster's Criterion, it follows that  $X(\cdot)$  is ergodic. Hence the queueing network is *e*-stable.

Under the assumptions (4.30) and (4.31) on the interarrival times, it follows from Proposition 4.7 that the sets  $A = \{x : |x| \le \kappa\}, \kappa > 0$ , are uniformly small. It therefore follows from Theorem 4.17 that, under (4.30) and (4.31), a queueing network will be *e*-stable if either its associated fluid limits or associated fluid model is stable.

In Chapter 5, we will employ Theorems 4.16 and 4.17 to demonstrate the stability/*e*-stability for queueing networks with different disciplines. In each case, the main part of the argument consists of showing the stability of an associated fluid model. The amount of work involved depends on the discipline. Here, as an elementary illustration of the procedure, we show the following.

**Example 1.** Assume that (4.30)-(4.31) holds for an HL queueing network with  $\sum_{i} \rho_j < 1$ . Then, the queueing network is e-stable.

*Proof.* By Proposition 4.7 and Theorem 4.17, it suffices to show that an associated fluid model is stable. We do this for the basic fluid model, which, by Proposition 4.12, is always associated with the network. By Part (a) of Proposition 4.11, this fluid model is stable. Consequently, the queueing network is e-stable.

The assumption  $\sum_{j} \rho_{j} < 1$  is sufficiently strong so that the above example can be proven directly, with a bit of work. For this, one can apply the Multiplicative Foster's Criterion to the total workload of the queueing network. (The total workload consists of all future work required of all jobs currently in the network.) As one should expect, the assumption that the discipline be HL is not necessary here, although one then needs to reformulate the definition of the underlying Markov process given in Section 4.1 and its accompanying state space. For less elementary examples, a direct proof of stability or *e*-stability will be quite tedious, if feasible.

#### 4.5. Appendix

The purpose of this section is to serve as an appendix for Section 4.1 and to go into more detail on background material that was omitted there. We cover here four topics. We first discuss the connection between Borel right processes and piecewise-deterministic Markov processes, which was only mentioned briefly in Section 4.1. We then go into more detail on three topics connected with recurrence that were mentioned in the subsection on Harris recurrence. In Proposition 4.18, we show the equivalence of Harris recurrence and of positive Harris recurrence for a Markov process X and its *R*-chain  $\tilde{X}$ , and that X and  $\tilde{X}$  have the same stationary measures. We next summarize how one can demonstrate Theorem 4.1 for a process X by employing its *R*-chain. We then summarize the argument showing the existence of a stationary measure for a discrete time Markov process with a petite set A that satisfies the discrete time analog of (4.19).

#### Borel right processes and PDPs

We first state the definition of a Borel right process, and then show that the class of PDPs that are associated with HL queueing networks are Borel right processes. For the definition of Borel right, we follow Section 27 of [Da93]. Other references are [Ge79], [Kn84], and [Sh88]; [Da93] relies on the approach given in [Sh88].

We assume that the  $\sigma$ -algebras associated with the continuous time Markov process X satisfy the completeness and right continuity conditions given in (4.5) and (4.6). The conditions on page 77 of [Da93] state that  $X(\cdot)$  is Borel right if, in addition:

- (a) The state space S is a Lusin space.
- (b) X has a semigroup  $P^t$  that maps B(S) into B(S).
- (c) The sample paths  $t \mapsto X(t)$  are a.s. right continuous.

(d) If f is an  $\alpha$ -excessive function for  $P^t$ , for some  $\alpha \ge 0$ , then the sample path  $t \mapsto f(X(t))$  is a.s. right continuous.

We proceed to explain the terms that are employed above, and show that these four conditions are satisfied for the Markov processes X underlying the HL queueing networks we have introduced. The first three conditions are quite general, and do not pose any serious constraints. Condition (d) is needed for the regularity of the process. In our setting, none is difficult to show.

A topological space is called a Lusin space if it is homeomorphic to a Borel subset of a compact metric space. As pointed out in [Sh88], a locally compact Hausdorff space with countable basis is a Lusin space, since its one point compactification is compact and metrizable. The space S introduced in Section 4.1 satisfies these properties, and so is a Lusin space. In (b), B(S) denotes the bounded measurable functions on S. The property holds since  $P^t$  is a probability transition kernel. As mentioned in Section 4.1, this is showed in [Da93]. The argument consists of writing  $P^t f$ ,  $f \in B(S)$ , as the limit of the *n*-fold iterates of  $G^t f$ , where  $G^t$  is the truncated operator obtained by stopping X at the time of its first jump, and employing the measurability of  $G^t$ . In (c) and (d), "a.s." means almost surely with respect to all initial probability measures  $\mu$ . Since the sample paths of X are right continuous by construction, property (c) is automatic.

A function f is  $\alpha$ -super-mean-valued, for some  $\alpha \geq 0$ , if  $f(x) \geq 0$  and  $e^{-\alpha t}P^t f(x) \leq f(x)$  for all  $t \geq 0$  and  $x \in S$ . The function f is  $\alpha$ -excessive if it is  $\alpha$ -super-mean-valued and  $e^{-\alpha t}P^t f(x) \uparrow f(x)$  as  $t \downarrow 0$ . In our setting, it is easy to see that (d) holds for the larger family of functions where  $P^t f(x) \to f(x)$  as  $t \downarrow 0$ . To see this, let t < s be close enough so that no jumps occur in (t, s] along a given sample path. Then, since  $X(\cdot)$  is piecewise deterministic,

$$f(X(s)) = f(P^{s-t}(X(t))) = f(P^{s-t}(x)),$$

for x = X(t). As  $s \downarrow t$ , the last quantity converges to f(x) = f(X(t)), which shows  $f(X(\cdot))$  is right continuous, and hence shows (d).

By (i) of Theorem 9.4 of [Ge79], under the conditions (a)–(c), the assumption that X is strong Markov is equivalent to condition (d). So, (d) may be replaced by

#### (d') The process X is strong Markov.

As mentioned in Section 4.1, PDPs are strong Markov, and therefore so are our processes X that are associated with HL queueing networks. The demonstration of (d) is of course quicker than (d') for such processes, but most readers will presumably find the latter condition more familiar.

#### Recurrence for a Markov process and its R-chain

Let X(t),  $t \ge 0$ , be a Markov process and  $\tilde{X}(n)$ , n = 0, 1, 2, ..., be its *R*-chain. The process X inherits certain properties from  $\tilde{X}$ , which we used in Section 4.1. We demonstrate them now in Proposition 4.18. For this, we employ the following standard result from discrete time Markov process theory, which can be found, for instance, in Chapter 10 of [MeT93d]: if a Markov process Y(n), n = 0, 1, 2, ..., is Harris recurrent, then it has a stationary measure, and this measure is unique up to a constant multiple. (We will outline part of the argument in the last subsection.) Much of the proof for the proposition is from [MeT93a].

**Proposition 4.18.** The process X is Harris recurrent, respectively positive Harris recurrent, if and only if  $\tilde{X}$  is. In either case, X and  $\tilde{X}$  have the same stationary measures. Consequently, X has a stationary measure, which is unique up to a constant multiple.

*Proof.* The existence and uniqueness of a stationary measure for X follows immediately from the preceding statement in Proposition 4.18, and the comment before the proposition about discrete time Markov processes.

It is immediate from the characterization of recurrence in (4.10) that when  $\hat{X}$  is recurrent with respect to the measure  $\varphi$ , then so is X. For the other direction, it suffices to show that for  $\varphi(A) > 0$ ,

$$P_x(\tilde{\tau}_A < \infty) = 1$$
 for all  $x$ ,

where  $\tilde{\tau}_A$  is the hitting time of A for  $\tilde{X}$  constructed as in (4.13). This follows immediately from (4.9) and the formula

$$P_x(\tilde{\tau}_A < \infty) = 1 - E_x[e^{-\eta_A}], \qquad (4.113)$$

which is given in Theorem 2.3 of [MeT93a]. Intuitively, (4.113) is not difficult to see since, conditioning on the process X, the number of indices n, at which  $\tilde{X}(n) = X(\sigma_n) \in A$ , will be Poisson with mean  $\eta_A$ , and so

$$P_x(\tilde{\tau}_A = \infty \mid \sigma(X)) = e^{-\eta_A}$$
 for all x

Suppose now that a measure  $\pi$  is stationary for  $P^t$ . Then,  $\pi P^t = \pi$ , and integration against  $e^{-t}$  on both sides gives  $\pi R = \pi$ . So,  $\pi$  is also stationary for R.

It remains to show that if  $\pi$  is stationary for R, then it is also stationary for  $P^t$ . Assuming  $\pi$  is stationary for R, we first show that  $\pi^t \stackrel{\text{def}}{=} \pi P^t$ , t > 0, is also stationary for R, that is,

$$\pi^t = \pi^t R. \tag{4.114}$$

Arguing as in Theorem 3.1 of [MeT93a], for  $A \in \mathcal{S}$ , one has

$$\pi P^t R(A) = \int_0^\infty e^{-s} \pi P^t P^s(A) ds = \int_0^\infty e^{-s} \pi P^s P^t(A) ds$$
$$= \pi R P^t(A) = \pi P^t(A),$$

where the first and third equalities come from reversing the order of integration, and the fourth holds since  $\pi$  is stationary for R. So, (4.114) also holds.

Since the stationary measure for R is unique up to a constant multiple, and both  $\pi$  and  $\pi^t$  are stationary for R, one has

$$\pi^t = c(t)\pi$$
 for  $t \ge 0$ ,

for some measurable function  $c(t) \geq 0$ . Since mass is conserved, one must have  $\pi^t = \pi$  when  $\tilde{X}$  is positive Harris recurrent (and hence  $\pi(S) < \infty$ ). So, in this case,  $\pi$  is stationary for  $P^t$ . The same holds in general, because  $P^{s+t} = P^s \circ P^t$ , and so c(s+t) = c(s)c(t), which implies  $c(t) = e^{Ct}$  for some C. Therefore, for  $A \in \mathscr{S}$ ,

$$\pi(A) = \pi R(A) = \int_0^\infty e^{-t} \pi^t(A) dt = \pi(A) \int_0^\infty e^{(C-1)t} dt.$$

which implies C = 0, and so, in fact,  $\pi^t = \pi$ , as desired.

The following proposition is a variant of Proposition 4.18, and will be used in the last subsection. It employs the resolvent

$$K(x,A) = \sum_{n=1}^{\infty} 2^{-n} P^n(x,A) \quad \text{for } x \in S, \ A \in \mathscr{S},$$
(4.115)

for a discrete time Markov process Y. (Typically, summation is chosen to start at n = 0, with the coefficients modified accordingly.) The corresponding discrete time Markov process, with transition function K, will be referred to as a Kchain. On account of Proposition 4.19, it will be easier to work with the K-chain than with Y directly.

**Proposition 4.19.** Suppose that a discrete time Markov process Y has a small set A that satisfies

$$P_x(\tau_A < \infty) = 1 \quad for \ all \ x \in S. \tag{4.116}$$

Then, A also satisfies (4.116) for the K-chain, with respect to which A will be small with  $m_0 = 1$ . Moreover, Y and its K-chain have the same stationary measures.

*Proof.* The argument that uses (4.113) can also be used to show (4.116) holds for the K-chain. (The number of indices in A, after conditioning on Y, will now be binomial instead of Poisson.) It is easy to see that A will also be small for the K-chain: one can choose  $m_0 = 1$  and small measure  $\nu = 2^{-m'_0}\nu'$ , where  $\nu'$ is the small measure for Y and  $m'_0$  is the time at which it is employed.

Suppose now that a measure  $\pi$  is stationary for P. Then,  $\pi P = \pi$ , and summation against  $2^{-n}$  on both sides gives  $\pi K = \pi$ , so  $\pi$  is also stationary for K. For the other direction, assume that  $\pi K = \pi$ . One has the string of equalities

$$\pi P = \pi K P = 2\pi R - \pi P = 2\pi - \pi P.$$

Comparison of the first and last terms then implies  $\pi P = \pi$ , as desired. The first and third equalities follow from the stationarity of  $\pi$  with respect to K, and the second equality follows quickly from the definition of K.

#### Proof of Theorem 4.1

We summarize here the proof of Theorem 4.1. Our approach will be to first state a discrete time analog of Theorem 4.1, Theorem 4.20, and say a little about how it is shown. We will then summarize how Theorem 4.1 follows from a slight modification of Theorem 4.20 and Proposition 4.18. At various points, this subsection relies on ideas from [MeT93a].

Theorem 4.20 is similar to Theorem 4.1. Differences are that the set A is no longer required to be closed and, in (4.118),  $\tau_A(\delta)$  is replaced by  $\tau_A$ . (These simpler quantities suffice since  $Y(\tau_A) \in A$  is automatic and  $\tau_A(\delta) = \tau_A$  for  $\delta \leq 1$ .) In one direction of Part (a), the set A is chosen to be small (rather than just petite), since this is needed in the next subsection. The state space  $(S, \mathscr{S})$  is not assumed to have a topological structure, although the  $\sigma$ -algebra  $\mathscr{S}$  is assumed to be countably generated.

**Theorem 4.20.** (a) If a discrete time Markov process Y is Harris recurrent, then there exists a small set A with

$$P_x(\tau_A < \infty) = 1 \quad for \ all \ x \in S. \tag{4.117}$$

Conversely, if (4.117) holds for a petite set A, then Y is Harris recurrent. (b) Suppose the discrete time Markov process Y is Harris recurrent. Then, Y is positive Harris recurrent if and only if there exists a petite set A for which

$$\sup_{x \in A} E_x[\tau_A] < \infty. \tag{4.118}$$

We discuss briefly the reasoning for the different parts of Theorem 4.20. As in the case of Theorem 4.1, the argument for the converse direction of Part (a) is elementary. It consists of repeatedly restarting Y at an increasing sequence of random times, between which Y has at least some probability  $\epsilon > 0$  of hitting a specified set B, for which  $\varphi(B) > 0$ , where  $\varphi$  is the irreducibility measure. Here, in the discrete time setting, less theoretical justification is required, since the strong Markov property and the measurability of hitting times of sets are elementary.

The argument for the other direction of Theorem 4.20(a), namely the construction of a small set satisfying (4.117), is longer, and requires cleverness. It is sometimes referred to as Orey's C-Set Theorem. For a proof, one can consult pages 18-19 in [Nu84] or Theorem 5.2.1, on page 107 of [MeT93d]. The construction involves setting  $\nu = \psi \mathbf{1}\{D\}$ , where  $\psi$  is a maximal irreducibility measure for Y with total mass 1, and D is an appropriately defined "high density" set. We note here that in the setting of Section 4.1, where  $\mathscr{S}$  is generated by a locally compact, separable metric, one can choose the set A so that it is also closed. (By Proposition 5.2.4(iii) of [MeT93d], there is a small set A' with  $\psi(A') > 0$ , for the maximal irreducibility measure  $\psi$ , and therefore a closed subset A of A' for which  $\psi(A) > 0$  as well.)

In Part (b) of Theorem 4.20, the argument that positive Harris recurrence follows from the existence of a petite set A satisfying (4.118) is quite quick, if one

assumes the following formula in (4.119). It is of interest in its own right, and says that the stationary measure  $\pi$  of a Harris recurrent discrete time Markov process Y satisfies

$$\pi(B) = \int_A E_x \left[ \sum_{n=0}^{\tau_A - 1} \mathbf{1}\{Y(n) \in B\} \right] \pi(dx) \quad \text{for } B \in \mathscr{S},$$
(4.119)

for any  $A \in \mathscr{S}$  with  $\psi(A) > 0$ . Often, the indices in the sum are taken from n = 1 to  $\tau_A$  (as in Theorem 10.0.1 in [MeT93d]). Setting B = S, we will employ the special case,

$$\pi(S) = \int_{A} E_x[\tau_A] \pi(dx).$$
 (4.120)

The formula (4.119) may be motivated by the following informal argument. For  $x \in S$  and  $A, B \in \mathscr{S}$ , with  $\psi(A) > 0$ , set

$$\pi_A(B) = \pi(A \cap B)$$
 and  ${}^AP(x,B) = P(x,B \cap A^c).$  (4.121)

Then, (4.119) may be rewritten as  $\pi = \pi_A \sum_{n=0}^{\infty} {}^{A}P^n$ . Assume that  $(I - {}^{A}P)^{-1} = \sum_{n=0}^{\infty} {}^{A}P^n$  exists. Then, this is equivalent to

$$\pi(I - {}^{A}P) = \pi_A,$$

that is,

$$\pi^A P = \pi_{A^c}$$

which follows immediately from the definition in (4.121) and the stationarity of  $\pi$ .

In order to employ (4.120), we first note that  $\pi(A) < \infty$  must hold if the set A is petite with respect to any measure  $\nu$ . Otherwise, it is not difficult to check that, from the definition of petite,  $\pi(B) = \infty$  must hold for any B with  $\nu(B) > 0$ . Since  $\pi$  is assumed to be  $\sigma$ -finite, this is not possible. Applying (4.118) to (4.120), one therefore obtains

$$\pi(S) \le \pi(A) \sup_{x \in A} E_x[\tau_A] < \infty.$$

So, Y is positive Harris recurrent, as desired.

The argument for the other direction of Theorem 4.20(b), which involves the construction of a petite set satisfying (4.118) is, not surprisingly, longer. The result is stated in Theorem 11.0.1 of [MeT93d]. It involves the use of *regular sets*, with Theorem 11.1.4 providing the key decomposition of S. We note here again that, in the topological setting of Section 4.1, one can choose A so that it is also closed. (In Theorem 11.0.1, one can choose a regular set A' with  $\psi(A') > 0$ , and therefore a closed regular subset A of A' for which  $\psi(A) > 0$ ; both A' and A will be petite.)

This completes our discussion of Theorem 4.20. We now summarize how Theorem 4.1 follows from a slight modification of Theorem 4.20 and Proposition 4.18. As in Theorem 4.20, there are two parts to show, each with two

directions. Recall that X denotes a continuous time Markov process as in Section 4.1 and  $\tilde{X}$  denotes the corresponding *R*-chain.

The argument that the Harris recurrence of X follows from the existence of a petite set satisfying (4.19) is elementary, and was summarized in Section 4.1. To show the other direction of Theorem 4.1(a), we first note that, by Proposition 4.18, if X is Harris recurrent, then so is  $\tilde{X}$ . By Theorem 4.20(a), there exists a small set A for  $\tilde{X}$ , with corresponding time  $m_0$ , for which (4.117) is satisfied. As mentioned in the outline of the proof of Theorem 4.20, A can be chosen so that it is also closed. Choosing a in (4.18) to be the  $m_0$ -fold convolution of the exponential distribution, it follows that A is petite for X. Using (4.13), it is easy to see that since (4.117) holds for  $\tilde{X}$ , its analog (4.19) holds for X. So, A has the properties stated in Theorem 4.1(a).

In order to show Part (b) of Theorem 4.1, we again employ Theorem 4.20 and Proposition 4.18. We also need the following comparisons between the bounds on the hitting times of petite sets A, in (4.20) and (4.118), for the processes X and  $\tilde{X}$ . These comparisons are taken from Section 4 of [MeT93a].

**Proposition 4.21.** (a) Suppose that for the continuous time Markov process X, (4.20) is satisfied for some closed petite set A. Then, there is a petite set A' for X, such that (4.118) is satisfied for the R-chain  $\tilde{X}$ . (b) Suppose that for the R-chain  $\tilde{X}$  of X, (4.118) is satisfied for some petite set A of  $\tilde{X}$ . Then, there is a closed petite set A' for the process X which satisfies (4.20).

We note that, in [MeT93a], it is shown that the set A in Part (b) is regular for X, with  $\psi(A) > 0$  for the maximal irreducibility measure  $\psi$ . The existence of a closed petite subset A' satisfying (4.20) then follows. (This application of regularity is analogous to that mentioned at the end of the discussion of the proof of Theorem 4.20.)

Using Proposition 4.21, we now show Theorem 4.20(b). Suppose that (4.20) is satisfied for some closed petite set A. Then, the petite set A' in Part (a) of the proposition must have  $\pi(A') < \infty$ , for the reasons mentioned in the demonstration of Theorem 4.20(b). Here,  $\pi$  is the stationary measure for both X and  $\tilde{X}$ . Since A' satisfies (4.118), one may apply the formula (4.120), as in the demonstration of Theorem 4.20(b), to conclude that  $\pi(S) < \infty$ . Hence, X is positive Harris recurrent, as desired.

For the other direction of Theorem 4.1(b), we assume that X is positive Harris recurrent. By Proposition 4.18,  $\tilde{X}$  is also positive Harris recurrent, and so by Theorem 4.20(b), there is a petite set A of  $\tilde{X}$  that satisfies (4.118). Applying Proposition 4.21(b), it follows that there exists a closed petite set A', for the process X, which satisfies (4.20). This completes our discussion of Theorem 4.1.

#### Existence and uniqueness of stationary measures

One of the fundamental properties of discrete time Harris recurrent Markov processes is the existence of a stationary measure that is unique up to a constant multiple. This is a standard result in discrete time Markov process theory and can be found in a number of places, such as [Nu84] and [MeT93d]. A short

account is given in [Dur96], which we follow here in summarizing the ideas behind the proof. Because of Proposition 4.18, this result immediately extends to continuous time Markov processes.

On account of Part (a) of Theorem 4.20, in order to show the existence of a stationary measure for a discrete time Harris recurrent Markov process, it suffices to show the following result. We will discuss uniqueness of the stationary measure afterwards.

**Theorem 4.22.** Suppose that Y is a discrete time Markov process with a small set A satisfying (4.117). Then, there exists a stationary measure  $\pi$ .

The reasoning for Theorem 4.22 can be broken into two main steps. The first step consists of showing that the conclusion of the theorem holds when S contains a *recurrent atom*  $\alpha$  for the process Y. By this, we mean that  $P_y(\tau_\alpha < \infty) = 1$  for all y, where  $\tau_\alpha \stackrel{\text{def}}{=} \tau_{\{\alpha\}}$ . (The use here of the term recurrent atom is not standard.) The second step shows that there is an appropriate "lumping together" of points (sometimes referred to as "splitting" in the literature) that allows one to construct a process  $\bar{Y}$  with such an atom and which has the same recurrence behavior as Y.

Before beginning, we note that, because of Proposition 4.19, one can assume without loss of generality that  $m_0 = 1$  for the small set A. Also, by scaling the corresponding small measure  $\nu$  by  $\epsilon = 1/\nu(S)$ , one can replace the inequality (4.18) by

$$\nu(B) \le \epsilon P(y, B) \quad \text{for } y \in A, \ B \in \mathscr{S}, \tag{4.122}$$

where  $\nu$  is now a probability measure.

For the first step, we assume that S has a recurrent atom  $\alpha$ . Then,  $P_{\alpha}(\tau_{\alpha} < \infty) = 1$ , and it is not difficult to show that the measure  $\pi$  defined by

$$\pi(B) = E_{\alpha} \left[ \sum_{n=0}^{\tau_{\alpha}-1} \mathbf{1}\{Y(n) \in B\} \right]$$
(4.123)

is  $\sigma$ -finite. (See Exercise 6.8 on page 331 of [Dur96].)

We claim that  $\pi$  is stationary for the process Y. The proof is essentially the same as that for Markov chains, which is given for Theorem 4.3, on page 303 of [Dur96]. It can be motivated by using the "cycle trick", noting that the number of visits to any  $y \in S$  over the time set  $\{0, \ldots, \tau_{\alpha} - 1\}$  is always the same as over  $\{1, \ldots, \tau_{\alpha}\}$ , and taking expectations over both sides. It follows from this that  $\pi P = \pi$ . Formula (4.123) is a special case of (4.119), with  $A = \{\alpha\}$ . (Note that the representation of  $\pi$  in (4.119) is not explicit, whereas the right side of (4.123) does not involve  $\pi$ .)

For the second step of the proof, one wishes to compare the given Markov process Y with one having a recurrent atom. The small set A will be used to construct the atom, and (4.117) will be used to show the atom is recurrent. We begin by appending a point  $\alpha$  to the state space S, thus creating a new space  $\overline{S}$  and a corresponding Borel  $\sigma$ -algebra  $\overline{\mathscr{I}}$ ,

$$\bar{S} = S \cup \{\alpha\} \quad \text{and} \quad \bar{\mathscr{I}} = \{B, B \cup \{\alpha\} : B \in \mathscr{S}\}.$$

One can define a probability transition kernel  $\bar{P}$  on  $(\bar{S}, \bar{\mathscr{I}})$  by

$$\bar{P}(y, \{\alpha\}) = \begin{cases} \epsilon & \text{for } y \in A, \\ 0 & \text{for } y \in S - A, \end{cases}$$

$$\bar{P}(y, B) = \begin{cases} P(y, B) - \epsilon \nu(B) & \text{for } y \in A, B \in \mathscr{S}, \\ P(y, B) & \text{for } y \in S - A, B \in \mathscr{S}, \end{cases}$$

$$\bar{P}(\alpha, B) = \int \bar{P}(y, B) \nu(dy) & \text{for } B \in \bar{\mathscr{S}}, \end{cases}$$
(4.124)

where  $\nu$  and  $\epsilon > 0$  are chosen as in (4.122). We denote by  $\bar{Y}$  the corresponding Markov process.

We also find it convenient to introduce the probability transition kernel V on  $(\bar{S}, \bar{\mathscr{I}})$ , with

$$V(y, \{y\}) = 1 \text{ for } y \in S, \quad V(\alpha, B) = \nu(B) \text{ for } B \in \mathscr{S}.$$

$$(4.125)$$

Note that  $V(y, \alpha) = 0$  for all  $y \in \overline{S}$ . One can check that

$$V\overline{P} = \overline{P}$$
 and  $(\overline{P}V)|_S = P.$  (4.126)

Here,  $|_S$  denotes the restriction to S in both the domain and range. As usual, we are employing the convention that the transition kernel on the left is the first to be applied to a given initial measure. We will demonstrate a version of (4.126) below (4.128).

The construction of  $\bar{P}$  can be motivated as follows. One wishes to "lump together" points in S in some way so as to form an atom  $\alpha$ , while not changing, in essence, the transition rule P. If one is to lump together points according to some weight  $\nu$ , transitions to and from  $\alpha$ , by the new transition rule  $\bar{P}$ , must both be done in a manner consistent with this weight. Since A is assumed to be small, with  $m_0 = 1$  and with (4.122) holding, one can choose  $\bar{P}(y, \{\alpha\})$  as on the top two lines of (4.124); the bottom line of (4.124) respects the weight  $\nu$ . The third and fourth lines of (4.124) define  $\bar{P}$  according to the old transition rule P on the mass that has not been directed to  $\alpha$ .

To develop some feel for (4.124), one can consider the case where S is discrete. (In this case, there is of course no need to lump together points to create an atom.) Denoting the transition densities by p(y, y') and setting  $q(y) = \nu(\{y\})$ , (4.124) becomes

$$\bar{p}(y,\alpha) = \begin{cases} \epsilon & \text{for } y \in A, \\ 0 & \text{for } y \in S - A, \end{cases}$$
$$\bar{p}(y,z) = \begin{cases} p(y,z) - \epsilon q(z) & \text{for } y \in A, \ z \in S \\ p(y,z) & \text{for } y \in S - A, \ z \in S, \end{cases}$$
(4.127)
$$\bar{p}(\alpha,z) = \sum_{i} \bar{p}(y_{i},z)q(y_{i}) & \text{for } z \in \bar{S}. \end{cases}$$

Here  $\{y_i\}$  denotes an enumeration of the points in S. Similarly, one can define v by

$$v(y,y) = 1, v(\alpha, y) = q(y) \text{ for } y \in S.$$

The equations in (4.126) become

$$(v\bar{p})(y,z) = \bar{p}(y,z)$$
 and  $(\bar{p}v)(y,z) = p(y,z),$  (4.128)

with  $y, z \in \overline{S}$  in the first equation and  $y, z \in S$  in the second. The first equation follows from the invariance of v on  $y \in S$  and

$$(v\bar{p})(\alpha, z) = \sum_{i} \bar{p}(y_i, z)q(y_i) = \bar{p}(\alpha, z).$$

For the second equation, one has

$$\begin{split} (\bar{p}v)(y,z) &= \sum_{i} \bar{p}(y,y_{i})v(y_{i},z) + \bar{p}(y,\alpha)v(\alpha,z), \\ &= \bar{p}(y,z) + \bar{p}(y,\alpha)q(z) \end{split}$$

for  $y, z \in S$ . For both  $y \in A$  and  $y \in S - A$ , it follows from the definition of  $\bar{p}$  that the last line equals p(y, z), as desired. The equations in (4.126) can be derived similarly.

We still need to show that the atom  $\alpha$  is recurrent. For this, it suffices to show

$$\sum_{n} \bar{P}^{n}(x, \{\alpha\}) = \infty \quad \text{for all } x \in \bar{S}.$$
(4.129)

We employ the string of inequalities

$$\sum_{n} \bar{P}^{n}(x, \{\alpha\}) = \sum_{n} V(\bar{P}V)^{n-1} \bar{P}(x, \{\alpha\}) \ge \epsilon \sum_{n} V(\bar{P}V)^{n-1}(x, A)$$
$$\ge \epsilon \min_{x \in S} \sum_{n} (\bar{P}V)^{n-1}(x, A) \ge \epsilon \min_{x \in S} \sum_{n} P^{n-1}(x, A) = \infty.$$

The first equality follows from the first half of (4.126), the first inequality follows from the first line of (4.124), and the second inequality follows from (4.125). The last inequality is a consequence of the second half of (4.126), since restricting the domain decreases the corresponding integrals. (Because of (4.125), one actually has equality.) The equality at the end follows from Borel-Cantelli and (4.117), since A will be hit infinitely often with probability 1. So, (4.129) holds.

We now tie together the two previous steps involving the creation and manipulation of the atom  $\alpha$ . If one assumes that the Markov process Y given in Theorem 4.22 has a petite set A satisfying (4.117), it follows that the process  $\bar{Y}$ defined by (4.124) has a recurrent atom at  $\alpha$ . The measure  $\bar{\pi}$  given in (4.123) for  $\bar{Y}$  is then stationary for  $\bar{Y}$ . We claim that the measure  $\pi \stackrel{\text{def}}{=} (\bar{\pi}V)|_S$ , given by the restriction of  $\bar{\pi}V$  to S, is stationary for Y. Theorem 4.22 follows immediately from this. To show the claim, it suffices to verify the string of equalities

$$\pi P = (\bar{\pi}V)|_S P = (\bar{\pi}\bar{P}V)|_S = (\bar{\pi}V)|_S = \pi.$$
(4.130)

Here, the third equality follows from the stationarity of  $\bar{\pi}$  and the first and fourth follow from the definition of  $\pi$ . The second equality follows from

$$(\bar{\pi}V)|_{S} P = (\bar{\pi}V)|_{S} (\bar{P}V)|_{S} = (\bar{\pi}V\bar{P}V)|_{S} = (\bar{\pi}\bar{P}V)|_{S},$$
(4.131)

where one applies first the second half of (4.126) and last the first half of (4.126). The middle equality in (4.131) holds since  $\bar{\pi}V(\alpha) = 0$ , and so no mass is lost by the restriction  $(\bar{\pi}V)|_S$ . This demonstrates (4.131), and hence (4.130) and the stationarity of  $\pi$ , as desired.

One can also show that the stationary measure of a discrete time Harris recurrent Markov process is unique, up to a constant multiple. As before, on account of Part (a) of Theorem 4.20, it suffices to show the following version.

**Theorem 4.23.** Suppose that Y is a discrete time Markov process with a small set satisfying (4.117). Then, up to a constant multiple, the stationary measure of Y is unique.

We note that some sort of condition, corresponding to (4.117), is of course necessary. Without it, as in the countable state space setting, states may not communicate and there may be many stationary measures.

We summarize here the argument for Theorem 4.23. As in the demonstration of the existence of a stationary measure, we can assume without loss of generality that  $m_0 = 1$  for the small set A, because of Proposition 4.19. The following argument, for  $m_0 = 1$ , can be found in Theorem 6.7, on page 331 of [Dur96].

As before, one employs the process  $\overline{Y}$  corresponding to (4.124), which has a recurrent atom. Let  $\rho$  be a  $\sigma$ -finite measure on  $(S, \mathscr{S})$ , and denote by  $\rho^{\overline{S}}$  its extension to  $(\overline{S}, \overline{\mathscr{S}})$  given by

$$\rho^S(B) = \rho(B \cap S) \quad \text{for } B \subseteq \overline{S}.$$

If  $\rho$  is stationary for Y, one can show that  $\bar{\rho} = \rho^{\bar{S}} \bar{P}$  is stationary for the process  $\bar{Y}$ . This is done in Lemma 6.6, on page 331 of [Dur96].

In the countable state space setting when all states communicate, it is not difficult to show that  $\bar{\rho} = \bar{\rho}(\alpha)\bar{\pi}$ , where  $\bar{\pi}$  is the stationary measure for  $\bar{Y}$  defined in (4.123). This is done in Theorem 4.4, on page 305 of [Dur96], and the argument in the general state space setting is essentially the same. Set  $\pi = (\bar{\pi}V)|_S$ . One then has

$$\rho = \rho P = \rho((\bar{P}V)|_S) = (\rho^S \bar{P}V)|_S \qquad (4.132)$$
$$= (\bar{\rho}V)|_S = (\bar{\rho}(\alpha)\bar{\pi}V)|_S = \bar{\rho}(\alpha)\pi.$$

Here, the first equality follows from the stationarity of  $\rho$ , the fourth and sixth equalities from the definitions of  $\bar{\rho}$  and  $\pi$ , and the second equality from the second half of (4.126). The third equality holds since  $\bar{\rho}(\{\alpha\}) = 0$ . By (4.132),  $\rho = \bar{\rho}(\alpha)\pi$ , which shows that  $\rho$  is a constant multiple of  $\pi$ , as desired.

### Chapter 5

# Applications and some further theory

Theorem 4.16, from Section 4.4, gives conditions under which an HL queueing network will be stable. Theorem 4.17, from the same section, gives similar conditions under which a network will be e-stable. In the first three sections of Chapter 5, we will demonstrate stability/e-stability for three families of queueing networks by using these criteria. In Section 5.1, we do this for a family of a networks that includes single class networks, and in Section 5.2, we do this for two families of SBP reentrant lines, FBFS and LBFS. In Section 5.3, we apply the same criteria to FIFO networks of Kelly type.

In Sections 5.4 and 5.5, we address other topics connected with stability. Global stability is introduced in Section 5.4. The condition says that the network is stable irrespective of the HL discipline that is employed. It will follow directly from Theorem 4.16 that a sufficient condition for global stability is the stability of the associated basic fluid model. For two stations, there is a developed theory for these fluid models, which we summarize. Important underlying concepts include virtual stations and the push start phenomenon. We also discuss briefly rate stability and global rate stability for queueing networks, and the corresponding concept of weak stability for fluid models.

In Section 5.5, we investigate the converse direction to Theorem 4.16. Namely, does queueing network stability, under reasonable side conditions, imply fluid model stability? In particular, are queueing network and fluid model stability in some sense equivalent? Most of this section is spent on two examples that show this is not always the case. Virtual stations and push starts are again useful tools in this context.

Our approach in Sections 5.1–5.3 will be based on the following considerations involving Theorems 4.16 and 4.17. The conditions in the theorems are of two types, where (a) one needs the bounded sets  $A = \{x : |x| \le \kappa\}, \kappa > 0$ , to be either petite or uniformly small and (b) one needs the fluid limits or fluid model, which is associated with the queueing network, to be stable. As noted in Section 4.4, the latter of the two stability conditions is the more tractible one, and is the one that is typically applied in practice. In the next three sections, we will demonstrate the stability of fluid models that are associated with the queueing networks there; the desired results then follow by applying Theorems 4.16 and 4.17. The first two cases are fairly quick, and the third takes a bit longer to show.

In order to avoid repetition later on, we will, in each case, explicitly assume that

$$A = \{x : |x| \le \kappa\} \text{ is petite for each } \kappa > 0.$$
(5.1)

This condition suffices, in Theorem 4.16, for stability of the queueing network, once the fluid model has been shown to be stable. In order to conclude that the queueing network is *e*-stable, by using Theorem 4.17, one needs to replace (5.1) here with the assumption that, for each  $\kappa > 0$ ,

$$A = \{x : |x| \le \kappa\} \text{ is uniformly small on } [s_1, s_2], \text{ for some } 0 < s_1 < s_2.$$
(5.2)

Whether the above sets A are either petite or uniformly small may depend on the discipline of the queueing network. We recall that the conditions (4.30) and (4.31) on the interarrival times are sufficient for both (5.1) and (5.2) under all HL disciplines, and that (4.30) alone is sufficient for (5.1) when  $|\mathcal{A}| = 1$ . In this chapter, as was the case in Chapter 4, it will be implicitly assumed that the interarrival and service times have finite means.

We recall that in Section 4.3, we defined regular points of a fluid model solution  $\mathfrak{X}(\cdot)$  to be those times t at which the derivatives of all components of  $\mathfrak{X}(\cdot)$  exist. Since the components of  $\mathfrak{X}(\cdot)$  are all Lipschitz continuous, the derivatives at regular points determine  $\mathfrak{X}(\cdot)$ . In our computations in Sections 5.1–5.3, we will often restrict our attention to regular points, without always explicitly saying so.

The material in this chapter relies on a number of papers. The material in the first three sections is mostly from [Br98a], [DaWe96], and [Br96a]. The material in Section 5.4 is mostly from [DaV00], and that in Section 5.5 is mostly from [Da96], [Br99], [DaHV04], and [GaH05].

#### 5.1. Single class networks

In Section 2.5, we gave the explicit formula (2.44) for the stationary distributions of subcritical single class HL queueing networks with exponentially distributed interarrival and service times, which we referred to as Jackson networks. These networks are stable (in fact, *e*-stable). Does stability still hold when the interarrival and service times are not exponential? This was considered a difficult question, and was answered in the affirmative in a number of papers under various assumptions on the interarrival and service time distributions ([Bo86], [Si90], [BaF94], [ChTK94], [MeD94]). Fluid models, together with Theorem 4.16, provide a quick proof of stability for these networks.

To show stability of the associated fluid models, we employ the proof given in [Br98a]; another argument was given in [Da95]. The proof relies on the inequality

$$D_k'(t) \ge \lambda_k + \epsilon \tag{5.3}$$

for some  $\epsilon > 0$  and all k, at all regular points of the fluid model solutions  $\mathfrak{X}(\cdot)$  where  $Z_k(t) > 0$ . It follows from (4.53) and (4.55) that (5.3) is satisfied for all subcritical single class fluid models.

Since the restriction to one class per station is otherwise not used in the proof of stability of the fluid model, we will instead assume that the HL queueing networks we consider here satisfy the related condition

$$R_k(t) \ge m_k(\lambda_k + \epsilon), \tag{5.4}$$

for some  $\epsilon > 0$  and all k, whenever the  $k^{\text{th}}$  class is not empty. Recall that, as in the first part of Section 4.1,  $R_k(t)$  is the proportion of service at station jallocated to the first job in class k. For subcritical single class networks, (5.4) holds for small enough  $\epsilon$ , since  $m_k \lambda_k < 1$ . We will show shortly that (5.3), for fluid limits, follows from (5.4). Needless to say, (5.4) is a severe restriction, and is not satisfied by the usual multiclass disciplines. (In [Br98a], the condition is used to show that any HL queueing network can be "stabilized" by inserting sufficiently quick single class stations between visits to classes.)

The following theorem is the main result in this section.

**Theorem 5.1.** Any HL queueing network satisfying (5.1) and (5.4) is stable. Consequently, any subcritical single class HL queueing network satisfying (5.1) is stable.

We will employ Theorem 4.16 to show Theorem 5.1. As the associated fluid model in Theorem 4.16, we choose the fluid model consisting of the basic fluid model equations (4.50)-(4.55), together with (5.3). We already know, from Proposition 4.12, that (on the set G) every fluid limit of the queueing network satisfies the basic fluid model equations. So, in order to show the fluid model is associated with the queueing network in Theorem 5.1, we only need to verify that (5.3) is satisfied by all fluid limits.

The reasoning for (5.3) is analogous to that in Proposition 4.12 for the other fluid model equations. By (5.4), for given x,

$$T_k^x(t_2) - T_k^x(t_1) \ge m_k(\lambda_k + \epsilon)(t_2 - t_1)$$

for all k and  $t_1 \leq t_2$ , if  $Z_k(t) > 0$  on  $[t_1, t_2]$ . For the same reason,

$$\frac{1}{a_n} T_k^{x_n}(a_n t_2) - \frac{1}{a_n} T_k^{x_n}(a_n t_1) \ge m_k(\lambda_k + \epsilon)(t_2 - t_1)$$
(5.5)

for any sequence of pairs  $(a_n, x_n)$  satisfying (4.71), if  $Z_k^{x_n}(t) > 0$  on  $[a_n t_1, a_n t_2]$ . On the other hand, for any fluid limit  $\bar{\mathfrak{X}}(\cdot)$ , its component  $\bar{Z}(\cdot)$  is continuous. So, if it is assumed that  $\bar{Z}_k(t) > 0$  on an interval  $[t_1, t_2]$ , then it is bounded away from 0 on this interval. Convergence to  $\overline{Z}(\cdot)$  is u.o.c., and so for sufficiently large  $n, Z_k^{x_n}(t) > 0$  on  $[a_n t_1, a_n t_2]$ . Together with (5.5), this implies

$$\bar{T}_k(t_2) - \bar{T}_k(t_1) \ge m_k(\lambda_k + \epsilon)(t_2 - t_1).$$

By (4.55), the inequality is equivalent to

$$D_k(t_2) - D_k(t_1) \ge (\lambda_k + \epsilon)(t_2 - t_1),$$

which is, of course, equivalent to (5.3).

Since the system of fluid model equations (4.50)-(4.55) together with (5.3) is associated with the queueing network in Theorem 5.1, it suffices to demonstrate the following result in order to show Theorem 5.1.

#### **Theorem 5.2.** Any fluid model satisfying (5.3) is stable.

We break the proof of Theorem 5.2 into two lemmas. The first lemma gives a lower bound on the rate of departures from any class, empty or not.

**Lemma 5.3.** Assume that a fluid model satisfies (5.3). Then,

$$D(t_2) - D(t_1) \ge \lambda(t_2 - t_1)$$
(5.6)

for  $0 \le t_1 \le t_2$ .

In order to show Lemma 5.3, we require some notation. At each time  $t \geq 0$ , let  $\mathcal{K}_0(t)$  denote those classes where  $Z_k(t) = 0$ , let  $\mathcal{K}_+(t)$  denote its complement, and let  $|\mathcal{K}_0(t)|$  and  $|\mathcal{K}_+(t)|$  be the number of elements in each set. The subscripts 0 and + will denote the restrictions to  $\mathcal{K}_0(t)$  and  $\mathcal{K}_+(t)$ , respectively, for vectors such as  $\alpha$ ,  $\lambda$ , D'(t), and Z'(t). Similarly,  $P_0$  and  $P_+$  will denote the  $|\mathcal{K}_0(t)| \times |\mathcal{K}_0(t)|$  and  $|\mathcal{K}_+(t)| \times |\mathcal{K}_0(t)|$  matrices obtained by these restrictions.

Proof of Lemma 5.3. By (5.3),

$$D'(t)_+ \ge \lambda_+. \tag{5.7}$$

In order to show (5.6), it therefore suffices to show

$$D'(t)_0 \ge \lambda_0. \tag{5.8}$$

First note that  $I_0 - P_0^T$  is invertible, with

$$Q_0 \stackrel{\text{def}}{=} (I_0 - P_0^T)^{-1} = I_0 + P_0^T + (P_0^T)^2 + \cdots$$

having nonnegative entries. This is the analog of (1.2). Also, the analog of (1.6) holds,

$$\lambda_0 = \alpha_0 + P_0^T \lambda_0 + P_+^T \lambda_+,$$

which implies

$$\lambda_0 = Q_0(\alpha_0 + P_+^T \lambda_+). \tag{5.9}$$

On the other hand, combining (4.50) and (4.51), and taking derivatives, one gets

$$Z'(t) = \alpha + (P^T - I)D'(t).$$

Restriction of the coordinates to  $\mathcal{K}_0(t)$  implies

$$Z'(t)_0 = \alpha_0 + (P_0^T - I_0)D'(t)_0 + P_+^T D'(t)_+.$$

Multiplying by  $Q_0$  and applying (5.9), one therefore gets

$$Q_0 Z'(t)_0 = \lambda_0 - D'(t)_0 + Q_0 P_+^T (D'(t)_+ - \lambda_+).$$

Since  $Z(t)_0 = 0$ , if t is a regular point, one must have  $Z'(t)_0 = 0$ . Hence,  $Q_0 Z'(t)_0 = 0$  there. Applying this to the left side of the above equality, and (5.3) to the last term on the right side of the equality, implies (5.8), as desired.

To show Theorem 5.2, we will employ the Lyapunov function

$$f(t) = e^T Q Z(t). (5.10)$$

The function f counts the "average" number of present and future visits within the network, under the transition matrix P, for jobs already in the network at time t. By combining (4.50) and (4.51), one has

$$Z(t) = Z(0) + \alpha t - (I - P^T)D(t),$$

which, after multiplying both sides by Q, gives

$$QZ(t) = QZ(0) + \lambda t - D(t).$$

Substitution into (5.10) shows that

$$f(t) = f(0) + te^T \lambda - e^T D(t).$$
(5.11)

By applying Lemma 5.3 to f(t), we will obtain the following result, Lemma 5.4. Since f(t) and |Z(t)| are at most bounded multiples of one another, Theorem 5.2 follows immediately from Lemma 5.4.

**Lemma 5.4.** Assume that (5.3) is satisfied. Then,

$$f(t) \le [f(0) - \epsilon t]^+ \quad for \ all \ t \ge 0.$$
 (5.12)

*Proof.* Lemma 5.3 implies that

$$e^{T}(D(t_{2}) - D(t_{1})) \ge (t_{2} - t_{1})e^{T}\lambda$$

for  $0 \le t_1 \le t_2$ . So, by (5.11), f(t) is nonincreasing. Consequently, if  $Z(t_0) = 0$  for a given  $t_0$ , then f(t) = 0 for all  $t \ge t_0$ .

Consider now Z(t) for

$$t < t_0 \stackrel{\text{def}}{=} \inf\{t : Z(t) = 0\}.$$

By (5.3),  $D'_k(t) \geq \lambda_k + \epsilon$  for some k (which depends on t). Together with Lemma 5.3, this implies that

$$e^T D(t) \ge t(e^T \lambda + \epsilon) \quad \text{for } t < t_0.$$

Plugging this into (5.11) implies (5.12) for  $t < t_0$  as well. (One can instead, if one wishes, give a somewhat different proof by using (4.60).)

#### 5.2. FBFS and LBFS reentrant lines

The static buffer priority (SBP), first-buffer-first-served (FBFS), and last-bufferfirst-served (LBFS) disciplines were introduced in Chapter 1. We review their definitions here. For SBP disciplines, classes at each station are assigned a strict ranking, with jobs of higher ranked (or priority) classes always being served before jobs of lower ranked classes, irrespective of when they arrived at the station. In this section, we assume the disciplines are preemptive, so that arriving higher ranked jobs interrupt lower ranked jobs currently in service; when the service of such higher ranked jobs has been completed, service of the lower ranked jobs continues where it left off. The first job of each class always receives all of the service allocated to the class, so the discipline is HL.

We focus here on two SBP disciplines for networks that are reentrant lines, FBFS and LBFS. For the FBFS discipline, jobs at earlier classes along the route have priority over later classes. For the LBFS discipline, the priority is reversed, with jobs at later classes along the route having priority over earlier classes. In both cases, we assign classes the values  $k = 1, \ldots, K$  according to the order of their appearance along the route. With this ordering, classes with smaller k have priority under FBFS and classes with larger k have priority under LBFS.

In order to specify the evolution of an SBP network, one requires another equation in addition to the basic queueing network equations (4.42)-(4.47). This is given by

$$t - T_k^+(t)$$
 can only increase when  $Z_k^+(t) = 0$  for  $k = 1, \dots, K$ , (5.13)

for all  $t \ge 0$ . Here,  $Z_k^+(t)$  denotes the sum of the queue lengths at the station j = s(k) of classes having priority at least as great as k, and  $T_k^+(t)$  denotes the corresponding sum of cumulative service times. It is easy to verify (5.13). (Recall that the discipline is assumed to be preemptive.) Note that in this setting, (4.46) is redundant, since it is equivalent to (5.13) when k is the lowest ranked class at its station.

Arguing as in the last part of the proof of Proposition 4.12, it is not difficult to show that (5.13) is satisfied for all fluid limits of SBP queueing networks. We already know from Proposition 4.12 that the basic fluid model equations (4.50)– (4.55) are satisfied by all fluid limits of the queueing network. So, the fluid model given by the equations (4.50)–(4.55) and (5.13) is associated with the SBP queueing network with corresponding parameters. We refer to these equations as the *SBP fluid model equations*, and to the corresponding fluid model as the *SBP* fluid model. When the priority scheme among the different classes at each station corresponds to the FBFS or LBFS disciplines, we refer to these equations as the *FBFS fluid model equations* or the *LBFS fluid model equations*, respectively. The respective fluid models are defined analogously. We note that (5.13) is equivalent to

$$(T_k^+)'(t) = 1$$
 when  $Z_k^+(t) > 0$  for  $k = 1, \dots, K$ , (5.14)

at all regular points t.

We saw in Chapter 3 that there exist subcritical SBP reentrant lines that are not stable. This is not the case for FBFS and LBFS reentrant lines. Our goal in this section is to show the following two results. Both are done in [DaWe96], whose reasoning we follow. [LuK91] showed analogous results for discrete deterministic systems.

**Theorem 5.5.** Any subcritical FBFS reentrant line satisfying (5.1) is stable.

**Theorem 5.6.** Any subcritical LBFS reentrant line satisfying (5.1) is stable.

As in Section 5.1, we will use Theorem 4.16 to demonstrate stability for the disciplines of interest. Since we have already shown that the FBFS and LBFS fluid models are associated with the reentrant lines in Theorems 5.5 and 5.6, it suffices to demonstrate the following two results.

Theorem 5.7. Any subcritical FBFS fluid model is stable.

Theorem 5.8. Any subcritical LBFS fluid model is stable.

The remainder of the section will be devoted to demonstrating Theorems 5.7 and 5.8. We observe that, in both cases, it suffices to show stability, as in (4.61), but for fluid model solutions that also satisfy |Z(0)| = 1. This additional assumption will simplify the bookkeeping somewhat. As is typically the case for fluid models,  $\tilde{\mathfrak{X}}(t) \stackrel{\text{def}}{=} \mathfrak{X}(ct)/c$ , c > 0, also satisfies the same fluid model equations as  $\mathfrak{X}(t)$ . It follows from this that the conditions (4.61), with and without |Z(0)| = 1, are equivalent.

For the proofs of both Theorem 5.7 and 5.8, we find it convenient to set  $d_k(t) = D'_k(t)$  for the departure rate from a class k in the fluid model. As mentioned earlier, we only need to consider the behavior of solutions  $\mathfrak{X}(\cdot)$  at regular points t.

Proof of Theorem 5.7. We will use induction to prove that, for each k = 1, ..., K, there exists a  $t_k \ge 0$  so that, for any fluid model solution with |Z(0)| = 1,

$$Z_{\ell}(t) = 0$$
 on  $[t_k, \infty)$ , for  $\ell = 1, \dots, k$ .

For the induction step, we assume that  $Z_{\ell}(t) = 0$  on  $[t_{k-1}, \infty)$  for  $\ell = 1, \ldots, k-1$ . It follows that for  $t \ge t_{k-1}$ ,

$$d_{k-1}(t) = d_{k-2}(t) = \dots = d_1(t) = \alpha_1.$$
(5.15)

Set  $\mathcal{H}_k = \{\ell \le k : s(\ell) = s(k)\}$ . If  $Z_k(t) > 0$ , then by (4.55) and (5.14),

$$\sum_{\ell \in \mathcal{H}_k} m_\ell d_\ell(t) = (T_k^+)'(t) = 1$$

So, for  $Z_k(t) > 0$  and  $t \ge t_{k-1}$ ,

$$d_k(t) = \mu_k \left( 1 - \alpha_1 \sum_{\ell \in \mathcal{H}_k \setminus \{k\}} m_\ell \right).$$
(5.16)

It follows from (5.15) and (5.16) that

$$Z'_k(t) = d_{k-1}(t) - d_k(t) = \alpha_1 - \mu_k \left( 1 - \alpha_1 \sum_{\ell \in \mathcal{H}_k \setminus \{k\}} m_\ell \right).$$

Since  $\alpha_1 \sum_{\ell \in \mathcal{H}_k} m_\ell < 1$  by assumption, it is not difficult to see that the right side of this equation is strictly negative. Setting

$$t'_{k} = t_{k-1} + \frac{Z_{k}(t_{k-1})}{\mu_{k}(1 - \alpha_{1} \sum_{\ell \in \mathcal{H}_{k} \setminus \{k\}} m_{k}) - \alpha_{1}},$$

it follows that  $Z_k(t) = 0$  for  $t \ge t'_k$ .

Since |Z(0)| = 1 is assumed, one has  $Z_{k-1}(t) \leq |Z(t)| \leq 1 + \alpha_1 t$  for all t. So, we can choose  $t_k \geq t'_k$  independently of the particular fluid model solution  $\mathfrak{X}(\cdot)$ , so that  $Z_k(t) = 0$  for  $t \geq t_k$ . Since, by induction, this holds for all k, it follows that for any fluid model solution with |Z(0)| = 1, one has Z(t) = 0 for  $t \geq N$  and some fixed N. So, the fluid model is stable.

We now demonstrate Theorem 5.8.

Proof of Theorem 5.8. We will show that for some N, one must have  $Z_k(t) = 0$  for  $t \ge N$  and all k, for any fluid model solution with |Z(0)| = 1. To show this, assume that at a given k and t,  $Z_k(t) > 0$ , with  $Z_\ell(t) = 0$  for  $\ell = k + 1, \ldots, K$ . It follows that if t is a regular point,

$$d_k(t) = d_{k+1}(t) = \dots = d_K(t).$$
 (5.17)

Set  $\mathcal{H}_k = \{\ell \ge k : s(\ell) = s(k)\}$ . Since  $Z_k(t) > 0$ , it follows from (5.17), (4.55), and (5.14), that

$$d_K(t) \sum_{\ell \in \mathcal{H}_k} m_\ell = \sum_{\ell \in \mathcal{H}_k} m_\ell d_\ell(t) = (T_k^+)'(t) = 1.$$

So,  $d_K(t) = 1 / \sum_{\ell \in \mathcal{H}_k} m_{\ell}$ . Because the system is subcritical,

$$\alpha_1 \sum_{\ell \in \mathcal{H}_k} m_\ell \le \alpha_1 \max_j \sum_{\ell \in \mathcal{C}(j)} m_\ell = \max_j \rho_j \stackrel{\text{def}}{=} \rho_{\max} < 1.$$

In particular,  $d_K(t) \ge \alpha_1 / \rho_{\text{max}} > \alpha_1$ .

Set f(t) = |Z(t)|. At regular points of f(t),

$$f'(t) = \alpha_1 - d_K(t).$$

From the previous paragraph, we know this is at most  $\alpha_1(1 - 1/\rho_{\text{max}}) < 0$  when  $Z(t) \neq 0$ . For |Z(0)| = 1, it follows that Z(t) = 0 for  $t \geq N$ , where

$$N = 1/\alpha_1 (1/\rho_{\max} - 1).$$

So, the fluid model is stable.

#### 5.3. FIFO networks of Kelly type

We saw, in Section 2.5, that the explicit formula (2.44) for the stationary distribution of subcritical single class HL networks generalizes to the related formula for subcritical FIFO networks of Kelly type in (2.4) and (2.9). In both cases, the interarrival and service times of the network were assumed to be exponentially distributed. We showed, in Section 5.1, that subcritical single class queueing networks are stable under more general distributions by using the machinery of fluid limits. Here, we do the same for FIFO networks of Kelly type. At the end of the section, we briefly discuss similar behavior of HLPPS queueing networks, which are an HL variant of PS networks.

In order to specify the evolution of such queueing networks, one requires an equation corresponding to the FIFO discipline, in addition to the basic queueing network equations (4.42)-(4.47). The equation we employ is

$$D_k(t + W_j(t)) = Z_k(0) + A_k(t), \quad k = 1, \dots, K,$$
(5.18)

for all  $t \ge 0$ . To verify (5.18), note that  $t + W_j(t)$  is the time at which the service of the last of the jobs currently at j will be completed, since jobs arriving after time t will have lower priority under the FIFO discipline.

Together, (4.42)–(4.47) and (5.18) form the FIFO queueing network equations; the corresponding 6-tuple  $\mathfrak{X}(\cdot)$  is the FIFO queueing network process. One can check that the triple  $(E(\cdot), \Gamma(\cdot), \Phi(\cdot))$ , together with

$$\{D_k(t) \text{ for } t \le W_j(0), \ k = 1, \dots, K\},$$
 (5.19)

determines  $\mathfrak{X}(\cdot)$ , for all  $t \geq 0$ , if  $\mathfrak{X}(\cdot)$  evolves according to the FIFO queueing network equations. The information in (5.19) serves the role of the *initial data* for solutions of these equations. This additional information is needed, since Z(0) is by itself not enough to determine the order in which the original jobs are served.

Arguing as in the proof of Proposition 4.12, it is not difficult to show that (5.18) is satisfied for all fluid limits of FIFO queueing networks. (Recall that the component  $\overline{D}(\cdot)$  of a fluid limit is continuous and nondecreasing.) We already know from Proposition 4.12 that the basic fluid model equations (4.50)–(4.55) are satisfied by all fluid limits of the queueing network. So, the fluid model given by the equations (4.50)–(4.55) and (5.18) is associated with the FIFO queueing network.

We refer to these equations as the *FIFO fluid model equations*, and to the corresponding fluid model as the *FIFO fluid model*. Solutions of the fluid model equations are denoted by the 6-tuple  $\mathfrak{X}(\cdot)$ . The *initial data* is again given by (5.19). Using the basic fluid model equations, one can show that

$$\sum_{k \in \mathcal{C}(j)} m_k D_k(t) = t \quad \text{for } t \le W_j(0)$$
(5.20)

must hold; (5.20) serves as a consistency condition on the initial data. In keeping with the definition for queueing networks, we say that a FIFO fluid model is of

Kelly type, if  $m_k = m_\ell$  whenever  $k, \ell \in \mathcal{C}(j)$  for some j. We will then write  $m_j^s$  for  $m_k$ .

We saw, in Chapter 3, that there exist subcritical FIFO queueing networks that are not stable. On the other hand, subcritical FIFO queueing networks of Kelly type that have exponentially distributed interarrival and service times are stable, as was shown in Chapter 2. The following result from [Br96a] shows that stability continues to hold for more general interarrival and service times. We follow the presentation that is given there.

**Theorem 5.9.** Any subcritical FIFO queueing network of Kelly type satisfying (5.1) is stable.

As in the previous two sections, we will use Theorem 4.16 to demonstrate this stability. Since we already know that FIFO fluid models are associated with the FIFO queueing networks, it suffices to demonstrate the following result.

Theorem 5.10. Any subcritical FIFO fluid model of Kelly type is stable.

#### Demonstration of Theorem 5.10

In order to demonstrate Theorem 5.10, we introduce a form of entropy. Let

$$h(x) = x \log x, \quad x \ge 0 \tag{5.21}$$

and

$$h_k(x) = \lambda_k h(x/\lambda_k) = x \log(x/\lambda_k), \quad x \ge 0, \tag{5.22}$$

for  $k = 1, \ldots, K$ . Note that h(x) and  $h_k(x)$  are convex, with

$$h(0) = h(1) = 0, \quad h_k(0) = h_k(\lambda_k) = 0, \quad h'(1) = h'_k(\lambda_k) = 1.$$
 (5.23)

The basic tool for analyzing the asymptotic behavior of Z(t) will be the *entropy* function  $\mathcal{H}(t)$ ,

$$\mathcal{H}(t) = \sum_{k} \int_{t}^{t+W_{j}(t)} h_{k}(D_{k}'(r)) dr, \quad t \ge 0.$$
(5.24)

Since  $D(\cdot)$  and  $W(\cdot)$  are Lipschitz continuous, it is not difficult to check that  $\mathcal{H}(\cdot)$  is also Lipschitz continuous. We will analyze  $\mathcal{H}'(t)$  at regular points t.

One can think of  $\mathcal{H}(t)$  as measuring the "distance" at time t from the "equilibrium"  $D'_k(r) \equiv \lambda_k$ , averaged over the above values of r. One can check that, for  $\rho < e$ , such equilibria only occur when Z(r) = 0. (For  $\rho = e$ , there are other solutions.) We will not need this fact here, although it motivates our approach.

Since  $h_k(\cdot)$  is convex and  $\rho < e$ , the following lemma is not difficult to show with the help of Jensen's Inequality. We will use here the expression

$$m_j^s \sum_{k \in \mathcal{C}(j)} D'_k(t) = 1 \quad \text{on } \{t : W_j(t) \neq 0\},$$
(5.25)

which follows from (4.53)-(4.55).

**Lemma 5.11.** Suppose  $\mathfrak{X}(\cdot)$  is any FIFO fluid model solution of Kelly type, with  $\rho_j \leq 1$  for all j. Then,  $\mathcal{H}(t) \geq 0$  for all t.

*Proof.* Rewriting (5.24) gives

$$\mathcal{H}(t) = \sum_{j \in F_t} \lambda_j^{\Sigma} \int_t^{t+W_j(t)} \sum_{k \in \mathcal{C}(j)} (\lambda_k / \lambda_j^{\Sigma}) h(D'_k(r) / \lambda_k) \, dr, \qquad (5.26)$$

where

$$F_t = \{j : W_j(t) \neq 0\}$$
 and  $\lambda_j^{\Sigma} = \sum_{k \in \mathcal{C}(j)} \lambda_k.$ 

By Jensen's Inequality, (5.26) is at least

$$\sum_{j \in F_t} \lambda_j^{\Sigma} \int_t^{t+W_j(t)} h\left(\sum_{k \in \mathcal{C}(j)} D'_k(r) / \lambda_j^{\Sigma}\right) dr.$$

On account of (5.25) and  $\rho_j \leq 1$ , which holds for all j, the integrand

$$h\left(\sum_{k\in\mathcal{C}(j)} D'_k(r)/\lambda_j^{\Sigma}\right) = h(1/\rho_j) \ge 0$$
(5.27)

at regular points  $r \in [t, t + W_j(t)]$ . It follows from (5.26)–(5.27) that  $\mathcal{H}(t) \geq 0$  for all t.

If we knew that  $\mathcal{H}'(t) \leq -c_1$ ,  $c_1 > 0$ , whenever  $\mathcal{H}(t) > 0$ , it would follow immediately that  $\mathcal{H}(t) = 0$  for  $t \geq \mathcal{H}(0)/c_1$ . We will instead show the weaker inequality that, for appropriate  $c_2, c_3 > 0$ ,

$$\mathcal{H}(t) - \mathcal{H}(t + c_2 W^M(t)) \ge c_3 W^M(t) \quad \text{for all } t,$$
(5.28)

where  $W^{M}(t) = \max_{i} W_{i}(t)$ . From this, we will show that

$$\mathcal{H}(t) = 0 \quad \text{for } t \ge c_4 \mathcal{H}(0), \tag{5.29}$$

and some  $c_4$  not depending on  $\mathfrak{X}(\cdot)$ . It will then follow quickly that

$$Z(t) = 0 \quad \text{for } t \ge c_5 |Z(0)| \tag{5.30}$$

and appropriate  $c_5$ , which implies Theorem 5.10.

Proposition 5.12 is the most important step in deriving (5.28). The representation for  $\mathcal{H}'(t)$  given on the right side of (5.31) will enable us to compute upper bounds on  $\mathcal{H}'(t)$  without too much difficulty. As usual, statements on  $\mathcal{H}'(t)$  refer to regular points t.

**Proposition 5.12.** For any FIFO fluid model of Kelly type,

$$\mathcal{H}'(t) = \sum_{k} [h_k(A'_k(t)) - h_k(D'_k(t))] - \sum_{j} \frac{1}{m_j^s} h(1 + W'_j(t)).$$
(5.31)
*Proof.* Differentiation of (5.24) gives

$$\mathcal{H}'(t) = \sum_{k} [(1 + W'_{j}(t))h_{k}(D'_{k}(t + W_{j}(t))) - h_{k}(D'_{k}(t))].$$
(5.32)

By (5.18), this

$$= \sum_{k} [(1 + W'_{j}(t))h_{k}(A'_{k}(t)/(1 + W'_{j}(t))) - h_{k}(D'_{k}(t))].$$

Using the definition of  $h_k(\cdot)$ , one can check this

$$=\sum_{k} [h_k(A'_k(t)) - h_k(D'_k(t))] - \sum_{j} \left(\sum_{k \in \mathcal{C}(j)} A'_k(t)\right) \log(1 + W'_j(t)). \quad (5.33)$$

If  $Y'_j(t) \neq 0$  and  $W'_j(t)$  exists, it follows from (4.54) that  $W'_j(t) = 0$ . For each j, the summand on the right side of (5.33) is therefore equal to

$$\left(\sum_{k \in \mathcal{C}(j)} A'_k(t) + \frac{1}{m_j^s} Y'_j(t)\right) \log(1 + W'_j(t)).$$
(5.34)

On the other hand, combining (4.52), (4.53), and (4.55), one gets

$$t + W_j(t) = W_j(0) + m_j^s \sum_{k \in \mathcal{C}(j)} A_k(t) + Y_j(t).$$

Plugging the derivative of the above quantity into the first factor in (5.34), shows that (5.34) is equal to  $h(1 + W'_j(t))/m_j^s$ . Consequently, (5.33) equals

$$\sum_{k} [h_k(A'_k(t)) - h_k(D'_k(t))] - \sum_{j} \frac{1}{m_j^s} h(1 + W'_j(t)).$$

Together with (5.32), this implies (5.31).

The following proposition provides bounds for the two terms on the right side of (5.31), and hence for  $\mathcal{H}'(t)$ .

Proposition 5.13. For any FIFO fluid model of Kelly type, both

$$\sum_{k} [h_k(A'_k(t)) - h_k(D'_k(t))] \le \sum_{k} Z'_k(t)$$
(5.35)

and, for some  $c_6 > 0$ ,

$$\sum_{j} \frac{1}{m_{j}^{s}} h(1 + W_{j}'(t)) \ge \sum_{k} Z_{k}'(t) + c_{6} \sum_{j} (W_{j}'(t))^{2}.$$
 (5.36)

Consequently,

$$\mathcal{H}'(t) \le -c_6 \sum_j (W'_j(t))^2.$$
 (5.37)

The proof of (5.36) is just a few lines. The proof of (5.35) is a bit longer. It relies on the convexity of  $h_k(\cdot)$  and on Jensen's Inequality, as well as on the general equality

$$Z(t) = Z(0) + (I - P^T)(\lambda t - D(t)), \qquad (5.38)$$

which follows from (4.51), (4.50), and (1.6). The inequality (5.35) reflects the randomness present in the mean transition matrix P. For instance, for reentrant lines that are closed (i.e., there are no arrivals to or departures from the system), this randomness is absent and both sides of (5.35) are equal to 0. (The sum on the left side telescopes, with all terms cancelling.)

Proof of Proposition 5.13. The inequality (5.37) immediately follows from Proposition 5.12 and (5.35)–(5.36). For (5.36), note that since h(1) = 0, h'(1) = 1, h''(x) = 1/x, and  $W'_i(t)$  is bounded,

$$h(1 + W'_j(t)) \ge W'_j(t) + c_7(W'_j(t))^2$$

for some  $c_7 > 0$ . Also, by (4.57),

$$W_j'(t) = m_j^s \sum_{k \in \mathcal{C}(j)} Z_k'(t).$$

Combining these two expressions and summing over j shows (5.36).

To show (5.35), note that by (4.50),

$$h_k(A'_k(t)) = h_k\left(\alpha_k + \sum_{\ell} P_{\ell,k} D'_{\ell}(t)\right)$$
(5.39)

for each t, which can be rewritten as

$$\lambda_k h \bigg( \lambda_k^{-1} \bigg[ \alpha_k + \sum_{\ell} (\lambda_\ell P_{\ell,k}) (\lambda_\ell^{-1} D'_\ell(t)) \bigg] \bigg).$$

By (1.6),  $\lambda_k^{-1}(\alpha_k + \sum_{\ell} \lambda_{\ell} P_{\ell,k}) = 1$ . So, by Jensen's Inequality and h(1) = 0, this is \_\_\_\_\_

$$\leq \alpha_k h(1) + \sum_{\ell} \lambda_{\ell} P_{\ell,k} h(\lambda_{\ell}^{-1} D_{\ell}'(t)) = \sum_{\ell} P_{\ell,k} h_{\ell}(D_{\ell}'(t)).$$
(5.40)

It follows from (5.39)–(5.40) that, for each k,

$$h_k(A'_k(t)) \le \sum_{\ell} P_{\ell,k} h_\ell(D'_\ell(t)).$$

Summation over k shows that

$$\sum_{k} [h_k(A'_k(t)) - h_k(D'_k(t))] \le -\sum_{k} \left(1 - \sum_{\ell} P_{k,\ell}\right) h_k(D'_k(t)).$$
(5.41)

Since  $h_k(\cdot)$  is convex with  $h_k(\lambda_k) = 0$  and  $h'_k(\lambda_k) = 1$ , the right side of (5.41) is

$$\leq -\sum_{k} \left(1 - \sum_{\ell} P_{k,\ell}\right) (D'_k(t) - \lambda_k) = \sum_{k} Z'_k(t), \qquad (5.42)$$

with the equality following from (5.38). The inequality (5.35) is an immediate consequence of (5.41) and (5.42).

Let  $\tau_j(t)$  denote the additional time, starting at t, until a station j is next empty. We will need the following general bound for showing (5.28).

Lemma 5.14. For each subcritical fluid model,

$$\tau_j(t) \le c_2 W^M(t) \quad \text{for all } t \tag{5.43}$$

and some constant  $c_2$ .

*Proof.* On account of (4.57), (5.43) is equivalent to

$$\tau_j(t) \le c_8 |Z(t)|,\tag{5.44}$$

for appropriate  $c_8$ . By the general equality (4.63),

$$CMQ(Z(t') - Z(t)) = (\rho - e)(t' - t) + Y(t') - Y(t)$$

for  $t' \geq t$ . Therefore, for each j,

$$m_j^s \sum_{k \in \mathcal{C}(j)} Z_k(t') \le (CMQZ(t))_j + (\rho_j - 1)(t' - t) + Y_j(t') - Y_j(t).$$

This implies (5.44), with  $c_8 = (CMQZ(t))_j / (1 - \rho_j)$ .

The bound in (5.28) follows from (5.37), Jensen's Inequality, and (5.43).

Proof of (5.28). By (5.37) and Jensen's Inequality,

$$\mathcal{H}(t) - \mathcal{H}(t') \ge c_6 \int_t^{t'} (W_j'(r))^2 dr \ge \frac{c_6}{t' - t} (W_j(t') - W_j(t))^2 \tag{5.45}$$

for  $0 \le t < t'$  and any j. Suppose that  $W_j(t) \ne 0$  for given t and j. Setting  $t' = t + \tau_j(t)$ , it follows from (5.43) and (5.45) that

 $\mathcal{H}(t) - \mathcal{H}(t + \tau_j(t)) \ge c_3 (W_j(t))^2 / W^M(t),$ 

where  $c_3 = c_6/c_2$ . Choosing j so that  $W_j(t) = W^M(t)$ , it follows from the monotonicity of  $\mathcal{H}(\cdot)$  that

$$\mathcal{H}(t) - \mathcal{H}(t + c_2 W^M(t)) \ge c_3 W^M(t),$$

which is (5.28).

We now finish the demonstration of Theorem 5.10.

315

Proof of Theorem 5.10. We will iterate along the times  $t_{i+1} = t_i + c_2 W^M(t_i)$ ,  $i = 0, 1, 2, \ldots$ , where  $t_0 = 0$ . This gives

$$\mathcal{H}(0) - \mathcal{H}(t_i) \ge c_3 t_i / c_2 \quad \text{for all } i,$$

by (5.28). Since  $\mathcal{H}(t_i) \geq 0$ , it follows that

$$t_{\infty} \stackrel{\text{def}}{=} \lim_{i \to \infty} t_i \le c_2 \mathcal{H}(0) / c_3.$$

Consequently, by the continuity of W(t),  $W(t_{\infty}) = 0$ . Moreover, because of (5.24),  $\mathcal{H}(t_{\infty}) = 0$ , which implies that  $\mathcal{H}(t) = 0$  for  $t \ge t_{\infty}$ . This is equivalent to (5.29). It follows from this and (5.28) that

$$W(t) = 0 \quad \text{for } t \ge t_{\infty}.$$

Since D(t) is Lipschitz continuous, it is not difficult to see that

$$\mathcal{H}(0) \le c_9 W^M(0)$$

for appropriate  $c_9$ . Together, the last three displays imply

$$W(t) = 0$$
 for  $t \ge c_{10} W^M(0)$ 

with  $c_{10} = c_2 c_9 / c_3$ . It follows from this and (4.57) that

$$Z(t) = 0 \quad \text{for } t \ge c_5 |Z(0)|,$$

for appropriate  $c_5$ , which is (5.30). Theorem 5.10 follows.

#### HLPPS queueing networks

The head-of-the-line proportional processor sharing (HLPPS) discipline is a variant of processor sharing, which was discussed in Chapter 2. Under this discipline, all nonempty classes present at a station are served simultaneously, with the fraction of time spent serving a class being proportional to the number of jobs of the class currently there, and all of the service going into the first job of each class. The discipline is clearly HL. When the service times are exponentially distributed, the queueing network process  $\mathfrak{X}(\cdot)$  of an HLPPS network coincides with that of the corresponding processor sharing network, where all jobs of a class receive equal service instead of the first receiving all of the service. The HLPPS discipline can be thought of as a simpler variant of processor sharing that exhibits the HL property. It can be preferable to processor sharing in certain situations when a penalty is attached to sharing service among too many jobs.

The following result is shown in [Br96b].

**Theorem 5.15.** Any subcritical HLPPS queueing network satisfying (5.1) is stable.

As in the previous examples, Theorem 4.16 can be employed to demonstrate stability of these networks. In order to use the accompanying fluid model machinery, we observe that the HLPPS property can be expressed as

$$T(t) = \int_0^t Z^P(s) \, ds \tag{5.46}$$

for all  $t \ge 0$ , where

$$Z_k^P(s) = \begin{cases} Z_k(s)/Z_j^{\Sigma}(s) & \text{for } Z_j^{\Sigma}(s) > 0, \\ 0 & \text{for } Z_j^{\Sigma}(s) = 0, \end{cases}$$

and

$$Z_j^{\Sigma}(s) = \sum_{k \in \mathcal{C}(j)} Z_k(s).$$

The *HLPPS queueing network equations* are then (4.42)-(4.47) together with (5.46). The *HLPPS fluid model equations* are (4.50)-(4.55) together with

$$T'_k(t) = Z^P_k(t)$$
 when  $Z^{\Sigma}_j(t) > 0, \quad k = 1, \dots, K.$  (5.47)

Arguing as in the proof of Proposition 4.12, it is not difficult to show that (5.47) is satisfied for all fluid limits of HLPPS queueing networks. So, the fluid model given by (4.50)-(4.55) and (5.47) is associated with the HLPPS queueing network. In order to show Theorem 5.15, it therefore suffices to show its fluid model analog.

## Theorem 5.16. Any subcritical HLPPS fluid model is stable.

The demonstration of Theorem 5.16 exhibits similarities to that of Theorem 5.10 for FIFO fluid models of Kelly type. The argument employs an entropy function

$$\mathcal{H}(t) = \sum_{k} Z_k(t) \log(D'_k(t)/\lambda_k), \quad t \ge 0.$$

This entropy function is equivalent to

$$\sum_{k} m_k Z_j^{\Sigma}(t) h_k(D'_k(t)), \quad t \ge 0,$$

which is similar to that in (5.24).

As before, the goal is to show (5.29), from which (5.30) will follow. In the present setting, one can show that

$$\mathcal{H}'(t) \le -c_{11} < 0$$

until  $\mathcal{H}(t) = 0$ , in place of (5.28). The individual steps of the argument differ from those for Theorem 5.10. We note that Theorem 5.16 (and hence Theorem 5.15) holds when  $m_k = m_\ell$  for  $s(k) = s(\ell)$  is not assumed, unlike its FIFO analog. This might be expected, because the stationary distribution for processor sharing networks given by (2.7) and (2.9) also does not require this condition.

#### 5.4. Global stability

In the last chapter and in the first three sections of this chapter, we have addressed the question on when an HL queueing network is stable. Our main technique has been the employment of fluid models: when an associated fluid model of a queueing network is stable, so is the queueing network. In this section, we introduce a different notion of stability, global stability. As before, fluid models will provide the main technique for demonstrating global stability.

We say that a queueing network is *globally stable* if it remains stable when the discipline is replaced by any HL discipline. That is, under any HL discipline (e.g., an SBP or FIFO discipline), the resulting network, with its given routing, and interarrival and service distributions, is positive Harris recurrent.

Global stability is clearly a more restrictive requirement, in general, than is stability. For example, the subcritical two-station reentrant lines, with route given in Figure 3.1, may or may not be stable, depending on the discipline. The FBFS discipline is stable, by Theorem 5.5 (provided (5.1) holds). On the other hand, the Lu-Kumar network, which has the priority scheme (4,1) and (2,3) at the two stations, is unstable for the range of mean service times given in Theorem 3.2.

We recall that the basic fluid model equations (4.50)-(4.55) do not reflect the discipline of a queueing network. By Proposition 4.12, each HL queueing network is associated with its basic fluid model. The following result is therefore a direct consequence of Theorem 4.16.

**Proposition 5.17.** Assume that a given HL queueing network satisfies (5.1) and that its basic fluid model is stable. Then, the queueing network is globally stable.

As we have seen earlier in this chapter, it is easier to study the stability of fluid models rather than directly investigating the stability of the corresponding queueing networks. We will take this approach here, and study the stability of the basic fluid model. A complete, explicit theory is given in [DaV00] when the network has two stations and routing is deterministic. Most of this section is devoted to providing a summary of these results. The remainder of the section briefly presents related results on weak stability and global weak stability.

We note that our terminology is somewhat different than that in [DaV00], and in related work [Ha97] and [DaHV99]. There, the authors employ the term "fluid networks" rather than "fluid models", and employ the term "global stability" for when the fluid networks are stable. The latter definition corresponds to stability for the basic fluid model here, although somewhat different fluid model equations are used in those papers. Another difference arises from the usage of the term "queueing network" in our lectures, which assumes that a discipline has already been assigned; in the above references, this is not assumed.

#### Virtual stations and push starts

An appealing theory for the stability of basic fluid models with two stations is developed in [DaV00] for networks with deterministic routing, with necessary



FIG. 5.1. The five classes along the route are labelled in the order of their appearance; the mean service time of each is given. The priority scheme is (1,3,4) at station 1 and (5,2) at station 2.

and sufficient conditions for stability being given. Key ingredients include the concepts of *virtual stations* and *push starts*, in terms of which stability can be phrased. The theory does not extend to networks with three or more stations.

We will later give a systematic definition of virtual stations and push starts, but the concepts are better illustrated by an example. We will employ the example from [DaHV04], which is a reentrant line with route given in Figure 5.1.

The queueing network has two stations, with three and two classes each, as illustrated in the figure. The mean service times at the classes are

$$m_1 = m_3 = m_5 = 0.4$$
 and  $m_2 = m_4 = 0.1$ , (5.48)

and jobs are assumed to enter the network at rate 1. The network is therefore subcritical. The actual distributions of the interarrival and service times will not be important to us. A preemptive SBP discipline is assigned, with priority scheme (1,3,4) at station 1 and (5,2) at station 2. That is, the discipline is FBFS at station 1 and LBFS at station 2. We will exhibit a virtual station for this queueing network.

We also consider the fluid model obtained by adding the SBP equation (5.13) to the basic fluid model equations, where the classes have the same priorities as above. The SBP fluid model thus obtained is associated with the above queueing network; this was shown in Section 5.2. Examination of the basic fluid model suffices presently, although we will need the SBP fluid model later when examining push starts. In order to exhibit a virtual station for either fluid model, we will need to employ the fluid limits from the above queueing network.

The interaction between classes 3 and 5 is important for understanding the evolution of this queueing network. If one assumes that  $Z_3(0) = 0$  or  $Z_5(0) = 0$ , it then follows that

$$Z_3(t) = 0 \text{ or } Z_5(t) = 0 \quad \text{for each } t \ge 0.$$
 (5.49)

That is, if at least one of these two classes is initially empty, then this condition persists for all time. The reason is that since class 3 has higher priority than class 4, no job from the latter class can enter class 5 as long as class 3 is not empty. Moreover, if class 5 is not empty, then no job can enter class 3, since class 5 has higher priority than class 2. This behavior relies on the assumption that the discipline is preemptive. (The corresponding behavior for classes 2 and 4 of the Lu-Kumar network was used in the proof of Theorem 3.2.)

It follows from (5.49) that, under such initial data, classes 3 and 5 can never be served simultaneously. Consequently,

$$T_3(t_2) - T_3(t_1) + T_5(t_2) - T_5(t_1) \le t_2 - t_1$$
(5.50)

whenever  $t_1 \leq t_2$ , or equivalently,

$$T_3'(t) + T_5'(t) \le 1 \tag{5.51}$$

wherever the derivative is defined. One can think of classes 3 and 5 as forming a "virtual station", with their service being constrained, as in (5.51), as if they actually belonged to a single station.

Let  $\mathfrak{X}(\cdot)$  be a fluid limit obtained from a sequence  $(a_n, x_n)$  as in (4.71), with  $Z_3^{x_n}(0) = a_n$  and  $Z_5^{x_n}(0) = 0$ . Then, (5.50) holds for each term of the sequence, and so the same is also true for  $\overline{T}(\cdot)$ . On the other hand,  $\mathfrak{X}(\cdot)$  is a solution of the SBP fluid model equations. So, classes 3 and 5 form a "virtual station" for the SBP fluid model as well. Note that not all SBP fluid model solutions need satisfy (5.50) or (5.51), since fluid mass can possibly be served and pass through a class k, with  $Z_k(t) = 0$  nonetheless holding over an entire time interval. This contrasts with the behavior for the associated queueing network.

For the SBP fluid model (and hence for the basic fluid model) to be stable,

$$m_3 + m_5 < 1 \tag{5.52}$$

needs to hold. The argument is similar to that for Part (c) of Proposition 4.11, and relies on (5.50). (One can employ the analogs of (4.62) and (4.63), but with C,  $\rho$ , and Y augmented to include the virtual station.) For either fluid model with the parameters given by (5.48),  $m_3 + m_5 = 0.8 < 1$ , which does not preclude stability. However, if  $m_5$  is replaced by  $m'_5 = 0.7$ , then  $m_3 + m'_5 =$ 1.1 > 1, and so the resulting fluid model will not be stable even though the network is still subcritical. One can then show, in fact, that some solutions satisfy  $\lim \inf_{t\to\infty} Z(t)/t \ge 1/11$ . As we will see, both fluid models with the original service rates are already not stable. For this, we will need to use push starts.

The push start phenomenon relies on the presence of a virtual station, and provides a stability condition which is an amplification of that provided by the virtual station. It is caused by a higher priority class that shares a station with one of the classes of the virtual station, and always receives at least a fixed proportion of the service at that station. In contrast to virtual stations, it is a fluid model phenomenon, rather than a queueing network phenomenon.

In the setting of the example in Figure 5.1, the stability condition (5.52) can be replaced by

$$\rho_{\text{push}} = \frac{m_3}{1 - m_1} + m_5 < 1, \tag{5.53}$$

by using push starts. We first note that if one deletes class 1 from the reentrant line in Figure 5.1, then the four-class reentrant line that remains is equivalent to the Lu-Kumar network in Section 3.1. If one retains the labelling of the original five-class reentrant line, then one can check that the bound (5.52) is needed for the associated SBP fluid model to be stable, for the same reasons as for the five-class fluid model.

Returning to the SBP fluid model for the five-class network in Figure 5.1, we observe that class 1 has the highest priority at its station. Since fluid enters the reentrant line at rate 1 and  $m_1 < 1$ , class 1 empties in finite time and remains empty thereafter. In keeping class 1 empty, station 1 spends proportion  $m_1 = 0.4$  of its effort in processing fluid at class 1. The remaining proportion  $1 - m_1$  of its effort can be spent on fluid in classes 3 and 5. This occurs for all solutions of the SBP fluid model.

Since class 1 remains empty, its sole effect on the remainder of the system is to reduce the amount of effort available at station 1 for the other two classes. Removing this class and expanding the service times at classes 3 and 4 by the factor  $1/(1 - m_1)$  to compensate for this reduced effort, one can show with a little work that the resulting SBP fluid model is identical to the four-station SBP fluid model mentioned above, but with the service times at classes 3 and 4 expanded by  $1/(1 - m_1)$ . The push start stability condition (5.53) therefore replaces the condition (5.52), as desired. With the choice of service times given in (5.48),  $\rho_{\text{push}} = \frac{2}{3} + \frac{2}{5} > 1$ , and so (5.53) is violated. Hence, the SBP fluid model in Figure 5.1 is not stable. This also implies that the corresponding basic fluid model is not stable.

#### Systematic presentation of virtual stations and push starts

We will provide an abridged version of the construction given in [DaV00], referring sometimes to the virtual station and push start example given earlier for motivation. In order to simplify matters somewhat, we will restrict consideration to networks with just a single deterministic route, i.e., to reentrant lines.

We employ the following terminology. An excursion is a maximal set of consecutive classes along the route that belong to a single station. A last class of an excursion is the last class visited there, and a first class denotes all of the remaining classes; if the excursion contains only one class, then it has no first class. A set S of excursions is strictly separating if it contains no consecutive excursions and does not contain the first excursion. For each such set S, the virtual station V(S) consists of the classes in the excursions of S, together with the first classes of excursions for which the immediately preceding excursion is not in S. Also, let  $k_1, k_2, \ldots, k_L$  denote the last classes visited for each of the L excursions, and let  $F^{\leq}(\ell)$  denote all of the classes along the route visited up to and including  $k_{\ell}$ ; then  $F^{\leq}(\ell) \stackrel{\text{def}}{=} F^{\leq}(\ell) - \{k_{\ell}\}$  is called a *push start set*. One sets  $V_j(S), F_j^{\leq}(\ell)$ , and  $F_j^{\leq}(\ell), j = 1, 2$ , equal to the corresponding classes restricted to the stations 1 and 2, respectively.

For the example in Figure 5.1, there are four excursions consisting of the sets of classes,  $\{1\}$ ,  $\{2\}$ ,  $\{3, 4\}$ , and  $\{5\}$ . The virtual stations are  $\{2\}$ ,  $\{2, 5\}$ ,  $\{3, 4\}$ ,

and  $\{3, 5\}$ . The only one that is not a subset of a station, and is therefore of interest, is  $V(\{5\}) = \{3, 5\}$ . For future reference, we record that

$$V({5}) = {3,5}, \quad F^{\leq}(2) = {1,2}, \quad F^{<}(2) = {1}.$$
 (5.54)

We now state the main result in [DaV00], restricted to reentrant lines. Here, we use the abbreviation  $m(A) = \sum_{k \in A} m_k$  for any set A of classes. The rate at which fluid enters the first class is denoted by  $\alpha_1$ .

**Theorem 5.18.** A two-station basic fluid model is stable if and only if

$$\rho_j < 1 \quad for \ j = 1, 2,$$
(5.55)

and for each strictly separating set S and  $\ell = 1, \ldots, L$ ,

$$\sum_{j=1}^{2} \frac{\alpha_1 m(V_j(S) \setminus F_j^{\leq}(\ell))}{1 - \alpha_1 m(F_j^{\leq}(\ell))} < 1.$$
(5.56)

In [DaV00], it is also shown that if the left side of (5.56) is strictly greater than 1 for some strictly separating set S and some  $\ell$ , then there exists a fluid model solution whose fluid mass  $|Z(t)| \to \infty$  as  $t \to \infty$ . (The authors actually consider the limit for the "work in progress", i.e., the total workload, which is equivalent to this limit.)

We note that, in order to be of interest in (5.56), virtual stations need to include classes from both stations. Otherwise, (5.56) follows from the subcriticality of each station.

One can interpret the example in Figure 5.1 in terms of Theorem 5.18. Substitution of (5.54) into (5.56) reduces the latter to the inequality in (5.53). Since this is violated for the choice of m in the example, Theorem 5.18 implies that the basic fluid model is not stable. There is, moreover, a fluid model solution with  $|Z(t)| \to \infty$  as  $t \to \infty$ . We saw earlier that such a solution is given by the SBP discipline with priorities (1,3,4) and (5,2) at the two stations. For arbitrary two-station fluid models, the calculation of the left side of (5.56) for all strictly separating sets S and all  $\ell$  will in general be tedious.

There is, for arbitrary two-station reentrant lines, a strong connection between stability of the associated basic fluid model and stability of the corresponding family of fluid models with SBP disciplines. Namely, when the basic fluid model is not stable, there must also exist an SBP fluid model that is not stable. Another way of phrasing this is the following. Let  $\mathcal{D} \subset \mathbf{R}_{+}^{K}$  denote the set of service time vectors, with coordinates  $m_{k}$ , on which the basic fluid model is stable for a given route and choice of  $\alpha_{1}$ . (In [DaHV99] and [DaV00],  $\mathcal{D}$  is referred to as the global stability region.) Similarly, let  $\mathcal{D}_{S} \subset \mathbf{R}_{+}^{K}$  denote the set on which all fluid models with SBP disciplines are stable. Clearly,  $\mathcal{D} \subseteq \mathcal{D}_{S}$ . The above assertion is that, in fact,

$$\mathcal{D} = \mathcal{D}_S. \tag{5.57}$$

This is shown in [DaV00], while demonstrating Theorem 5.18.

We also point out that, as a consequence of Theorem 5.18, the region  $\mathcal{D}$  is monotone. That is, reducing service times maintains stability. This follows immediately from the conditions (5.55) and (5.56).

The proofs of the two directions of Theorem 5.18 are of different levels of difficulty. The necessity of the conditions (5.55) and (5.56) is reasonably straightforward. If (5.56) is violated for some value of  $\ell$ , one can make the same basic type of argument we sketched for the SBP fluid model in Figure 5.1, once an appropriate SBP discipline has been chosen. One wants a discipline that (a) assigns the highest priority to classes in  $F^{<}(\ell)$ , with classes in  $F^{<}(\ell)$  being ordered according to the FBFS discipline, (b) assigns a high priority to the remaining classes in V(S), and (c) assigns a low priority to the other classes, including the class in  $F^{\leq}(1) \setminus F^{<}(1)$ . (The particular priority is not important in each of (b) and (c).) Using Theorem 5.5, one can then show that  $F^{<}(\ell)$  will be empty after large times, like the first class in Figure 5.1. By applying the same type of argument we sketched for that network, one can also show that once all of the classes in either  $V_1(S)$  or  $V_2(S)$  are empty, this condition persists for all time. This argument heavily uses the structure of V(S), which was defined with appropriate "gaps" between its classes so that service at  $V_1(S)$  will prevent service at  $V_2(S)$  and vice versa, because of the presence of low priority classes in between. More detail is given in [DaV00] and in [Ha97], which looks at a related problem. (The reference [DaV96] cited in [DaV00] never appeared due to an unrelated flaw in extending push starts to the queueing network setting. This flaw also affects the example on page 756 in [Da96].)

The demonstration of the sufficiency of (5.55) and (5.56) is more involved and requires most of the work in [DaV00]. The paper employs linear programming techniques to construct a piecewise linear Lyapunov function, from which the stability of the fluid model will follow. (Piecewise linear Lyapunov functions have also been employed in [BoZ92], [DoM94], and [DaWe96].) The duality between minimum flows and maximum cuts is used to obtain the explicit formulation in (5.55) and (5.56). We will not go into details here.

It is natural to ask whether Theorem 5.18 extends to fluid models with more than two stations. [DaHV99] shows this is not the case, in general, by analyzing the fluid models whose routing is given by Figure 5.2. Not only is there no



FIG. 5.2. The basic fluid model for this three-station reentrant line has irregular behavior with regard to stability, when m and the discipline are varied. This behavior is not present for two-station reentrant lines.

analog of Theorem 5.18, but  $\mathcal{D} \neq \mathcal{D}_S$  and  $\mathcal{D}$  is not monotone. There is presently no developed theory for stability for more than two stations.

#### Rate and global stability

We conclude this section with two other types of stability. We will say that an HL queueing network is *rate stable* (or *pathwise stable*) if for any given initial state x,

$$\lim_{t \to \infty} Z^x(t)/t = 0 \quad \text{a.s.}$$
(5.58)

(Alternative definitions are often given. For instance,  $\lim_{t\to\infty} D^x(t)/t = \lambda$  a.s. is equivalent to (5.58).) An HL queueing network will be globally rate stable if (5.58) continues to hold irrespective of the discipline. Rate stability for queueing networks differs from stability in that only the first order behavior of  $Z^x(t)$ enters into (5.58); an unstable queueing network might conceivably be rate stable, with  $\lim_{t\to\infty} |Z^x(t)| = \infty$  a.s., but with  $|Z^x(t)| = o(t)$ .

As is the case for stability, fluid models and fluid limits may be employed to demonstrate rate stability. A fluid model is *weakly stable*, if for each solution of the fluid model equations with Z(0) = 0, one has Z(t) = 0 for all  $t \ge 0$ . Clearly, stability of a fluid model implies weak stability. Also, as was done in the context of global stability, the basic fluid model may be employed to demonstrate global rate stability. (In the literature, the term *globally weakly stable* is used when the basic fluid model is weakly stable.)

For queueing networks, rate stability and global rate stability are less satisfying properties than are stability and global stability, but they are easier to show. One has the following analog of Theorem 4.16.

**Theorem 5.19.** Assume that an associated fluid model of a given HL queueing network is weakly stable. Then, the queueing network is rate stable.

*Proof.* Suppose on the contrary that the queueing network is not rate stable. Then,

$$\limsup_{t \to \infty} |Z^x(t)|/t > 0$$

for some  $\omega \in G$ , where G is given in (4.70). Let  $a_n \to \infty$ , as  $n \to \infty$ , be a sequence on which  $\liminf_{t\to\infty} |Z^x(a_n)|/a_n > 0$ . The sequence  $(a_n, x)$  satisfies (4.71). So, there is a subsequence  $(a_{i_n}, x)$  along which  $\mathfrak{X}^x(a_{i_n}t)/a_{i_n}$  has a limit  $\overline{\mathfrak{X}}(\cdot)$  that satisfies the associated fluid model equations.

Since the initial state is constant,  $\overline{Z}(0) = 0$ . Because the fluid model is assumed to be weakly stable, it follows that  $\overline{Z}(t) \equiv 0$ . In particular,  $\overline{Z}(1) = 0$ . Therefore,

$$\lim_{n \to \infty} |Z^x(a_{i_n})| / a_{i_n} = 0,$$

which is a contradiction. So, the queueing network is, in fact, rate stable.  $\Box$ 

By Proposition 4.12, the basic fluid model is associated with its queueing network. The following corollary is therefore an immediate consequence of Theorem 5.19. Both Theorem 5.19 and the corollary are related to Theorem 4.1 in [Ch95].

**Corollary 1.** Assume that the basic fluid model of a given HL queueing network is weakly stable. Then, the queueing network is globally rate stable.

The following weak stability analog of Theorem 5.18 is given in [DaV00]. As before, we restrict the result to reentrant lines from queueing networks with deterministic routing. We employ the same notation as before involving strictly separating sets of excursions, virtual stations, and push start sets. Note that the conditions (5.59) and (5.60) for weak stability are the same as (5.55) and (5.56) in Theorem 5.18, except that strict inequalities in (5.55) and (5.56) are replaced by inequalities.

Theorem 5.20. A two-station basic fluid model is weakly stable if and only if

$$\rho_j \le 1 \quad for \ j = 1, 2,$$
(5.59)

and for each strictly separating set S and  $\ell = 1, \ldots, L$ ,

$$\sum_{j=1}^{2} \frac{\alpha_1 m(V_j(S) \setminus F_j^{\leq}(\ell))}{1 - \alpha_1 m(F_j^{<}(\ell))} \le 1.$$
(5.60)

The proof of Theorem 5.20 is not spelled out there, but, according to [DaV00], is analogous to that of Theorem 5.18.

### 5.5. Relationship between QN and FM stability

The material in the first four sections of this chapter has relied heavily on the stability of fluid models that are associated with a given queueing network. In the first three sections, stability of such fluid models enabled us to demonstrate stability for a number of disciplines when the queueing network is subcritical. In the last section, this approach was applied to global stability.

We have so far avoided the question in the opposite direction, of whether stability of a queueing network implies the stability of its associated fluid model. On account of Theorem 4.16, this would imply that the two concepts of stability are equivalent, modulo certain side conditions. If the above implication is not correct, how "close" are the two concepts? Besides being aesthetically pleasing, a two-directional relationship would allow reduction of questions involving the stability of queueing networks to the less complex setting of fluid models. Results, such as Theorem 5.18 of the previous section, would also take on added significance. Of course, for disciplines such as those in the first three sections of the chapter, this equivalence is already clear, if both the queueing networks and fluid models are stable whenever they are subcritical.

The question should not be taken in its most naive form. For instance, the basic fluid model for a queueing network (which has no equations specifying the discipline) need not be stable even if the queueing network is. So, the fluid model needs to include an appropriate equation (or equations) corresponding to the discipline; we have already seen that there are often "canonical" equations that

suggest themselves in this context. Also, one should exclude certain "exotic" disciplines. For instance, if the priority rule favoring different classes is allowed to change with the total queue length |Z|, such a discipline might consist of priority rules that are decreasingly stable as  $|Z| \to \infty$ . The fluid model would then correspond to the limiting rule, which is not stable, whereas the queueing network itself could be stable.

As we will see, even for certain "standard" disciplines, the above two forms of stability are not equivalent: there exist stable queueing networks whose fluid models are not stable. Moreover, there are no general results in this direction. Nonetheless, one must work to produce such examples, which seem to be, in some sense, "borderline". So, at this point, one can claim that the reduction to fluid models "works well in practice". The same should hold for the basic fluid model in the context of global stability. For global weak stability, there is, in fact, a partial result, which we mention at the end of the section.

This section is divided into four parts. We first present an elementary condition for the instability of a queueing network from [Da96]. We then present examples of stable queueing networks with unstable fluid models from [Br99] and [DaHV04], which together constitute most of the section. We conclude with the global weak stability result mentioned above, which is from [GaH05].

### An elementary condition for instability

Proposition 5.21 gives an elementary condition for the instability of a queueing network in terms of its fluid limits. The result relies on Proposition 4.11 and is a variant of a result from [Da96]. Related results, with more involved conditions, are given in [Me95] and [PuR00]. Similar reasoning also shows that an HL queueing network with a supercritical station is unstable; the result is included in Proposition 5.21.

**Proposition 5.21.** (a) Assume that for every fluid limit  $\overline{\mathfrak{X}}(\cdot)$  of a given HL queueing network, with  $\overline{Z}(0) = 0$ , that  $\overline{Z}(\delta) \neq 0$  for some fixed  $\delta > 0$ . Then, for every initial state x,

$$\liminf_{t \to \infty} |Z^x(t)|/t > 0 \quad on \ G.$$

$$(5.61)$$

(b) Assume that for a given HL queueing network,  $\rho_j > 1$  at some j. Then, for some  $\epsilon > 0$  and every initial state x,

$$\liminf_{t \to \infty} |Z^x(t)|/t \ge \epsilon \quad on \ G. \tag{5.62}$$

*Proof.* The argument for (5.61) is almost the same as that used in the proof of Theorem 5.19. Suppose on the contrary that, for some  $\omega \in G$ ,

$$\liminf_{t \to \infty} |Z^x(\delta t)|/t = 0,$$

where  $\delta > 0$  is chosen as in the statement of the proposition. Let  $a_n \to \infty$ , as  $n \to \infty$ , be a sequence on which this limit holds. Since the sequence  $(a_n, x)$ satisfies (4.71), by Proposition 4.12, there is a subsequence  $(a_{i_n}, x)$  along which  $\mathfrak{X}^{x}(a_{i_{n}}t)/a_{i_{n}}$  has a limit  $\overline{\mathfrak{X}}(\cdot)$ . Since the initial state is constant,  $\overline{Z}(0) = 0$ , and so, by assumption,  $\overline{Z}(\delta) \neq 0$ . Therefore,

$$\lim_{t \to \infty} |Z^x(\delta a_{i_n})| / a_{i_n} > 0,$$

which is a contradiction. Consequently, (5.61) holds.

Suppose  $\rho_j > 1$  for some j. The basic fluid model is associated with the queueing network. By Part (d) of Proposition 4.11,  $|Z(1)| \ge \epsilon$  for some  $\epsilon > 0$  and all solutions of the basic fluid model. Reasoning analogous to that for (5.61) then implies (5.62).

The assumption in Part (a) of Proposition 5.21, that  $Z(\delta) \neq 0$  for all fluid limits with  $\overline{Z}(0) = 0$ , is unfortunately too strong for most applications, as is the assumption, in Part (b), that  $\rho_j > 1$  for some j. One can, in fact, wonder whether the proposition has any applications to subcritical networks. Note, for instance that, under Z(0) = 0, the conditions  $Z(t) \equiv 0$  and  $D(t) = \lambda t$  are equivalent for any fluid model. Since none of the stations, in this case, is overloaded if the network is subcritical, these equations provide a solution for fluid models such as the basic fluid model and the other fluid models that have appeared in this chapter. So, the assumptions in Part (a) will not be satisfied if one considers all fluid model solutions (rather than just fluid limits).

The situation is different if one considers only the fluid limits of the queueing network. For instance, a subcritical queueing network might contain supercritical virtual stations, as in the previous section. In this setting,  $\bar{Z}(\delta) = 0$  will no longer be possible for any fluid limit. Hence, Part (a) of Proposition 5.21 will be applicable. Other situations where  $\bar{Z}(\delta) \neq 0$  for all fluid limits will also occur.

#### An example of a stable queueing network with unstable fluid model

In this subsection and the next, we will present two examples of stable queueing networks with unstable fluid models. In this subsection, the example consists of a network with routing that is a modification of that given in Figure 5.3.

The network portrayed in Figure 5.3 is a reentrant line with three stations, each possessing two classes. The discipline is a preemptive SBP, with priority



FIG. 5.3. The six classes along the route are labelled in the order of their appearance; the mean service times are given in (5.63). The priority scheme is (6,1), (5,2), and (3,4) at the three stations.



FIG. 5.4. In the modified network, jobs leaving class 2 are randomly routed to one of the L stations, (3, l). After service at the two classes at the station, jobs are routed to class 5. In the figure, L = 3.

scheme (6,1), (5,2), and (3,4) at the three stations. Interarrival and service times are assumed to be exponentially distributed, with the interarrival times having mean 1 and the service times having means

$$m_2 = m_3 = m_6 = 3/4, \quad m_1 = m_5 = \gamma, \quad m_4 = \gamma/L^2,$$
 (5.63)

where  $\gamma \in (0, 1/8)$  and  $L \in \mathbb{Z}_+$ . (One can, for example, set  $\gamma = 1/16$ .) The queueing network is subcritical with

$$\rho_1 = \rho_2 = \frac{3}{4} + \gamma < \frac{7}{8} \quad \text{and} \quad \rho_3 = \frac{3}{4} + \frac{\gamma}{L^2} < \frac{7}{8}.$$
(5.64)

The process Z(t) corresponding to the queueing network is Markov because of the SBP discipline and the exponential interarrival and service times.

The modified queueing network we will employ is defined by "splitting" the station 3 into L separate two-class stations,  $(3, 1), \ldots, (3, L)$ , one of which is randomly chosen along the route. That is, the higher priority class of each of these stations is entered with probability 1/L, by a job leaving class 2. After service at this class is completed, the job passes to the lower priority class of this station, and then to class 5 of the original network; the queueing network otherwise evolves as before (see Figure 5.4). The new service times are exponentially distributed with means  $\frac{3}{4}L$  and  $\gamma/L$ . The modified network is subcritical, with traffic intensity again given by (5.64). The process Z(t) corresponding to the modified queueing network is again Markov. The original network can be obtained by "collapsing" the stations  $(3, 1), \ldots, (3, L)$  into a single station 3.

Both the original and modified queueing networks have preemptive SBP disciplines. As in Section 5.2, the fluid models consisting of the basic fluid equations, together with the SBP equation (5.13), are associated with these queueing networks. In [Br99], the stability of the modified queueing network and its fluid model are characterized as follows.

**Theorem 5.22.** (a) For sufficiently large L, the modified queueing network defined above is stable. (b) For any L, the associated fluid model is not stable. In particular, there is a solution of the fluid model, with

$$\liminf_{t \to \infty} |Z(t)|/t = 1/3.$$
(5.65)

Sketch of proof. The argument for Part (b) consists of explicitly constructing a solution of the fluid model equations that satisfies (5.65). The construction is similar to that for the simpler two-station, four-class network in Example 2 of the second part of Section 4.3. Rather than doing this here, we will instead motivate the evolution of the solution. The interested reader can refer to page 825 of [Br99] for a precise treatment.

The above solution is constructed so as to take the same values at each of the stations  $(3, \ell), l = 1, \ldots, L$ . The sum of the contributions of these stations therefore undergoes the same evolution as the corresponding solution for the original network, which we henceforth consider. For the original network,  $m_5 \leq m_6$ . Since class 5 is the higher priority class at station 2, this implies that the fluid mass there is always passed to class 6 at least as fast as it can be served at class 6. Also, since  $m_2 = m_3$ , fluid mass for this solution is also passed from class 2 to class 3 as fast as it can be served at class 3, provided no mass is concurrently served at the higher priority class 5. Such will be the case for this particular solution, as can easily be verified by checking its explicit construction.

Using these two observations, one can see that this solution will become a solution of the two-station network obtained by deleting station 2, if one combines the mass at class 2 with that at class 3, and the mass at class 5 with that at class 6. The resulting network is just the Lu-Kumar network in Figure 3.1, with priorities (6, 1) and (3, 4) at the remaining stations 1 and 3, and service times

$$m_3 = m_6 = 3/4, \quad m_1 = \gamma, \quad m_4 = \gamma/L^2.$$
 (5.66)

The classes 3 and 6 are the high priority classes at their stations. For the same reasons as given between (5.49) and (5.52) for the fluid model considered there, the classes 3 and 6 form a virtual station, with

$$\rho_{\text{virtual}} = 3/2 > 1.$$
(5.67)

So, the two-station fluid model is not stable. For our particular solution, fluid mass will only be processed at this virtual station at 2/3 the rate at which it arrives there, and so 1/3 will be a lower bound for the limit in (5.65). From the actual construction of the solution, the limit in fact equals 1/3, as claimed in Part (b) of the theorem.

The argument for Part (a) of Theorem 5.22 is more involved. The result might also be true for the original queueing network, but one needs to employ the modified queueing network, with large L, to obtain the bounds that are used here.

The basic reason for the different behavior in Parts (a) and (b) of the theorem is the different behavior at the stations  $(3, \ell)$ . For the unstable fluid model solution that was just discussed, as in the examples of the unstable queueing networks in Chapter 3, the flow of mass through the network is "cyclic", with, in particular, an increasingly large periodic buildup of mass at the different stations. This includes the stations  $(3, \ell), \ \ell = 1, \ldots, L$ . (Or equivalently, the station 3 for the original network.)

At the stations  $(3, \ell), \ell = 1, \ldots, L$ , the behavior of the queueing network is different. One can show that, for large L, the effective service rate at class 2 is slower than the combined service rate at the classes  $(3, \ell), \ell = 1, \ldots, L$ , when the combined number of jobs at the classes  $(3, \ell)$  is large. The basic idea is that for large L, the probability is close to 1 that, relatively frequently, one of the classes, say  $(3, \ell_0)$ , becomes empty. Once this occurs, service begins on the jobs at the quick low priority class  $(4, \ell_0)$  that follows  $(3, \ell_0)$ . The served jobs from there continue to the high priority class 5, which interrupts service at class 2. This interruption prevents jobs from entering any of the  $(3, \ell)$  classes until class 5 empties, which only occurs after it stops receiving jobs from the different  $(4, \ell)$ classes. As additional  $(3, \ell)$  classes empty, this allows service to begin at the corresponding  $(4, \ell)$  classes. Without this interference from class 5, the service rate at class 2 is the same as the combined service rate of all of the  $(3, \ell)$  classes, which is 4/3. This interference, however, creates idle periods for class 2, which causes it to have a slower effective service rate than the combined service rate of the  $(3, \ell)$  classes. This service rate is also slower than the combined service rate of the  $(3, \ell)$  stations, since service at the  $(4, \ell)$  classes is quick.

The above behavior creates a strong bias for the total number of jobs  $Z_3^{\Sigma}(t)$  at all of the  $(3, \ell)$  stations to drift toward 0. Using this, one can show that  $Z_3^{\Sigma}(t)$  typically remains close to 0 on the relevant time scale after it first hits 0. It is therefore reasonable to guess that the presence of the  $(3, \ell)$  stations should have only a negligible effect on the evolution of the queueing network on this time scale, and that the qualitative behavior of the network should not change if these stations are omitted. This is, in fact, correct. By using fluid limits, one can make this statement precise. (The fluid limits used in [Br99] are slightly more general than those introduced in Section 4.3. They are mentioned briefly in the discussion after Theorem 4.16.)

After a fixed time (depending on the initial state), all such fluid limits will have no mass at station 3. They will satisfy the SBP fluid model consisting of the remaining stations 1 and 2, whose classes have priorities (6,1) and (5,2), and service times

$$m_2 = m_6 = 3/4$$
 and  $m_1 = m_5 = \gamma.$  (5.68)

This SBP discipline is LBFS. Since  $\rho_1 = \rho_2 < 7/8 < 1$ , one knows from Theorem 5.6 that this two-station fluid model is stable. It follows that the fluid limits for the entire modified queueing network are stable. From this, it in turn follows that the modified queueing network is stable. So, Part (a) holds.

The queueing network in Theorem 5.22 is an example of a stable SBP queueing network for which the associated fluid model consisting of the basic fluid model equations and (5.13) is not stable. The discipline here is an SBP discipline and not an "exotic" discipline one would wish to exclude from consideration. This example therefore casts doubt on a robust equivalence between queueing network and fluid model stability.

Despite this example, one can still ask whether there is some general correspondence between queueing network stability and some notion similar to fluid model stability. This question is, of course, vague, and there are different possible approaches. One approach is to focus on the stability of fluid limits instead of fluid models. There is no known correspondence here either. Moreover, the set of fluid limits for a queueing network will be difficult to describe in general.

Another approach is more philosophical. How does one know that the fluid model employed in Theorem 5.22 is the "right one"? Perhaps fluid limits of the queueing network automatically satisfy further "hidden" fluid model equations, and under these additional equations, all fluid model solutions will be stable. Maybe the same is true for queueing networks in general. It is unclear how to disprove such a thesis. However, even if the thesis is correct, one will still need a way of finding such hidden equations, in order for fluid models to provide a general practical framework for reformulating queueing network stability.

The literature on related work includes two papers, [FoK99] and [StR99], that give examples of stable networks with unstable fluid limits for polling models. The model in [FoK99] consists of two stations and two servers, which switch back and forth between the stations when the work is exhausted; there is also a switchover period. The paper also introduces a less restrictive criterion of stability under which the fluid limits are stable.

# An example where stability depends on the distributions of the queueing network

At the end of the last subsection, we wondered how one could discount the possibility that an unstable fluid model, which is associated with a stable queueing network, merely lacks equations that are implicit in the evolution of the queueing network. One convincing response to this would be to find two queueing networks, one stable and the other not, that differ only in their interarrival and service time distributions, but for which everything else, including the interarrival and service time means, is the same. The queueing networks would then have the same fluid model, which would show that stability of a queueing network cannot always be determined at the fluid model level.

This approach is taken in [DaHV04]. The queueing network that is analyzed is the two-station SBP reentrant line in Figure 5.1 of the previous section. It has priority scheme (1, 3, 4) and (5, 2) at the two stations, interarrival rate 1, and mean service times given by (5.48). One version of the queueing network is assumed to have deterministic interarrival and service times. Hence, there is no randomness in the evolution of the network. The other version of the queueing network is assumed to have exponentially distributed interarrival and service times. Both versions are assumed to be nonpreemptive.

In order to describe the evolution of the deterministic network, abbreviated notation is employed in [DaHV04] to designate certain specific states. One uses the 6-tuples,  $(z_1, \ldots, z_5; a)$ , where  $z_k$  is the number of jobs in class k and a is the remaining interarrival time until the next job enters the network. Only certain states occurring at the instant of a service completion are designated this way, and they are viewed at the time t- just prior to the completion of service. For

example, by

$$(0, 0, 0, 1, 0; a), (5.69)$$

one means the "state" at a time t- if service of the class 4 customer is completed at time t. Strictly speaking, these are not states for the state space we introduced in Section 4.1, but they suffice for describing the evolution of the network from the specific "states" mentioned above. One can, in particular, verify that the deterministic network starting from (5.69), with  $a \in (0, 0.1]$ , returns to this state exactly one unit of time later. Thus, such a trajectory forms an *orbit*; for a given  $a \in (0, 0.1]$ , this orbit is called an *a orbit*.

Employing this terminology, it is shown in [DaHV04] that the queueing network with deterministic distributions has the following behavior.

**Theorem 5.23.** For any initial state of the nonpreemptive deterministic queueing network specified above, there exists a finite time at which the network enters an a orbit, with  $a \in (0, 0.1]$ .

On the other hand, in [DaHV04], it is also shown that the following result holds for the queueing network with exponentially distributed interarrival and service times.

**Theorem 5.24.** For any initial state of the nonpreemptive exponential queueing network specified above,

$$|Z(t)| \to \infty \quad as \ t \to \infty, \tag{5.70}$$

with probability 1.

The exponential queueing network in Theorem 5.24 is unstable. (This is also true for the preemptive version of the discipline.) On the other hand, the trajectories of the deterministic queueing network in Theorem 5.23 all eventually enter a fixed bounded set. Since not all states communicate with one another, its underlying (deterministic) Markov process is not positive Harris recurrent, and so the network is not stable in the sense we are using in these lectures. Nevertheless, the two examples show distinctly different behavior with regard to their "stability", despite having interarrival and service time distributions with the same means, and therefore not being distinguishable at the fluid model level.

In [DaHV04], it is also shown that the above deterministic queueing network, but with preemptive rather than nonpreemptive discipline, is unstable, with  $|Z(t)| \to \infty$  linearly as  $t \to \infty$ , for appropriate initial states. So, whether or not a queueing network is preemptive can also influence its stability.

The natural question arises as to whether the behavior of the nonpreemptive deterministic network is an anomaly. Simulations in [DaHV04] seem to indicate that when the deterministic interarrival and service times are replaced by those with uniform distributions having the same means as before, the (nonpreemptive) network is stable when the width of the uniform distributions is 0.001, but is not stable when the width is 0.1.

It is tricky to attempt to formulate rules as to which distributions should be stable and which should not, for more general networks. On the basis of the above examples, it is tempting to say that both a more deterministic distribution and a nonpreemptive discipline should be "good" with regard to stability, whereas a more random distribution and a preemptive discipline should be "bad". But, the three-station example in Theorem 5.22 of the previous subsection consists of a stable preemptive queueing network with exponential distributions and a fluid model that is not stable, which make this less clear. In [DaHV04], it is suggested that exponential distributions should be "bad" for a large family of two-station networks, based on the belief that the proof of Theorem 5.24 should generalize.

Not surprisingly, the argument for Theorem 5.23 is primarily computational, on account of the system's deterministic evolution. The reader is referred to [DaHV04] for details. The argument for Theorem 5.24 is more involved. In the remainder of this subsection, we will discuss some of the ideas behind the proof of the theorem that relate to virtual stations and push starts.

Virtual stations and push starts were introduced in the previous section in the context of the associated fluid model for the preemptive version of the queueing network in Theorem 5.24. Since the push start condition (5.53) is violated for the choice of service times in (5.48), one already knows from the previous section that the fluid model is not stable. One would like to be able to apply similar reasoning to show that the queueing network in Theorem 5.24 is unstable.

We recall that for the same queueing network, but where the discipline is preemptive, the classes 3 and 5 form a virtual station. As explained in the previous section, the virtual station owes its presence to the constraint in (5.49), which does not allow simultaneous service at the two classes if either is initially empty.

Unfortunately, for the nonpreemptive version of interest to us here, (5.49) need not hold. For instance, a job at class 4 can complete service there and continue to class 5 after another job has already arrived at class 3. However, at most one job at class 5 can coexist with jobs at class 3; the same is true for jobs at class 5 coexisting with those at class 3. For the nonpreemptive network, one can therefore replace (5.49) with

$$(Z_3(t) - 1)^+ = 0 \text{ or } (Z_5(t) - 1)^+ = 0 \text{ for each } t \ge 0,$$
 (5.71)

assuming (5.49) holds at t = 0. In contrast to (5.49), (5.71) allows simultaneous service at classes 3 and 5. One might hope that there is not "too much" such joint service, so that the classes together still behave like a virtual station, up to a small error.

There are also difficulties in applying the push start condition in (5.53) to the queueing network. The inequality applies to the fluid model, where it relies on the proportion of effort devoted to class 1 being constant over time. For the queueing network (either preemptive or nonpreemptive), there may or may not be a job there receiving service at a given time. So, service at class 1 may or may not interfere with service at class 3. The reasoning leading up to (5.53) will therefore not hold in the strict sense. However, if there is sufficient independence between the times of external arrivals to class 1 and times when class 3 is not empty, one might expect the same behavior to hold up to a small error. Since the arrivals to the network are Poisson, one might expect that to be the case in the present setting. It is certainly not the case for the deterministic network in Theorem 5.23.

Much of the work in [DaHV04] is devoted to carrying out the ideas that are summarized in the last two paragraphs, in order to demonstrate Theorem 5.24. The queueing network itself must be analyzed, which involves deriving estimates based on its random evolution. However, the virtual station and push start behavior of the associated fluid model provide guidance for these steps.

### Global rate stability

The examples in the previous two subsections cast doubt on the equivalence, under general conditions, of queueing network and fluid model stability. This does not preclude analogous positive results in the context of global stability, though. In particular, the presence of a basic fluid model that is not stable might imply the existence of an associated queueing network that is not stable, under *some* discipline. This is an open question even for two-station reentrant lines. If one knows this equivalence in the two-station reentrant line setting, one can then apply Theorem 5.18 to obtain necessary and sufficient conditions on the global stability of such reentrant lines.

At the end of Section 5.4, we introduced the concepts of rate stability, global rate stability, and weak stability. As in the two previous subsections, one can attempt to show the equivalence of rate stability for queueing networks and weak stability for fluid models. One does not meet with greater success here than with stability. The deterministic queueing network in Theorem 5.23, for example, is rate stable, but its fluid model (with auxiliary equation (5.13)) is not.

The question of global rate stability is more approachable. We recall, from the corollary to Theorem 5.19, that an HL queueing network is globally rate stable if its basic fluid model is weakly stable. The following partial converse is shown in [GaH05] for a family of queueing networks. The family consists of two-station networks with deterministic routing. The interarrival and service distributions there are assumed to satisfy a large deviation condition.

# **Theorem 5.25.** Assume that a two-station queueing network with the above properties is globally rate stable. Then, its basic fluid model is weakly stable.

The proof of Theorem 5.25 is by contradiction. The basic idea is to show the existence of a fluid model solution that diverges linearly to infinity, and to construct a discipline for which the sample paths of the queueing network almost surely "track" this solution. The large deviation condition is employed in this construction.

The queueing networks in the theorem differ, in two ways, from the HL queueing networks we defined in Section 4.1 and have been employing since.

Changes in service allocation are allowed between the arrival and departure times of jobs at stations, in order to facilitate tracking. More seriously, the discipline that was constructed need not be time homogeneous. It should be possible to modify the construction so as to eliminate both of these differences.

Since the discipline of a queueing network is not reflected in the basic fluid model equations, the corollary to Theorem 5.19 continues to hold in the setting of the queueing networks in Theorem 5.25. Together with the theorem, this implies the equivalence of global rate stability for these two-station queueing networks and weak stability for their basic fluid models. As an immediate consequence of this and Theorem 5.20, one obtains the following necessary and sufficient conditions for the global rate stability of two-station queueing networks. Since Theorem 5.20 was stated for reentrant lines, we make the same restriction here. We also make the same assumptions on the networks as were made in Theorem 5.25.

**Theorem 5.26.** A two-station reentrant line with the above properties is globally rate stable if and only if both (5.59) and (5.60) hold.

# Acknowledgments

The author wishes to thank the following probabilists, with whom he consulted at various points: John Baxter, Jim Dai, Michael Harrison, John Hasenbein, Haya Kaspi, Thomas Kurtz, Sean Meyn, and Ruth Williams. He also thanks John Baxter for his technical help in preparing the manuscript. The author thanks the National Science Foundation for its continued support.

# Bibliography

- [AnAFKLL96] Andrews, M., Awerbuch, B., Fernandez, A., Kleinberg, J., Leighton, T., and Liu, Z. (1996). Universal stability results for greedy contention-resolution protocols. In *Symposium on Computer Science*, 1996. MR1450636
- [As03] Asmussen, S. (2003). Applied Probability and Queues, second edition. Springer, New York. MR1978607
- [AtN78] Athreya, K.B. and Ney, P. (1978). A new approach to the limit theory of recurrent Markov chains. *Trans. Amer. Math. Soc.* 245, 493–501. MR0511425
- [AzKR67] Azema, J., Kaplan-Duflo, M. and Revuz, D. (1967). Mesure invariante sur les classes récurrentes des processus de Markov. Z. Wahrscheinlichkeitstheorie verw. Geb. 8, 157–181. MR0222955
- [BaB99] Baccelli, F. and Bonald, T. (1999). Window flow control in FIFO networks with cross traffic. *Queueing Systems* 32, 195–231. MR1720554
- [BaF94] Baccelli, F. and Foss, S.G. (1994). Ergodicity of Jackson-type queueing networks I. Queueing Systems 17, 5–72. MR1295406
- [Ba76] Barbour, A.D. (1976). Networks of queues and the method of stages. Adv. Appl. Prob. 8, 584–591. MR0440729
- [BaCMP75] Baskett, F., Chandy, K.M., Muntz, R.R., and Palacios, F.G. (1975). Open, closed and mixed networks of queues with different classes of customers. J. Assoc. Comput. Mach. 22, 248–260. MR0365749
- [BoKRSW96] Borodin, A., Kleinberg, J., Raghavan, P., Sudan, M., and Williamson, D.P. (1996). Adversarial queueing theory. In Symposium on Computer Science, 1996. MR1427534
- [Bo86] Borovkov, A.A. (1986). Limit theorems for queueing networks. Theory of Probability and its Applications 31, 413–427. MR0866868
- [BoZ92] Botvich, D.D. and Zamyatin, A.A. (1992). Ergodicity of conservative communication networks. *Rapport de Recherche* **1772**, INRIA.
- [Br94a] Bramson, M. (1994). Instability of FIFO queueing networks. Ann. Appl. Probab. 4, 414–431. (Correction 4 (1994), 952.) MR1272733
- [Br94b] Bramson, M. (1994). Instability of FIFO queueing networks with quick service times. Ann. Appl. Probab. 4, 693–718. MR1284981
- [Br95] Bramson, M. (1995) Two badly behaved queueing networks. Stochastic Networks, 105–116. IMA Math. Appl. 71. Springer, New York. MR1381007

- [Br96a] Bramson, M. (1996). Convergence to equilibria for fluid models of FIFO queueing networks. Queueing Systems 22, 5–45. MR1393404
- [Br96b] Bramson, M. (1996). Convergence to equilibria for fluid models of headof-the-line proportional processor sharing queueing networks. *Queueing* Systems 23, 1–26. MR1433762
- [Br98a] Bramson, M. (1998). Stability of two families of queueing networks and a discussion of fluid limits. *Queueing Systems* 28, 7–31. MR1628469
- [Br98b] Bramson, M. (1998). State space collapse with application to heavy traffic limits for multiclass queueing networks. Queueing Systems 30, 89– 148. MR1663763
- [Br99] Bramson, M. (1999). A stable queueing network with unstable fluid model. Ann. Appl. Probab. 9, 818–853. MR1722284
- [Br01] Bramson, M. (2001). Earliest-due-date, first-served queueing networks. Queueing Systems 39, 79–102. MR1865459
- [BrD01] Bramson, M. and Dai, J.G. (2001). Heavy traffic limits for some queueing networks. Ann. Appl. Probab. 11, 49–90. MR1825460
- [Bu56] Burke, P.J. (1956). The output of a queueing system. Operat. Res. 4, 699–704. MR0083416
- [ChTK94] Chang, C.S., Thomas, J.A., and Kiang, S.H. (1994). On the stability of open networks: a unified approach by stochastic dominance. *Queueing* Systems 15, 239–260. MR1266795
- [Ch95] Chen, H. (1995). Fluid approximations and stability of multiclass queueing networks: work-conserving disciplines. Ann. Appl. Probab. 5, 637–655. MR1359823
- [ChM91] Chen, H. and Mandelbaum, A. (1991). Discrete flow networks: Bottleneck analysis and fluid approximations. *Math. Oper. Res.* 16, 408–446. MR1106809
- [ChY01] Chen, H. and Yao, D.D. (2001). Fundamentals of Queueing Networks. Springer, New York. MR1835969
- [Da95] Dai, J.G. (1995). On positive Harris recurrence of multiclass queueing networks: a unified approach via fluid limit models. Ann. Appl. Probab. 5, 49–77. MR1325041
- [Da96] Dai, J.G. (1996). A fluid limit model criterion for instability of multiclass queueing networks. Ann. Appl. Probab. 6, 751–757. MR1410113
- [DaHV99] Dai, J.G., Hasenbein, J.J., and Vande Vate, J.H. (1999). Stability of a three-station fluid network. Queueing Systems 33, 293–325. MR1742573
- [DaHV04] Dai, J.G., Hasenbein, J.J., and Vande Vate, J.H. (2004). Stability and instability of a two-station queueing network. Ann. Appl. Probab. 14, 326–377. MR2023024
- [DaM95] Dai, J.G. and Meyn, S.P. (1995). Stability and convergence of moments for multiclass queueing networks via fluid limit models. *IEEE Trans. Automat. Control* 40, 1889–1904. MR1358006
- [DaV96] Dai, J.G. and Vande Vate, J.H. (1996). Virtual stations and the capacity of two-station queueing networks. Preprint. MR1410113
- [DaV00] Dai, J.G. and Vande Vate J.H. (2000). The stability of two-station multitype fluid networks. Operations Research 48, 721–744. MR1792776

- [DaWa93] Dai, J.G. and Wang, Y. (1993). Nonexistence of Brownian models of certain multiclass queueing networks. Queueing Systems 13, 41–46. MR1218843
- [DaWe96] Dai, J.G. and Weiss, G. (1996). Stability and instability of fluid models for re-entrant lines. *Math. of Oper. Res.* 21, 115–134. MR1385870
- [Da84] Davis, M.H.A. (1984). Piecewise deterministic Markov processes: a general class of nondiffusion stochastic models. J. Roy. Statist. Soc. Ser. B 46, 353–388. MR0790622
- [Da93] Davis, M.H.A. (1993). Markov Models and Optimization. Chapman & Hall, London. MR1283589
- [Do53] Doob, J.L. (1953). Stochastic Processes. Wiley, New York. MR0058896
- [DoM94] Down, D. and Meyn, S.P. (1994). Piecewise linear test functions for stability of queueing networks. Proc. 33rd Conference on Decision and Control, 2069–2074.
- [Du96] Dumas, V. (1996). Approches fluides pour la stabilité et l'instabilité de réseaux de files d'attente stochastiques à plusieurs classes de clients. Doctoral thesis, Ecole Polytechnique.
- [Du97] Dumas, V. (1997). A multiclass network with non-linear, nonconvex, non-monotonic stability conditions. Queueing Systems 25, 1–43. MR1458584
- [Dur96] Durrett, R. (1996). Probability: Theory and Examples, second edition. Duxbury Press, Belmont. MR1609153
- [Fi89] Filonov, Y.P. (1989). A criterion for the ergodicity of discrete homogeneous Markov chains. (Russian). Ukrain. Mat. Zh. 41, 1421–1422; translation in Ukrainian Math. J. 41, 1223–1225. MR1034693
- [Fo91] Foss, S.G. (1991). Ergodicity of queueing networks. Sib. Math. J. 32, 183–202. MR1142079
- [FoK04] Foss, S.G. and Konstantopoulos, T. (2004). An overview of some stochastic stability methods. J. Oper. Res. Soc. Japan 47, 275–303. MR2174067
- [FoK99] Foss, S.G. and Kovalevskii, A. (1999). A stability criterion via fluid limits and its application to a polling system. Queueing Systems 32, 131– 168. MR1720552
- [Fo53] Foster, F.G. (1953). On the stochastic matrices associated with certain queueing processes. Ann. Math. Stat. 24, 355–360. MR0056232
- [GaH05] Gamarnik, D. and Hasenbein, J.J. (2005). Instability in stochastic and fluid queueing networks. Ann. Appl. Probab. 15, 1652–1690. MR2152240
- [Ge79] Getoor, R. (1979). Transience and recurrence of Markov processes. In Séminaire de Probabilités XIV (J. Azéma, M. Yor editors), 397–409. Springer, Berlin. MR0580144
- [Ha56] Harris, T.E. (1956). The existence of stationary measures for certain Markov processes. Proc. 3rd Berkeley Symp., Vol. II, 113–124. MR0084889
- [Ha97] Hasenbein, J.J. (1997). Necessary conditions for global stability of multiclass queueing networks. Oper. Res. Lett. 21, 87–94. MR1482493
- [Hu94] Humes, C. (1994). A regular stabilization technique: Kumar-Seidman revisited. IEEE Trans. Automat. Control 39, 191–196. MR1258702

- [Ja54] Jackson, R.R.P. (1954). Queueing systems with phase-type service. Operat. Res. Quart. 5, 109–120.
- [Ja63] Jackson, R.R.P. (1963). Jobshop-like queueing systems. Mgmt. Sci. 10, 131–142.
- [KaM94] Kaspi, H. and Mandelbaum, A. (1994). On Harris recurrence in continuous time. Math. Oper. Res. 19, 211–222. MR1290020
- [Ke75] Kelly, F.P. (1975). Networks of queues with customers of different types. J. Appl. Probab. 12, 542–554. MR0388571
- [Ke76] Kelly, F.P. (1976). Networks of queues. Adv. Appl. Prob. 8, 416–432. MR0415800
- [Ke79] Kelly, F.P. (1979). Reversibility and Stochastic Networks. Wiley, New York. MR0554920
- [Kn84] Knight, F.B. (1984). On the ray topology. Séminaire de Probabilités (Strasbourg) 18, 56–69. Springer, Berlin. MR0770948
- [Ku93] Kumar, P.R. (1993). Re-entrant lines. Queueing Systems 13, 87–110. MR1218845
- [KuS90] Kumar, P.R. and Seidman, T.I. (1990). Dynamic instabilities and stabilization methods in distributed real-time scheduling of manufacturing systems. *IEEE Trans. Automat. Control* 35, 289–298. MR1044023
- [LuK91] Lu, S.H. and Kumar, P.R. (1991). Distributed scheduling based on due dates and buffer priorities. *IEEE Trans. Automat. Control* 36, 1406–1416.
- [MaM81] Malyshev, V.A. and Menshikov, M.V. (1981). Ergodicity, continuity, and analyticity of countable Markov chains. *Transactions of the Moscow Mathematical Society* 1, 1–47.
- [Me95] Meyn, S.P. (1995). Transience of multiclass queueing networks via fluid limit models. Ann. Appl. Probab. 5, 946–957. MR1384361
- [MeD94] Meyn, S.P. and Down, D. (1994). Stability of generalized Jackson networks. Ann. Appl. Probab. 4, 124–148. MR1258176
- [MeT93a] Meyn, S.P. and Tweedie, R.L. (1993). Generalized resolvents and Harris recurrence of Markov processes. *Contemporary Mathematics* 149, 227–250. MR1229967
- [MeT93b] Meyn, S.P. and Tweedie, R.L. (1993). Stability of Markov processes II: continuous-time processes and sampled chains. Adv. Appl. Prob. 25, 487–517. MR1234294
- [MeT93c] Meyn, S.P. and Tweedie, R.L. (1993). Stability of Markov processes III: Foster-Lyapunov criteria for continuous-time processes. Adv. Appl. Prob. 25, 518–548. MR1234295
- [MeT93d] Meyn, S.P. and Tweedie, R.L. (1993). Markov Chains and Stochastic Stability. Springer, London. MR1287609
- [MeT94] Meyn, S.P. and Tweedie, R.L. (1994). State-dependent criteria for convergence of Markov chains. Ann. Appl. Probab. 4, 149–168. MR1258177
- [Mu72] Muntz, R.R. (1972). Poisson Departure Processes and Queueing Networks, IBM Research Report RC 4145. A shortened version appeared in Proceedings of the Seventh Annual Conference on Information Science and Systems, Princeton, 1973, 435–440.

- [Ne82] Newell, G.F. (1982). Applications of Queueing Theory. Chapman and Hall. MR0737381
- [Nu78] Nummelin, E. (1978). A splitting technique for Harris recurrent Markov chains. Z. Wahrscheinlichkeitsth. verw. Geb. 43, 309–318. MR0501353
- [Nu84] Nummelin, E. (1984). General Irreducible Markov Chains and Non-Negative Operators. Cambridge University Press, Cambridge. MR0776608
- [Or71] Orey, S. (1971). Lecture Notes on Limit Theorems for Markov Chain Transition Probabilities. Van Nostrand Reinhold, London. MR0324774
- [PeK89] Perkins, J.R. and Kumar, P.R. (1989). Stable distributed real-time scheduling of flexible manufacturing/assembly/disassembly systems. *IEEE Trans. Automat. Control* 34, 139–148. MR0975573
- [PuR00] Puhalskii, A. and Rybko, A. (2000). Nonergodicity of a queueing network under nonstability of its fluid model. *Probl. Inf. Transm.* 36, 23–41. MR1746007
- [Re57] Reich, E. (1957). Waiting times when queues are in tandem. Ann. Math. Statist. 28, 768–773. MR0093060
- [Re92] Resnick, S.I. (1992). Adventures in Stochastic Processes. Birkhäuser, Boston. MR1181423
- [RyS92] Rybko, A.N. and Stolyar, A.L. (1992). Ergodity of stochastic processes that describe functioning of open queueing networks. *Problems Inform. Transmission* 28, 3–26 (in Russian). MR1189331
- [Se94] Seidman, T.I. (1994). "First come, first served" can be unstable! IEEE Trans. Automat. Control 39, 2166–2170. MR1295752
- [Se99] Serfozo, R. (1999). Introduction to Stochastic Networks. Springer, New York. MR1704237
- [Sh88] Sharpe, M. (1988). General Theory of Markov Processes. Academic Press, San Diego. MR0958914
- [Si90] Sigman, K. (1990). The stability of open queueing networks. Stoch. Proc. Applns. 35, 11–25. MR1062580
- [St95] Stolyar, A.L. (1995). On the stability of multiclass queueing networks: a relaxed sufficient condition via limiting fluid processes. *Markov Process. Related Fields* 1, 491–512. MR1403094
- [StR99] Stolyar, A.L. and Ramakrishnan, K.K. (1999). The stability of a flow merge point with non-interleaving cut-through scheduling disciplines. In *Proceedings of INFOCOM 99.*
- [Wa82] Walrand, J. (1982). Discussant's report on "Networks of quasi-reversible nodes", by F.P. Kelly. Applied Probability - Computer Science: The Interface, Volume 1, 27–29. Birkhäuser, Boston. MR0718495
- [Wa83] Walrand, J. (1983). A probabilistic look at networks of quasi-reversible queues. IEEE Trans. Info. Theory, IT-29, 825–836. MR0733189
- [Wa88] Walrand, J. (1988). An Introduction to Queueing Networks. Prentice-Hall, Englewood Cliffs.
- [Wi98] Williams, R.J. (1998). Diffusion approximations for open multiclass queueing networks: sufficient conditions involving state space collapse. Queueing Systems 30, 27–88. MR1663759

# Index

adversarial queueing, 221  $\alpha$ -excessive function, 251, 292

balance equations, 192 detailed, 193 partial, 210 Borel right, 251, 252, 291 bounded set, 257 buffer, 173

class, 173 classical networks, 186 clearing policy, 225 constituency matrix, 267 critical, 180, 266, 270 cumulative idle time, 267 service time process, 266 customers, 171, 172

Debut Theorem, 251, 252 deterministic routing, 173 discipline, 171, 177

entropy function, 311e-stability, 180, 256, 280, 290 ergodic, 180, 255 Erlang distribution, 190, 201, 202 excursion, 321 external arrival, 181 process, 266 rate, 176, 267

## FIFO

network of Kelly type, 190, 310 node of Kelly type, 188 filtration, 250 natural, 250 first-buffer-first-served (FBFS), 177, 307first-in, first-out (FIFO), 177, 186, 188, 228, 233, 235, 249, 268 first-in-system, first-out (FISFO), 178, 250 fluid limit, 243, 265, 275, 276 associated, 276, 277 model, 276 fluid model, 180, 182, 243, 265, 269 basic, 269 equations, 182, 269 auxiliary, 269 basic, 269 FBFS, 307 FIFO, 310 LBFS, 307 SBP, 307 with delay, 270, 289 without delay, 269 solution, 269 fluid network, 182, 234 Foster's Criterion, 257 Generalized, 259 Multiplicative, 259, 281, 286 general distributions, 205 Harris recurrent, 178, 252, 295

positive, 171, 178, 252, 255 positive, 171, 178, 179, 252, 295 head-of-the-line (HL), 177 head-of-the-line proportional processor sharing (HLPPS), 316 heavy traffic limits, 182, 221, 265 homogeneous node, 188, 196 of Kelly type, 195, 196 homogeneous queueing network, 196 of Kelly type, 197, 200 hydrodynamic scaling, 182

infinite server (IS), 186, 189, 190
instability, 180, 221, 226-228, 233, 235, 237, 326
irreducibility measure, 252

Jackson network, 176 generalized, 176 jobs, 171, 172

Kelly type, 238 Kumar-Seidman network, 225

last-buffer-first-served (LBFS), 177, 307
last-in, first-out (LIFO), 178, 186, 190
last-in-system, first-out (LISFO), 178
Lu-Kumar network, 221, 222
deterministic, 222
random, 223
Lusin space, 292

M/M/1 queue, 171, 172, 187 Martin boundary, 230 maximal irreducibility measure, 253 mean routing matrix, 176, 267 service time, 176, 267 transition matrix, 176, 267 method of stages, 190, 202

node, 186, 187 nominal load, 179 non-idling, 176 nonpreemptive, 177 nonuniqueness (of fluid model solutions), 272 norm, 247, 257

orbit, 332

petite, 254, 303 phase-type distributions, 205  $\varphi$ -irreducible, 252 piecewise-deterministic Markov process (PDP), 250, 291 policy, 171, 177 preemptive, 177 resume, 177 primitive triple, 266 processor sharing (PS), 177, 186, 190, 316 push start, 319–321, 333 set, 321 quasi-reversible, 191, 207, 209, 210 queueing network, 173, 178 closed, 177 equations, 180, 266 auxiliary, 268 basic, 268 multiclass, 176, 177 open, 177 process, 267 single class, 176, 177, 303 R-chain, 253, 292 reentrant line, 173 refined class, 202 regular point, 270, 303 set, 296 residual interarrival times, 247 service times, 247 resolvent, 253, 294 reversed process, 192 reversible, 192 round robin, 177 routing process, 266 Rybko-Stolyar network, 221, 226 service rate, 176, 267 simple queue, 172

small, 254 uniformly, 256, 261, 303 stability, 171, 179, 183, 271, 276, 280 asymptotic, 289

343

global, 224, 318 global rate, 324, 334 global weak, 324 nonconvex region, 239 nonmonotone region, 240, 242, 324pathwise, 324 rate, 324 weak, 324 stage, 202 state space countable, 171, 179, 180 uncountable, 178, 206 static buffer priority (SBP), 177, 221, 226, 249, 268, 307, 327 stationary, 192 distribution, 180, 192 measure, 172, 252, 297 strictly separating, 321 strong Markov, 251 subcritical, 180, 270 supercritical, 180, 326 symmetric node, 201 total arrival rate, 179, 267 traffic equations, 179 intensity, 171, 179, 267 unstable, see instability virtual station, 319, 321, 333 work, 178 conserving, 176 workload immediate, 224, 267 A(t), 181 $(a_n, x_n), 276$  $\mathcal{A}, 176$  $\alpha$ , 176, 267  $(\alpha, M, P), 277$  $\beta(i, n), 196, 201$ C, 267C(j), 173D(t), 181

 $\delta(i, n), 196$ E(t), 181, 266 $(E(\cdot), \Gamma(\cdot), \Phi(\cdot)), 277$ e, 180, 267  $e_{\ell}, 249$  $\eta_A, 252$  $\mathcal{F}_t, 250$  $\mathcal{F}_t^0, 250 \\ G, 276$  $\Gamma$ , 266  $\gamma_k, 249, 266$ h(x), 311H(t), 311 $\lambda$ , 179, 267 M, 176, 267  $m^s$ , 188  $m_k, 176, 267$  $\mathcal{M}, 276$  $\mu_k, 176$  $n_x, 198$ P, 176, 267 $P^t, 251$  $P_{\mu}, 250$  $\Phi$ , 181, 266  $\phi(n), 196, 201$  $\phi^k$ , 249, 266  $\pi, 172$ Q, 177, 267q(x), 192q(x, y), 192, 198, 204, 212 $\hat{q}(x), 195$  $\hat{q}(x,y), 193, 195, 204, 210$  $\rho, 179, 267$ S, 181, 191, 192, 211 $(S, \mathscr{S}), 247$  $S_0, 188$  $S_e, 190, 203$  $S_{\infty}, 206$ s(i), 202s(k), 173 $\mathfrak{s}_i(x), 204, 209$ T(t), 181 $\tau_A, 252$  $\tau_A(\delta), 254$ u, 247, 276 v, 247, 276

v(i), 202	$\xi_k,249$
W(t), 181, 267	$\xi_k(i), \ 266$
x , 247	Y(t),  267
x(i), 202 $\hat{x}(i), 267, 260$	Z(t), 172, 181
$\bar{\mathfrak{X}}(t), 201, 203$ $\bar{\mathfrak{X}}(t), 276$	z, 276